# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **FORM 10-Q**

x QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the period ended September 30, 2011

OR

o TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

to

**Commission file number 1-8993** 

# WHITE MOUNTAINS INSURANCE GROUP, LTD.

(Exact name of Registrant as specified in its charter)

Bermuda

(State or other jurisdiction of incorporation or organization)

**94-2708455** (I.R.S. Employer Identification No.)

80 South Main Street, Hanover, New Hampshire (Address of principal executive offices)

**03755-2053** (Zip Code)

Registrant's telephone number, including area code: (603) 640-2200

Indicate by check mark whether the Registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months, and (2) has been subject to such filing requirements for the past 90 days. Yes x No o

Indicate by check mark whether the Registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months. Yes x No o

Indicate by check mark whether the Registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See definitions of "large accelerated filer," "accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer x

Accelerated filer o

Non-accelerated filer o

Smaller reporting company o

Indicate by check mark whether the Registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No x

As of October 28, 2011, 7,630,717 common shares with a par value of \$1.00 per share were outstanding (which includes 73,500 restricted common shares that were not vested at such date).

**Table of Contents** 

WHITE MOUNTAINS INSURANCE GROUP, LTD.

**Table of Contents** 

Page No.

<u>PART I.</u> <u>FINANCIAL INFORMATION</u>

<u>Item 1.</u> <u>Financial Statements (Unaudited)</u>

	Consolidated Statements of Operations and Comprehensive Income, Three and Nine Months Ended September 30, 2011 and 2010	2
	Consolidated Statements of Changes in Equity, Nine Months Ended September 30, 2011 and 2010	3
	Consolidated Statements of Cash Flows, Nine Months Ended September 30, 2011 and 2010	4
	Notes to Consolidated Financial Statements	5
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	
	Results of Operations – Three and Nine Months Ended September 30, 2011 and 2010	36
	<u>Liquidity and Capital Resources</u>	47
	Fair Value Considerations	54
	Non-GAAP Financial Measures	55
	Critical Accounting Estimates	56
	Forward-Looking Statements	56
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	56
<u>Item 4.</u>	Controls and Procedures	57
PART II.	OTHER INFORMATION	57
Items 1 throug	<u>h 6.</u>	57
<b>SIGNATURES</b>		58

# Table of Contents

# PART I. FINANCIAL INFORMATION.

# **Item 1. Financial Statements**

# WHITE MOUNTAINS INSURANCE GROUP, LTD. CONSOLIDATED BALANCE SHEETS

Accounts payable on unsettled investment purchases

CONSOLIDATED BALANCE SHEETS				
(Millions, except share and per share amounts)		tember 30, 2011 naudited	Do	ecember 31, 2010
Assets	U	naudited		
Fixed maturity investments, at fair value	\$	5,287.7	\$	5,786.1
Short-term investments, at amortized cost (which approximates fair value)		772.8		975.3
Common equity securities, at fair value		669.8		667.0
Convertible fixed maturity investments, at fair value		140.5		143.0
Other long-term investments		328.8		372.1
Total investments		7,199.6		7,943.5
Cash (restricted: \$393.3 and \$286.7)		557.7		395.0
Reinsurance recoverable on unpaid losses		2,581.2		2,344.0
Reinsurance recoverable on paid losses		30.9		63.1
Insurance and reinsurance premiums receivable		644.8		532.5
Funds held by ceding companies		131.6		118.7
Investments in unconsolidated affiliates		458.6		389.7
Deferred acquisition costs		202.9		176.1
Deferred tax asset		431.7		431.9
Ceded unearned insurance and reinsurance premiums		104.4		184.0
Accrued investment income		58.3		63.3
Accounts receivable on unsettled investment sales		20.3		39.5
Other assets		760.2		730.0
Assets held for sale		1,399.5		1,122.8
Total assets	\$	14,581.7	\$	14,534.1
Liabilities				
Loss and loss adjustment expense reserves	\$	5,882.0	\$	5,736.8
Unearned insurance and reinsurance premiums		976.4		938.7
Debt		669.0		818.8
Deferred tax liability		360.6		373.2
Accrued incentive compensation		169.9		157.4
Ceded reinsurance payable		188.3		221.1
Funds held under reinsurance treaties		39.3		85.8
A		4		22.2

57.1

22.3

Other liabilities	1,310.6	1,167.7
Liabilities held for sale	909.4	751.5
Total liabilities	10,562.6	10,273.3
Shareholders' equity and noncontrolling interests		
White Mountains' common shareholders' equity		
White Mountains' common shares at \$1 par value per share - authorized 50,000,000 shares; issued and		
outstanding 7,630,717 and 8,194,925 shares	7.6	8.2
Paid-in surplus	1,262.7	1,350.8
Retained earnings	2,004.0	2,175.6
Accumulated other comprehensive income, after-tax:		
Equity in unrealized gains from investments in unconsolidated affiliates	135.7	58.5
Net unrealized foreign currency translation gains	35.7	61.4
Other	(1.0)	(1.5)
Total White Mountains' common shareholders' equity	3,444.7	3,653.0
Noncontrolling interests		
Noncontrolling interest - OneBeacon Ltd.	269.8	295.0
Noncontrolling interest - Sirius Group Preference Shares	250.0	250.0
Noncontrolling interest - consolidated limited partnerships and A.W.G Dewar	54.6	62.8
Total noncontrolling interests	574.4	607.8
Total equity	4,019.1	4,260.8
Total liabilities and equity	\$ 14,581.7	\$ 14,534.1

See Notes to Consolidated Financial Statements

1

# **Table of Contents**

# WHITE MOUNTAINS INSURANCE GROUP, LTD. CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE INCOME (LOSS) Unaudited

(Millions, except per share amounts)  Revenues:	Septem 2011	2010	Septem	
		2010	2011	2010
Г				
Earned insurance and reinsurance premiums \$	490.6	\$ 498.1	\$ 1,436.7	\$ 1,772.9
Net investment income	42.8	48.9	138.1	157.4
Net realized and unrealized investment gains	1.7	11.6	<b>78.3</b>	104.7
Other revenue	(33.8)	(7.6)	(68.6)	(21.1
Total revenues	501.3	551.0	1,584.5	2,013.9
Expenses:				
Loss and loss adjustment expenses	277.1	253.4	914.0	1,175.9
Insurance and reinsurance acquisition expenses	107.0	102.1	296.2	370.9
Other underwriting expenses	67.0	75.0	211.5	245.9
General and administrative expenses	36.2	32.1	120.4	119.5
Interest expense on debt	12.8	13.2	38.8	44.2
Total expenses	500.1	475.8	1,580.9	1,956.4
Pre-tax income (loss) from continuing operations	1.2	75.2	3.6	57.5
Income tax (expense) benefit	.6	(11.7)	(1.9)	(10.3
Net income (loss) from continuing operations	1.8	63.5	1.7	47.2
Loss from sale of discontinued operations, net of tax	(18.2)	_	(18.2)	_
Net (loss) income from discontinued operations, net of tax	(11.9)	3.6	(10.7)	1.2
(Loss) income before equity in earnings of unconsolidated affiliates	(28.3)	67.1	(27.2)	48.4
Equity in earnings of unconsolidated affiliates	1.5	7.9	16.1	1.6
Net (loss) income	(26.8)	75.0	(11.1)	50.0
Net income (loss) attributable to noncontrolling interests	11.0	(24.3)	(21.2)	(36.4
Net (loss) income attributable to White Mountains' common				
shareholders	(15.8)	50.7	(32.3)	13.6
Comprehensive income (loss), net of tax:				
Change in equity in net unrealized gains from investments in				
unconsolidated affiliates	55.1	43.1	77.2	126.4
Change in foreign currency translation and other	(81.8)	126.9	(25.3)	45.9

Comprehensive (loss) income	(42.5)	220.7	19.6	185.9
Comprehensive income attributable to noncontrolling interests	_			_
Comprehensive (loss) income attributable to White Mountains' common				
shareholders	\$ (42.5)	\$ 220.7	\$ 19.6	\$ 185.9
Income (loss) per share attributable to White Mountains' common				
shareholders				
Basic income (loss) per share				
Continuing operations	\$ 1.81	\$ 5.58	\$ (.43)	\$ 1.44
Discontinued operations	(3.81)	.43	(3.63)	.13
Diluted income (loss) per share				
Continuing operations	\$ 1.81	\$ 5.58	\$ (.43)	\$ 1.44
Discontinued operations	(3.81)	.43	(3.63)	.13
Dividends declared and paid per White Mountains' common share	\$ _	\$ _	\$ 1.00	\$ 1.00

See Notes to Consolidated Financial Statements

2

White Mountains' Common Shareholders' Equity

# **Table of Contents**

# WHITE MOUNTAINS INSURANCE GROUP, LTD. CONSOLIDATED STATEMENTS OF CHANGES IN EQUITY Unaudited

(Millions)		Common areholders' equity	sha p	ommon ares and aid-in arplus		Retained earnings	comp inco	um. other prehensive ome (loss), fter-tax	Non- controlling interest
Balances at January 1, 2011	\$	3,653.0	\$	1,359.0	\$	2,175.6	\$	118.4	\$ 607.8
Net (loss) income		(32.3)		_		(32.3)		_	21.2
Other comprehensive income, after-tax		52.0				_		52.0	_
Dividends declared on common shares		(8.0)		_		(8.0)		_	_
Dividends to noncontrolling interests		_				_		_	(47.2)
Repurchases and retirements of common shares		(229.8)		(98.5)		(131.3)		_	_
Issuances of common shares		.9		.9		_		_	_
Net distributions to noncontrolling interests		_		_		_		_	(7.7)
Amortization of restricted share and option awards		8.9		8.9		_		_	.3
Balances at September 30, 2011	\$	3,444.7	\$	1,270.3	\$	2,004.0	\$	170.4	\$ <b>574.4</b>
	-		=						
		W			on Shar	reholders' Equit			
(Millions)	-	Common areholders'	sha p	ommon ires and aid-in irplus		Retained earnings	comp inco	um. other prehensive ome (loss), fter-tax	Non- controlling interest
(Millions) Balances at January 1, 2010	-		sha p	res and		Retained earnings 2,215.9	comp inco af	prehensive ome (loss), fter-tax	\$
(Millions) Balances at January 1, 2010	sha	reholders' equity	sha p sı	res and aid-in urplus		earnings	comp inco	prehensive ome (loss),	 controlling interest
	sha	reholders' equity	sha p sı	res and aid-in urplus		earnings	comp inco af	prehensive ome (loss), fter-tax	 controlling interest
Balances at January 1, 2010	sha	equity 3,657.4	sha p sı	res and aid-in urplus		earnings 2,215.9	comp inco af	prehensive ome (loss), fter-tax	 controlling interest 684.1
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17	sha	reholders' equity 3,657.4	sha p sı	res and aid-in urplus		earnings 2,215.9 (.4)	comp inco af	prehensive ome (loss), fter-tax (3.5)	 controlling interest 684.1
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer	sha	(.4)	sha p sı	res and aid-in urplus		2,215.9 (.4) (4.4)	comp inco af	prehensive ome (loss), fter-tax (3.5)	 controlling interest 684.1 (22.8)
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer  Net (loss) income	sha	17. (4) (10.4) 13.6	sha p sı	res and aid-in urplus		2,215.9 (.4) (4.4)	comp inco af	prehensive ome (loss), fter-tax (3.5)  (6.0)	 controlling interest 684.1 (22.8)
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer  Net (loss) income  Other comprehensive income, after-tax	sha	(.4) (10.4) 13.6 172.3	sha p sı	res and aid-in urplus		2,215.9 (.4) (4.4) 13.6	comp inco af	prehensive ome (loss), fter-tax (3.5)  (6.0)	 controlling interest 684.1 (22.8)
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer  Net (loss) income  Other comprehensive income, after-tax  Dividends declared on common shares	sha	(.4) (10.4) 13.6 172.3	sha p sı	res and aid-in urplus		2,215.9 (.4) (4.4) 13.6	comp inco af	prehensive ome (loss), fter-tax (3.5)  (6.0)	 (22.8) — 36.4 — —
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer  Net (loss) income  Other comprehensive income, after-tax  Dividends declared on common shares  Dividends to noncontrolling interests	sha	(4) (10.4) 13.6 172.3 (8.8)	sha p sı	res and aid-in urplus 1,445.0 — — — — — — — — — — — — — — — — — — —		(.4) (4.4) 13.6 — (8.8)	comp inco af	prehensive ome (loss), fter-tax (3.5)  (6.0)	 (22.8) — 36.4 — —
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer  Net (loss) income  Other comprehensive income, after-tax  Dividends declared on common shares  Dividends to noncontrolling interests  Repurchases and retirements of common shares	sha	(4) (10.4) 13.6 172.3 (8.8) (178.2)	sha p sı	res and aid-in urplus 1,445.0 — — — — — — — — — — — — — — — — — — —		(.4) (4.4) 13.6 — (8.8)	comp inco af	prehensive ome (loss), fter-tax (3.5)  (6.0)	 (22.8) — 36.4 — —
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer  Net (loss) income  Other comprehensive income, after-tax  Dividends declared on common shares  Dividends to noncontrolling interests  Repurchases and retirements of common shares  Issuances of common shares	sha	(4) (10.4) 13.6 172.3 (8.8) (178.2)	sha p sı	res and aid-in urplus 1,445.0 — — — — — — — — — — — — — — — — — — —		(.4) (4.4) 13.6 — (8.8)	comp inco af	prehensive ome (loss), fter-tax (3.5)  (6.0)	 (22.8) (22.8) 36.4 (80.5)
Balances at January 1, 2010  Cumulative effect adjustment - ASU 2009-17  Tax basis change due to intercompany transfer  Net (loss) income  Other comprehensive income, after-tax  Dividends declared on common shares  Dividends to noncontrolling interests  Repurchases and retirements of common shares  Issuances of common shares  Net distributions to noncontrolling interests	sha	(4) (10.4) 13.6 172.3 (8.8) — (178.2) .8	sha p sı	res and aid-in urplus 1,445.0		(.4) (4.4) 13.6 — (8.8)	comp inco af	prehensive ome (loss), fter-tax (3.5)  (6.0)	 (22.8) (22.8) 36.4 (80.5) (13.5)

See Notes to Consolidated Financial Statements

3

# **Table of Contents**

WHITE MOUNTAINS INSURANCE GROUP, LTD. CONSOLIDATED STATEMENTS OF CASH FLOWS Unaudited

> **Nine Months Ended** September 30,

(Millions)

2011 2010

Cash flows from operations:		
Net (loss) income	\$ (11.1)	\$ 50.0
Charges (credits) to reconcile net loss to net cash used for operations:		
Net realized and unrealized investment gains	(78.3)	(104.7)
Net loss (income) from discontinued operations	10.7	(1.2)
Net loss on sale of discontinued operation - AutoOne	18.2	_
Other operating items:		
Net change in loss and loss adjustment expense reserves	244.5	(237.4)
Net change in reinsurance recoverable on paid and unpaid losses	(209.2)	255.7
Net change in unearned insurance and reinsurance premiums	86.4	39.5
Net change in funds held by ceding companies	(17.2)	14.5
Net change in deferred acquisition costs	(31.2)	36.1
Net change in ceded unearned premiums	<b>76.5</b>	(157.1)
Net change in funds held under reinsurance treaties	(46.0)	(15.1)
Net change in insurance and reinsurance premiums receivable	(85.4)	(100.7)
Net change in ceded reinsurance payable	(76.1)	198.7
Net change in other assets and liabilities, net	141.2	(26.5)
	22.0	(40.2)
Net cash provided from (used for) operations - continuing operations	23.0	(48.2)
Net cash provided from operations - discontinued operations	17.5	 27.2
Net cash provided from (used for) operations	40.5	(21.0)
		( 12)
Cash flows from investing activities:		
Net change in short-term investments	112.6	139.3
Sales of fixed maturity and convertible fixed maturity investments	2,905.7	1,625.4
Maturities, calls and paydowns of fixed maturity and convertible fixed maturity investments	1,015.3	1,426.0
Sales of common equity securities	125.4	110.2
Distributions and redemptions of other long-term investments	90.0	26.5
Purchases of other long-term investments	(30.6)	(71.0)
Contribution to discontinued operations	(97.1)	(16.8)
Purchases of common equity securities	(189.0)	(211.0)
Purchases of fixed maturity and convertible fixed maturity investments	(3,497.6)	(2,663.8)
Purchases of consolidated and unconsolidated affiliates, net of cash acquired	_	(4.9)
Sales of consolidated and unconsolidated affiliates	_	166.6
Net change in unsettled investment purchases and sales	54.0	(18.7)
Net acquisitions of property and equipment	(3.3)	 (3.2)
Net cash provided from investing activities - continuing operations	485.4	504.6
Net cash provided from (used for) investing activities - discontinued operations		
ivet cash provided from (used for) investing activities - discontinued operations	173.7	 (21.7)
Net cash provided from investing activities	659.1	482.9
Cash flows from financing activities:		
Repayment of debt	(161.6)	(14.0)
Repurchase of debt	_	(197.3)
Cash dividends paid to the Company's common shareholders	(8.0)	(8.8)
Cash dividends paid to OneBeacon Ltd.'s noncontrolling common shareholders	(37.9)	(71.2)
Cash dividends paid on Sirius Group Preference Shares	(9.4)	(9.4)
Common shares repurchased	(229.7)	(178.2)
OneBeacon Ltd. common shares repurchased and retired	_	(10.5)
Proceeds from issuances of common shares	.9	.8
Net cash used for financing activities - continuing operations	(445.7)	(488.6)
Net cash provided from (used for) financing activities - discontinued operations	<u></u> _	
Net cash used for financing activities	(445.7)	 (488.6)
Effect of exchange rate changes on cash	(2.0)	6.0
Not change in each during the period	251.9	(20.7)
Net change in cash during the period		(20.7)
Net change in cash from discontinued operations  Cash balances at beginning of period (excludes restricted cash balances of \$286.7 and \$217.1 and	(191.2)	(5.5)
AutoOne cash of \$4.7 and \$3.9)	103.7	120.0
Cash balances at end of period (excludes restricted cash balances of \$393.3 and \$273.1 and AutoOne		
cash of \$0(1) and \$4.4)	<b>\$</b> 164.4	\$ 93.8
Supplemental cash flows information:		
Interest paid	\$ (38.5)	\$ (49.6)
Net income tax payments to national governments	<b>\$</b> (6.4)	\$ (10.6)

<sup>(1)</sup> AutoOne cash of 5.5 is included in assets held for sale at September 30, 2011.

See Notes to Consolidated Financial Statements

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

#### **Note 1. Summary of Significant Accounting Policies**

#### **Basis of Presentation**

These interim consolidated financial statements include the accounts of White Mountains Insurance Group, Ltd. (the "Company" or the "Registrant") and its subsidiaries (collectively, with the Company, "White Mountains") and have been prepared in accordance with generally accepted accounting principles in the United States ("GAAP"). The Company is an exempted Bermuda limited liability company whose principal businesses are conducted through its property and casualty insurance and reinsurance subsidiaries and affiliates. The Company's headquarters is located at 14 Wesley Street, Hamilton, Bermuda HM 11, its principal executive office is located at 80 South Main Street, Hanover, New Hampshire 03755-2053 and its registered office is located at Clarendon House, 2 Church Street, Hamilton, Bermuda HM 11. White Mountains' reportable segments are OneBeacon, Sirius Group (formerly White Mountains Re) and Other Operations. As discussed further in **Note 2**, on October 7, 2011 White Mountains completed its sale of Esurance Holdings, Inc. and its subsidiaries ("Esurance Insurance") and Answer Financial Inc. and its subsidiaries ("AFI") (collectively, "Esurance") and the business that Esurance Insurance cedes to Sirius Group. Esurance has been presented as discontinued operations and assets and liabilities held for sale in the financial statements. Prior year amounts have been reclassified to conform to the current year's presentation. (See **Note 14** for discontinued operations and assets and liabilities held for sale).

The OneBeacon segment consists of OneBeacon Insurance Group, Ltd. ("OneBeacon Ltd."), an exempted Bermuda limited liability company that owns a family of U.S. based property and casualty insurance companies (collectively "OneBeacon"), most of which operate in a multi-company pool. OneBeacon is a specialty property and casualty insurance writer that offers a wide range of insurance products through independent agencies, regional and national brokers, wholesalers and managing general agencies. As of September 30, 2011 and December 31, 2010, White Mountains owned 75.5% and 76.0% of OneBeacon Ltd.'s outstanding common shares. As discussed further in **Note 2**, OneBeacon entered into a definitive agreement on August 30, 2011 to sell the AutoOne business. Accordingly, effective for September 30, 2011, AutoOne has been presented as discontinued operations and assets and liabilities held for sale in the financial statements. Prior year income statement and cash flow amounts have been reclassified to conform to the current year's presentation.

The Sirius Group segment (previously defined as the White Mountains Re segment) consists of Sirius International Insurance Group, Ltd. (formerly White Mountains Re Ltd.), an exempted Bermuda limited liability company, and its subsidiaries (collectively, "Sirius Group"). Sirius Group provides insurance and reinsurance products for property, accident and health, aviation and space, trade credit, marine, casualty, and agriculture and certain other exposures on a worldwide basis through its subsidiaries, Sirius International Insurance Corporation ("Sirius International", formerly WMRe Sirius) and Sirius America Insurance Company ("Sirius America", formerly WMRe America). Sirius Group also specializes in the acquisition and management of run-off insurance and reinsurance companies both in the United States and internationally through its White Mountains Re Solutions division. Sirius Group also includes Scandinavian Reinsurance Company, Ltd. ("Scandinavian Re") and Central National Insurance Company of Omaha ("Central National"), which was acquired during the first quarter of 2010 (see **Note 2**), both of which are in run-off.

White Mountains' Other Operations segment consists of the Company and its intermediate holding companies, its wholly-owned investment management subsidiary, White Mountains Advisors LLC ("WM Advisors"), White Mountains' investment in common shares and warrants to purchase common shares of Symetra Financial Corporation ("Symetra"), the consolidated results of Tuckerman Capital, LP fund ("Tuckerman Fund I"), White Mountains' variable annuity reinsurance business, White Mountains Life Reinsurance (Bermuda) Ltd. ("WM Life Re"), which is in run-off, as well as other entities not included in other segments. The Other Operations segment also included White Mountains' investment in Lightyear Delos Acquisition Corporation ("Delos") prior to its disposition in December 2010.

White Mountains' discontinued operations consist of Esurance Insurance, AFI, the business that Esurance Insurance ceded to Sirius Group and AutoOne. Esurance Insurance writes personal auto insurance directly to customers in 30 states through its website and over the phone and also sells other lines of personal insurance for unaffiliated insurance companies. Esurance Insurance also writes personal auto policies through select online agents and provides other insurance products through partnerships with industry leading online providers. Esurance Insurance earns commissions and fees by referring to unaffiliated insurance companies those shoppers that it cannot underwrite because of pricing or underwriting eligibility. AFI is one of the largest independent personal insurance agencies in the United States. AFI sells insurance online and through call centers for both Esurance Insurance and unaffiliated companies utilizing a comparison quoting platform. AutoOne was formed by OneBeacon in 2001 to provide products and services to automobile assigned risk markets primarily in New York and New Jersey.

All significant intercompany transactions have been eliminated in consolidation. These interim financial statements include all adjustments considered necessary by management to fairly present the financial position, results of operations and cash flows of White Mountains that are of a normal recurring nature. These interim financial statements may not be indicative of financial results for the full year and should be read in conjunction with the Company's 2010 Annual Report on Form 10-K.

5

#### **Table of Contents**

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Refer to the Company's 2010 Annual Report on Form 10-K for a complete discussion regarding White Mountains' significant accounting policies.

#### Noncontrolling Interests

Noncontrolling interests consist of the ownership interests of noncontrolling parties in consolidated subsidiaries and are presented separately as a component of equity on the balance sheet.

The percentage of the noncontrolling ownership interests in OneBeacon Ltd. at September 30, 2011 and December 31, 2010 was 24.5% and 24.0%. The increase in the noncontrolling ownership percentage during the first nine months of 2011 was the result of the issuance of 630,000 restricted common shares of OneBeacon Ltd. to its CEO in May 2011 (see **Note 12**).

On May 24, 2007, Sirius International Group, Ltd. ("Sirius International Group" formerly known as White Mountains Re Group, Ltd.), an intermediate holding company of Sirius Group, issued 250,000 non-cumulative perpetual preference shares with a \$1,000 per share liquidation preference (the "Sirius

Group Preference Shares"). Proceeds of \$245.7 million, net of \$4.3 million of issuance costs and commissions, were received from the issuance. The Sirius Group Preference Shares are included in noncontrolling interests on the balance sheet.

At September 30, 2011 and December 31, 2010, the noncontrolling equity interest in limited partnerships that are consolidated with White Mountains (Tuckerman Fund I, the Prospector Offshore Fund and the Prospector Turtle Fund) was \$51.8 million and \$59.7 million. At September 30, 2011 and December 31, 2010, the noncontrolling equity interest in A.W.G. Dewar Inc, a subsidiary of OneBeacon, was \$2.2 million and \$3.1 million.

#### **Recently Adopted Changes in Accounting Principles**

Disclosures about Fair Value Measurements

White Mountains adopted Accounting Standards Update ("ASU") 2010-06, *Improving Disclosures about Fair Value Measurements* (included in Accounting Standards Codification ("ASC") 820-10), effective March 31, 2010. The ASU clarifies existing disclosure requirements for fair value measurements and requires the disclosure of the amounts and nature of transfers in and out of Level 1 and Level 2 measurements; purchase, sale, issuance and settlement activity for Level 3 measurements presented on a gross rather than a net basis; fair value measurements by Level presented on a more disaggregated basis, by asset or liability class; and more detailed disclosures about inputs and valuation techniques for Level 2 and Level 3 measurements for interim and annual reporting periods. White Mountains has expanded its fair value disclosures to meet the requirements of the ASU effective for the period ended March 31, 2010 (see **Note 5**).

Transfers of Financial Assets and Amendments to FIN 46R

On June 12, 2009, the FASB issued ASU 2009-16, *Accounting for Transfers of Financial Assets*, *an amendment of FASB Statement No. 140* (included in ASC 860) and ASU 2009-17, *Amendments to FIN46(R)* (included in ASC 810). Both ASU 2009-16 and ASU 2009-17 became effective as of the beginning of the first annual reporting period that began after November 15, 2009. White Mountains adopted the new guidance on January 1, 2010.

ASU 2009-16 eliminates the concept of a qualifying special-purpose entity ("QSPE"). Under the new guidance, the appropriateness of de-recognition of assets held by an entity formerly considered a QSPE is evaluated based on whether or not the transferor has surrendered control of the transferred assets. The evaluation must consider any continuing involvement by the transferor. White Mountains did not have any entities that were considered a QSPE under guidance prior to the amendments to ASC 860 and there was no effect resulting from adoption.

ASU 2009-17 amends ASC 810-10 to clarify the application of consolidation accounting for entities for which the controlling financial interest might not be solely indentified through voting rights. Under the new guidance a variable interest represents a controlling financial interest in a variable interest entity ("VIE") when it has both of the following: (1) the power to direct the activities of a variable interest entity that most significantly impact the entity's economic performance and (2) the obligation to absorb losses of the entity that could potentially be significant to the VIE or the right to receive benefits from the entity that could potentially be significant to the VIE. A reporting entity must assess whether it has an implicit financial responsibility to ensure that a VIE operates as designed when determining if it has the power to direct the activities of the VIE that most significantly affect the entity's economic performance. ASC 810-10 as amended requires ongoing reassessments of whether a reporting entity is the primary beneficiary of a VIE.

Upon adoption, White Mountains determined that its ownership interest in Tuckerman Fund II did not meet the criteria for consolidation under the revised guidance for variable interest entities and, accordingly, effective January 1, 2010, White Mountains deconsolidated its investment in Tuckerman Fund II. Upon deconsolidation, White Mountains made the fair value election for its investment in Tuckerman Fund II and recognized an adjustment to decrease opening retained earnings by \$0.4 million.

6

#### **Table of Contents**

#### **Recently Issued Accounting Pronouncements**

Goodwill Impairment

On September 15, 2011, the FASB issued ASU 2011-08, *Testing Goodwill for Impairment* (ASC 350). The ASU amends the guidance that requires an entity to test goodwill for impairment on at least an annual basis using a two-step quantitative test. The new guidance permits an entity to first assess facts and circumstances to determine whether it is more likely than not that the fair value of a reporting unit is less than its carrying amount. If the entity determines on the basis of this assessment that it is not more likely than not that the fair value of the reporting unit is less than its carrying amount, then performance of the two-step quantitative test is not required. ASU 2011-08 is effective for annual and interim goodwill impairment tests performed for fiscal years beginning after December 15, 2011. Early adoption is permitted. White Mountains expects that adoption will have no material effect upon its financial position, results of operations or cash flows.

Comprehensive Income

On June 16, 2011, the FASB issued ASU 2011-05, *Presentation of Comprehensive Income* (ASC 220). The ASU requires presentation of the components of net income, the total of other comprehensive income and total comprehensive income to be presented on a single statement. The option of presenting other comprehensive income in the statement of changes in equity has been eliminated by the ASU. ASU 2011-05 is effective for fiscal years and interim periods within those fiscal years ending after December 15, 2011. White Mountains already presents comprehensive income in accordance with the requirements of the ASU, so there will be no effect upon adoption.

Fair Value Measurements and Disclosures

On May 12, 2011, the FASB issued ASU 2011-04, *Amendments to Achieve Common Fair Value Measurement and Disclosure Requirements in U.S. GAAP and IFRS* (ASC 820). The ASU clarifies existing guidance with respect to the concepts of highest and best use and valuation premise and measuring instruments classified within a reporting entity's shareholders' equity. The ASU also clarifies disclosure requirements, requiring disclosure of quantitative information about unobservable inputs used in Level 3 fair value measurements. The ASU also amends existing guidance. In circumstances where a reporting entity manages a portfolio of financial assets and liabilities based on the net market and counterparty credit risk exposures, the ASU permits determination of the fair value of those instruments to be based on the net risk exposure. In addition, the ASU permits the application of premiums or discounts to be applied in

a fair value measurement to the extent that market participants would consider them in valuing the financial instruments. The ASU also expands the required disclosures for Level 3 measurements, requiring that reporting entities provide a narrative description of the sensitivity of Level 3 fair value measurements to changes in unobservable inputs and the interrelationships between those inputs, if any. ASU 2011-04 is effective for interim and annual periods beginning after December 15, 2011. White Mountains is currently evaluating the effect the adoption of ASU 2011-04 will have on its financial position and results of operations.

#### Policy Acquisition Costs

On October 13, 2010, the FASB issued ASU 2010-26, *Accounting for Costs Associated with Acquiring or Renewing Insurance Contracts* (ASC 944). The new standard changes the types of policy acquisition costs that are eligible for deferral. Specifically, the new guidance limits deferrable costs to those that are incremental direct costs of contract acquisition and certain costs related to acquisition activities performed by the insurer, such as underwriting, policy issuance and processing, medical and inspection costs and sales force contract selling. The ASU defines incremental direct costs as those costs that result directly from and were essential to the contract acquisition and would not have been incurred absent the acquisition. Accordingly, under the new guidance, deferrable acquisition costs are limited to costs related to successful contract acquisitions. Acquisition costs that are not eligible for deferral are to be charged to expense in the period incurred.

ASU 2010-26 is effective for interim periods and annual fiscal years beginning after December 15, 2011 and may be applied prospectively or retrospectively. White Mountains expects that adoption will not have a material effect upon its financial position, results of operations or cash flows.

7

#### **Table of Contents**

#### **Note 2. Significant Transactions**

#### Sale of Esurance

On October 7, 2011, White Mountains completed its sale of Esurance Insurance and AFI to The Allstate Corporation (see Note 15).

#### **Self Tender Offer**

On September 21, 2011, White Mountains completed a "modified Dutch auction" self-tender offer whereby White Mountains repurchased 327,872 of its common shares at \$415 per share. The total cost of the tender offer was \$136.5 million, including \$0.4 million of fees and expenses.

#### Sale of AutoOne

On August 30, 2011, OneBeacon entered into a definitive agreement (the "Purchase Agreement") to sell the AutoOne business to Interboro Holdings, Inc. ("Interboro"). OneBeacon formed AutoOne in 2001 to provide products and services to automobile assigned risk markets primarily in New York and New Jersey. OneBeacon will transfer to the buyer AutoOne Insurance Company ("AOIC") and AutoOne Select Insurance Company ("AOSIC"), which will contain the assets, liabilities (including loss reserves and unearned premiums), and the equity capital of the business, and will transfer substantially all of the AutoOne infrastructure including staff, systems and office space. The transaction is expected to close in the fourth quarter of 2011, subject to regulatory approvals. As a result of entering into the Purchase Agreement, AutoOne is now reported as discontinued operations. See **Note 14**.

During the three months ended September 30, 2011, OneBeacon recorded an after-tax charge of \$18.2 million reflecting the estimated loss on sale of the AutoOne business.

#### **OneBeacon Personal Lines Sale**

In July 2010, OneBeacon completed the sale of its traditional personal lines business (the "Personal Lines Transaction") to Tower Group, Inc. ("Tower") for consideration of \$166.6 million. The Personal Lines Transaction included two insurance companies, York Insurance Company of Maine ("York") and Massachusetts Homeland Insurance Company ("MHIC"), through which the majority of the traditional personal lines business was written on a direct basis. Subsequent to the transaction, OneBeacon cedes to Tower, on a 100% quota share basis, traditional personal lines business not directly written by York and MHIC; and OneBeacon assumes, on a 100% quota share basis, non-traditional personal lines business written directly by York. The Personal Lines Transaction also included two attorneys-in-fact managing the reciprocal insurance exchanges ("reciprocals") that wrote the traditional personal lines business in New York and New Jersey, the surplus notes issued by the New York and New Jersey reciprocals, and the remaining renewal rights to certain other traditional personal lines insurance policies. The sale of the two attorneys-in-fact and the transfer of the surplus notes triggered deconsolidation of the reciprocals by White Mountains. OneBeacon and Tower also entered into a Transition Services Agreement ("TSA"), pursuant to which OneBeacon provides certain services to Tower during a three-year term.

#### **Acquisition of Central National**

On February 26, 2010, Sirius Group acquired Central National for \$5 million in cash. Central National ceased writing business in 1989 and was operated under the control of the Nebraska Department of Insurance since 1990. The transaction resulted in a gain of \$12.8 million recorded in other revenues.

8

#### **Table of Contents**

# Note 3. Loss and Loss Adjustment Expense Reserves

The following table summarizes the loss and loss adjustment expense ("LAE") reserve activities of White Mountains' insurance and reinsurance subsidiaries for the three and nine months ended September 30, 2011 and 2010:

	Septem		September 30,			
Millions	 2011	2010		2011		2010
Gross beginning balance	\$ 5,642.4	\$ 6,198.1	\$	5,736.8	\$	6,379.2
Less beginning reinsurance recoverable on unpaid losses	 (2,193.0)	(2,748.3)	)	(2,344.0)		(2,771.5)
Net loss and LAE reserves	3,449.4	3,449.8		3,392.8		3,607.7
	(=0.0)	(0.4 <b>-</b> )		(50.0)		(0= 0)
Less: Beginning net loss and LAE reserves for AutoOne (1)	(58.6)	(81.7)	)	(68.2)		(95.8)
Loss and LAE reserves acquired — Central National	_	_		_		17.6
Loss and LAE reserves sold — OneBeacon Personal Lines (2)	_	_		_		(231.0)
Loss and LAE incurred relating to:						
Current year losses	307.8	284.7		972.2		1,244.9
Prior year losses	 (30.7)	(31.3)	)	(58.2)		(69.0)
Total incurred losses and LAE	277.1	253.4		914.0		1,175.9
Accretion of fair value adjustment to loss and LAE reserves	2.1	2.1		6.2		6.3
Foreign currency translation adjustment to loss and LAE reserves	(25.5)	39.9		5.7		6.7
Loss and LAE paid relating to:						
Current year losses	(112.8)	(156.3)		(237.9)		(403.9)
Prior year losses	 (230.9)	(211.8)		(711.8)		(788.1)
Total loss and LAE payments	(343.7)	(368.1)	)	(949.7)		(1,192.0)
Plus: Ending net loss and LAE reserves for AutoOne (1)	_	82.9		_		82.9
Net ending balance	3,300.8	3,378.3		3,300.8		3,378.3
Plus ending reinsurance recoverable on unpaid losses	2,581.2	2,592.1		2,581.2		2,592.1
Gross ending balance	\$ 5,882.0	\$ 5,970.4	\$	5,882.0	\$	5,970.4

<sup>(1)</sup> Adjustment is to present loss and LAE reserve activities from continuing operations. At September 30, 2011, loss and LAE reserves for AutoOne have been reclassified as liabilities held for sale. Loss and LAE reserve balances for AutoOne prior to September 30, 2011 were not reclassified. See **Note 14**.

Loss and LAE incurred relating to prior year losses for the three and nine months ended September 30, 2011

During the three and nine months ended September 30, 2011, White Mountains experienced \$30.7 million and \$58.2 million of net favorable loss reserve development.

For the three and nine months ended September 30, 2011, OneBeacon had net favorable loss reserve development of \$9.8 million and \$25.0 million. The favorable loss reserve development was primarily due to lower than expected severity on losses related to professional liability lines, multiple peril liability lines and other general liability lines.

For the three and nine months ended September 30, 2011, Sirius Group had net favorable loss reserve development of \$20.9 million and \$33.2 million, primarily attributable to property lines.

Loss and LAE incurred relating to prior year losses for the three and nine months ended September 30, 2010

During the three and nine months ended September 30, 2010, White Mountains experienced \$31.3 million and \$69.0 million of net favorable loss reserve development.

For the three and nine months ended September 30, 2010, OneBeacon had net favorable loss reserve development of \$17.8 million and \$42.1 million. The favorable loss reserve development was primarily due to lower than expected severity on losses related to professional liability lines, commercial package business and other general liability lines.

For the three and nine months ended September 30, 2010, Sirius Group had net favorable loss reserve development of \$13.5 million and \$26.9 million. The favorable loss reserve development at Sirius Group was primarily due to short-tailed lines of business, primarily property, accident and health, and aviation.

9

#### **Table of Contents**

Fair value adjustment to loss and LAE reserves

In connection with purchase accounting for the acquisitions of Scandinavian Re and Stockbridge Insurance Company ("Stockbridge"), White Mountains was required to adjust loss and LAE reserves and the related reinsurance recoverables to fair value on their respective acquired balance sheets. The net reduction to loss and LAE reserves is being recognized through an income statement charge ratably with and over the period the claims are settled.

Sirius Group recognized \$2.1 million and \$6.2 million for the three and nine months ended September 30, 2011, and \$2.1 million and \$6.3 million for the three and nine months ended September 30, 2010. As of September 30, 2011, the pre-tax un-accreted adjustment was \$14.9 million.

<sup>(2)</sup> In the second quarter of 2010, \$231.0 of net loss and LAE reserves related to the Personal Lines Transaction were reclassified to liabilities held for sale. The Personal Lines Transaction closed in July 2010.

#### **Note 4. Third Party Reinsurance**

In the normal course of business, White Mountains' insurance and reinsurance subsidiaries may seek to limit losses that may arise from catastrophes or other events by reinsuring with third party reinsurers. White Mountains remains liable for risks reinsured in the event that the reinsurer does not honor its obligations under reinsurance contracts.

#### OneBeacon

At September 30, 2011, OneBeacon had \$13.7 million of reinsurance recoverables on paid losses and \$2,405.7 million (gross of \$166.6 million in purchase accounting adjustments) that will become recoverable if claims are paid in accordance with current reserve estimates. The collectability of balances due from OneBeacon's reinsurers is critical to OneBeacon's financial strength because reinsurance contracts do not relieve OneBeacon of its primary obligation to its policyholders. OneBeacon is selective with its reinsurers, placing reinsurance with only those reinsurers having a strong financial condition. OneBeacon monitors the financial strength of its reinsurers on an ongoing basis. Uncollectible amounts historically have not been significant.

The following table provides a listing of OneBeacon's top reinsurers, excluding industry pools and associations, based upon recoverable amounts, the percentage of total paid and unpaid reinsurance recoverables and the reinsurer's A.M Best Company, Inc. ("A.M. Best") rating.

Top Reinsurers (Millions)	Balance at September 30, 2011 % of To		% of Total	A.M. Best Rating(1)
National Indemnity Company and General Reinsurance Corporation (2)	\$	1,565.2	65%	A++
Hanover Insurance Company		110.1	5%	A
Tokio Marine and Nichido Fire (3)		55.8	2%	A++
Tower Insurance Company		53.7	2%	A-
Munich Reinsurance America		31.5	1%	A+

- (1) A.M. Best ratings as detailed above are: "A++" (Superior, which is the highest of fifteen ratings), "A+" (Superior, which is the second highest of fifteen ratings), "A" (Excellent, which is the fourth highest of fifteen ratings).
- (2) Includes \$198.3 of Third Party Recoverables (as defined below), which NICO (as defined below) would pay under the terms of the NICO Cover (as defined below) if they are unable to collect from third party reinsurers.
- (3) Includes \$29.3 of reinsurance recoverables from various third party reinsurers that are guaranteed by Tokio Marine and Nichido Fire under the terms of a 100% quota share reinsurance agreement between Houston General Insurance Company and Tokio Marine and Nichido Fire.

Immediately prior to White Mountains' acquisition of OneBeacon, the seller caused OneBeacon to purchase two reinsurance contracts from subsidiaries of Berkshire Hathaway Inc. ("Berkshire"): a full risk-transfer cover from National Indemnity Company ("NICO") for up to \$2.5 billion in old asbestos and environmental ("A&E") claims and certain other exposures (the "NICO Cover") and an adverse loss reserve development cover (the "GRC Cover") from General Reinsurance Corporation ("GRC") for up to \$570.0 million, comprised of \$400.0 million of adverse loss reserve development on losses occurring in years 2000 and prior and \$170.0 million of reserves ceded as of the date of the OneBeacon acquisition. The NICO Cover and GRC Cover, which were contingent on and occurred contemporaneously with the OneBeacon acquisition, were put in place in lieu of a seller guarantee of loss and LAE reserves and are therefore accounted for under GAAP as a seller guarantee.

Under the terms of the NICO Cover, NICO receives the economic benefit of reinsurance recoverables ("Third Party Recoverables") from certain of OneBeacon's third party reinsurers in existence at the time the NICO Cover was executed. As a result, the Third Party Recoverables serve to protect the \$2.5 billion limit of NICO coverage for the benefit of OneBeacon. White Mountains estimates that on an incurred basis, net of Third Party Recoverables, as of September 30, 2011 it has used approximately \$2.3 billion of the coverage provided by NICO. Through September 30, 2011, \$1.4 billion of these incurred losses have been paid by NICO. Since entering into the NICO Cover, approximately 8% (\$187.7 million), of the \$2.3 billion of utilized coverage from NICO related to uncollectible Third Party Recoverables. To the extent that actual experience differs from White Mountains' estimate of ultimate A&E losses and Third Party Recoverables, future losses could utilize some or all of the protection remaining under the NICO Cover.

#### 10

#### **Table of Contents**

Pursuant to the GRC Cover, OneBeacon is not entitled to recover losses to the full contract limit if such losses are reimbursed by GRC more quickly than anticipated at the time the contract was signed. OneBeacon intends to only seek reimbursement from GRC for claims which result in payment patterns similar to those supporting its recoverables recorded pursuant to the GRC Cover. The economic cost of not submitting certain other eligible claims to GRC is primarily the investment spread between the rate credited by GRC and the rate achieved by OneBeacon on its own investments. This cost, if any, is expected to be nominal. During the three and nine months ended September 30, 2011, \$13.6 million and \$60.7 million was collected under the GRC Cover.

During the third quarter of 2011, OneBeacon completed a new ground-up study of its legacy asbestos and environmental exposures. The previous study was based on experience through 2007. Reasonable estimates of potential adverse scenarios continue to be within the \$2.5 billion reinsurance cover issued by NICO. The point estimate of incurred losses ceded to NICO increased from \$2.2 billion to \$2.3 billion.

Effective May 1, 2011, OneBeacon renewed its property catastrophe reinsurance program through April 30, 2012. The program provides coverage for OneBeacon's property business as well as certain acts of terrorism. Under the program, the first \$50.0 million of losses resulting from any single catastrophe are retained by OneBeacon. The next \$175.0 million of losses resulting from the catastrophe are reinsured with OneBeacon co-participating in the losses. OneBeacon retains 26% of the losses from \$50.0 million up to \$100.0 million and 10% of the losses from \$100.0 million up to \$175.0 million. Any loss above \$225.0 million is retained by OneBeacon. In the event of a catastrophe, OneBeacon's property catastrophe reinsurance program is reinstated for the remainder of the original contract term by paying a reinstatement premium that is based on the percentage of coverage reinstated and the original property catastrophe coverage premium.

OneBeacon had entered into a 30% quota share agreement with a group of reinsurers that ran from January 1, 2009 through December 31, 2009, and had renewed the agreement effective January 1, 2010 through July 1, 2010, the closing date of the Personal Lines Transaction. Through June 30, 2010, OneBeacon ceded \$25.6 million of written premiums from its Northeast homeowners business written through OneBeacon Insurance Company ("OBIC") and

its subsidiary companies, along with Adirondack Insurance Exchange ("Adirondack Insurance") and New Jersey Skylands Insurance Agency ("NJSIA") in New York and New Jersey, respectively.

#### Sirius Group

At September 30, 2011, Sirius Group had \$17.2 million of reinsurance recoverables on paid losses and \$342.1 million of reinsurance that will become recoverable if claims are paid in accordance with current reserve estimates. Because reinsurance contracts do not relieve Sirius Group of its obligation to its ceding companies, the collectability of balances due from its reinsurers is critical to Sirius Group's financial strength. Sirius Group monitors the financial strength of its reinsurers on an ongoing basis. The following table provides a listing of Sirius Group's top reinsurers based upon recoverable amounts, the percentage of total paid and unpaid reinsurance recoverables and the reinsurers' A.M. Best ratings.

	В	alance at		A.M. Best	
Top Reinsurers (Millions)	Septer	September 30, 2011		Rating (1)	% Collateralized
General Reinsurance Corporation	\$	43.3	12%	A++	2%
Olympus (2)		40.8	11%	NR-5	100%
Swiss Re Group		40.0	11%	A	4%
Lloyds of London(3)		31.3	9%	A	6%
Michigan Catastrophic Claims Association		14.6	4%	N/A(4)	—%

- (1) A.M. Best ratings as detailed above are: "NR-5" (Not formally followed), "A++" (Superior, which is the highest of fifteen ratings), and "A" (Excellent, which is the third highest of fifteen ratings).
- (2) Non-U.S. insurance entity. Balances are fully collateralized through funds held, letters of credit or trust agreements.
- (3) Represents the total of reinsurance recoverables due to Sirius Group from all Lloyds Syndicates.
- (4) Michigan Catastrophic Claims Association ("MCCA") is a non-profit unincorporated association of which every insurance company that sells automobile coverage in Michigan is required to be a member. A.M. Best does not rate MCCA. Sirius Group acquired its recoverable from MCCA pursuant to its acquisition of Stockbridge.

11

#### **Table of Contents**

#### **Note 5. Investment Securities**

White Mountains' invested assets consist of securities and other long-term investments held for general investment purposes. The portfolio of investment securities includes fixed maturity investments, convertible fixed maturities and equity securities which are all classified as trading securities. Trading securities are reported at fair value as of the balance sheet date. Realized and unrealized investment gains and losses on trading securities are reported pre-tax in revenues. White Mountains' investments in debt securities, including mortgage-backed and asset-backed securities, are generally valued using industry standard pricing models. Key inputs include benchmark yields, benchmark securities, reported trades, issuer spreads, bids, offers, credit ratings and prepayment speeds. Income on mortgage-backed and asset-backed securities is recognized using an effective yield based on anticipated prepayments and the estimated economic life of the securities. When actual prepayments differ significantly from anticipated prepayments, the estimated economic life is recalculated and the remaining unamortized premium or discount is amortized prospectively over the remaining economic life.

Realized investment gains and losses resulting from sales of investment securities are accounted for using the specific identification method. Premiums and discounts on all fixed maturity investments are accreted or amortized to income over the anticipated life of the investment. Short-term investments consist of money market funds, certificates of deposit and other securities which, at the time of purchase, mature or become available for use within one year. Short-term investments are carried at amortized cost, which approximated fair value as of September 30, 2011 and December 31, 2010.

Other long-term investments primarily comprise White Mountains' investments in hedge funds and private equity funds.

# **Net Investment Income**

Pre-tax net investment income for the three and nine months ended September 30, 2011 and 2010 consisted of the following:

	Three Months Ended September 30,					Nine Months Ended September 30,			
Millions	2	2011		2010	2011			2010	
Investment income:									
Fixed maturity investments	\$	38.5	\$	46.9	\$	127.0	\$	149.9	
Short-term investments		.9		.8		3.2		3.7	
Common equity securities		3.6		2.7		10.1		7.1	
Convertible fixed maturity investments		1.3		.4		3.9		4.4	
Other long-term investments		1.2		1.1		1.9		1.6	
Interest on funds held under reinsurance treaties		.2		(.5)		(.6)		(2.2)	
Total investment income	<u></u>	45.7		51.4		145.5		164.5	
Less third-party investment expenses		(2.9)		(2.5)		(7.4)		(7.1)	
Net investment income, pre-tax	\$	42.8	\$	48.9	\$	138.1	\$	157.4	

# Net Realized and Unrealized Investment Gains and Losses

	September 30,				Nine Months Ended September 30,			
Millions		2011		2010		2011		2010
Net realized investment gains (losses), pre-tax	\$	37.1	\$	31.4	\$	109.6	\$	93.7
Net unrealized investment gains (losses), pre-tax		(35.4)		(19.8)		(31.3)		11.0
Net realized and unrealized investment gains (losses), pre-tax		1.7		11.6		78.3		104.7
Income taxes attributable to realized and unrealized investment		(1.1)		(22.7)		(23.7)		(24.5)

80.2

54.6

# 12

#### **Table of Contents**

Net realized investment gains (losses)

Net realized investment gains (losses) for the three and nine months ended September 30, 2011 and 2010 consisted of the following:

	Three Mor Septem			Nine Months Ended September 30,					
Millions	2011		2010		2011		2010		
Fixed maturity investments	\$ 14.0	\$	34.7	\$	33.6	\$	68.1		
Short-term investments	_		_		_		_		
Common equity securities	23.1		(13.7)		41.0		6.7		
Convertible fixed maturity investments	(.7)		6.2		6.3		19.9		
Other long-term investments	.7		4.2		28.7		(1.0)		
Net realized investment gains (losses), pre-tax	37.1	-	31.4	-	109.6		93.7		
Income taxes attributable to realized investment gains									
(losses)	(8.0)		(4.9)		(31.7)		(20.5)		
Net realized investment gains (losses), after-tax	\$ 29.1	\$	26.5	\$	77.9	\$	73.2		

Net unrealized investment gains (losses)

The following table summarizes changes in the carrying value of investments measured at fair value:

	Three Months Ended September 30, 2011					Nine Months Ended September 30, 2011						
Millions		Net unrealized gains (losses)		Net foreign exchange gains (losses)		Total changes in fair value reflected in earnings		Net unrealized gains (losses)		Net foreign exchange gains (losses)		Total changes in fair value reflected in earnings
Fixed maturities	\$	3.5	\$	67.5	\$	71.0	\$	12.6	\$	77.7	\$	90.3
Short-term investments		(.1)		(.4)		(.5)		(.1)		(1.4)		(1.5)
Common equity securities		(101.0)		(.9)		(101.9)		(95.1)		(1.8)		(96.9)
Convertible fixed maturity investments		(11.2)		_		(11.2)		(19.3)		_		(19.3)
Other long-term investments		(1.2)		8.4		7.2		(8.4)		4.5		(3.9)
Net unrealized investment gains (losses),												
pre-tax		(110.0)		74.6		(35.4)		(110.3)		79.0		(31.3)
Income taxes attributable to unrealized												
investment gains (losses)		26.4		(19.5)		6.9		28.5		(20.5)		8.0
Net unrealized investment gains (losses),												
after-tax	\$	(83.6)	\$	55.1	\$	(28.5)	\$	(81.8)	\$	58.5	\$	(23.3)
				e Months Ended tember 30, 2010						Months Ended tember 30, 2010		
Millions		Net unrealized gains (losses)		Net foreign exchange gains		Total changes in fair value reflected in		Net unrealized gains (Josses)		Net foreign exchange gains		Total changes in fair value reflected in earnings
<u>Millions</u> Fixed maturities	\$	unrealized gains (losses)		Net foreign exchange gains (losses)	\$	changes in fair value reflected in earnings	\$	unrealized		Net foreign exchange gains (losses)	\$	changes in fair value reflected in earnings
		unrealized gains	Sep	Net foreign exchange gains	\$	changes in fair value reflected in	\$	unrealized gains (losses)	Sept	Net foreign exchange gains	\$	changes in fair value reflected in
Fixed maturities Short-term investments		unrealized gains (losses)	Sep	Net foreign exchange gains (losses) (81.2)	\$	changes in fair value reflected in earnings (77.9)	\$	unrealized gains (losses)	Sept	Net foreign exchange gains (losses) (53.6)	\$	changes in fair value reflected in earnings (17.2)
Fixed maturities		unrealized gains (losses)	Sep	Net foreign exchange gains (losses)	\$	changes in fair value reflected in earnings (77.9) 1.4	\$	unrealized gains (losses) 36.4	Sept	Net foreign exchange gains (losses)	\$	changes in fair value reflected in earnings (17.2)
Fixed maturities Short-term investments Common equity securities Convertible fixed maturity investments		unrealized gains (losses)  3.3  64.3	Sep	Net foreign exchange gains (losses) (81.2)	\$	changes in fair value reflected in earnings (77.9)  1.4 53.5	\$	unrealized gains (losses)  36.4  —  31.8	Sept	Net foreign exchange gains (losses) (53.6)	\$	changes in fair value reflected in earnings (17.2) .6 22.1
Fixed maturities Short-term investments Common equity securities		unrealized gains (losses)  3.3  64.3  2.8	Sep	Net foreign exchange gains (losses)  (81.2)  1.4  (10.8)	\$	changes in fair value reflected in earnings (77.9) 1.4 53.5 2.8	\$	unrealized gains (losses)  36.4  — 31.8 (8.3)	Sept	Net foreign exchange gains (losses) (53.6) .6 (9.7)	\$	changes in fair value reflected in earnings (17.2) .6 22.1 (8.3)
Fixed maturities Short-term investments Common equity securities Convertible fixed maturity investments Other long-term investments		unrealized gains (losses)  3.3  64.3  2.8	Sep	Net foreign exchange gains (losses)  (81.2)  1.4  (10.8)	\$	changes in fair value reflected in earnings (77.9) 1.4 53.5 2.8	\$	unrealized gains (losses)  36.4  — 31.8 (8.3)	Sept	Net foreign exchange gains (losses) (53.6) .6 (9.7)	\$	changes in fair value reflected in earnings (17.2) .6 22.1 (8.3)
Fixed maturities Short-term investments Common equity securities Convertible fixed maturity investments Other long-term investments Net unrealized investment gains (losses),		3.3 — 64.3 2.8 9.0	Sep	Net foreign exchange gains (losses)  (81.2)  1.4  (10.8)  (8.6)	\$	changes in fair value reflected in earnings (77.9) 1.4 53.5 2.8 .4	\$	unrealized gains (losses)  36.4  —  31.8  (8.3)  20.5	Sept	ember 30, 2010  Net foreign exchange gains (losses)  (53.6) .6 (9.7) — (6.7)	\$	changes in fair value reflected in earnings (17.2)622.1 (8.3)13.8
Fixed maturities Short-term investments Common equity securities Convertible fixed maturity investments Other long-term investments Net unrealized investment gains (losses), pre-tax		3.3 — 64.3 2.8 9.0	Sep	Net foreign exchange gains (losses)  (81.2)  1.4  (10.8)  (8.6)	\$	changes in fair value reflected in earnings (77.9) 1.4 53.5 2.8 .4	\$	unrealized gains (losses)  36.4  —  31.8  (8.3)  20.5	Sept	ember 30, 2010  Net foreign exchange gains (losses)  (53.6) .6 (9.7) — (6.7)	\$	changes in fair value reflected in earnings (17.2)622.1 (8.3)13.8
Fixed maturities Short-term investments Common equity securities Convertible fixed maturity investments Other long-term investments Net unrealized investment gains (losses), pre-tax Income taxes attributable to unrealized		3.3 (losses) 64.3 2.8 9.0	Sep	tember 30, 2010  Net foreign exchange gains (losses)  (81.2) 1.4 (10.8) — (8.6)	\$	changes in fair value reflected in earnings (77.9) 1.4 53.5 2.8 .4 (19.8)	\$	unrealized gains (losses)  36.4   31.8 (8.3) 20.5	Sept	ember 30, 2010  Net foreign exchange gains (losses)  (53.6) .6 (9.7) — (6.7)	\$	changes in fair value reflected in earnings (17.2)622.1 (8.3)13.8
Fixed maturities Short-term investments Common equity securities Convertible fixed maturity investments Other long-term investments Net unrealized investment gains (losses), pre-tax Income taxes attributable to unrealized investment gains (losses)		3.3 (losses) 64.3 2.8 9.0	Sep	tember 30, 2010  Net foreign exchange gains (losses)  (81.2) 1.4 (10.8) — (8.6)	\$	changes in fair value reflected in earnings (77.9) 1.4 53.5 2.8 .4 (19.8)	\$	unrealized gains (losses)  36.4   31.8 (8.3) 20.5	Sept	ember 30, 2010  Net foreign exchange gains (losses)  (53.6) .6 (9.7) — (6.7)	\$	changes in fair value reflected in earnings (17.2)622.1 (8.3)13.8

# **Table of Contents**

The following table summarizes the amount of total pre-tax gains (losses) included in earnings attributable to unrealized investment gains (losses) for Level 3 investments for the three and nine months ended September 30, 2011 and 2010:

13

	Three Mon Septem			ded		
Millions	 2011	2010		2011	2010	
Fixed maturities	\$ (13.5)	\$ 3.3	\$	(14.9)	\$	9.5
Common equity securities	(17.5)	2.5		(17.0)		(20.0)

Convertible fixed maturities	_	_	_	_
Other long-term investments	(3.2)	8.5	(6.5)	37.4
Total unrealized investment gains (losses), pre-tax -			·	
Level 3 investments	\$ (34.2)	\$ 14.3	\$ (38.4)	\$ 26.9

#### **Investment Holdings**

The cost or amortized cost, gross unrealized investment gains and losses, net foreign currency gains and losses and carrying values of White Mountains' fixed maturity investments as of September 30, 2011 and December 31, 2010, were as follows:

	September 30, 2011									
Millions	Cost or amortized cost			Gross unrealized gains		Gross unrealized losses		Net foreign currency gains (losses)		Carrying value
U.S. Government and agency obligations	\$	347.1	\$	5.8	\$	(.2)	\$	.4	\$	353.1
Debt securities issued by corporations		1,990.0		76.6		(11.9)		(8.)		2,053.9
Municipal obligations		2.4		_		_		_		2.4
Mortgage-backed and asset-backed securities		2,200.8		29.8		(4.9)		14.9		2,240.6
Foreign government, agency and provincial obligations		653.0		14.1		(.2)		(.9)		666.0
Preferred stocks		82.3		3.3		(11.0)		_		74.6
Total fixed maturity investments(1)	\$	5,275.6	\$	129.6	\$	(28.2)	\$	13.6	\$	5,390.6

(1) Carrying value includes \$102.9 that is classified as assets held for sale relating to AutoOne discontinued operations.

	December 31, 2010										
Millions	Cost or amortized cost		Gross unrealized gains		Gross unrealized losses		Net foreign currency gains (losses)			Carrying value	
U.S. Government and agency obligations	\$	385.9	\$	13.6	\$	(8.)	\$	_	\$	398.7	
Debt securities issued by corporations		2,149.2		87.9		(21.1)		(37.9)		2,178.1	
Municipal obligations		3.3		.1		(.1)		_		3.3	
Mortgage-backed and asset-backed securities		2,082.0		18.2		(15.6)		(12.9)		2,071.7	
Foreign government, agency and provincial obligations		1,053.6		7.7		(6.6)		(8.3)		1,046.4	
Preferred stocks		81.9		6.1		_		(.1)		87.9	
Total fixed maturity investments	\$	5,755.9	\$	133.6	\$	(44.2)	\$	(59.2)	\$	5,786.1	

The cost or amortized cost, gross unrealized investment gains and losses, net foreign currency gains and losses and carrying values of White Mountains' common equity securities, convertible fixed maturities and other long-term investments as of September 30, 2011 and December 31, 2010, were as follows:

		Cost or		Gross		Gross	Net foreign	
	a	mortized	unrealized			unrealized	currency	Carrying
Millions		cost		gains		losses	gains (losses)	value
Common equity securities	\$	661.0	\$	35.4	\$	(26.6)	\$ _	\$ 669.8
Convertible fixed maturities	\$	143.7	\$	4.4	\$	(7.6)	\$ _	\$ 140.5
Other long-term investments	\$	290.2	\$	56.2	\$	(15.2)	\$ (2.4)	\$ 328.8

Santambar 20 2011

				Dec	ember 31, 2010				
	Cost or amortized	Gross unrealized			Gross unrealized	Net foreign currency			Carrying
Millions	cost	gains			losses	gains (losses)			value
Common equity securities	\$ 561.2	\$	105.5	\$	(1.5)	\$	1.8	\$	667.0
Convertible fixed maturities	\$ 126.9	\$	16.2	\$ (.1)		\$	_	\$	143.0
Other long-term investments	\$ 329.5	\$ 62.4		\$ (13.0)		\$	(6.8)	\$	372.1

14

# **Table of Contents**

#### Other long-term investments

White Mountains holds investments in hedge funds and private equity funds, which are included in other long-term investments. The fair value of these investments has been estimated using the net asset value of the funds. At September 30, 2011, White Mountains held investments in 17 hedge funds and 33 private equity funds. The decrease in the fair value of hedge funds and private equity funds is due to net redemptions during the period. The largest investment in a single fund was \$31.4 million at September 30, 2011. The following table summarizes investments in hedge funds and private equity interests by investment objective and sector at September 30, 2011 and December 31, 2010:

	Septembe	r 30,	2011		December	.010	
Millions Hedge funds	Fair Value	_	Unfunded Commitments	_	Fair Value		Unfunded Commitments
Long/short equity	\$ 49.0	\$	_	\$	50.4	\$	_
Long/short credit & distressed	32.3		_		34.1		_
Long diversified strategies	22.0		_		24.0		_
Long/short equity REIT	14.7		_		25.2		_
Long/short equity activist	11.7		_		16.9		_
Long bank loan	.5		_		5.1		_
Total hedge funds	130.2				155.7		_

Private equity funds				
Distressed residential real estate	31.4	_	49.2	_
Energy infrastructure & services	29.5	9.9	24.2	10.8
Multi-sector	28.0	8.2	26.0	10.6
Manufacturing/Industrial	17.0	_	17.9	_
Private equity secondaries	11.3	4.0	10.4	4.4
Real estate	10.2	3.3	9.1	4.6
International multi-sector, Europe	8.0	4.7	10.5	5.3
International multi-sector, Asia	3.6	2.7	4.9	2.7
Insurance	3.6	41.3	3.9	41.3
Venture capital	2.5	.5	2.2	1.0
Healthcare	2.3	7.0	1.5	8.0
Total private equity funds	147.4	81.6	159.8	88.7
Total hedge and private equity funds included in				
other long-term investments	\$ 277.6	\$ 81.6	\$ 315.5	\$ 88.7

Redemption of investments in certain hedge funds is subject to restrictions including lock-up periods where no redemptions or withdrawals are allowed, restrictions on redemption frequency and advance notice periods for redemptions. Amounts requested for redemptions remain subject to market fluctuations until the redemption effective date, which generally falls at the end of the defined redemption period. The following summarizes the September 30, 2011 fair value of hedge funds subject to restrictions on redemption frequency and advance notice period requirements for investments in active hedge funds:

					N	otice Period				
Millions Redemption frequency	3	30-59 days notice		60-89 days notice		90-119 days notice		120+ days notice		Total
Monthly	\$	_	\$	_	\$	_	\$	6.0	\$	6.0
Quarterly		26.1		29.7		19.6				75.4
Semi-annual		_		5.1		_		14.7		19.8
Annual		21.9		_		6.6		.5		29.0
Total	\$	48.0	\$	34.8	\$	26.2	\$	21.2	\$	130.2

Certain of the hedge fund investments in which White Mountains is invested are no longer active and are in process of disposing of their underlying investments. Distributions from such funds are remitted to investors as the fund's underlying investments are liquidated. At September 30, 2011, distributions of \$3.2 million were outstanding from these investments. The actual amount of the final distribution remittances remain subject to market fluctuations. The date at which such remittances will be received is not determinable at September 30, 2011.

15

# **Table of Contents**

White Mountains has also submitted redemption requests for certain of its investments in active hedge funds. At September 30, 2011 redemptions of \$2.0 million are outstanding. The date at which such remittances will be received is not determinable at September 30, 2011. Redemptions are recorded as receivables when approved by the hedge funds and no longer subject to market fluctuations.

Investments in private equity funds are generally subject to a "lock-up" period during which investors may not request a redemption. Distributions prior to the expected termination date of the fund may be limited to dividends or proceeds arising from the liquidation of the fund's underlying investments. In addition, certain private equity funds provide an option to extend the lock-up period at either the sole discretion of the fund manager or upon agreement between the fund and the investors. At September 30, 2011, investments in private equity funds were subject to lock-up periods as follows:

Millions	1-3	1-3 years		3 – 5 years		5 – 10 years		10 years	Total
Private Equity Funds — expected lock-up			· ·					_	 
period remaining	\$	47.3	\$	_	\$	83.1	\$	17.0	\$ 147.4

Fair value measurements at September 30, 2011

White Mountains' invested assets measured at fair value include fixed maturity securities, common and preferred equity securities, convertible fixed maturity securities and other long-term investments which primarily consist of hedge funds and private equity funds. Fair value measurements reflect management's best estimate of the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. Fair value measurements fall into a hierarchy with three levels based on the nature of the inputs. Fair value measurements based on quoted prices in active markets for identical assets are at the top of the hierarchy ("Level 1"), followed by fair value measurements based on observable inputs that do not meet the criteria for Level 1, including quoted prices in inactive markets and quoted prices in active markets for similar, but not identical instruments ("Level 2"). Measurements based on unobservable inputs, including a reporting entity's estimates of the assumptions that market participants would use are at the bottom of the hierarchy ("Level 3").

White Mountains uses quoted market prices or other observable inputs to estimate fair value for the vast majority of its investment portfolio. Investments valued using Level 1 inputs include fixed maturities, primarily U.S. Treasury securities, publicly-traded common equities and short-term investments, which include U.S. Treasury Bills. Investments valued using Level 2 inputs consist of fixed maturities, including corporate debt, state and other government debt, convertible fixed maturity securities and mortgage-backed and asset-backed securities. Fair value estimates for investments classified as Level 3 measurements include investments in hedge funds, private equity funds and certain investments in fixed maturities and common equity securities. Fair value measurements for securities for which observable inputs are unavailable are estimated using industry standard pricing models and observable inputs such as benchmark interest rates, matrix pricing, market comparables, broker quotes, issuer spreads, bids, offers, credit ratings, prepayment speeds and other relevant inputs. In circumstances where quoted prices or observable inputs are adjusted to reflect management's best estimate of fair value, such fair value measurements are considered a lower level measurement in the fair value hierarchy.

White Mountains employs a number of procedures to assess the reasonableness of the fair value measurements for its other long-term investments, including obtaining and reviewing each hedge fund's and private equity fund's audited annual financial statements and periodically discussing each fund's pricing with the fund manager. However, since the fund managers do not provide sufficient information to evaluate the pricing inputs and methods for each underlying investment, the inputs are considered to be unobservable. Accordingly, the fair values of White Mountains' investments in hedge funds and private equity funds have been classified as Level 3 measurements. The fair value of White Mountains' investments in hedge funds and private equity funds has been determined using net asset value.

In addition to the investments described above, White Mountains has \$66.1 million and \$77.8 million of investment-related liabilities recorded at fair value and included in other liabilities as of September 30, 2011 and December 31, 2010. These liabilities relate to securities that have been sold short by limited partnerships in which White Mountains has investments and is required to consolidate under GAAP. All of the liabilities included have a Level 1 designation.

16

#### **Table of Contents**

## Fair Value Measurements by Level

The following tables summarize White Mountains' fair value measurements for investments at September 30, 2011 and December 31, 2010, by level. The fair value measurements for derivative assets associated with White Mountains' variable annuity reinsurance business are presented in **Note 8**.

		_		September					
Millions	Fair va	alue	Leve	l 1 Inputs	Level 2 I	iputs	Level 3 Inputs		
Fixed maturities:	_		_		_		_		
US Government and agency obligations	\$	353.1	\$	343.5	\$	9.6	\$	_	
Debt securities issued by corporations:									
Consumer		719.0		_		719.0		_	
Industrial		425.3		_		425.3		_	
Financials		243.4		5.7		212.2		25.5	
Communications		211.0				211.0			
Energy		118.0		_		118.0		_	
Basic materials		160.1		_		160.1		_	
Utilities		148.3		_		148.3		_	
Technology		25.3		_		25.3		_	
Diversified		3.5		_		3.5		_	
Total debt securities issued by									
corporations:		2,053.9		5.7		2,022.7		25.5	
Municipal obligations		2.4				2.4		_	
Mortgage-backed and asset-backed securities		2,240.6		_		2,108.7		131.9	
Foreign government, agency and provincial									
obligations		666.0		64.5		601.5			
Preferred stocks		74.6				15.1		59.5	
Total fixed maturities(2)		5,390.6		413.7		4,760.0		216.9	
Short-term investments		772.8		759.0		13.8		_	
Common equity securities:									
Financials		191.6		159.4		.5		31.7	
Consumer		166.0		164.8		1.2		51.7	
Basic materials		108.4		107.2		1.2			
Energy		57.5		57.5					
Utilities		40.7		40.7					
Technology		28.6		27.6		1.0		_	
Other		77.0		25.3		51.3		.4	
Total common equity securities		669.8		582.5		55.2	<u></u>	32.1	
<b>^</b> •									
Convertible fixed maturity investments		140.5		_		140.5		_	
Other long-term investments(1)		292.2		_		_		292.2	
	\$	7,265.9	\$	1,755.2	\$	4,969.5	\$	541.2	

<sup>(1)</sup> Excludes carrying value of \$36.6 associated with other long-term investment limited partnerships accounted for using the equity method.

17

$T_{2}$	പം	of	Con	tents
Iai	JIE	UI	CUL	items

	December 31, 2010										
Millions	Fair value	Level 1 Inputs	Level 2 Inputs	Level 3 Inputs							

<sup>(2)</sup> Carrying value includes \$102.9 that is classified as assets held for sale relating to AutoOne discontinued operations.

ixed maturities:						
US Government and agency obligations	\$ 398.7	\$	388.9	\$ 9.8	\$	_
Debt securities issued by corporations:						
Consumer	769.0		_	769.0		_
Industrial	511.3		_	511.3		_
Financials	265.8		6.3	259.5		-
Communications	226.3		_	226.3		_
Basic materials	125.3		_	125.3		_
Utilities	106.7		_	106.7		_
Energy	151.5		_	151.5		_
Technology	22.1		_	22.1		_
Total debt securities issued by						
corporations:	2,178.0		6.3	2,171.7		_
Municipal obligations	3.3		_	3.3		
Mortgage-backed and asset-backed securities	2,071.7		_	2,014.7		57.
Foreign government, agency and provincial						
obligations	1,046.5		82.6	963.9		-
Preferred stocks	87.9		_	16.5		71.
Total fixed maturities	 5,786.1		477.8	 5,179.9		128.
hort-term investments	975.3		878.7	96.6		-
Common equity securities:						
Financials	221.6		152.6	1.2		67.
Consumer	123.3		122.5	.8		_
Basic materials	98.8		97.1	1.7		_
Energy	60.2		60.2	_		_
Utilities	47.6		44.6	_		3.
Technology	31.8		30.4	1.4		_
Other	83.7		30.5	52.8		
Total common equity securities	667.0	,	537.9	57.9	,	71.
Convertible fixed maturity investments	143.0		_	143.0		_
Other long-term investments(1)	330.2		_			330.

<sup>(1)</sup> Excludes carrying value of \$41.9 associated with other long-term investment limited partnerships accounted for using the equity method.

#### Debt securities issued by corporations

The following table summarizes the ratings of the corporate debt securities held in White Mountains' investment portfolio as of September 30, 2011 and December 31, 2010:

Millions	September 30, 2011	December 31, 2010				
AAA	\$ _	\$	_			
AA	217.4		210.0			
A	765.0		833.7			
BBB	1,059.8		1,109.1			
BB	5.5		24.5			
Other	6.2		.7			
Debt securities issued by corporations	\$ 2,053.9	\$	2,178.0			

18

#### **Table of Contents**

Mortgage-backed, Asset-backed Securities

White Mountains purchases commercial and residential mortgage-backed securities with the goal of maximizing risk adjusted returns in the context of a diversified portfolio. White Mountains' non-agency commercial mortgage-backed portfolio ("CMBS") is generally short tenor, fixed rate and structurally senior, with more than 20 points of subordination on average for fixed rate CMBS and more than 55 points of subordination on average for floating rate CMBS as of September 30, 2011. In general, subordination represents the percentage principal loss on the underlying collateral that would be absorbed by other securities lower in the capital structure before the more senior security incurs any loss. White Mountains believes these levels of protection will mitigate the risk of loss tied to the refinancing challenges facing the commercial real estate market. As of September 30, 2011, on average approximately 2% of the underlying loans were reported as non-performing for all non-agency CMBS held by White Mountains. White Mountains is not an originator of residential mortgage loans and did not hold any residential mortgage-backed securities ("RMBS") categorized as sub-prime as of September 30, 2011. White Mountains' investments in hedge funds and private equity funds contain negligible amounts of sub-prime mortgage-backed securities at September 30, 2011. White Mountains considers sub-prime mortgage-backed securities as those that have underlying loan pools that exhibit weak credit characteristics, or those that are issued from dedicated sub-prime shelves or dedicated second-lien shelf registrations (i.e., White Mountains considers investments backed primarily by second-liens to be a sub-prime risk regardless of credit scores or other metrics).

White Mountains categorizes mortgage-backed securities as "non-prime" (also called "Alt A" or "A-") if they are backed by collateral that has overall credit quality between prime and sub-prime based on White Mountains' review of the characteristics of their underlying mortgage loan pools, such as credit scores and financial ratios. White Mountains' non-agency residential mortgage-backed portfolio is generally of moderate average life and structurally senior. White Mountains does not own any collateralized debt obligations, including residential mortgage-backed collateralized debt obligations.

			tember 30, 2011		December 31, 2010							
Millions	F	air Value		Level 2		Level 3		Fair Value		Level 2		Level 3
Mortgage-backed securities:												
Agency:												
GNMA	\$	1,268.0	\$	1,136.1	\$	131.9	\$	1,143.6	\$	1,102.7	\$	40.9
FNMA		268.9		268.9		_		234.9		234.9		_
FHLMC		56.6		56.6		_		39.1		39.1		_
Total Agency(1)		1,593.5		1,461.6		131.9		1,417.6		1,376.7		40.9
Non-agency:												
Residential		68.0		68.0		_		90.6		74.5		16.1
Commercial		243.5		243.5		_		87.0		87.0		_
Total Non-agency		311.5		311.5		_		177.6		161.5		16.1
Total mortgage-backed securities		1,905.0		1,773.1		131.9		1,595.2		1,538.2		57.0
Other asset-backed securities:		ĺ		,				,		,		
Credit card receivables		77.5		77.5		_		248.3		248.3		_
Vehicle receivables		235.5		235.5		_		228.1		228.1		_
Other		22.6		22.6		_		.1		.1		_
Total other asset-backed securities		335.6		335.6		_	_	476.5	_	476.5		
Total mortgage and asset-backed securities	\$	2,240.6	\$	2,108.7	\$	131.9	\$		\$	2,014.7	\$	57.0

<sup>(1)</sup> Represents publicly traded mortgage-backed securities which carry the full faith and credit guaranty of the U.S. government (i.e., GNMA) or are guaranteed by a government sponsored entity (i.e., FNMA, FHLMC).

#### Non-agency Mortgage-backed Securities

The security issuance years of White Mountains' investments in non-agency RMBS and non-agency CMBS securities as of September 30, 2011 are as follows:

			Security Issuance Year												
Millions	Fai	r Value	2003		2005		2006		2007		2009		2010		2011
Non-agency RMBS	\$	68.0	\$ 3.7	\$	_	\$	24.3	\$	29.5	\$	_	\$	10.5	\$	_
Non-agency CMBS		243.5	2.4		14.7		_		17.1		11.3		2.2		195.8
Total	\$	311.5	\$ 6.1	\$	14.7	\$	24.3	\$	46.6	\$	11.3	\$	12.7	\$	195.8
	<del></del>		 				-								

19

#### **Table of Contents**

Non-agency Residential Mortgage-backed Securities

The classification of the underlying collateral quality and the tranche levels of White Mountains' non-agency RMBS securities are as follows as of September 30, 2011:

Millions	F	air Value	Sup	oer Senior (1)	Senior (2)			Subordinate(3)
Prime	\$	49.9	\$	23.2	\$	26.7	\$	_
Non-prime		18.1		17.2		.9		_
Sub-prime		_		_		_		_
Total	\$	68.0	\$	40.4	\$	27.6	\$	

<sup>(1)</sup> At issuance, Super Senior were rated AAA by Standard & Poor's or Aaa by Moody's and were senior to other AAA or Aaa bonds.

# Non-agency Commercial Mortgage-backed Securities

The amount of fixed and floating rate securities and their tranche levels of White Mountains' non-agency CMBS securities are as follows as of September 30, 2011:

Millions	Fair Value	5	Super Senior (1)	Senior(2)	Subordinate(3)
Fixed rate CMBS	\$ 201.8	\$	24.3	\$ 177.5	\$ _
Floating rate CMBS	41.7		31.8	9.9	_
Total	\$ 243.5	\$	56.1	\$ 187.4	\$ _

<sup>(2)</sup> At issuance, Senior were rated AAA by Standard & Poor's or Aaa by Moody's and were senior to non-AAA or non-Aaa bonds.

<sup>(3)</sup> At issuance, Subordinate were not rated AAA by Standard & Poor's or Aaa by Moody's and were junior to AAA or Aaa bonds.

- (1) At issuance, Super Senior were rated AAA by Standard & Poor's or Aaa by Moody's and were senior to other AAA or Aaa bonds.
- (2) At issuance, Senior were rated AAA by Standard & Poor's or Aaa by Moody's and were senior to non-AAA or non-Aaa bonds.
- (3) At issuance, Subordinate were not rated AAA by Standard & Poor's or Aaa by Moody's and were junior to AAA or Aaa bonds.

#### Rollforward of Fair Value Measurements by Level

White Mountains uses quoted market prices where available as the inputs to estimate fair value for its investments in active markets. Such measurements are considered to be either Level 1 or Level 2 measurements, depending on whether the quoted market price inputs are for identical securities (Level 1) or similar securities (Level 2). Level 3 measurements for fixed maturities, common equity securities, convertible fixed maturities and other long-term investments at September 30, 2011 are comprised of securities for which the estimated fair value has not been determined based upon quoted market price inputs for identical or similar securities.

The following table summarizes the changes in White Mountains' fair value measurements by level for the nine months ended September 30, 2011:

Millions	_	Level 1 vestments	Iı	Level 2 rvestments	Fixed aturities	Common equity ecurities	Convertible fixed maturities		ner long- term estments	Total
Balance at January 1, 2011	\$	1,894.4	\$	5,477.4	\$ 128.4	\$ 71.2	\$ _	\$	330.2(1)	\$ 7,901.6(1)
Total realized and unrealized gains										
(losses)		(50.5)		115.9	(10.6)	(4.9)	_		22.5	72.4
Foreign currency gains (losses) through										
OCI and other revenue		.3		(84.4)	(4.4)	1.6	_		(5.0)	(91.9)
Amortization/Accretion		2.4		(40.7)	(.1)		_			(38.4)
Purchases		6,690.7		3,987.4	212.8	19.7	_		27.9	10,938.5
Sales		(6,782.1)		(4,596.9)	1.6	(55.5)	_		(83.4)	(11,516.3)
Transfers in		_		111.8	1.0	_	_		_	112.8
Transfers out		_		(1.0)	(111.8)	_	_		_	(112.8)
Balance at September 30, 2011	\$	1,755.2	\$	4,969.5	\$ 216.9	\$ 32.1	\$ 	\$	292.2(1)	\$ 7,265.9(1)

<sup>(1)</sup> Excludes carrying value of \$36.6 and \$41.9 at September 30, 2011 and January 1, 2011 associated with other long-term investment limited partnerships accounted for using the equity method.

20

# **Table of Contents**

 $Fair\ Value\ Measurements-transfers\ between\ levels-Nine-month\ period\ ended\ September\ 30,2011$ 

During the first nine months of 2011, seven securities which had been classified as Level 3 measurements in the prior period were recategorized as Level 2 measurements because quoted market prices for similar securities that were considered reliable and could be validated against an alternative source were available at September 30, 2011. These measurements comprise "Transfers out" of Level 3 and "Transfers in" to Level 2 of \$111.8 million in fixed maturities for the period ended September 30, 2011. One security that was classified as a Level 2 investment in the prior period was priced with unobservable inputs during the current period and represents the "Transfers in" of \$1 million in Level 3 investments. The fair value of this security was estimated using industry standard pricing models, in which management selected inputs using its best judgment. The security is considered to be Level 3 because the measurements are not directly observable. At September 30, 2011, the estimated fair value for this security determined using the industry standard pricing models was \$0.8 million less than the estimated fair value based upon quoted prices provided by a third party.

# Note 6. Debt

White Mountains' debt outstanding as of September 30, 2011 and December 31, 2010 consisted of the following:

Millions	mber 30, 2011	Dec	ember 31, 2010
OBH Senior Notes, at face value	\$ 269.9	\$	419.9
Unamortized original issue discount	(.2)		(.3)
OBH Senior Notes, carrying value	 269.7		419.6
Sirius Group Senior Notes, at face value	400.0		400.0
Unamortized original issue discount	(.7)		(8.)
Sirius Group Senior Notes, carrying value	 399.3		399.2
WTM Bank Facility	_		_
Total debt	\$ 669.0	\$	818.8

#### **OBH Senior Notes**

On April 20, 2011, through a cash tender offer, OneBeacon U.S. Holdings, Inc. ("OBH") purchased and retired \$150.0 million aggregate principal amount of OBH Senior Notes for \$161.6 million, which resulted in a \$12.0 million loss, including transaction fees.

On June 1, 2010, through a cash tender offer, OBH purchased and retired \$156.4 million aggregate principal amount of OBH Senior Notes for an aggregate purchase price of \$165.4 million, which resulted in a \$9.6 million loss, including transaction fees.

During 2010, in addition to the cash tender offer, OBH repurchased and retired \$29.7 million of outstanding OBH Senior Notes for \$30.8 million, which resulted in a \$1.2 million loss and OBIC purchased \$1.1 million of outstanding OBH Senior Notes for \$1.1 million.

Bank Facility

On August 12, 2011 White Mountains entered into a new revolving credit facility with a total commitment of \$375.0 million (the "WTM Bank Facility") with a syndicate of lenders administered by Bank of America. The WTM Bank Facility replaces the previous revolving credit facility which had a total commitment of \$475.0 million. As of September 30, 2011, the WTM Bank Facility was undrawn. During the third quarter of 2011, White Mountains recorded \$2.2 million of interest expense on the WTM Bank Facility, primarily related to costs associated with entering into the new facility.

**Debt Covenants** 

At September 30, 2011, White Mountains was in compliance with all debt covenants.

21

#### **Table of Contents**

#### Note 7. Income Taxes

The Company and its Bermuda domiciled subsidiaries are not subject to Bermuda income tax under current Bermuda law. In the event there is a change in the current law such that taxes are imposed, the Company and its Bermuda domiciled subsidiaries would be exempt from such tax until March 31, 2035, pursuant to the Bermuda Exempted Undertakings Tax Protection Act of 1966. The Company has subsidiaries and branches that operate in various other jurisdictions around the world that are subject to tax in the jurisdictions in which they operate. The jurisdictions in which the Company's subsidiaries and branches are subject to tax are Australia, Belgium, Canada, Germany, Gibraltar, Luxembourg, the Netherlands, Singapore, Sweden, Switzerland, the United Kingdom and the United States.

White Mountains' effective tax rate for the three and nine months of 2011 was not meaningful as pre-tax income was near break-even (\$1.2 million and \$3.6 million). The effective tax rate differed from the U.S. statutory rate of 35% due to income generated in jurisdictions other than the United States.

White Mountains' income tax expense for the three months and nine months ended September 30, 2010 represented effective tax rates of 15.6% and 17.9%. The effective tax rate for the three and nine months ended September 30, 2010 was lower than the U.S. statutory rate of 35% due to income generated in jurisdictions other than the United States and recognition of a tax benefit related to the Personal Lines Transaction, partially offset by an increase in the valuation allowance for the insurance reciprocals.

In arriving at the effective tax rate for the three and nine months ended September 30, 2011 and 2010, White Mountains forecasted the change in unrealized investment gains (losses) and realized investment gains (losses) for the years ending December 31, 2011 and 2010 and included these gains (losses) in the effective tax rate calculation pursuant to ASC 740-270.

White Mountains records a valuation allowance against deferred tax assets if it becomes more likely than not that all or a portion of a deferred tax asset will not be realized. In determining whether a valuation allowance, or change therein, is warranted, White Mountains considers factors such as prior earnings history, expected future earnings, carry-back and carry-forward periods and strategies that if executed would result in the realization of a deferred tax asset. As of September 30, 2011, the net U.S. deferred tax assets were \$362.6 million which includes \$88.2 million recorded as assets held for sale. During the next twelve months, it is possible that certain planning strategies will no longer be sufficient to utilize the entire deferred tax asset, which could result in material changes to White Mountains' valuation allowance on deferred tax assets and tax expense. Utilization of the deferred tax asset is dependent on future profitability and generation of net capital gains.

White Mountains classifies all interest and penalties on unrecognized tax benefits as part of income tax expense. With few exceptions, White Mountains is no longer subject to U.S. federal, state or non-U.S. income tax examinations by tax authorities for years before 2005.

The IRS is conducting an examination of income tax returns for 2005 and 2006 for certain U.S. subsidiaries of OneBeacon and Esurance. On January 5, 2011, White Mountains received a revised Form 4549-A (Income Tax Discrepancy Adjustments) from the IRS relating to the examination of tax years 2005 and 2006. The estimated total assessment, including interest and utilization of alternative minimum and foreign tax credit carryovers, is \$18.7 million. White Mountains disagrees with the proposed adjustments and intends to defend its position. The timing of the resolution of these issues is uncertain; however, it is reasonably possible that the resolution could occur within the next 12 months. An estimate of the range of potential outcomes cannot be made at this time. When ultimately settled, White Mountains does not expect the resolution of this examination to result in a material change to its financial position.

On July 28, 2011, the IRS commenced an examination on income tax returns for 2007, 2008 and 2009 for certain U.S. subsidiaries of OneBeacon.

On October 26, 2011, Sirius Re Holdings, Inc. (formerly White Mountains Re Holdings, Inc.) received and signed the IRS Revenue Agent's Report for 2006 - 2007, which contains no proposed adjustments by the Internal Revenue Service.

The IRS also examined the U.S. income tax return filed by WM Belvaux S.à r.l., a Luxembourg subsidiary, for tax year 2007. On May 3, 2011, the exam was completed with no proposed adjustments.

22

White Mountains has entered into agreements to reinsure death and living benefit guarantees associated with certain variable annuities in Japan. At September 30, 2011 and December 31, 2010, the total guarantee value was approximately \( \frac{\pmathbf{2}}{2} \) and \( \frac{\pmathbf{2}}{2} \) billion (approximately \( \frac{\pmathbf{3}}{2} \). 0 billion at exchange rates on that date) and \( \frac{\pmathbf{2}}{2} \) and \( \frac{\pmathbf{2}}{2} \) of the guarantee value at September 30, 2011 and December 31, 2010. The following table summarizes the pre-tax operating results of WM Life Re for the three and nine months ended September 30, 2011 and 2010:

	Three Mon Septem	led	Nine Months Ended September 30,					
Millions	2011	2010		2011		2010		
Fees, included in other revenues	\$ 8.4	\$ 7.7	\$	24.3	\$	22.3		
Change in fair value of variable annuity liability, included in								
other revenues	(164.4)	(53.5)		(148.0)		(204.7)		
Change in fair value of derivatives, included in other revenues	148.9	(15.7)		105.4		114.7		
Foreign exchange, included in other revenues	14.8	10.4		17.1		16.0		
Other investment income and gains (losses)	.5	(.2)		(.1)		(.7)		
Total revenues	 8.2	(51.3)		(1.3)		(52.4)		
Change in fair value of variable annuity death benefit liabilities,								
included in other expenses	(5.6)	.3		(3.1)		(5.8)		
Death benefit claims paid, included in other expenses	(.7)	(.6)		(2.4)		(1.4)		
General and administrative expenses	(1.1)	(1.4)		(3.2)		(5.0)		
Pre-tax income (loss)	\$ .8	\$ (53.0)	\$	(10.0)	\$	(64.6)		

All of White Mountains' variable annuity reinsurance liabilities were classified as Level 3 measurements at September 30, 2011. The following table summarizes the changes in White Mountains' variable annuity reinsurance liabilities and derivative instruments for the three and nine month periods ended September 30, 2011:

	Variable Annuity (Liabilities)	Derivative Instruments								
Millions	Level 3		Level 3 (1) Level 2 (1)(2)			Level 1 (3)			Total (4)	
Balance at January 1, 2011	\$ (610.2)	\$	275.3	\$	72.2	\$	_	\$	347.5	
Purchases	_		5.0		_		_		5.0	
Realized and unrealized gains (losses)	(151.1)		27.1		67.5		10.8		105.4	
Transfers in (out)	_		_		_		_		_	
Sales/settlements	_		(29.1)		(64.2)		(10.8)		(104.1)	
Balance at September 30, 2011	\$ (761.3)	\$	278.3	\$	75.5	\$		\$	353.8	

	iabilities)	Derivative Instruments									
Millions	 Level 3		Level 3 (1)		el 2 (1)(2)	Level 1 (3)			Total (4)		
Balance at June 30, 2011	\$ (591.3)	\$	232.7	\$	48.2	\$	(11.9)	\$	269.0		
Purchases	_		_						_		
Realized and unrealized gains (losses)	(170.0)		45.6		75.7		27.6		148.9		
Transfers in (out)	_		_		_				_		
Sales/settlements	_		_		(48.4)		(15.7)		(64.1)		
Balance at September 30, 2011	\$ (761.3)	\$	278.3	\$	75.5	\$	_	\$	353.8		

Variable Annuity

23

# **Table of Contents**

The following summarizes realized and unrealized derivative gains (losses) recognized in other revenues for the three and nine months ended September 30, 2011 and 2010 and the carrying values, included in other assets, at September 30, 2011 and December 31, 2010, by type of instrument:

	Three Mor	nths Ended Nine Months Ended							Carrying Value				
	 Septem	ber 30	,	September 30,				September 30,			December 31,		
Millions	2011		2010 2011		2011	2010		2011			2010		
Fixed income/Interest rate	\$ 29.8	\$	(25.5)	\$	17.3	\$	(1.9)	\$	50.3	\$	43.9		
Foreign exchange	63.0		33.0		32.8		99.6		212.0		225.3		
Equity	 56.1		(23.2)		55.3		17.0		91.5		78.3		
Total	\$ 148.9	\$	(15.7)	\$	105.4	\$	114.7	\$	353.8	\$	347.5		

WM Life Re enters into both over-the-counter ("OTC") and exchange traded derivative instruments to economically hedge the liability from the variable annuity benefit guarantee. In the case of OTC derivatives, WM Life Re has exposure to credit risk for amounts that are uncollateralized by counterparties. WM Life Re's internal risk management guidelines establish net counterparty exposure thresholds that take into account OTC counterparties' credit ratings. WM Life Re has entered into master netting agreements with certain of its counterparties whereby the collateral provided (held) is calculated on a net basis. The following summarizes collateral provided to WM Life Re from counterparties:

<sup>(1)</sup> Comprises over-the-counter instruments.

<sup>(2)</sup> Comprises interest rate swaps, total return swaps, foreign currency forward contracts, and bond forwards. Fair value measurement based upon bid/ask pricing quotes for similar instruments that are actively traded, where available. Swaps for which an active market does not exist have been priced using observable inputs including the swap curve and the underlying bond index.

<sup>(3)</sup> Comprises exchange traded equity index, foreign currency and interest rate futures. Fair value measurements based upon quoted prices for identical instruments that are actively traded.

<sup>(4)</sup> In addition to derivative instruments, WM Life Re held cash, short-term and fixed maturity investments of \$419.7 at September 30, 2011 posted as collateral to its reinsurance counterparties.

Millions	Sept	ember 30, 2011	Dec	ember 31, 2010
Short-term investments	\$	72.2	\$	52.4
Fixed maturity securities		58.1		48.6
Total	\$	130.3	\$	101.0

Collateral held by or provided by WM Life Re in the form of fixed maturity securities comprise U.S. Treasury securities, which are recorded at fair value. Collateral in the form of short-term investments consists of money-market instruments, carried at amortized cost, which approximates fair value. The following summarizes the value, collateral provided (held) by WM Life Re and net exposure to credit losses on OTC derivative instruments recorded within other assets:

Millions		ber 30, 2011	Dec	cember 31, 2010
OTC derivative instruments(1)	\$	363.4	\$	351.5
Collateral held		(130.3)		(101.0)
Collateral provided		22.6		25.3
Net exposure to credit losses on fair value of OTC instruments	\$	255.7	\$	275.8

(1) Value of OTC derivative instruments as of September 30, 2011 and December 31, 2010 excludes adjustments for counterparty credit risk of \$(9.6) and \$(4.0) included in fair value under GAAP.

The following table summarizes uncollateralized amounts due under WM Life Re's OTC derivative contracts:

Millions	Uncollat balanc Septembe	e as of	S&P Rating(1)
Royal Bank of Scotland	\$	86.5	A
Bank of America		49.7	A
Citigroup (3)		46.7	A
Barclays		32.2	A+
Nomura		15.6	BBB+
JP Morgan		14.7	A+
BNP Paribas		<b>7.</b> 5	AA
Goldman Sachs(2)		2.8	A
Total	\$	255.7	

- (1) Standard & Poor's ("S&P") ratings as detailed above are: "AA" (Very Strong, which is the third highest of twenty-one creditworthiness ratings), "A+" (which is the fifth highest of twenty-one creditworthiness ratings), "A" (which is the sixth highest of twenty-one creditworthiness ratings) and "BBB+" (Adequate, which is the eighth highest of twenty-one creditworthiness ratings).
- (2) Collateral provided (held) calculated under master netting agreement.

24

#### **Table of Contents**

The OTC derivative contracts are subject to restrictions on liquidation of the instruments and distribution of proceeds under collateral agreements. In addition, WM Life Re held cash, short-term and fixed maturity investments posted as collateral to its reinsurance counterparties. The additional collateral comprises the following:

Millions	Septe	mber 30, 2011	Ι	December 31, 2010
Cash	\$	393.3	\$	291.1
Short-term investments		14.5		31.6
Fixed maturity investments		11.9		3.3
Total	\$	419.7	\$	326.0

# Note 9. Earnings (Loss) Per Share

Basic earnings (loss) per share amounts are based on the weighted average number of common shares outstanding including unvested restricted shares that are considered participating securities. Diluted earnings (loss) per share amounts are based on the weighted average number of common shares including unvested restricted shares and the net effect of potentially dilutive common shares outstanding. The following table outlines the Company's computation of earnings (loss) per share from continuing operations for the three and nine months ended September 30, 2011 and 2010 (see **Note 14** for earnings per share amounts for discontinued operations):

	 Three Mor Septem 2011	 	 Nine Mont Septem 2011	 
Basic and diluted earnings per share numerators (in millions):	 2011	 2010	 2011	 2010
Net income (loss) attributable to White Mountains' common shareholders	\$ (15.8)	\$ 50.7	\$ (32.3)	\$ 13.6
Less: net (income) loss from discontinued operations	30.1	(3.6)	28.9	(1.2)
Net income (loss) attributable to White Mountains' common shareholders from				
continuing operations	14.3	47.1	(3.4)	12.4
Dividends declared and paid	_	_	(8.0)	(8.8)
Undistributed earnings (loss)	\$ 14.3	\$ 47.1	\$ (11.4)	\$ 3.6
Basic earnings per share denominators (in thousands):				

Average common shares outstanding during the period	7,845	8,332	7,902	8,547
8 8 1				,
Average unvested restricted shares (1)	 74	 99	 68	97
Basic earnings (loss) per share denominator	7,919	8,431	7,970	8,644
Diluted earnings per share denominator (in thousands):				
Average common shares outstanding during the period	7,845	8,332	7,902	8,547
Average unvested restricted shares (1)	74	99	68	97
Average outstanding dilutive options to acquire common shares (2)	_	_	_	1
Diluted earnings (loss) per share denominator	 7,919	8,431	7,970	8,645
Basic earnings per share (in dollars):				
Net income (loss) attributable to White Mountains' common shareholders	\$ 1.81	\$ 5.58	\$ (0.43)	\$ 1.44
Dividends declared and paid	_	_	(1.00)	(1.00)
Undistributed earnings (loss) per share	\$ 1.81	\$ 5.58	\$ (1.43)	\$ 0.44
Diluted earnings per share (in dollars)				
Net income (loss) attributable to White Mountains' common shareholders	\$ 1.81	\$ 5.58	\$ (0.43)	\$ 1.44
Dividends declared and paid		_	(1.00)	(1.00)
Undistributed earnings (loss) per share	\$ 1.81	\$ 5.58	\$ (1.43)	\$ 0.44

<sup>(1)</sup> Restricted shares outstanding vest either in equal annual installments or upon a stated date (see Note 12).

25

#### **Table of Contents**

#### **Note 10. Segment Information**

White Mountains has determined that its reportable segments are OneBeacon, Sirius Group and Other Operations. As a result of the Esurance Sale, the results of operations for Esurance have been classified as discontinued operations and are now presented, net of related income taxes, in the statement of comprehensive income. In addition, the OneBeacon segment has been presented excluding the results of operations of AutoOne, which has been classified as discontinued operations. Prior year amounts have been reclassified to conform to the current period's presentation (See **Note 14**).

White Mountains has made its segment determination based on consideration of the following criteria: (i) the nature of the business activities of each of the Company's subsidiaries and affiliates; (ii) the manner in which the Company's subsidiaries and affiliates are organized; (iii) the existence of primary managers responsible for specific subsidiaries and affiliates; and (iv) the organization of information provided to the Board of Directors. Significant intercompany transactions among White Mountains' segments have been eliminated herein. Financial information for White Mountains' segments follows:

No.				_	m . 1		
Millions	Unel	Beacon	Sirius Group		Operations		 Total
Three months ended September 30, 2011							
Earned insurance and reinsurance premiums	\$	258.6	\$	232.0	\$		\$ 490.6
Net investment income		16.1		22.1		4.6	42.8
Net realized and unrealized investment gains (losses)		(47.4)		64.7		(15.6)	1.7
Other revenue		.1		(28.1)		(5.8)	(33.8)
Total revenues		227.4		290.7		(16.8)	501.3
Losses and loss adjustment expenses		146.6		130.5		_	277.1
Insurance and reinsurance acquisition expenses		58.6		48.4		_	107.0
Other underwriting expenses		38.7		28.3		_	67.0
General and administrative expenses		2.5		7.4		26.3	36.2
Interest expense on debt		4.1		6.5		2.2	12.8
Total expenses		250.5		221.1		28.5	500.1
Pre-tax income (loss)	\$	(23.1)	\$	69.6	\$	(45.3)	\$ 1.2

Millions	OneBeacon			Sirius Group	Other Operations	Total
Three months ended September 30, 2010	'					
Earned insurance and reinsurance premiums	\$	293.6	\$	204.5	\$ _	\$ 498.1
Net investment income		21.6		23.0	4.3	48.9
Net realized and unrealized investment gains (losses)		51.6		(51.8)	11.8	11.6
Other revenue		12.7		43.3	(63.6)	(7.6)
Total revenues		379.5		219.0	(47.5)	551.0
Losses and loss adjustment expenses		155.2		98.2		253.4
Insurance and reinsurance acquisition expenses		64.3		37.8	_	102.1
Other underwriting expenses		50.5		24.5	_	75.0
General and administrative expenses		2.1		4.8	25.2	32.1
Interest expense on debt		6.4		6.7	.1	13.2
Total expenses		278.5		172.0	25.3	475.8
Pre-tax income (loss)	\$	101.0	\$	47.0	\$ (72.8)	\$ 75.2

<sup>(2)</sup> The diluted earnings (loss) per share denominator for the three and nine months ended September 30, 2011 and for the three months ended September 30, 2010 does not include common shares issuable upon exercise of the Non-Qualified Options as they are anti-dilutive to the calculation.

Millions	On	neBeacon	Other Sirius Group Operations			 Total
Nine months ended September 30, 2011						
Earned insurance and reinsurance premiums	\$	751.2	\$ 685.5	5	\$ —	\$ 1,436.7
Net investment income		55.8	68.3	3	14.0	138.1
Net realized and unrealized investment gains (losses)		(13.3)	102.6	5	(11.0)	78.3
Other revenue		(10.6)	(55.2	7)	(2.3)	(68.6)
Total revenues		783.1	800.7	7	.7	1,584.5
Losses and loss adjustment expenses		414.9	499.1	Ī		 914.0
Insurance and reinsurance acquisition expenses		161.1	135.3	L	_	296.2
Other underwriting expenses		132.1	79.4	1	_	211.5
General and administrative expenses		7.4	23.3	3	89.7	120.4
Interest expense on debt		16.4	19.0	5	2.8	38.8
Total expenses		731.9	756.5	5	92.5	1,580.9
Pre-tax income (loss)	\$	51.2	\$ 44.2	2	\$ (91.8)	\$ 3.6

Millions	OneBeacon		Sirius Group		Other Operations			Total
Nine months ended September 30, 2010			'					
Earned insurance and reinsurance premiums	\$	1,134.7	\$	638.2	\$	_	\$	1,772.9
Net investment income		74.9		71.3		11.2		157.4
Net realized and unrealized investment gains (losses)		79.6		13.7		11.4		104.7
Other revenue		1.9		36.4		(59.4)		(21.1)
Total revenues		1,291.1		759.6		(36.8)		2,013.9
Losses and loss adjustment expenses		719.3		456.6				1,175.9
Insurance and reinsurance acquisition expenses		248.3		122.6		_		370.9
Other underwriting expenses		176.6		69.3		_		245.9
General and administrative expenses		9.4		23.2		86.9		119.5
Interest expense on debt		23.4		20.1		.7		44.2
Total expenses		1,177.0	,	691.8		87.6	-	1,956.4
Pre-tax income (loss)	\$	114.1	\$	67.8	\$	(124.4)	\$	57.5

27

#### **Table of Contents**

#### **Note 11. Investments in Unconsolidated Affiliates**

White Mountains' investments in unconsolidated affiliates represent investments in other companies in which White Mountains has a significant voting and economic interest but does not control the entity.

Millions	ember 30, 2011	Dec	ember 31, 2010
Symetra common shares	\$ 449.0	\$	350.4
Symetra warrants	7.7		37.1
Total investment in Symetra	 456.7		387.5
Pentelia Capital Management	1.9		2.2
Total investments in unconsolidated affiliates	\$ 458.6	\$	389.7

#### Symetra

At September 30, 2011 and December 31, 2010, White Mountains owned 17.4 million common shares of Symetra and warrants to acquire an additional 9.5 million common shares. White Mountains accounts for its investment in common shares of Symetra using the equity method and accounts for its Symetra warrants as derivatives with changes in fair value recognized as a gain or loss through other revenues. White Mountains uses a Black Scholes valuation model to determine the fair value of the Symetra warrants. The major assumptions used in valuing the Symetra warrants at September 30, 2011 were a risk free rate of 0.37%, volatility of 31.64%, an expected life of 2.83 years, a strike price of \$11.49 per share and a share price of \$8.15 per share.

In January 2010, Symetra completed an initial public offering at a price of \$12.00 per share, with 25.3 million shares sold by Symetra and 9.7 million shares sold by existing shareholders. White Mountains did not sell any of its shares in the offering. As a result of the offering, White Mountains' fully converted ownership in Symetra decreased to approximately 20% during the first quarter of 2010. The issuance of the new Symetra shares at a price below its adjusted book value per share diluted White Mountains' investment in Symetra's common shares, resulting in a \$16.0 million decrease to White Mountains' carrying value in Symetra.

The following table summarizes amounts recorded by White Mountains relating to its investment in Symetra for the three months ended September 30, 2011 and 2010:

				Ionths End ber 30, 201					onths End er 30, 201	
	C	ommon				C	ommon			
Millions		Shares	W	arrants	Total		Shares	Wa	rrants	Total
Carrying value of investment in Symetra as of June 30	\$	388.5	\$	32.5	\$ 421.0	\$	344.9	\$	27.6	\$ 372.5
Equity in earnings of Symetra (1)		1.6			1.6		8.3(4	4)	_	8.3
Net unrealized gains from Symetra's fixed maturity portfolio (2)		<b>59.9</b>		_	<b>59.9</b>		46.9		_	46.9
Dividends received		(1.0)		_	(1.0)		(.9)		_	(.9)

(Decrease) increase in value of warrants	 	(24.8)	(24.8)	 _	(10.7)	(10.7)
Carrying value of investment in Symetra as September 30 (3)	\$ 449.0	\$ 7.7	\$ 456.7	\$ 399.2	\$ 16.9	\$ 416.1

- (1) Equity in earnings excludes tax expense of \$0.1 and \$0.7.
- (2) Net unrealized gains excludes tax expense of \$4.8 and \$3.8.
- (3) Includes White Mountains' equity in net unrealized gains from Symetra's fixed maturity portfolio of \$147.7 and \$120.7 as of September 30, 2011 and 2010, which exclude tax expense of \$12.3 and \$9.8.
- (4) Includes a \$0.1 loss from the dilutive effect of Symetra's quarterly dividend and the issuance of restricted shares by Symetra.

28

#### **Table of Contents**

The following table summarizes amounts recorded by White Mountains relating to its investment in Symetra for the nine months ended September 30, 2011 and 2010:

			onths Ende ber 30, 201				ths Ended er 30, 2010	-	
Millions	ommon Shares	W	arrants	Total	ommon Shares	Wa	rrants		Total
Carrying value of investment in Symetra as of December 31	\$ 350.4	\$	37.1	\$ 387.5	\$ 269.2	\$	38.5	\$	307.7
Equity in earnings of Symetra (1)	17.5		_	17.5	2.1(4)		_		2.1
Net unrealized gains from Symetra's fixed maturity portfolio (2)	84.0		_	84.0	129.7(5)		_		129.7
Dividends received	(2.9)		_	(2.9)	(1.8)		_		(1.8)
(Decrease) increase in value of warrants	_		(29.4)	(29.4)	_		(21.6)		(21.6)
Carrying value of investment in Symetra as September 30 (3)	\$ 449.0	\$	7.7	\$ 456.7	\$ 399.2	\$	16.9	\$	416.1

- (1) Equity in earnings excludes tax expense of \$1.4 and \$0.7.
- (2) Net unrealized gains excludes tax expense of \$6.8 and \$3.8.
- (3) Includes White Mountains' equity in net unrealized gains from Symetra's fixed maturity portfolio of \$147.7 and \$120.7 as of September 30, 2011 and 2010, which excludes tax expense of \$24.2 and \$9.8.
- (4) Includes a \$17.9 loss from the dilutive effect of Symetra's initial public offering and a \$0.7 loss from the dilutive effect of Symetra's quarterly dividend and issuance of restricted shares by Symetra.
- (5) Includes a \$1.9 gain from the effect of Symetra's initial public offering.

During the three and nine months ended September 30, 2011, White Mountains received cash dividends from Symetra of \$1.0 million and \$2.9 million on its common share investment that was recorded as a reduction of White Mountains' investment in Symetra. During the three and nine months ended September 30, 2011, White Mountains also received cash dividends from Symetra of \$0.6 million and \$1.6 million on its investment in Symetra warrants that was recorded in net investment income.

The aggregate value of White Mountains' investment in common shares of Symetra was \$141.8 million based upon the quoted market price of \$8.15 at September 30, 2011. White Mountains believes that the difference in the market value and the carrying value of Symetra's common shares is temporary.

#### Note 12. Employee Share-Based Incentive Compensation Plans

White Mountains' Long-Term Incentive Plan (the "WTM Incentive Plan") provides for grants of various types of share-based and non share-based incentive awards to key employees and service providers of the Company and certain of its subsidiaries. White Mountains' share-based compensation incentive awards consist of performance shares, restricted shares and stock options.

#### **Share-Based Compensation Based on White Mountains Common Shares**

#### WTM Performance Shares

Performance shares are conditional grants of a specified maximum number of common shares or an equivalent amount of cash. Performance share awards vest, subject to the attainment of performance goals, at the end of a three-year period and are valued based on the market value of common shares at the time awards are paid. The following table summarizes performance share activity for the three and nine months ended September 30, 2011 and 2010 for WTM performance shares granted under the WTM Incentive Plan and phantom WTM performance shares granted under subsidiary plans ("WTM Phantom Share Plans"):

	7	hree Months	Ended September 30,		1	Nine Months	Ended September 30,		
	2011		20	10	2011	l	201	0	
Million and John and	Target Performance Shares	Accrued	Target Performance Shares	Accrued	Target Performance Shares	Accrued			ccrued
Millions, except share amounts	Outstanding	Expense	Outstanding	Expense	Outstanding	Expense		E	xpense
Beginning of period	150,064	\$ 56	.5 163,220	\$ 20.2	163,184	\$ 29	<b>9.4</b> 172,454	\$	12.2
Shares expired (1)	_			_	(51,131)		— (49,127)		_
New grants	_			_	37,675		<b>—</b> 47,880		_
Assumed forfeitures and cancellations (2)	_		- (36)	_	336		<b>(.6)</b> (8,023)		_
Expense recognized	_	(1	.0)	3.0	_	20	6.7		11.0
Ending September 30,	150,064	\$ 55	.5 163,184	\$ 23.2	150,064	\$ 55	<b>5.5</b> 163,184	\$	23.2

<sup>(1)</sup> There were no payments made in 2011 and 2010 for the 2008-2010 and 2007-2009 performance cycles; those performance shares did not meet the threshold performance goals and expired.

(2) Amounts include changes in assumed forfeitures, as required under GAAP.

#### **Table of Contents**

If 100% of the outstanding WTM performance shares had vested on September 30, 2011, the total additional compensation cost to be recognized would have been \$23.5 million, based on accrual factors at September 30, 2011 (common share price and payout assumptions).

Performance Shares granted under the WTM Incentive Plan

The following table summarizes WTM performance shares outstanding and accrued expense for WTM performance shares awarded under the WTM Incentive Plan at September 30, 2011 for each performance cycle:

	Target WTM Performance Shares	Accrued
Millions, except share amounts	Outstanding	 Expense
Performance cycle:		
2009 – 2011	60,410	\$ 38.1
2010 – 2012	42,890	7.9
2011 – 2013	37,675	4.5
Sub-total	140,975	50.5
Assumed forfeitures	(3,524)	(1.3)
Total at September 30, 2011	137,451	\$ 49.2

Phantom Performance Shares granted under WTM Phantom Share Plans

The following table summarizes WTM phantom performance shares outstanding and accrued expense for grants made under WTM Phantom Share Plans at September 30, 2011 for each performance cycle:

Millions, except share amounts  Performance cycle:	Target WTM Phantom Performance Shares Outstanding	_	Accrued Expense
2009 – 2011	7,947	\$	5.2
2010 – 2012	4,990		1.2
2011 – 2013(1)	_		_
Sub-total	12,937		6.4
Assumed forfeitures	(324)		(.1)
Total at September 30, 2011	12,613	\$	6.3

<sup>(1)</sup> All performance shares for the 2011-2013 performance cycle were granted from the WTM Incentive Plan

#### Restricted Shares

At September 30, 2011 and 2010, the Company had 73,500 and 98,750 unvested restricted shares outstanding. The following outlines the unrecognized compensation cost associated with the outstanding restricted share awards for the three and nine months ended September 30, 2011 and 2010:

	Three Months Ended September 30,							Nine Months Ended September 30,								
	20	2011 20				<u>.</u>	20		2010							
Millions, except share amounts	Restricted Shares	Iss	mortized ue Date ir Value	Restricted Shares		namortized Issue Date Fair Value	Restricted Shares	I	namortized ssue Date Fair Value	Restricted Shares	]	namortized ssue Date Fair Value				
Non-vested,																
Beginning of period	73,500	\$	18.9	98,750	\$	20.9	46,250	\$	14.1	91,900	\$	23.7				
Issued	_		_	_		_	27,250		9.9	19,750		6.7				
Vested	_		_	_		_	_		_	(12,650)		_				
Forfeited	_		_	_		_	_		_	(250)		(.1)				
Modified(1)	_		_	_		_	_		_	`—		(3.3)				
Expense recognized	_		(2.8)	_		(3.3)	_		(7.9)	_		(9.4)				
Non-vested at September 30	73,500	\$	16.1	98,750	\$	17.6	73,500	\$	16.1	98,750	\$	17.6				

<sup>(1)</sup> During the first quarter of 2007, White Mountains issued 15,000 restricted shares to the Company's Chairman and CEO in connection with his hiring that would vest in the event of a change in control of the Company before January 20, 2012. During 2010, the Compensation Committee modified the vesting terms so that the 15,000 restricted shares time vest in three equal annual installments beginning on January 20, 2013.

30

# **Table of Contents**

During the second quarter of 2011, White Mountains issued 250 restricted shares that vest on January 1, 2014. During the first quarter of 2011, White Mountains issued 27,000 restricted shares that vest on January 1, 2014. During the first quarter of 2010, White Mountains issued 19,750 restricted shares that vest on December 31, 2012. As of September 30, 2011, \$16.1 million is expected to be recognized ratably over the remaining vesting periods.

#### Non-Qualified Options

In January 2007, the Company issued 200,000 seven-year Non-Qualified Options to the Company's Chairman and CEO (the "original grant") that vest in equal annual installments over five years and that had an initial exercise price of \$650 per common share that escalated at an annual rate of 5% less the annual regular dividend rate (the "Escalator"). The fair value of the Non-Qualified Options was \$27.2 million at the grant date. The fair value of the Non-Qualified Options at the grant date was estimated using a closed-form option model using an expected volatility assumption of 29.7%, a risk-free interest rate

assumption of 1.1% (or 4.7% less the Escalator), a forfeiture assumption of 0%, an expected dividend rate assumption of 1.4% and a term assumption of seven years.

At the 2010 Annual General Meeting of Members held on May 26, 2010 (the "modification date"), the Company's shareholders approved the following amendments to the Non-Qualified Options (the "amended grant"): (1) extend the term of the Non-Qualified Options by three years to January 20, 2017; (2) freeze the exercise price at \$742 per common share, the exercise price on February 24, 2010; (3) extinguish 75,000 of the 200,000 Non-Qualified Options; and (4) limit the potential in-the-money value of the Non-Qualified Options in excess of \$100 million to 50% of the amount in excess of \$100 million. The fair value of the amended grant was \$4.4 million at the modification date, while the fair value of the original grant as of the modification date was \$3.5 million. The fair value of the amended grant was estimated using a closed-form option model using an expected volatility assumption of 34.0%, a risk-free interest rate assumption of 2.43%, a forfeiture assumption of 0%, an expected dividend rate assumption of 0.32% and a term assumption of 6.67 years. The fair value of the original grant as of the modification date was estimated using a closed-form option model using an expected volatility assumption of 41.0%, a risk-free interest rate assumption of 1.57%, a forfeiture assumption of 0%, an expected dividend rate assumption of 0.32% and a term assumption of 3.67 vears.

Prior to the modification, \$18.2 million of the original grant fair value had been amortized into income. In connection with the modification, White Mountains recognized \$8.7 million of the remaining \$9.0 million of unamortized option expense related to the original grant. As of the first quarter in 2011, the Non-Qualified Options have been fully amortized.

For the nine months ended September 30, 2011, White Mountains recognized a total of \$0.1 million of expense related to amortizing the Non-Qualified Options compared to \$10.1 million and \$11.5 million of expense in the three and nine months ended September 30, 2010.

#### Share-Based Compensation Based on OneBeacon Ltd. Common Shares

The OneBeacon Long-Term Incentive Plan (the "OneBeacon Incentive Plan") provides for grants of various types of share-based and non share-based incentive awards to key employees of OneBeacon Ltd. and certain of its subsidiaries. OneBeacon's share-based incentive awards consist of OneBeacon performance shares, stock options granted in connection with OneBeacon's initial public offering, restricted shares and restricted stock units ("RSUs").

#### OneBeacon Performance Shares

OneBeacon performance shares are conditional grants of a specified maximum number of common shares or an equivalent amount of cash. OneBeacon performance share awards vest, subject to the attainment of performance goals, at the end of a three-year period and are valued based on the market value of OneBeacon Ltd. common shares at the time awards are paid. The following table summarizes performance share activity for the three and nine months ended September 30, 2011 and 2010 for OneBeacon performance shares granted under the OneBeacon Incentive Plan:

		Three Months En	ded September 30,	Nine Months Ended September 30,								
	20:	11	2010				11		2010			
Millions, except share amounts	Target Performance Shares Outstanding	Accrued Expense	Target Performance Shares Outstanding		Accrued Expense	Target Performance Shares Outstanding		Accrued Expense	Target Performance Shares Outstanding		Accrued Expense	
							_			_		
Beginning of period	671,727	\$ 10.0	1,770,423	\$	15.2	1,464,295	\$	18.5	2,224,215	\$	15.1	
Payments and deferrals (1)(2)	_	_	(199,774)		(2.2)	(936,150)		(10.5)	(882,118)		(4.5)	
New awards	_	_	14,618		_	194,900		_	293,478		_	
Forfeitures and cancellations	(19,583)	.1	(20,438)		(.3)	(70,901)		(.2)	(70,746)		(.7)	
Expense recognized		(.5)			1.7			1.8			4.5	
Ending September 30,	652,144	\$ 9.6	1,564,829	\$	14.4	652,144	\$	9.6	1,564,829	\$	14.4	

OneBeacon performance share payments in 2011 for the 2008-2010 performance cycle were at 68.5% of target. OneBeacon performance shares payments in 2010 for the 2007-2009 performance cycle were at 14.2% of

31

# **Table of Contents**

If 100% of the outstanding OneBeacon performance shares had been vested on September 30, 2011, the total additional compensation cost to be recognized would have been \$3.7 million, based on accrual factors at September 30, 2011 (common share price and payout assumptions).

The following table summarizes OneBeacon performance shares outstanding awarded under the OneBeacon Incentive Plan at September 30, 2011 for each performance cycle:

	Target OneBeacon Performance Shares	Accrued
Millions, except share amounts	Outstanding	Expense
Performance cycle:		
2008 - 2010	259,102	\$ 7.2
2009 – 2011	251,557	1.9
2010 – 2012	151,563	.6
Sub-total	662,222	 9.7
Assumed forfeitures	(10,078)	(.1)
Total at September 30, 2011	652,144	\$ 9.6

# OneBeacon Non-Qualified Options

In November 2006, in connection with its initial public offering, OneBeacon Ltd. issued to its key employees 1,420,000 OneBeacon Non-Qualified Options to acquire OneBeacon Ltd. common shares at an above-market fixed exercise price. The following table summarizes option activity for the three and nine months ended September 30, 2011 and 2010:

Three Months Ended September 30,	Nine Months Ended September 30,

target. Amounts include deposits into OneBeacon's deferred compensation plan.

OneBeacon performance share payments also include accelerated payments resulting from the OneBeacon Personal Lines and Commercial Lines Transactions. The accelerated OneBeacon performance shares payments for the 2009-2011 and 2010-2012 performance cycles were on a pro rata basis and at a performance factor of 100%.

	20	11	201	10	201	1	2010		
Millions, except share amounts	Target Options Outstanding	Expense Amortized	Target Options Outstanding	Expense Amortized	Target Options Outstanding	Expense Amortized	Target Options Outstanding	Expense Amortized	
Beginning of period	750,130	\$ 4.8	879,783	\$ 4.1	768,652	\$ 4.5	1,015,610	\$ 3.6	
New awards	_	_	_	_	_	_	_	_	
Forfeitures and cancellations	(9,260)	_	_	_	(27,782)	_	(37,044)	_	
Vested and expired (1)		_	(43,218)	_		_	(142,001)	_	
Exercised	_	_	` _ `	_	_	_		_	
Expense recognized	_	.1	_	.2	_	.4	_	.7	
Ending September 30,	740,870	\$ 4.9	836,565	\$ 4.3	740,870	\$ 4.9	836,565	\$ 4.3	

(1) During the three and nine months ended September 30, 2010, as a result of the Commercial Lines Transaction and the Personal Lines Transaction, 43,218 and 142,001 options vested that were unexercised and expired.

The options vest in equal installments on each of the third, fourth and fifth anniversaries of their issuance and expire 5.5 years from the date of issuance. The fair value of each option award at grant was estimated using a Black-Scholes option pricing model using an expected volatility assumption of 30%, a risk-free interest rate assumption of 4.6%, a forfeiture assumption of 5%, an expected dividend rate assumption of 3.4% and an expected term assumption of 5.5 years. The options originally had a per share exercise price of \$30.00. On May 27, 2008, the OneBeacon Compensation Committee of the Board of Directors (the "OB Compensation Committee") amended the exercise price to \$27.97 as a result of the \$2.03 per share special dividend paid in the first quarter of 2008. On November 16, 2010, the OB Compensation Committee adjusted the exercise price to \$25.47 as a result of the \$2.50 per share special dividend paid in the third quarter of 2010. The compensation expense associated with the options and the incremental fair value of the award modification is being recognized ratably over the remaining period.

#### OneBeacon Restricted Shares

On May 25, 2011, OneBeacon issued 630,000 restricted shares to its CEO that vest one-fourth on each of February 22, 2014, 2015, 2016 and 2017. Concurrently with the grant of the restricted shares, 35,000 OneBeacon performance shares issued to OneBeacon's CEO for the 2011-2013 performance share cycle were forfeited and performance share awards to OneBeacon's CEO for the next 5 years will be reduced by a similar number of shares. The restricted shares contain dividend participation features, and therefore, are considered participating securities. As of September 30, 2011, there were 630,000 unvested restricted shares outstanding. For the three and nine months ended September 30, 2011, OneBeacon recognized \$0.4 million and \$0.5 million, respectively of expense. As of September 30, 2011, \$8.1 million is expected to be recognized ratably over the remaining vesting period.

32

#### **Table of Contents**

#### Restricted Stock Units

The Non-Qualified Options granted by OneBeacon Ltd., in connection with its initial public offering, did not include a mechanism in the options to reflect the contribution to total return from the regular quarterly dividend. As a result, during the first quarter of 2008, OneBeacon granted 116,270 RSUs to actively employed option holders. The RSUs vest one-third on each of November 9, 2009, 2010 and 2011 subject to, for each vesting tranche of units, the attainment of 4% growth in OneBeacon's book value per share from January 1, 2008 through the end of the calendar year immediately following the applicable vesting date. Upon vesting, the RSUs are mandatorily deferred into one of OneBeacon's non-qualified deferred compensation plans and will be paid out in 2012 in cash or shares at the discretion of the OB Compensation Committee. The performance goal for the first and second tranches was attained. The performance goal for the third tranche will be assessed at December 31, 2011. The expense associated with the RSUs is being recognized over the vesting period. OneBeacon recognized no expense for the three months and nine months ended September 30, 2010, OneBeacon recognized no expense for the three months ended September 30, 2010 and \$0.2 million of expense for the nine months ended September 30, 2010. As of September 30, 2011, there were 22,773 RSUs outstanding.

#### **Note 13. Fair Value of Financial Instruments**

White Mountains carries its financial instruments on its balance sheet at fair value with the exception of its fixed-rate, long-term indebtedness and the Sirius Group Preference Shares, which are recorded as noncontrolling interest. The following table summarizes the fair value and carrying value of financial instruments as of September 30, 2011 and December 31, 2010:

	Septembe	r 30, 2	2011	December 31, 2010				
Millions	 Fair Value		Carrying Value		Fair Value	•	Carrying Value	
OBH Senior Notes	\$ 276.6	\$	269.7	\$	435.1	\$	419.6	
Sirius Group Senior Notes	403.7		399.3		390.9		399.2	
Sirius Group Preference Shares	220.0		250.0		212.5	250.0		

## **Note 14. Discontinued Operations**

#### Esurance

On October 7, 2011, White Mountains completed the sale of Esurance Insurance and AFI to The Allstate Corporation (see **Note 15**). As a result of the transaction, Esurance Insurance, AFI and the business Esurance Insurance cedes to Sirius Group (collectively, "the Esurance Disposal Group") are reported as discontinued operations. Effective for the nine months ended September 30, 2011, the results of operations for the Esurance Disposal Group have been classified as discontinued operations and are presented, net of related income taxes, in the statement of comprehensive income. The assets and liabilities of the Esurance Disposal Group have been presented in the balance sheet as held for sale. Prior year amounts have been reclassified to conform to the current period's presentation.

The amounts reflected within discontinued operations differ from amounts previously presented within the Esurance segment. The segment results of operations for Esurance reported in previous periods included investment income and realized and unrealized investment gains and losses allocated to Esurance related to capital that was used by Esurance but contained within Sirius Group's legal entities. These allocations do not have separately identifiable cash flows and have therefore been excluded from amounts classified in discontinued operations and are now included in the Sirius Group segment. For the three and nine month periods ended September 30, 2011, \$0.9 million and \$(2.4) million of net investment income and realized and unrealized investment gains and losses that had been previously included in the Esurance segment were excluded from net income (loss) from discontinued operations. For the three

#### **Table of Contents**

#### AutoOne

On August 30, 2011, OneBeacon entered into an agreement to sell the AutoOne business to Interboro. AutoOne operates as a division within OneBeacon, offering products and services to automobile assigned risk markets. The transaction includes the sale of two insurance entities, AOIC and AOSIC, through which substantially all of the AutoOne business is written on a direct basis. For the nine months ended September 30, 2011, the results of operations for the AutoOne business have been classified as discontinued operations and are presented, net of related income taxes, in the statement of comprehensive income. Prior year results of operations have been reclassified to conform to the current period's presentation.

The AutoOne disposal group excludes investing and financing activities from amounts classified as discontinued operations. OneBeacon's investing and financing operations are conducted on an overall consolidated level and accordingly, there were no separately identifiable cash investing or financing cash flows associated with AutoOne. Prior to the sale of AutoOne, OneBeacon will ensure that the AutoOne legal entities hold an agreed upon level of invested assets and capital. The assets and liabilities associated with the AutoOne business as of September 30, 2011 have been presented in the balance sheet as held for sale sheet assuming the investing and financing steps required to effect the sale were completed as of the current balance sheet date. The prior year balance sheet has not been reclassified to conform to the current period's presentation because the assets and liabilities associated with AutoOne in the prior year would not provide a meaningful comparison to the assets and liabilities presented as held for sale at September 30, 2011.

The following summarizes the assets and liabilities associated with the businesses classified as held for sale:

Millions	Sej	otember 30, 2011	Γ	December 31, 2010
Assets held for sale				
Fixed maturity investments, at fair value	\$	585.0	\$	593.8
Short-term investments, at amortized cost (which approximates fair value)		224.9		131.0
Common equity securities, at fair value		_		43.0
Convertible fixed maturity investments, at fair value		_		17.2
Other long-term investments		_		4.0
Total investments		809.9		789.0
Cash		235.7		39.8
Reinsurance recoverable on paid and unpaid losses		34.6		34.4
Insurance premiums receivable		128.2		102.9
Deferred acquisition costs		42.8		34.7
Deferred tax asset		88.2		67.8
Other assets		60.1		54.2
Total assets held for sale	\$	1,399.5	\$	1,122.8
Liabilities held for sale				
Loss and loss adjustment expense reserves	\$	514.9	\$	450.4
Unearned insurance premiums		258.3		201.6
Accrued incentive compensation		31.5		19.1
Other liabilities		104.7		80.4
Total liabilities held for sale		909.4		751.5
Net assets held for sale	\$	490.1	\$	371.3

34

# **Table of Contents**

The following summarizes the results of operations, including related income taxes associated with the businesses classified as discontinued operations:

	Three Mor Septem			Nine Months Ended September 30,			
Millions, except per share amounts	2011	2010		2011		2010	
Revenues							
Earned insurance premiums	\$ 233.8	\$ 234.6	\$	698.7	\$	683.6	
Net investment income	3.8	5.8		11.8		15.4	
Net realized and unrealized investment gains (losses)	(6.4)	10.4		1.4		17.0	
Other revenue	17.5	15.3		52.4		44.7	
Total revenues	 248.7	 266.1		764.3		760.7	
Expenses							
Loss and loss adjustment expenses	180.5	172.7		518.7		508.0	
Insurance and reinsurance acquisition expenses	51.4	50.8		154.5		142.2	
Other underwriting expenses	32.3	21.4		76.3		66.9	
General and administrative expenses	1.7	12.1		37.5		36.5	
Total expenses	265.9	257.0		787.0		753.6	
Pre-tax income (loss)	(17.2)	9.1		(22.7)		7.1	
Income tax (expense) benefit	5.3	(5.5)		12.0		(5.9)	
Income (loss) from discontinued operations	 (11.9)	 3.6	-	(10.7)	-	1.2	
Loss from sale of AutoOne, net of tax	 (18.2)			(18.2)			

Net income (loss) from discontinued operations	\$ (30.1) \$	3.6 \$	(28.9) \$	1.2

Basic earnings (loss) per share amounts are based on the weighted average number of common shares outstanding including unvested restricted shares that are considered participating securities. Diluted earnings (loss) per share amounts are based on the weighted average number of common shares including unvested restricted shares and the net effect of potentially dilutive common shares outstanding. The following table outlines the computation of earnings (loss) per share for discontinued operations for the three and nine months ended September 30, 2011 and 2010:

	Three Mon Septem		Nine Months Ended September 30,				
	2011	2010	2011		2010		
Basic and diluted earnings (loss) per share numerators							
(in millions):							
Net income (loss)	\$ (30.1)	\$ 3.6	\$ (28.9)	\$	1.2		
Basic earnings (loss) per share denominators (in	 						
thousands):							
Average common shares outstanding during the period	7,845	8,332	7,902		8,547		
Average unvested restricted shares	74	99	68		97		
Basic earnings (loss) per share denominator	7,919	 8,431	7,970		8,644		
Diluted earnings (loss) per share denominator (in							
thousands):							
Average common shares outstanding during the period	7,845	8,332	7,902		8,547		
Average unvested restricted shares	74	99	68		97		
Average outstanding dilutive options to acquire common							
shares		_	_		1		
Diluted earnings (loss) per share denominator	7,919	 8,431	7,970		8,645		
Basic and diluted earnings (loss) per share (in dollars):	\$ (3.81)	\$ 0.43	\$ (3.63)	\$	0.13		

#### **Note 15. Subsequent Event**

#### Sale of Esurance

On October 7, 2011, White Mountains completed the sale of Esurance Insurance and AFI to The Allstate Corporation for a cash payment of \$1.01 billion, which was equal to \$700 million plus the estimated pro forma tangible book value at closing of the legal entities sold of approximately \$310 million. The final purchase price is subject to true-up of estimated tangible book value of the entities sold through the date of the closing. As a result of the transaction, the Esurance Disposal Group is reported as discontinued operations.

35

#### **Table of Contents**

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

The following discussion contains "forward-looking statements". White Mountains intends statements that are not historical in nature, which are hereby identified as forward-looking statements, to be covered by the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. White Mountains cannot promise that its expectations in such forward-looking statements will turn out to be correct. White Mountains' actual results could be materially different from and worse than its expectations. See "FORWARD-LOOKING STATEMENTS" for specific important factors that could cause actual results to differ materially from those contained in forward-looking statements.

The following discussion also includes five non-GAAP financial measures - adjusted comprehensive income (loss), adjusted book value per share, total adjusted capital and Esurance's adjusted expense ratio and adjusted combined ratio - that have been reconciled to their most comparable GAAP financial measures (see page 55). White Mountains believes these measures to be more relevant than comparable GAAP measures in evaluating White Mountains' financial performance and condition.

#### RESULTS OF OPERATIONS FOR THE THREE AND NINE MONTHS ENDED SEPTEMBER 30, 2011 AND 2010

#### Overview

White Mountains ended the third quarter of 2011 with an adjusted book value per share of \$436, down 3% for the third quarter and 1% for the first nine months of 2011, including dividends. White Mountains reported adjusted comprehensive loss of \$98 million and \$58 million for the third quarter and first nine months of 2011 compared to adjusted comprehensive income \$178 million and \$60 million for the third quarter and first nine months of 2010. The decline in adjusted book value per share in the third quarter of 2011 was driven by equity investment losses and foreign currency translation losses from the strengthening U.S. dollar, somewhat offset by profitable underwriting results at OneBeacon and the Sirius Group and the benefit of shares repurchased below adjusted book value per share. White Mountains' GAAP pre-tax total return on invested assets was -1.5% and 1.7% for the third quarter and first nine months of 2011, which included 1.2% and 0.2% of foreign currency losses, compared to 4.0% and 4.6% for the third quarter and first nine months of last year, which included 1.9% and 0.5% of foreign currency gains. Investment results for the third quarter of 2011 were driven by losses in the equity portfolio, as it was down 7.8%, compared to the S&P 500 index return of -13.9%. Adjusted book value per share decreased \$6 and \$1 in third quarter and first nine months of 2011 from foreign currency translation losses. In addition, share repurchases increased adjusted book value per share by \$1 and \$4 in the third quarter and first nine months of 2011.

OneBeacon's book value per share decreased 3% for the third quarter and increased 2% for the first nine months of 2011, including dividends. The decrease for the third quarter was driven by losses on equity investments and an after-tax charge of \$18 million for the estimated loss on the sale of AutoOne. OneBeacon's GAAP combined ratio was 94% for the third quarter of 2011 compared to 92% for the third quarter of last year, as higher catastrophe losses and lower favorable loss reserve development were mostly offset by improved non-catastrophe current accident year results. OneBeacon's GAAP combined ratio was 94% for the first nine months of 2011 compared to 101% for the first nine months of last year. The GAAP combined ratio for the first nine months of

2010 included unusually high non-catastrophe losses, primarily a number of large losses in businesses that OneBeacon has since exited and in its property and inland marine specialty businesses.

Sirius Group's GAAP combined ratio was 89% for the third quarter of 2011 compared to 78% for the third quarter of 2010. The increase was primarily due to worse current accident year underwriting results in the accident and health, marine and aviation lines and higher catastrophe losses. Catastrophe losses increased the combined ratio by 11 points in the third quarter of 2011 compared to 9 points in the third quarter of 2010. Catastrophe losses in the third quarter of 2011 were primarily composed of \$11 million in losses from hurricane Irene (5 points), \$7 million in losses from the June 2011 New Zealand earthquake (3 points) and \$6 million in losses from the July 2011 Copenhagen floods (2 points). Sirius Group's GAAP combined ratio was 104% for the first nine months of 2011 compared to 102% for the first nine months of 2010. Catastrophe losses increased the combined ratio by 27 points in the first nine months of 2011 compared to 28 points in the first nine months of 2010. For the first nine months of 2011, catastrophe losses were primarily composed of \$90 million in losses from the Japan earthquake and tsunami (13 points) and \$49 million in losses from the February and June New Zealand earthquakes (7 points), while catastrophe losses in the first nine months of 2010 were primarily from the Chile earthquake.

White Mountains' total net written premiums increased 12% to \$511 million in the third quarter and decreased 5% to \$1,590 million in the first nine months of 2011 compared to \$456 million and \$1,679 million in the third quarter and first nine months of 2010. Excluding the \$107 million of net written premiums in the first nine months of 2010 related to the exited business at OneBeacon, White Mountains' net written premiums increased \$82 million, or 5% in the first nine months of 2011. OneBeacon's specialty lines premiums increased 9% and 6% to \$297 million and \$822 million in the third quarter and first nine months of 2011. Sirius Group's net written premiums were \$214 million and \$768 million in the third quarter and first nine months of 2011, increases of 14% and 4% from the comparable 2010 periods, primarily due to increases in the accident and health and trade credit lines and to foreign exchange translation.

36

#### **Table of Contents**

During the third quarter of 2011, White Mountains completed a "modified Dutch auction" self-tender offer and repurchased 327,872 of its common shares at \$415 per share, for a total cost of \$136.5 million, including \$0.4 million of fees and expenses related to the tender offer. Including shares repurchased in the self-tender offer and shares repurchased under its share repurchase program, White Mountains repurchased a total of 593,451 of its common shares during the first nine months of 2011 for \$230 million at an average share price of \$387, which is 89% of White Mountains' adjusted book value per share of \$436 at September 30, 2011.

Also during the third quarter of 2011, OneBeacon completed a new ground-up study of its legacy asbestos and environmental exposures. The previous study was based on experience through 2007. Reasonable estimates of potential adverse scenarios continue to be within the \$2.5 billion reinsurance cover issued by NICO. The point estimate of incurred losses ceded to NICO increased from \$2.2 billion to \$2.3 billion. There was no impact to adjusted book value per share.

#### **Adjusted Book Value Per Share**

The following table presents White Mountains' adjusted book value per share and reconciles this non-GAAP measure to the most comparable GAAP measure. (See **NON-GAAP FINANCIAL MEASURES** on page 55).

	Sept. 30, 2011		June 30, 2011		Dec. 31, 2010		Sept. 30, 2010
Book value per share numerators (in millions):		_		_	•		
White Mountains' common shareholders' equity	\$	3,444.7	\$	3,620.5	\$	3,653.0	\$ 3,667.8
Benefits to be received from common share obligations under							
employee stock option plans(1)		_		_		_	_
GAAP book value per share numerator		3,444.7		3,620.5		3,653.0	3,667.8
Equity in net unrealized gains from Symetra's fixed maturity							
portfolio, net of applicable taxes		(135.7)		(80.6)		(58.5)	(110.9)
Adjusted book value per share numerator	\$	3,309.0	\$	3,539.9	\$	3,594.5	\$ 3,556.9
Book value per share denominators (in thousands of shares):							
Common shares outstanding		7,630.7		7,958.6		8,194.9	8,343.0
Common share obligations under employee stock option plans (1)		_		_		_	_
GAAP book value per share denominator		7,630.7		7,958.6		8,194.9	8,343.0
Unearned restricted common shares		(44.5)		(51.5)		(36.5)	(47.7)
Adjusted book value per share denominator		7,586.2		7,907.1		8,158.4	8,295.3
GAAP book value per share	\$	451.42	\$	454.92	\$	445.76	\$ 439.62
Adjusted book value per share	\$	436.18	\$	447.69	\$	440.59	\$ 428.79

<sup>(1)</sup> Assumes conversion of in-the money stock options

#### **Esurance Sale**

On October 7, 2011, White Mountains completed the sale of Esurance Insurance and AFI to The Allstate Corporation for a cash payment of \$1.01 billion, which was equal to \$700 million plus the estimated pro forma tangible book value at closing of the legal entities sold of approximately \$310 million (the "Esurance Sale"). The final purchase price is subject to true-up of estimated tangible book value of the entities sold through the date of the closing. As a result of the Esurance Sale, the Esurance Disposal Group is reported as discontinued operations (see **III. Discontinued Operations** on page 46).

Pro-forma for the Esurance Sale, adjusted book value per share was \$523 at September 30, 2011, or \$87 higher than actual adjusted book value per share. This pro-forma increase differs from White Mountains' previous estimate of an \$80 increase in adjusted book value per share due to certain deal-related compensation expenses that were recorded and reflected in adjusted book value per share prior to the fourth quarter and the reduced number of shares outstanding as a result of share repurchases completed in September 2011.

#### **Table of Contents**

#### **Review of Consolidated Results**

White Mountains' consolidated financial results for the three and nine months ended September 30, 2011 and 2010 follow:

		Three Mon Septem			Nine Months Ended September 30,				
Millions		2011	JCI 30,	2010		2011	ocr ou	2010	
Gross written premiums	\$	555.7	\$	576.3	\$	1,809.8	\$	2,221.6	
Net written premiums	\$	510.9	\$	456.0	\$	1,589.8	\$	1,679.2	
Revenues									
Earned insurance and reinsurance premiums	\$	490.6	\$	498.1	\$	1,436.7	\$	1,772.9	
Net investment income		42.8		48.9		138.1		157.4	
Net realized and unrealized investment gains		1.7		11.6		78.3		104.7	
Other revenue — foreign currency translation gains (losses)		(29.2)		43.1		(57.9)		22.9	
Other revenue — Tuckerman Fund I		8.0		9.0		18.9		18.4	
Other revenue — Symetra warrants		(24.7)		(10.7)		(29.3)		(21.6)	
Other revenue — other		12.1		(49.0)		(.3)		(40.8)	
Other revenue (losses)		(33.8)		(7.6)		(68.6)		(21.1)	
Total revenues		501.3		551.0		1,584.5		2,013.9	
Expenses			-						
Losses and LAE		277.1		253.4		914.0		1,175.9	
Insurance and reinsurance acquisition expenses		107.0		102.1		296.2		370.9	
Other underwriting expenses		67.0		75.0		211.5		245.9	
General and administrative expenses		27.1		22.5		97.5		97.0	
General and administrative expenses — Tuckerman Fund I		7.0		7.5		16.7		16.2	
Accretion of fair value adjustment to loss and LAE reserves		2.1		2.1		6.2		6.3	
Interest expense on debt		12.8		13.2		38.8		44.2	
Total expenses		500.1		475.8		1,580.9		1,956.4	
Pre-tax income		1.2		75.2		3.6		57.5	
Income tax expense		.6		(11.7)		(1.9)		(10.3)	
Net income from continuing operations	_	1.8		63.5		1.7		47.2	
Net loss on sale of AutoOne, net of tax		(18.2)		_		(18.2)		_	
Net income (loss) from discontinued operations, net of tax		(11.9)		3.6		(10.7)		1.2	
Equity in earnings of unconsolidated affiliates, net of tax		1.5		7.9		16.1		1.6	
Net income (loss)		(26.8)		75.0		(11.1)		50.0	
Net income (loss) attributable to noncontrolling interests		11.0		(24.3)		(21.2)		(36.4)	
Net income (loss) attributable to White Mountains' common	-			(= 1.15)		(==:=)		(331.1)	
shareholders		(15.8)		50.7		(32.3)		13.6	
Other comprehensive income (loss), net of tax		(26.7)		170.0		51.9		172.3	
Comprehensive income (loss)	-	(42.5)		220.7		19.6		185.9	
Comprehensive income attributable to noncontrolling interests		(12.5)							
Comprehensive income (loss) attributable to White							_		
Mountains' common shareholders		(42.5)		220.7		19.6		185.9	
Change in equity in net unrealized gains from Symetra's fixed		(12,0)				10.0		100.0	
maturity portfolio		(55.1)		(43.1)		(77.2)		(125.9)	
Adjusted comprehensive income (loss)	\$	(97.6)	\$	177.6	\$	(57.6)	\$	60.0	
Aujusteu comprehensive income (1088)	9	(37.0)	Φ	1//.0	Φ	(37.0)	ψ	00.0	

#### Consolidated Results —Three Months Ended September 30, 2011 versus Three Months Ended September 30, 2010

White Mountains' total revenues decreased 9% to \$501 million in the third quarter of 2011 compared to \$551 million in the third quarter of 2010, primarily due to foreign currency translation, lower earned insurance and reinsurance premiums, lower net investment income and lower investments gains. Earned premiums were down 2% to \$491 million in the third quarter of 2011 as a 12% decrease at OneBeacon, which was driven by the Commercial Lines Transaction, was partially offset by a 13% increase at Sirius Group. Excluding the \$44 million of earned premiums in the third quarter of 2010 related to the exited businesses at OneBeacon, White Mountains' earned premiums increased 8% in the third quarter of 2011. Net investment income was down 12% to \$43 million in the third quarter of 2011, due primarily to lower fixed maturity yields and a reduction in invested assets from the Personal Lines Transaction, OBH Senior Notes repurchases at OneBeacon and the Company's share repurchase program. White Mountains reported net realized and unrealized investment gains of \$2 million in the third quarter of 2011 compared to \$12 million in the third quarter of 2010 (see **Investment Returns** on page 45). Other revenues decreased to a loss of \$34 million in the third quarter of 2011 from a loss of \$8 million in the third quarter of 2010, due primarily to increased foreign currency translation losses and higher mark-to-market losses on the Symetra warrants, partially offset by lower losses reported by WM Life Re. WM Life Re reported essentially break-even results in the third quarter of 2011 compared to \$53 million of pre-tax losses in the third quarter of 2010.

38

#### **Table of Contents**

The third quarter of 2010 included a \$48 million charge recorded as a reduction of other revenues from lowered surrender assumptions used to calculate WM Life Re's variable annuity guarantee liability.

White Mountains' total expenses increased 5% to \$500 million in the third quarter of 2011 compared to \$476 million in the third quarter of 2010. Excluding \$44 million in expenses reported in the third quarter of 2010 associated with businesses that OneBeacon exited through the Commercial Lines Transaction, White Mountains' total expenses increased 16% in the third quarter of 2011, as losses and LAE expenses increased 22%, insurance and reinsurance acquisition expenses increased 16% and other underwriting expenses would have been essentially flat. Losses and LAE expenses included \$42 million of catastrophe losses in the third quarter of 2011, primarily related to hurricane Irene, compared to \$17 million of catastrophe losses in the third quarter of 2010.

White Mountains' income tax benefit for the third quarter of 2011 was not meaningful as pre-tax income was near break-even (\$1 million). White Mountains' effective tax rate for the third quarter of 2010 represented an effective tax rate of 15.6%, which differed from the U.S. statutory rate of 35% primarily due to income generated in jurisdictions other than the United States.

#### Consolidated Results — Nine Months Ended September 30, 2011 versus Nine Months Ended September 30, 2010

White Mountains' total revenues decreased 21% to \$1,585 million in the first nine months of 2011 compared to \$2,014 million in the first nine months of 2010, primarily due to lower earned insurance and reinsurance premiums, foreign currency translation, lower net investment income and lower investments gains, partially offset by lower losses at WM Life Re. Earned premiums were down 19% to \$1,437 million in the first nine months of 2011 as a 34% decrease at OneBeacon, which was driven by the Personal Lines Transaction and the Commercial Lines Transaction, was partially offset by a 7% increase at Sirius Group. Excluding the \$408 million of earned premiums in the first nine months of 2010 related to the exited businesses at OneBeacon, White Mountains' earned premiums increased 4% in the first nine months of 2011. Net investment income was down 12% to \$138 million in the first nine months of 2011, due primarily to lower fixed maturity yields and a reduction in invested assets from the Personal Lines Transaction, OBH Senior Notes repurchases at OneBeacon and share repurchases. White Mountains reported net realized and unrealized investment gains of \$78 million in the first nine months of 2011 compared to \$105 million in the first nine months of 2010. Both periods were significantly impacted by foreign currency translation on U.S. dollar-denominated investments at Sirius International, the effects of which are offset in other comprehensive income (see **Investment Returns** on page 45). Other revenues decreased to a loss of \$69 million in the first nine months of 2011 compared to \$2010. Other revenues included \$58 million in foreign currency translation gains in the first nine months of 2010. Other revenues also included a \$1 million loss from WM Life Re in the first nine months of 2011 compared to a \$52 million loss in the first nine months of 2010. Other revenues for the first nine months of 2010 also included a \$13 million pre-tax gain from the acquisition of Central National by the Sirius Group.

White Mountains' total expenses decreased 19% to \$1,581 million in the first nine months of 2011 compared to \$1,956 million in the first nine months of 2010. Losses and LAE expenses decreased \$262 million, or 22%, insurance and reinsurance acquisition expenses decreased \$75 million, or 20%, and other underwriting expenses decreased \$34 million, or 14%, all due primarily to the Personal Lines Transaction and Commercial Lines Transaction. Excluding \$464 million in expenses reported in the first nine months of 2010 related to the exited businesses at OneBeacon, White Mountains' total expenses increased 5% in the first nine months of 2011, as losses and LAE expenses, insurance and reinsurance acquisition expenses and other underwriting expenses increased 4%, 5% and 10%. Other underwriting expenses at Sirius Group increased 15%, primarily due to foreign currency translation. Interest expense on debt decreased 12% to \$39 million in the first nine months of 2011 compared to \$44 million in the first nine months of 2010, primarily due to reductions of outstanding debt resulting from repurchases of OBH Senior Notes.

White Mountains' effective tax rate for the first nine months of 2011 was not meaningful as pre-tax income was near break-even (\$4 million). White Mountains' income tax benefit for the first nine months of 2010 represented an effective tax rate of 17.9%, which differed from the U.S. statutory rate of 35% primarily due to income generated in jurisdictions other than the United States.

39

#### **Table of Contents**

#### I. Summary of Operations By Segment

White Mountains conducts its operations through three segments: (1) OneBeacon, (2) Sirius Group and (3) Other Operations. White Mountains manages all of its investments through its wholly-owned subsidiary, WM Advisors, therefore, a discussion of White Mountains' consolidated investment operations is included after the discussion of operations by segment. White Mountains' segment information is presented in **Note 10** to the Consolidated Financial Statements.

#### **OneBeacon**

OneBeacon's book value per share decreased 3% for the third quarter and increased 2% for the first nine months of 2011, including dividends. The decrease for the third quarter was driven by declines in equity investments and an after-tax charge of \$18 million for the estimated loss on the sale of AutoOne. On August 30, 2011, OneBeacon entered into an agreement to sell its AutoOne business to Interboro. AutoOne operates as a division within OneBeacon, offering products and services to automobile assigned risk markets. The transaction includes the sale of two insurance entities, AOIC and AOSIC, through which substantially all of the AutoOne business is written on a direct basis. The results of operations for the AutoOne business, net of related income taxes, are presented as discontinued operations in the statement of comprehensive income (see III. Discontinued Operations on page 46).

Financial results for OneBeacon for the three and nine months ended September 30, 2011 and 2010 follow:

	Three Mon Septem	ed	Nine Mont Septem		
Millions	 2011		2010	2011	2010
Gross written premiums	\$ 305.2	\$	361.9	\$ 865.6	\$ 1,303.8
Net written premiums	\$ 296.8	\$	268.8	\$ 822.3	\$ 943.0
Earned insurance and reinsurance premiums	\$ 258.6	\$	293.6	\$ <b>751.2</b>	\$ 1,134.7
Net investment income	16.1		21.6	55.8	74.9
Net realized and unrealized investment gains (losses)	(47.4)		51.6	(13.3)	79.6
Other revenue (losses)	.1		12.7	(10.6)	1.9
Total revenues	 227.4		379.5	783.1	1,291.1
Losses and LAE					

	146.6	155.2	414.9	719.3
Insurance and reinsurance acquisition expenses	58.6	64.3	161.1	248.3
Other underwriting expenses	38.7	50.5	132.1	176.6
General and administrative expenses	2.5	2.1	7.4	9.4
Interest expense on debt	4.1	6.4	16.4	23.4
Total expenses	250.5	278.5	731.9	1,177.0
Pre-tax income (loss)	\$ (23.1)	\$ 101.0	\$ 51.2	\$ 114.1

The following table presents OneBeacon's GAAP book value per share.

	Sept. 30,	June 30,	Dec. 31,	Sept. 30,
Millions, except per share amounts	 2011	2011	2010	2010
OneBeacon's common shareholders' equity	\$ 1,100.2	\$ 1,152.3	\$ 1,229.0	\$ 1,219.1
OneBeacon's common shares outstanding	 95.1	95.1	 94.4	 94.4
OneBeacon's GAAP book value per share	\$ 11.57	\$ 12.12	\$ 13.02	\$ 12.91

The following tables provide OneBeacon's GAAP ratios, net written premiums and earned insurance premiums for the three and nine months ended September 30, 2011 and 2010:

	Three Months Ended September 30, 2011						Nine Months Ended September 30, 2011					
GAAP Ratios:	 Specialty Other(1)			Total		Specialty		Specialty Other(1)		Other(1)		Total
Loss and LAE	58%		n/m		57%		57%		n/m		55%	
Expense	36%		n/m		37%		38%		n/m		39%	
Combined	94%		n/m		94%		95%		n/m		94%	
Net written premiums	\$ 297.2	\$	(.4)	\$	296.8	\$	821.5	\$	.8	\$	822.3	
Earned premiums	\$ 259.1	\$	(.5)	\$	258.6	\$	747.9	\$	3.3	\$	751.2	

40

#### **Table of Contents**

		Three Months Ended September 30, 2010						Nine Months Ended September 30, 2010							
GAAP Ratios:	S	pecialty		Other(1)		Total		Specialty		Other(1)		Total			
Loss and LAE		52%		60%		53%		56%		76%		63%			
Expense		38%		41%		39%		38%		38%		38%			
Combined		90%		101%		92%		94%		114%		101%			
Net written premiums	\$	273.5	\$	(4.7)	\$	268.8	\$	771.6	\$	171.4	\$	943.0			
Earned premiums	\$	249.6	\$	44.0	\$	293.6	\$	726.7	\$	408.0	\$	1,134.7			

<sup>(1)</sup> Other is primarily businesses that are now in run-off or have been sold prior to 2011. Accordingly, GAAP ratios for the three and nine months ended September 30, 2011 are not meaningful.

## OneBeacon Results - Three Months Ended September 30, 2011 versus Three Months Ended September 30, 2010

OneBeacon's GAAP combined ratio was 94% for the third quarter of 2011 compared to 92% for the third quarter of 2010. Catastrophe losses, principally related to hurricane Irene, increased the combined ratio by 5 points in the third quarter 2011, compared to a decrease of 1 point in the third quarter of last year from a change in estimated catastrophe losses incurred earlier in 2010. Prior year favorable loss reserve development in the third quarter of 2011 decreased the combined ratio by 4 points compared to 6 points in the third quarter of 2010.

OneBeacon's net written premiums increased 10% in the three months ended September 30, 2011 to \$297 million, compared to \$269 million for the three months ended September 30, 2010, primarily due to new business and improved retention in several lines, particularly within the government risk, entertainment and collector car and boat businesses.

During the third quarter of 2011, OneBeacon completed a new ground-up study of its legacy asbestos and environmental exposures. The previous study was based on experience through 2007. Reasonable estimates of potential adverse scenarios continue to be within the \$2.5 billion reinsurance cover issued by NICO. The point estimate of incurred losses ceded to NICO increased from \$2.2 billion to \$2.3 billion. There was no impact to book value.

**Reinsurance protection.** For the third quarter of 2011, OneBeacon's specialty lines' net combined ratio was higher than its gross combined ratio by 3 points, primarily due to the impact of the cost of facultative reinsurance, professional liability reinsurance and marine reinsurance, and to a lesser extent the cost of catastrophe reinsurance and casualty reinsurance. For the third quarter of 2010, OneBeacon's specialty lines' net combined ratio was higher than its gross combined ratio by 4 points, primarily due to the cost of facultative reinsurance, catastrophe reinsurance, casualty reinsurance and property reinsurance.

# OneBeacon Results - Nine Months Ended September 30, 2011 versus Nine Months September 30, 2010

OneBeacon's GAAP combined ratio was 94% for the first nine months of 2011 compared to 101% for the first nine months of last year. The GAAP combined ratio for the first nine months of 2010 included unusually high non-catastrophe losses, primarily a number of large losses in businesses that OneBeacon has since exited and in its property and inland marine specialty businesses. Catastrophe losses increased the combined ratio by 5 points in both the first nine months of 2011 and the first nine months of last year. Favorable loss reserve development in the first nine months of 2011 decreased the combined ratio by 3 points compared to 4 points in the first nine months of last year.

OneBeacon's net written premiums decreased 13% to \$822 million in the first nine months of 2011, reflecting decreases from the Commercial Lines Transaction and the Personal Lines Transaction. OneBeacon's specialty insurance net written premiums increased 6% in the first nine months of 2011 to \$822 million, primarily due to new business and improved retention in several lines, particularly within the accident, government risk, energy and collector car and boat businesses.

**Reinsurance protection.** For the first nine months of 2011, OneBeacon's specialty lines' net combined ratio was higher than its gross combined ratio by 4 points, primarily due to the impact of the cost of property reinsurance and facultative reinsurance, and also the cost of catastrophe reinsurance and marine reinsurance. For the first nine months of 2010, OneBeacon's specialty lines' net combined ratio was higher than its gross combined ratio by 3 points, primarily due to the cost of facultative reinsurance and catastrophe reinsurance.

#### **Table of Contents**

#### Sirius Group

Financial results and GAAP combined ratios for Sirius Group for the three and nine months ended September 30, 2011 and 2010 follow:

	Three Months Ended September 30,					Nine Months Ended September 30,				
(\$ in millions)		2011		2010		2011	2010			
Gross written premiums	\$	250.5	\$	214.4	\$	944.2	\$	917.7		
Net written premiums	\$	214.1	\$	187.2	\$	767.5	\$	736.2		
Earned insurance and reinsurance premiums	\$	232.0	\$	204.5	\$	685.5	\$	638.2		
Net investment income		22.1		23.0		68.3		71.3		
Net realized and unrealized investment gains (losses)		64.7		(51.8)		102.6		13.7		
Other revenue - foreign currency translation gains (losses)		(29.2)		43.1		(57.9)		22.9		
Other revenue		1.1		.2		2.2		13.5		
Total revenues		290.7		219.0		800.7		759.6		
Losses and LAE		130.5		98.2		499.1		456.6		
Insurance and reinsurance acquisition expenses		48.4		37.8		135.1		122.6		
Other underwriting expenses		28.3		24.5		79.4		69.3		
General and administrative expenses		5.3		2.7		17.1		16.9		
Accretion of fair value adjustment to losses and LAE										
reserves		2.1		2.1		6.2		6.3		
Interest expense on debt		6.5		6.7		19.6		20.1		
Total expenses		221.1		172.0		<b>756.5</b>		691.8		
Pre-tax income	\$	69.6	\$	47.0	\$	44.2	\$	67.8		
GAAP ratios:										
Losses and LAE		56%		48%		73%		72%		
Expense		33%		30%		31%		30%		
Combined		89%		78%		104%		102%		

# Sirius Group Results - Three Months Ended September 30, 2011 versus Three Months Ended September 30, 2010

Sirius Group's GAAP combined ratio was 89% for the third quarter of 2011 compared to 78% for the third quarter of 2010. The increase was primarily due to worse current accident year underwriting results in the accident and health, marine and aviation lines and higher catastrophe losses, partially offset by improved underwriting results in the trade credit line. Catastrophe losses increased the combined ratio by 11 points in the third quarter of 2011 compared to 9 points in the third quarter of 2010. Catastrophe losses in the third quarter of 2011 were primarily composed of \$11 million (5 points) of losses from hurricane Irene, \$7 million (3 points) of losses from the June 2011 New Zealand earthquake and \$6 million (2 points) of losses from the July Copenhagen floods. Catastrophe losses in the third quarter of 2010 were primarily composed of \$4 million of flood losses in Europe and \$4 million of losses from storms in Canada. The third quarter of 2011 included 9 points of favorable loss reserve development primarily attributable to the property lines, which was partially offset by approximately 2 points of related increases to profit commission expense. The third quarter of 2010 included 7 points (\$14 million) of favorable loss reserve development from short-tailed lines of business, primarily property, accident and health and aviation.

Sirius Group's gross written premiums increased 17% to \$251 million in the third quarter of 2011 from \$214 million in the third quarter of 2010, while net written premiums increased 14% to \$214 million in the third quarter of 2011 from \$187 million in the third quarter of 2010. Earned premiums increased 13% to \$232 million from \$205 million in the third quarter of 2010. These increases were primarily due to increases in the accident and health and trade credit lines and to foreign curency translation.

Sirius Group's total other revenues decreased to a loss of \$28 million in the third quarter of 2011 from a gain of \$43 million in the third quarter of 2010. Sirius Group's other revenues included \$29 million of foreign currency translation losses in the third quarter of 2011 compared to \$43 million of foreign currency translation gains in the third quarter of 2010.

42

# **Table of Contents**

Reinsurance protection. Sirius Group's reinsurance protection primarily consists of pro-rata and excess of loss protections to cover aviation, trade credit, and certain property exposures. Sirius Group's property reinsurance provides both proportional and non-proportional protections for exposures written in Europe, the Americas, Asia, the Middle East, and Australia. This reinsurance is designed to increase underwriting capacity where appropriate, and to reduce exposure both to large catastrophe losses and to a frequency of smaller loss events. Attachment points and coverage limits vary by region around the world. For 2011 and 2010, industry loss warranties ("ILW") for European wind exposures and floods with \$20 million of available limit were purchased, attaching at various levels.

In 2011 and 2010, Sirius Group purchased group excess of loss retrocessional coverage for its non-U.S. and non-Japan earthquake-related exposures. The non-U.S. and non-Japan earthquake cover was renewed for one year at April 1, 2011, providing \$61 million of reinsurance protection through partially placed coverage layers in excess of Sirius Group's retention of \$35 million. Sirius Group will recognize the full \$61 million recovery if a covered earthquake loss reaches \$110 million. At April 1, 2010, Sirius Group purchased coverage for \$65 million in excess of \$45 million, and this coverage applies to losses incurred from the February 2011 New Zealand earthquake. At September 30, 2011, losses incurred for the February 2011 New Zealand earthquake totaled \$42 million. Accordingly, Sirius Group has not ceded any losses to this coverage.

Sirius Group's net combined ratios for the third quarter of 2011 and the third quarter of 2010 were 9 points higher and 1 point lower than its gross combined ratios for those periods. The net combined ratio for the third quarter of 2011 was higher primarily due to the cost of the ILWs, aviation reinsurance, and other excess of loss protections with limited incurred loss cessions.

#### Sirius Group Results - Nine Months Ended September 30, 2011 versus Nine Months Ended September 30, 2010

Sirius Group's GAAP combined ratio was 104% for the first nine months of 2011 compared to 102% for the first nine months of 2010. Catastrophe losses increased the combined ratio by 27 points in the first nine months of 2011 compared to 28 points in the first nine months of 2010. For the first nine months of 2011, catastrophe losses were primarily composed of \$90 million (13 points) of losses from the Japan earthquake and tsunami, \$49 million (7 points) of losses from the February and June 2011 New Zealand earthquake and \$25 million (4 points) of losses from severe weather and tornados in the Midwestern United States, while catastrophe losses for the first nine months of 2010 were primarily due to the Chile earthquake. The first nine months of 2011 included 5 points of favorable loss reserve development, primarily attributable to property lines, which was partially offset by related premium adjustments and profit commission expense. The first nine months of 2010 included 4 points of favorable loss reserve development that was mostly from short-tailed lines of business, primarily property, accident and health, and aviation.

Sirius Group's recorded property losses from the earthquake and tsunami in Japan have been estimated based on information from ceding companies and using third party and internal catastrophe models, applying overall estimates of industry insured losses to Sirius Group's exposure information. As of September 30, 2011, 46% of Sirius Group's ultimate loss estimate for the Japan earthquake had been paid or reported. In relation to the Chile earthquake in 2010, at the same relative point in time (7 months post-event), approximately 79% of Sirius Group's ultimate loss estimate for the Chile earthquake had been paid or reported.

Sirius Group's gross written premiums increased 3% to \$944 million in the first nine months of 2011 from \$918 million in 2010, while net written premiums increased 4% to \$768 million in the first nine months of 2011 from \$736 million in 2010. These increases were primarily due to foreign exchange translation and increases in the accident and health and trade credit lines of business. Earned premiums increased 7% to \$686 million in first nine months of 2011 from \$638 million in the first nine months of 2010. In addition to the changes noted above for written premiums, earned premiums increased due to changes in business mix, most notably an increase in accident and health and trade credit premiums (which are recognized as earned premium over an extended period of time), somewhat offset by decreases in property earned premiums.

Sirius Group's other revenues consisted primarily of \$58 million of foreign currency translation losses recorded in the first nine months of 2011 compared to \$23 million of foreign currency translation gains in the first nine months of 2010. Additionally, during the first nine months of 2010, Sirius Group acquired Central National and recorded a pre-tax gain of approximately \$13 million in other revenues that reflects the excess of the fair value of the net assets acquired over the consideration paid.

**Reinsurance protection.** Sirius Group's net combined ratio for the first nine months of 2011 was higher than its gross combined ratio by 9 points. For the first nine months of 2010, the net combined ratio was lower than the gross combined ratio by 8 points. The 2011 net combined ratio was higher primarily due to the cost of the ILWs, aviation reinsurance, and other excess of loss protections with limited incurred loss cessions. The lower net combined ratio for the first nine months of 2010 was primarily due to significant retrocessional recoveries, including recovery of the full \$65 million limit under Sirius Group's non-U.S./non-Japan earthquake cover triggered by losses from the Chile earthquake.

43

# Table of Contents

#### **Other Operations**

Other Operations consists of the operations of the Company, the Company's intermediate holding companies, the consolidated results of the Tuckerman Fund I, WM Advisors and White Mountains' investments in unconsolidated affiliates. The Other Operations segment also includes the results of WM Life Re, which is in run-off.

A summary of White Mountains' financial results from its Other Operations segment for the three and nine months ended September 30, 2011 and 2010 follows:

	Three Months Ended September 30,					Nine Months Ended September 30,			
Millions		2011		2010		2011	2010		
Net investment income	\$	4.6	\$	4.3	\$	14.0	\$	11.2	
Net realized and unrealized investment gains (losses)		(15.6)		11.8		(11.0)		11.4	
Other revenue - Tuckerman Fund I		8.0		9.0		18.9		18.4	
Other revenue (losses) — Symetra warrants		(24.7)		(10.7)		(29.3)		(21.6)	
Other revenue (losses)		10.9		(61.9)		8.1		(56.2)	
Total revenues		(16.8)		(47.5)		.7		(36.8)	
General and administrative expenses - Tuckerman Fund									
I		7.0		7.5		16.7		16.2	
General and administrative expenses		19.3		17.7		73.0		70.7	
Interest expense — debt		2.2		.1		2.8		.7	
Total expenses		28.5		25.3		92.5		87.6	
Pre-tax loss	\$	(45.3)	\$	(72.8)	\$	(91.8)	\$	(124.4)	

White Mountains' Other Operations segment reported pre-tax loss in the third quarter and first nine months of 2011 of \$45 million and \$92 million, compared to \$73 million and \$124 million in the third quarter and first nine months of 2010. The decrease in pre-tax loss for the third quarter and first nine months of 2011 was driven by lower losses from WM Life Re, partially offset by higher mark-to-market losses on the Symetra warrants and lower net investment gains. WM Life Re reported essentially break-even results in the third quarter of 2011 and a \$10 million pre-tax loss in the first nine months of 2011 compared to \$53 million of pre-tax loss in the third quarter of 2010 and \$65 million of pre-tax loss in the first nine months of 2010. The value of White Mountains' investment in Symetra warrants decreased \$25 million and \$29 million in the third quarter and first nine months of 2011 compared to a decrease of \$11 million and \$21 million in the third quarter and first nine months of 2010.

During the third quarter of 2010, WM Life Re lowered the surrender assumptions used to calculate its variable annuity guarantee liability. WM Life Re's previous assumptions had reflected its expectation that surrenders would rise as the surrender charges in the underlying annuities decline. However, the persistent instability in financial and foreign exchange markets has kept surrenders low. The lower surrender assumption resulted in a \$48 million increase in WM Life Re's variable annuity guarantee liability, which was recorded as a reduction of other revenues in the third quarter of 2010.

As discussed on page 54 under "FAIR VALUE CONSIDERATIONS", in determining the fair value of the benefit guarantees that WM Life Re reinsures, management makes assumptions about future policyholder surrender rates. If actual surrenders are lower than the surrender assumptions used in WM Life Re's liability valuation model, the liability for the benefit guarantees would increase. For example, a 50% reduction in the surrender assumptions used in WM Life Re's valuation model at September 30, 2011 would have increased the fair value of the liability by approximately \$9 million.

44

#### **Table of Contents**

#### II. Summary of Investment Results

#### **Investment Returns**

For purposes of discussing rates of return, all percentages are presented gross of management fees and trading expenses in order to produce a more relevant comparison to benchmark returns, while all dollar amounts are presented net of any management fees and trading expenses. A summary of White Mountains' consolidated pre-tax investment results and gross investment returns versus typical benchmarks for the three and nine months ended September 30, 2011 and 2010 follows:

Pre-tax investment results		Three Mon Septem	 	 Nine Months Ended September 30,				
Millions		2011	2010	2011	2010			
Net investment income		42.8	\$ 48.9	\$ 138.1	\$	157.4		
Net realized and unrealized investment gains		1.7	11.6	78.3		104.7		
Net unrealized foreign currency gains (losses) on								
investments (reported in other comprehensive income)		(171.0)	259.9	(94.7)		104.5		
Pre-tax investment gains (losses) included in discontinued								
operations		(2.6)	16.2	13.2		32.4		
Total GAAP pre-tax investment gains (losses)		(129.1)	\$ 336.6	\$ 134.9	\$	399.0		

At September 30, 2011, White Mountains' investment portfolio included \$1.0 billion in non-U.S. dollar-denominated investments, most of which are held at Sirius International and are denominated in Swedish kronor or euros. The value of the investments in this portfolio is impacted by changes in the exchange rate between the U.S. dollar and the krona and between the U.S. dollar and the euro. During the third quarter and first nine months of 2011, the U.S. dollar strengthened 9% and 3% against the kronor. During the third quarter and first nine months of 2011, the U.S dollar strengthened 8% and was flat against the euro. These currency movements resulted in approximately \$96 million and \$15 million of pre-tax foreign currency investment losses for the three and nine months ended September 30, 2011, which are recorded as components of net realized and unrealized investment gains and unrealized foreign currency gains and losses on investments. During the third quarter and first nine months of 2010, the U.S. dollar weakened 14% and 6% against the kronor. During the third quarter and first nine months of September 30, 2010, the U.S. dollar weakened 5% against the euro, which resulted in \$173 million and \$32 million of pre-tax foreign currency gains for the three and nine months ended September 30, 2010.

Sirius International also holds a large portfolio of investments that are denominated in U.S. dollars, but its functional currency is the Swedish kronor. When Sirius International prepares its stand-alone GAAP financial statements, it translates its U.S. dollar-denominated investments to Swedish kronor and recognizes the related foreign currency translation gains or losses through income. When White Mountains consolidates Sirius International, it translates Sirius International's stand-alone GAAP financial statements to U.S. dollars and recognizes the foreign currency gains or losses arising from this translation, including those associated with Sirius International's U.S. dollar-denominated investments, through other comprehensive income. Since White Mountains reports its financial statements in U.S. dollars, there is no net effect to adjusted book value per share or to investment returns from foreign currency translation on its U.S. dollar-denominated investments at Sirius International. However, net realized and unrealized investment gains, foreign currency translation and other comprehensive income can be significantly affected during periods of high volatility in the foreign exchange rate between the U.S. dollar and the Swedish kronor, as was the case during the first and third quarter of 2011.

The amount of foreign currency translation on Sirius International's U.S. dollar denominated investments recognized as a decrease of other comprehensive income and an increase of net income was \$66 million and \$27 million for the third quarter and first nine months of 2011. The amount of foreign currency translation on Sirius International's U.S. dollar denominated investments recognized as an increase of other comprehensive income and a decrease of net income (loss) was \$101 million and \$48 million for the third quarter and first nine months of 2010.

	Three Months Ended September 30,		Nine Months En September 3	
	2011	2010	2011	2010
Fixed maturity investments	-0.5%	3.9%	2.7%	5.5%
Short-term investments	-0.2%	2.2%	.8%	.2%
Total fixed income	-0.5%	3.5%	2.5%	4.3%
Barclay's U.S. Intermediate Aggregate Index	2.3%	2.1%	5.0%	7.0%
Common stocks	-11.3%	9.6%	-6.7%	7.9%
Convertible securities	-8.5%	5.8%	-6.8%	8.2%
Other long-term investments	-0.1%	3.6%	6.8%	4.9%
Total equities, convertible securities and other long-				
term investments	-7.8%	6.9%	-2.8%	6.9%
S&P 500 Index (total return)	-13.9%	11.3%	-8.7%	3.9%
Total consolidated portfolio	-1.5%	4.0%	1.7%	4.6%

White Mountains' GAAP pre-tax total return on invested assets was -1.5% and 1.7% for the third quarter and first nine months of 2011, which included 1.2% and 0.2% of foreign currency losses, compared to 4.0% and 4.6% for the third quarter and first nine months of 2010, including 1.9% and 0.5% of foreign currency gains. White Mountains' fixed income portfolio returned -0.5%, which included -1.4% of foreign currency losses for the third quarter of 2011, underperforming the Barclays U.S. Intermediate Aggregate returns of 2.3%. White Mountains' high-quality, short-duration, fixed income portfolio (a duration of approximately 2 years, including short term investments, and an average credit rating of AA at September 30, 2011) trailed the longer-duration benchmark as rates fell during the third quarter of 2011. For the first nine months of 2011, the fixed income portfolio return of 2.5%, which included -0.2% of foreign currency losses also underperformed its benchmark for the same reason.

White Mountains' equity portfolio, approximately 16% of GAAP invested assets at September 30, 2011, returned -7.8% for the third quarter of 2011, compared to the S&P 500 Index return of -13.9%, while the equity portfolio returned -2.8% in the first nine months of 2011, compared to the S&P 500 Index return of -8.7%. Equity accounts managed by Prospector Partners have outperformed the S&P 500 by 2.8% in the third quarter and 0.7% for the first nine months of 2011. These accounts have overweight positions in the gold mining sector when compared to the S&P 500 Index and also contain convertible bond investments that generally limit downside price exposure to the underlying value of the fixed maturity component of the instrument. In addition, other long-term investments, which are composed principally of hedge funds and private equities, have outperformed the S&P 500 by 13.0% in the third quarter and 13.6% for the first nine months of 2011.

Net investment income was down 12% in each of the third quarter and first nine months of 2011, due primarily to lower fixed maturity yields and a reduction in invested assets from the Personal Lines Transaction, the OBH Senior Notes repurchases at OneBeacon and share repurchases.

#### **III. Discontinued Operations**

# Esurance

As a result of the Esurance Sale, the Esurance Disposal Group is presented as discontinued operations and the results of these businesses are shown as a single line item in the accompanying Consolidated Statement of Operations. In addition, the assets and liabilities associated with the business being sold are presented separately as single line items in the asset and liability sections of the Consolidated Balance Sheet. Prior year amounts have been reclassified to conform to the current period's presentation.

The amounts reflected within discontinued operations differ from amounts previously presented within the Esurance segment. The segment results of operations for Esurance reported in previous periods included investment income and realized and unrealized investment gains and losses allocated to Esurance related to capital that was used by Esurance but contained within Sirius Group's legal entities. These allocations do not have separately identifiable cash flows and have therefore been excluded from amounts classified in discontinued operations and are now included in the Sirius Group segment. For the three and nine month periods of September 30, 2011, \$1 million and \$(2) million of net investment income and realized and unrealized investment gains and losses that had been previously included in the Esurance segment were excluded from net income (loss) from discontinued operations. For the three and nine month period September 30, 2010, \$1 million and \$4 million of net investment income and realized and unrealized investment gains and losses that had been previously included in the Esurance segment were excluded from net income (loss) from discontinued operations.

Net income (loss) from discontinued operations, net of tax, related to the business subject to the Esurance Sale was \$(9) million and \$(6) million in the third quarter and first nine months of 2011 and \$7 million in both the third quarter and first nine months of 2010. The results for the first nine months of 2011 included a \$10 million after-tax charge (\$16 million pre-tax) from increased compensation accruals recorded as a result of the Esurance Sale.

46

# **Table of Contents**

Esurance's adjusted combined ratios for the three and nine months ended September 30, 2011 and 2010 follows:

	Three Months Ended September 30,			Nine Months I September 3				
(\$ in millions)	2011		2010		2011		2010	
Adjusted ratios(1):								
Losses and LAE	75%		70%		73%	)	73%	
Adjusted expense	29%		30%	29%		)	29%	
Adjusted combined	104%		100%		102%		102%	
Gross Written Premiums	\$ 230.8	\$	222.0	\$	675.1	\$	648.4	
Net Written Premiums	\$ 229.8	\$	221.2	\$	672.5	\$	645.9	
Earned Premiums	\$ 217.6	\$	210.3	\$	646.6	\$	618.9	

(1) Adjusted expense and combined ratios include acquisition expenses net of referral fee revenue. See "NON-GAAP FINANCIAL MEASURES" on page 55 for a reconciliation of Esurance's adjusted expense ratio and adjusted combined ratio to its GAAP expense and combined ratios.

Esurance reported an adjusted combined ratio of 104% for the third quarter of 2011 and compared to 100% for the third quarter of 2010, while the adjusted combined ratio for the first nine months of 2011 and 2010 was 102%. The loss and LAE ratio increased to 75% for the third quarter of 2011 from 70% for the third quarter of 2010, primarily due to severe weather, higher catastrophe losses and higher losses in material damage coverages.

#### AutoOne

On August 30, 2011, OneBeacon entered into an agreement to sell the AutoOne business to Interboro. AutoOne operates as a division within OneBeacon, offering products and services to automobile assigned risk markets. The transaction includes the sale of two insurance entities, AOIC and AOSIC, through which substantially all of the AutoOne business is written on a direct basis. For the nine-months ended September 30, 2011, the results of operations for the AutoOne business have been classified as discontinued operations and are presented, net of related income taxes, in the statement of comprehensive income. Prior year results of operations have been reclassified to conform to the current period's presentation. As of September 30, 2011, the assets and liabilities of the AutoOne business have been presented in the balance sheet as held for sale.

Net income (loss) from discontinued operations, net of tax, related to AutoOne was \$(3) million and \$(4) million in the third quarter and first nine months of 2011 and \$(4) million and \$(5) million in the third quarter and first nine months of 2010. The third quarter and first nine months of 2011 also included an \$18 million after-tax loss (\$28 million pre-tax) on the disposal of AutoOne.

# LIQUIDITY AND CAPITAL RESOURCES

#### **Operating Cash and Short-term Investments**

Holding company level. The primary sources of cash for the Company and certain of its intermediate holding companies are expected to be distributions and tax sharing payments received from its insurance and reinsurance operating subsidiaries, capital raising activities, net investment income and proceeds from sales and maturities of investments. The primary uses of cash are expected to be repurchases of the Company's and OneBeacon Ltd.'s common shares, payments on and repurchases/retirements of its debt obligations, dividend payments to holders of the Company's common shares, to noncontrolling interest holders of OneBeacon Ltd.'s common shares and to holders of the Sirius Group Preference Shares, purchases of investments, payments made to tax authorities, contributions to operating subsidiaries and operating expenses.

Operating subsidiary level. The primary sources of cash for White Mountains' insurance and reinsurance operating subsidiaries are expected to be premium collections, net investment income, proceeds from sales and maturities of investments, contributions from holding companies and capital raising activities. The primary uses of cash are expected to be claim payments, policy acquisition costs, purchases of investments, payments on and repurchases/retirements of its debt obligations, distributions and tax sharing payments made to holding companies and operating expenses.

Both internal and external forces influence White Mountains' financial condition, results of operations and cash flows. Claim settlements, premium levels and investment returns may be impacted by changing rates of inflation and other economic conditions. In many cases, significant periods of time, sometimes several years or more, may lapse between the occurrence of an insured loss, the reporting of the loss to White Mountains and the settlement of the liability for that loss. The exact timing of the payment of claims and benefits cannot be predicted with certainty. White Mountains' insurance and reinsurance operating subsidiaries maintain portfolios of invested assets with varying maturities and a substantial amount of cash and short-term investments to provide adequate liquidity for the payment of claims.

Management believes that White Mountains' cash balances, cash flows from operations, routine sales and maturities of investments and the liquidity provided by the WTM Bank Facility are adequate to meet expected cash requirements for the foreseeable future on both a holding company and operating subsidiary level.

47

#### **Table of Contents**

# **Dividend Capacity**

Under the insurance laws of the states and jurisdictions that White Mountains' insurance and reinsurance operating subsidiaries are domiciled, an insurer is restricted with respect to the timing and the amount of dividends it may pay without prior approval by regulatory authorities. Accordingly, there can be no assurance regarding the amount of such dividends that may be paid by such subsidiaries in the future. Following is a description of the dividend capacity of White Mountains' insurance and reinsurance operating subsidiaries:

# OneBeacon:

Generally, OneBeacon's top tier regulated insurance operating subsidiaries have the ability to pay dividends during any 12-month period without the prior approval of regulatory authorities in an amount equal to the greater of prior year statutory net income or 10% of prior year end statutory surplus, subject to the availability of unassigned funds. OneBeacon's top tier regulated insurance operating subsidiaries have the ability to pay approximately \$291 million of dividends during 2011 without prior approval of regulatory authorities, subject to the availability of unassigned funds. At June 30, 2011, OneBeacon's top tier regulated insurance operating subsidiaries had \$0.7 billion of unassigned funds. During the first nine months of 2011, OneBeacon's top tier regulated insurance operating subsidiaries distributed \$150 million of extraordinary dividends to their immediate parent.

During the first nine months of 2011, OneBeacon's unregulated insurance operating subsidiaries paid \$4 million of dividends to their immediate parent. At September 30, 2011, OneBeacon's unregulated insurance operating subsidiaries had \$7 million of net unrestricted cash, short-term investments and fixed maturity investments.

During the first nine months of 2011, OneBeacon Ltd. paid \$60 million of regular quarterly dividends and a \$95 million special dividend to its common shareholders. White Mountains received \$117 million of these dividends.

At September 30, 2011, OneBeacon Ltd. and its intermediate holding companies had \$216 million of net unrestricted cash, short-term investments and fixed maturity investments and \$27 million of common equity securities and convertible fixed maturity investments outside of its regulated and unregulated insurance operating subsidiaries.

#### Sirius Group:

During the third quarter of 2011, White Mountains announced a rebranding and reorganization (the "Reorganization") of its reinsurance business. Under the rebranding, which is being executed in the third and fourth quarters of 2011, White Mountains' reinsurance businesses adopted the Sirius name. For the Reorganization, which was completed in October 2011, Sirius America (formerly WMRe America) became a wholly owned subsidiary of Sirius International (formerly WMRe Sirius). In connection with the Reorganization, A.M. Best upgraded the financial strength rating of Sirius America from "A-" (Excellent, the fourth highest of fifteen ratings) to a group "A" (Excellent, the third highest of fifteen ratings), consistent with Sirius International's rating, and all financial strength ratings from the four major ratings agencies were affirmed with stable outlook. Additionally, A.M. Best upgraded the debt ratings to "bbb" (Adequate, the ninth highest of twenty-two ratings) on the Sirius Group Senior Notes and to "bb+" (Speculative, the eleventh highest of twenty-two ratings) from "bb" (Speculative, the twelfth highest of twenty-two ratings) on the Sirius Group Preference Shares. S&P upgraded the debt ratings to "BBB" (Adequate, the ninth highest of twenty-two ratings) from "BB-" (Adequate, the tenth highest of twenty-two ratings) and to "BB+" (Speculative, the eleventh highest of twenty-two ratings) from "BB" (Speculative, the twelfth highest of twenty-two ratings) on the Sirius Group Preference Shares.

In conjunction with the Reorganization, the following capital transactions occurred in October 2011:

- · Sirius America paid \$250 million to its immediate parent, which included \$67 million of dividends and \$183 million return of capital;
- · Sirius International paid \$169 million to an intermediate holding company of Sirius Group for a portion of Sirius America.
- · An intermediate holding company of Sirius Group contributed the remaining shares of Sirius America with a GAAP book value of \$436 million to Sirius International, which now owns 100% of the shares of Sirius America;
- · Sirius Group distributed \$425 million to White Mountains, which included approximately \$300 million that was freed up at Sirius International and Sirius America as a result of the Reorganization and the commutation of quota-share agreements between Sirius Group and Esurance as a result of the Esurance Sale;
- · White Mountains contributed common shares in Symetra with a GAAP book value of \$241 million (including equity in net unrealized gains from Symetra's fixed income portfolio of \$79 million) to an intermediate holding company of Sirius Group.

Subject to certain limitations under Swedish law, Sirius International is permitted to transfer all or a portion of its pre-tax income to its Swedish parent companies to minimize taxes (referred to as a group contribution). In 2011, Sirius International intends to transfer approximately \$70 million (based on the September 30, 2011 SEK to USD exchange rate) of its 2010 pre-tax income to its Swedish parent companies as a group contribution, of which \$37 million was transferred during the first nine months of 2011.

48

# **Table of Contents**

Sirius International has the ability to pay dividends subject to the availability of unrestricted statutory surplus. Historically, Sirius International has allocated the majority of its pre-tax income, after group contributions to its Swedish parent companies, to the Safety Reserve (see "Safety Reserve" below). At December 31, 2010, Sirius International had \$263 million (based on the December 31, 2010 SEK to USD exchange rate) of unrestricted statutory surplus, which is available for distribution in 2011. Prior to the Reorganization, Sirius International paid a cash dividend of \$67 million to its immediate parent during the second quarter of 2011. In connection with the Reorganization, Sirius International's unrestricted statutory surplus increased by \$436 million due to the contribution of the remaining shares of Sirius America in October.

Sirius America has the ability to pay dividends during any 12-month period without the prior approval of regulatory authorities in an amount set by formula based on the lesser of net investment income, as defined by statute, or 10% of statutory surplus, in both cases as most recently reported to regulatory authorities, subject to the availability of earned surplus. Based upon June 30, 2011 statutory surplus of \$779 million, Sirius America has the ability to pay \$78 million of dividends during 2011 without prior approval of regulatory authorities, subject to the availability of earned surplus. At June 30, 2011, Sirius America had \$100 million of earned surplus, prior to the \$67 million paid as part of the Reorganization.

Prior to the Reorganization, Sirius Group paid a cash dividend of \$85 million to its immediate parent during the first quarter of 2011.

At September 30, 2011, Sirius Group and its intermediate holding companies had \$54 million of net unrestricted cash, short-term investments and fixed maturity investments and \$15 million of other long-term investments outside of Sirius International, Sirius America, Scandinavian Re and Central National.

# Safety Reserve

Subject to certain limitations under Swedish law, Sirius International is permitted to transfer pre-tax amounts into an untaxed reserve referred to as a safety reserve. At September 30, 2011, Sirius International's safety reserve amounted to \$1.4 billion. Under GAAP, an amount equal to the safety reserve, net of a related deferred tax liability established at the Swedish tax rate of 26.3%, is classified as shareholders' equity. Generally, this deferred tax liability is only required to be paid by Sirius International if it fails to maintain predetermined levels of premium writings and loss reserves in future years. As a result of the indefinite deferral of these taxes, Swedish regulatory authorities do not apply any taxes to the safety reserve when calculating solvency capital under Swedish insurance regulations. Accordingly, under local statutory requirements, an amount equal to the deferred tax liability on Sirius International's safety reserve (\$368 million at September 30, 2011) is included in solvency capital. Access to the safety reserve is restricted to coverage of insurance losses. Access for any other purpose requires the approval of Swedish regulatory authorities. Similar to the approach taken by Swedish regulatory authorities, most major rating agencies generally include the \$1.4 billion balance of the safety reserve, without any provision for deferred taxes, in Sirius International's capital when assessing Sirius International's financial strength.

## **Other Operations:**

During the first nine months of 2011, WM Advisors did not pay any dividends to its immediate parent. At September 30, 2011, WM Advisors had approximately \$28 million of net unrestricted cash, short-term investments and fixed maturity investments.

At September 30, 2011, the Company and its intermediate holding companies had \$93 million of net unrestricted cash, short-term investments and fixed maturity investments and \$202 million of common equity securities and other long-term investments included in its Other Operations segment.

#### **Insurance Float**

Insurance float is an important aspect of White Mountains' insurance operations. Insurance float represents funds that an insurance or reinsurance company holds for a limited time. In an insurance or reinsurance operation, float arises because premiums are collected before losses are paid. This interval can extend over many years. During that time, the insurer or reinsurer invests the funds. When the premiums that an insurer or reinsurer collects do not cover the losses and expenses it eventually must pay, the result is an underwriting loss, which is considered to be the cost of insurance float. White Mountains calculates its insurance float by taking its net investment assets and subtracting its total adjusted capital. Although insurance float can be calculated using numbers determined under GAAP, insurance float is not a GAAP concept and, therefore, there is no comparable GAAP measure.

Insurance float can increase in a number of ways, including through acquisitions of insurance and reinsurance operations, organic growth in existing insurance and reinsurance operations and recognition of losses that do not cause a corresponding reduction in investment assets. Conversely, insurance float can decrease in a number of other ways, including sales of insurance and reinsurance operations, shrinking or run-off of existing insurance and/or reinsurance operations, the acquisition of operations that do not have substantial investment assets (e.g., an agency) and the recognition of gains that do not cause a corresponding increase in investment assets. White Mountains has historically obtained its insurance float primarily through acquisitions, as opposed to organic growth. It is White Mountains' intention to generate low-cost float over time through a combination of acquisitions and/or by organic growth in its existing insurance and reinsurance operations. However, White Mountains will seek to increase its insurance float organically only when market conditions allow for an expectation of generating underwriting profits.

49

#### **Table of Contents**

Certain operational leverage metrics can be measured with ratios that are calculated using insurance float. There are many activities that do not change the amount of insurance float at an insurance company but can have a significant impact on the company's operational leverage metrics. For example, investment gains and losses, foreign currency gains and losses, debt issuances and repayments, common and preferred share issuances and repurchases and dividends paid to shareholders are all activities that do not change insurance float but that can meaningfully impact operational leverage metrics.

The following table illustrates White Mountains' consolidated insurance float position and four operational leverage ratios based on insurance float and net investment assets as of September 30, 2011 and December 31, 2010:

(\$ in millions)	Sej	ptember 30, 2011	1	December 31, 2010
Total investments	\$	7,199.6	\$	7,943.5
Consolidated limited partnership investments(1)		(72.7)		(91.5)
Cash		557.7		395.0
Investments in unconsolidated affiliates		458.6		389.7
Equity in net unrealized gains from Symetra's fixed maturity portfolio		(147.7)		(63.7)
Net investment assets classified within assets held for sale		1,045.6		829.2
Accounts receivable on unsettled investment sales		20.3		39.5
Accounts payable on unsettled investment purchases		(57.1)		(22.3)
Interest-bearing funds held by ceding companies (2)		87.9		78.9
Interest-bearing funds held under reinsurance treaties (3)		(10.3)		(52.5)
Net investment assets	\$	9,081.9	\$	9,445.8
Total White Mountains' common shareholders' equity	\$	3,444.7	\$	3,653.0
Noncontrolling interest—OneBeacon Ltd.		269.8		295.0
Noncontrolling interest—Sirius Group Preference Shares		250.0		250.0
Debt		669.0		818.8
Total capital (1)		4,633.5		5,016.8
Equity in net unrealized gains from Symetra's fixed maturity portfolio, net of applicable				
taxes		(135.7)		(58.5)
Total adjusted capital	\$	4,497.8	\$	4,958.3
Insurance float	\$	4,584.1	\$	4,487.5
Insurance float as a multiple of total adjusted capital		1.0x		0.9x
Net investment assets as a multiple of total adjusted capital		2.0x		1.9x
Insurance float as a multiple of White Mountains' common shareholders' equity		1.3x		1.2x
Net investment assets as a multiple of White Mountains' common shareholders' equity		2.6x		2.6x

<sup>(1)</sup> The noncontrolling interest arising from White Mountains' investments in consolidated limited partnerships has not been included in insurance float or total capital because White Mountains does not benefit from the return on or have the ability to utilize the net assets supporting this noncontrolling interest.

During the first nine months of 2011, insurance float increased by \$97 million, primarily due to sales and settlements of derivative assets at WM Life Re and the portion of the \$230 million of catastrophe losses reported by White Mountains in the first nine months of 2011 that were unpaid as of September 30, 2011. These catastrophe losses reduce White Mountains' total capital, creating a temporary increase in insurance float when they are first recorded, which will reverse and decrease insurance float as the catastrophe losses are paid in the future. These increases were partially offset by a decrease in insurance float at OneBeacon due to the continued run-off of reserves as a result of the Commercial Lines Transaction.

<sup>(2)</sup> Excludes funds held by ceding companies from which White Mountains does not receive interest credits.

<sup>(3)</sup> Excludes funds held by White Mountains under reinsurance treaties for which White Mountains does not provide interest credits.

## **Table of Contents**

#### **Financing**

The following table summarizes White Mountains' capital structure as of September 30, 2011 and December 31, 2010:

(\$ in millions)	Sep	tember 30, 2011	December 31, 2010
OBH Senior Notes, carrying value	\$	269.7	\$ 419.6
Sirius Group Senior Notes, carrying value		399.3	399.2
WTM Bank Facility		_	_
Total debt		669.0	818.8
Noncontrolling interest—OneBeacon Ltd.		269.8	295.0
Noncontrolling interest—Sirius Group Preference Shares		250.0	250.0
Total White Mountains' common shareholders' equity		3,444.7	3,653.0
Total capital(1)		4,633.5	5,016.8
Equity in net unrealized gains from Symetra's fixed maturity portfolio		(135.7)	(58.5)
Total adjusted capital	\$	4,497.8	\$ 4,958.3
Total debt to total adjusted capital		15%	 17%
Total debt and Sirius Group Preference Shares to total adjusted capital		20%	 22%

<sup>(1)</sup> The noncontrolling interest arising from White Mountains' investments in consolidated limited partnerships has not been included in total capital because White Mountains does not benefit from the return on or have the ability to utilize the assets supporting this noncontrolling interest.

Management believes that White Mountains generally has the flexibility and capacity to obtain funds externally as needed through debt or equity financing on both a short-term and long-term basis. However, White Mountains can provide no assurance that, if needed, it would be able to obtain additional debt or equity financing on satisfactory terms, if at all.

The Company's 2010 Annual Report on Form 10-K contains a full discussion of White Mountains' debt obligations as of December 31, 2010.

On April 21, 2011, OBH repurchased and retired \$150 million aggregate principal amount of the OBH Senior Notes for an aggregate purchase price of \$162 million through a cash tender offer.

On August 12, 2011, White Mountains entered into a new revolving credit facility with a total commitment of \$375 million (the "WTM Bank Facility") with a syndicate of lenders administered by Bank of America, N.A. The WTM Bank Facility replaces White Mountains' previous revolving credit facility, which had a total commitment of \$475 million. As of September 30, 2011, the WTM Bank Facility was undrawn.

The WTM Bank Facility contains various affirmative, negative and financial covenants that White Mountains considers to be customary for such borrowings, including certain minimum net worth and maximum debt to capitalization standards. Failure to meet one or more of these covenants could result in an event of default, which ultimately could eliminate availability under these facilities and result in acceleration of principal repayment on any amounts outstanding. At September 30, 2011, White Mountains was in compliance with all of the covenants under the WTM Bank Facility and anticipates it will continue to remain in compliance with these covenants for the foreseeable future.

White Mountains provides an irrevocable and unconditional guarantee as to the payment of principal and interest on the OBH Senior Notes. In consideration of this guarantee OneBeacon pays White Mountains a guarantee fee equal to 25 basis points per annum on the outstanding principal amount of the OBH Senior Notes. If White Mountains' voting interest in OneBeacon Ltd.'s common shares ceases to represent more than 50% of all their voting securities, OneBeacon Ltd. will seek to redeem, exchange or otherwise modify the senior notes in order to fully and permanently eliminate White Mountains' obligations under the guarantee. In the event that White Mountains' guarantee is not eliminated, the guarantee fee will increase over time up to a maximum guarantee fee of 425 basis points.

The OBH Senior Notes and the Sirius Group Senior Notes were issued under indentures that contain restrictive covenants which, among other things, limit the ability of the Company, OBH, Sirius International Group and their respective subsidiaries to create liens and enter into sale and leaseback transactions and consolidate, merge or transfer their properties and assets. The indentures do not contain any financial ratios or specified levels of net worth or liquidity to which the Company, OBH or Sirius International Group must adhere. At September 30, 2011, White Mountains was in compliance with all of the covenants under the OBH Senior Notes and the Sirius Group Senior Notes, and anticipates it will continue to remain in compliance with these covenants for the foreseeable future.

#### 51

# **Table of Contents**

# **Share Repurchase Programs**

On September 21, 2011, White Mountains completed a "modified Dutch auction" self-tender offer and repurchased 327,872 of its common shares at \$415 per share. The total cost of the share repurchases was \$136.5 million, including fees and expenses related to the tender offer.

In 2006, White Mountains' board of directors authorized the Company to repurchase up to 1 million of its common shares, from time to time, subject to market conditions. In August 2010, White Mountains' board of directors authorized the Company to repurchase an additional 600,000 common shares, for a total of 1.6 million shares authorized. Shares may be repurchased on the open market or through privately negotiated transactions. This program does not have a stated expiration date. During the first nine months of 2011, White Mountains repurchased and retired 265,579 of its common shares under this program for \$93 million at an average share price of \$351. During the first nine months of 2010, the Company repurchased and retired 347,409 of its common shares under this program for \$117 million at an average share price of \$338. Since the inception of this program in 2006, the Company has repurchased and retired 1,363,315 common shares for \$516 million. At September 30, 2011, 236,685 shares may still be repurchased under this program.

Including shares repurchased through its self-tender offer and shares repurchased under its share repurchase program, White Mountains repurchased a total of 593,451 of its common shares during the first nine months of 2011 for \$230 million at an average share price of \$387, which is 89% of White Mountains' adjusted book value per share of \$436 at September 30, 2011.

#### **Cash Flows**

Detailed information concerning White Mountains' cash flows during nine months ended September 30, 2011 and 2010 follows:

#### Cash flows from operations for the nine months ended September 30, 2011 and 2010

Net cash flows from operations was a source of \$40 million and a use of \$21 million in the first nine months of 2011 and 2010, respectively. Net cash flows from operations reflect a source of approximately \$35 million and a source of approximately \$5 million of net payments in the first nine months of 2011 and 2010 to purchase/sell derivative instruments and to fund collateral trusts under the terms of existing derivative contracts at WM Life Re. In addition, net cash flows for operations in the first nine months of 2010 were adversely impacted by the run-off of reserves related to the Commercial Lines Transaction at OneBeacon and declining net investment income, primarily due to a decrease in the overall invested asset base. In addition, White Mountains does not believe that these trends will have a meaningful impact on its future liquidity or its ability to meet its future cash requirements.

#### Cash flows from investing and financing activities for the nine months ended September 30, 2011

Financing and Other Capital Activities

During the first quarter of 2011, the Company declared and paid an \$8 million cash dividend to its common shareholders.

During the first nine months of 2011, the Company repurchased and retired 593,640 of its common shares for \$230 million.

During the first nine months of 2011, OneBeacon Ltd. declared and paid \$155 million of cash dividends to its common shareholders, including \$60 million of regular quarterly dividends and a \$95 million special dividend. White Mountains received a total of \$117 million of these dividends.

During the first nine months of 2011, OBH repurchased and retired a portion of the outstanding OBH Senior Notes for \$162 million.

During the first nine months of 2011, OneBeacon paid a total of \$13 million of interest on the OBH Senior Notes.

During the first nine months of 2011, Sirius Group declared and paid \$85 million of cash dividends to its immediate parent.

During the first nine months of 2011, Sirius Group paid \$26 million of interest on the Sirius Group Senior Notes and \$9 million of dividends on the Sirius Group Preference Shares.

During the first nine months of 2011, White Mountains contributed \$10 million to WM Life Re.

52

# **Table of Contents**

# Cash flows from investing and financing activities for the nine months ended September 30, 2010

Financing and Other Capital Activities

During the first quarter of 2010, the Company declared and paid a \$9 million cash dividend to its common shareholders.

During the first nine months of 2010, the Company repurchased and retired 957,412 of its common shares for \$379 million through its share repurchase program and 3,909 of its common shares for \$1 million outside of the share repurchase program.

During the first nine months of 2010, OneBeacon Ltd. declared and paid \$296 million of cash dividends to its common shareholders, including \$60 million of regular quarterly dividends and a \$236 million special dividend. White Mountains received a total of \$225 million of these dividends.

During the first nine months of 2010, OBH repurchased and retired a portion of the outstanding OBH Senior Notes for \$196 million and OneBeacon Insurance Company purchased a portion of the outstanding OBH Senior Notes for \$1 million. In addition, during the first quarter of 2010, OneBeacon repaid in full the \$14 million outstanding under the Atlantic Specialty Note.

During the first nine months of 2010, OneBeacon paid a total of \$18 million of interest on the OBH Senior Notes.

During the first nine months of 2010, Sirius Group declared and paid \$50 million of cash dividends to its immediate parent.

During the first nine months of 2010, Sirius Group paid \$26 million of interest on the Sirius Group Senior Notes and \$9 million of dividends on the Sirius Group Preference Shares.

During the first nine months of 2010, White Mountains contributed \$45 million to WM Life Re.

During the first quarter of 2010, Sirius Group acquired Central National from Drum Financial Corporation for \$5 million in cash.

During the third quarter of 2010, OneBeacon completed the Personal Lines Transaction and received consideration of approximately \$167 million.

53

#### **Table of Contents**

#### FAIR VALUE CONSIDERATIONS

White Mountains carries certain financial instruments at fair value with changes therein recognized in earnings. Assets and liabilities carried at fair value include investment securities, derivative instruments (both exchange traded and over the counter instruments) and reinsurance assumed liabilities associated with variable annuity benefit guarantees. Valuation of assets and liabilities measured at fair value require management to make estimates and apply judgment to matters that may carry a significant degree of uncertainty. In determining its estimates of fair value, White Mountains uses a variety of valuation approaches and inputs. Whenever possible, White Mountains estimates fair value using valuation methods that maximize the use of quoted prices and other observable inputs. Where observable inputs are not available, the estimated fair value is based upon internal pricing models using assumptions that include inputs that may not be observable in the marketplace but which reflect management's best judgment given the circumstances and consistent with what other market participants would use when pricing such instruments. Where appropriate, assets and liabilities measured at fair value have been adjusted for the effect of counterparty credit risk.

As of September 30, 2011, approximately 93% of the investment portfolio recorded at fair value was priced based upon quoted market prices or other observable inputs. White Mountains uses brokers and outside pricing services to assist in determining fair values. For investments in active markets, White Mountains uses the quoted market prices provided by the outside pricing service to determine fair value. The outside pricing services used by White Mountains have indicated that they will only provide prices where observable inputs are available. White Mountains' process to validate the market prices obtained from the outside pricing sources includes, but is not limited to, periodic evaluation of model pricing methodologies and monthly analytical reviews of certain prices. White Mountains also periodically performs back-testing of selected sales activity to determine whether there are any significant differences between the market price used to value the security prior to sale and the actual sale price. See **Note 5** for further discussion of fair value measurements for investment securities.

#### WM Life Re

White Mountains has entered into agreements to reinsure death and living benefit guarantees associated with certain variable annuities in Japan. White Mountains carries the benefit guarantees at fair value. The fair value of the guarantees is estimated using actuarial and capital market assumptions related to the projected discounted cash flows over the term of the reinsurance agreement. The valuation uses assumptions about surrenders/lapse rates, market volatilities and other factors, and includes a risk margin which represents the additional compensation a market participant would require to assume the risks related to the business. The selection of surrender/lapse rates, market volatility assumptions, risk margins and other factors require the use of significant management judgment. Assumptions regarding policyholder behavior, including surrender and lapse rates, are generally unobservable inputs and significantly impact the fair value estimate. The cost of reinsuring these benefit guarantees may be greater than expected. Market conditions including, but not limited to, changes in interest rates, equity indices, market volatility and foreign currency exchange rates as well as variations in actuarial assumptions regarding policyholder behavior may result in significant fluctuations in the fair value of the liabilities associated with these guarantees that could materially affect results of operations. All of White Mountains' variable annuity reinsurance liabilities (\$761 million) were classified as Level 3 measurements at September 30, 2011.

WM Life Re projects future surrender rates by year for policies based on a combination of actual experience and expected policyholder behavior. Actual policyholder behavior, either individually or collectively, may differ from projected behavior as a result of a number of factors such as the level of the account value versus guarantee value and applicable surrender charge, views of the primary insurance company's financial strength and ability to pay the guarantee at maturity, annuitants' need for money in a prolonged recession and time remaining to receive the guarantee at maturity. Policyholder behavior is especially difficult to predict given that the types of contracts reinsured by WM Life Re are relatively new to the Japanese market and the recent financial turmoil is unprecedented for this type of product in the Japanese market. Actual policyholder behavior may differ materially from WM Life Re's projections.

As of September 30, 2011, WM Life Re's annual surrender assumptions vary from 0.3% currently to 3% depending on the level of account value versus guarantee value; at the current levels of account value, the average is approximately 0.8% per annum. The potential increase in the fair value of the liability due to a change in current actuarial assumptions is as follows:

	Increase in fair value of liability				
Millions	September 30, 2011		December 31, 2010		
Decrease 50%	\$ 9 million	\$	12 million		
Decrease 100% (to zero surrenders)	\$ 19 million	\$	24 million		

The amounts in the table above could increase in the future if the fair value of the variable annuity guarantee liability changes due to factors other than the surrender assumptions (e.g., a decline in the ratio of the annuitants' aggregate account values to their aggregate guarantee values).

54

# **Table of Contents**

#### NON-GAAP FINANCIAL MEASURES

This report includes five non-GAAP financial measures that have been reconciled to their most comparable GAAP financial measures. White Mountains believes these measures to be more relevant than comparable GAAP measures in evaluating White Mountains' results of operations and financial condition.

Adjusted comprehensive income is a non-GAAP financial measure that excludes the change in equity in net unrealized gains and losses from Symetra's fixed maturity portfolio, net of applicable taxes, from comprehensive income. In the calculation of comprehensive income under GAAP, fixed maturity investments are marked-to-market while the liabilities to which those assets are matched are not. Symetra attempts to earn a "spread" between what it earns on its investments and what it pays out on its products. In order to try to fix this spread, Symetra invests in a manner that tries to match the duration and cash flows of its investments with the required cash outflows associated with its life insurance and structured settlements products. As a result, Symetra typically earns the same spread on in-force business whether interest rates fall or rise. Further, at any given time, some of Symetra's structured settlement obligations may extend 40 or 50 years into the future, which is further out than the longest maturing fixed maturity investments regularly available for purchase in the market (typically 30 years). For these long-dated products, Symetra is unable to fully match the obligation with assets until the remaining expected payout schedule comes within the duration of securities available in the market. If at that time, these fixed maturity investments have yields that are lower than the yields expected when the structured settlement product was originally priced, the spread for the product will shrink and Symetra will ultimately harvest lower returns for its shareholders. GAAP comprehensive income increases when rates decline, which would suggest an increase in the value of Symetra - the opposite of what is happening to the intrinsic value of the business. Therefore, White Mountains' management and Board of Directors use adjusted comprehensive income when assessing Symetra's quarterly financial performance. In addition, this measure is typically the predominant component of change in adjusted book value per share,

Adjusted book value per share is a non-GAAP measure which is derived by expanding the GAAP calculation of book value per White Mountains common share to exclude equity in net unrealized gains and losses from Symetra's fixed maturity portfolio, net of applicable taxes. In addition, the number of common shares outstanding used in the calculation of adjusted book value per share are adjusted to exclude unearned restricted common shares, the compensation cost of which, at the date of calculation, has yet to be amortized. The reconciliation of adjusted book value per share to GAAP book value per share is included on page 37.

Total capital at White Mountains is comprised of White Mountains' common shareholders' equity, debt and noncontrolling interest in OneBeacon Ltd and the Sirius Group Preference Shares. Total adjusted capital excludes the equity in net unrealized gains and losses from Symetra's fixed maturity portfolio, net of applicable taxes, from total capital. The reconciliation of total capital to total adjusted capital is included on page 50.

Esurance's adjusted expense ratio and adjusted combined ratio are non-GAAP financial measures. To calculate the adjusted expense ratio and adjusted combined ratio, acquisition expenses are reduced by referral fee revenue. Referral fee revenue, which is recorded as a component of other revenues under GAAP, represents fees that Esurance receives for referring customers for whom it does not write policies to other insurance carriers and aggregators. Management believes that Esurance's adjusted expense ratio and adjusted combined ratio are better measures to evaluate Esurance's underwriting results than its GAAP expense ratio and GAAP combined ratio because the expenses that are incurred to acquire policyholders at Esurance, particularly advertising expenses, also lead to referral fee revenue. The reconciliation of Esurance's adjusted expense ratio and adjusted combined ratio to its GAAP expense ratio and GAAP combined ratio follows:

	Three Months Ended Sept. 30,		Nine Months Ended	Sept. 30,
	2011	2010	2011	2010
GAAP expense ratio	31%	32%	31%	31%
Referral fees	(2)%	(2)%	(2)%	(2)%
Adjusted expense ratio	29%	30%	29%	29%
GAAP combined ratio	106%	102%	104%	104%
Referral fees	(2)%	(2)%	(2)%	(2)%
Adjusted combined ratio	104%	100%	102%	102%
	55			

# **Table of Contents**

## CRITICAL ACCOUNTING ESTIMATES

Refer to the Company's 2010 Annual Report on Form 10-K for a complete discussion regarding White Mountains' critical accounting estimates.

#### FORWARD-LOOKING STATEMENTS

The information contained in this report may contain "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. All statements, other than statements of historical facts, included or referenced in this report which address activities, events or developments which White Mountains expects or anticipates will or may occur in the future are forward-looking statements. The words "will", "believe," "intend," "expect," "anticipate," "project," "estimate," "predict" and similar expressions are also intended to identify forward-looking statements. These forward-looking statements include, among others, statements with respect to White Mountains':

- · change in adjusted book value per share or return on equity;
- · business strategy;
- · financial and operating targets or plans;
- · incurred loss and loss adjustment expenses and the adequacy of its loss and LAE reserves and related reinsurance;
- · projections of revenues, income (or loss), earnings (or loss) per share, dividends, market share or other financial forecasts;
- expansion and growth of its business and operations; and
- · future capital expenditures.

These statements are based on certain assumptions and analyses made by White Mountains in light of its experience and perception of historical trends, current conditions and expected future developments, as well as other factors believed to be appropriate in the circumstances. However, whether actual results and developments will conform to its expectations and predictions is subject to a number of risks and uncertainties that could cause actual results to differ materially from expectations, including:

- the risks associated with Item 1A of the Company's 2010 Annual Report on Form 10-K;
- claims arising from catastrophic events, such as hurricanes, earthquakes, floods, fires, terrorist attacks or severe winter weather;
- · the continued availability of capital and financing;

- · general economic, market or business conditions;
- · business opportunities (or lack thereof) that may be presented to it and pursued;
- · competitive forces, including the conduct of other property and casualty insurers and reinsurers;
- changes in domestic or foreign laws or regulations, or their interpretation, applicable to White Mountains, its competitors or its customers;
- · an economic downturn or other economic conditions adversely affecting its financial position;
- · recorded loss reserves subsequently proving to have been inadequate; and
- · actions taken by ratings agencies from time to time, such as financial strength or credit ratings downgrades or placing ratings on negative watch;
- · other factors, most of which are beyond White Mountains' control.

Consequently, all of the forward-looking statements made in this report are qualified by these cautionary statements, and there can be no assurance that the actual results or developments anticipated by White Mountains will be realized or, even if substantially realized, that they will have the expected consequences to, or effects on, White Mountains or its business or operations. White Mountains assumes no obligation to publicly update any such forward-looking statements, whether as a result of new information, future events or otherwise.

#### Item 3. Quantitative and Qualitative Disclosures About Market Risk.

Refer to White Mountains' 2010 Annual Report on Form 10-K and in particular **Item 7A.** - "Quantitative and Qualitative Disclosures About Market Risk". As of September 30, 2011, there were no material changes in the market risks as described in White Mountains' most recent Annual Report.

56

#### **Table of Contents**

#### **Item 4. Controls and Procedures.**

The Principal Executive Officer ("PEO") and the Principal Financial Officer ("PFO") of White Mountains have evaluated the effectiveness of its disclosure controls and procedures (as defined in Rule 13a-15(e) and 15d-15(e) of the Exchange Act) as of the end of the period covered by this report. Based on that evaluation, the PEO and PFO have concluded that White Mountains' disclosure controls and procedures are adequate and effective.

There were no significant changes with respect to the Company's internal control over financial reporting or in other factors that materially affected, or are reasonably likely to materially affect, internal control over financial reporting during the quarter ended September 30, 2011.

#### Part II. OTHER INFORMATION

# Item 1. Legal Proceedings.

# **Tribune Company**

In June 2011, Deutsche Bank Trust Company Americas, Law Debenture Company of New York and Wilmington Trust Company (collectively referred to as the "Plaintiff"), in their capacity as trustees for certain senior notes issued by the Tribune Company (the "Tribune"), filed lawsuits in various jurisdictions against numerous defendants including certain of White Mountains' subsidiaries in their capacity as former shareholders of the Tribune seeking recovery of the proceeds from the sale of common stock of the Tribune in connection with the Tribune's leveraged buyout in 2007. The Plaintiff seeks recovery of the proceeds received by the former Tribune shareholders on the basis that the Tribune purchased or repurchased its common shares without receiving fair consideration at a time when it was, or as a result of the purchases of shares, was rendered, insolvent. Certain subsidiaries of White Mountains, received a total of approximately \$39 million for the Tribune common stock it tendered in connection with the leveraged buyout. The various cases are now pending before a Multi District Litigation panel ("MDL") in the federal district court in N.Y. The MDL is considering a motion to consolidate these actions and a hearing on that motion is scheduled for December 1, 2011.

# Item 1A. Risk Factors

There have been no material changes in the Registrant's risk factors since the Registrant's most recently filed

Form 10-K.

# Item 2. Issuer Purchases of Equity Securities.

Months	Total Number of Shares Purchased	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Plan (1)	Maximum Number of Shares that May Yet Be Purchased Under the Plan (1)
July 1- 31, 2011				236,685
August 1 - 31, 2011	_	_	_	236,685
September 1 - 30, 2011	327,872(2)	\$ 415.00	_	236,685
Total	327,872	\$ 415.00		236,685

- (1) On November 17, 2006, White Mountains' board of directors authorized the Company to repurchase up to 1 million of its common shares, from time to time, subject to market conditions. On August 26, 2010, White Mountains' board of directors authorized the Company to repurchase up to an additional 600,000 common shares, for a total authorization of 1.6 million shares. Shares may be repurchased on the open market or through privately negotiated transactions. The repurchase authorization does not have a stated expiration.
- (2) On September 21, 2011, White Mountains completed a "modified Dutch auction" self-tender offer and repurchased 327,872 of its common shares at \$415 per share, for a total cost of \$136 million, including fees and expenses related to the tender offer. This tender offer was authorized by White Mountains' board of directors in August 2011 and does not impact the previously announced repurchase authorization referred to above.

# Item 3. Defaults Upon Senior Securities.

None.

#### Item 4. Other Information.

None.

57

# **Table of Contents**

#### Item 5. Exhibits.

- (a) Exhibits
  - \$375,000,000 Credit Agreement, dated August 12, 2011 among the Company, as the Borrower, Bank of America, N.A., as Administrative Agent, Swing Line Lender and Issuing Lender, and the other lenders party hereto. (\*)
  - Statement Re Computation of Per Share Earnings \*\*\*
  - 31.1 Principal Executive Officer Certification Pursuant to Rule 13a-14 (a) of the Securities Exchange Act of 1934, as Amended.
  - 31.2 Principal Financial Officer Certification Pursuant to Rule 13a-14 (a) of the Securities Exchange Act of 1934, as Amended.
  - 32.1 Principal Executive Officer Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. \*
  - Principal Financial Officer Certification Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the
     Sarbanes-Oxley Act of 2002. \*
  - The following financial information from White Mountains' Quarterly Report on Form 10Q for the quarter ended September 30, 2011 formatted in XBRL: (i) Consolidated Balance Sheets, September 30, 2011 and December 31, 2010; (ii) Consolidated Statements of Operations and Comprehensive Income (Loss), Three and Nine Months Ended September 30, 2011 and 2010; (iii) Consolidated Statements of Changes in Equity, Nine Months Ended September 30, 2011 and 2010; (iv) Consolidated Statements of Cash Flows, Nine Months Ended September 30, 2011 and 2010; and (v) Notes to Consolidated Financial Statements.\*\*
- \* Included herein
- \*\* Furnished herewith
- \*\*\* Not included as an exhibit as the information is contained elsewhere within this report. See **Note 9** of the Notes to Consolidated Financial Statements.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

<u>WHITE MOUNTAINS INSURANCE GROUP, LTD.</u> (Registrant)

Date: October 28, 2011 By: /s/ J. Brian Palmer

J. Brian Palmer

Vice President and Chief Accounting Officer

# **CREDIT AGREEMENT**

Dated as of August 12, 2011

among

# WHITE MOUNTAINS INSURANCE GROUP, LTD.,

as the Borrower,

# BANK OF AMERICA, N.A.,

as Administrative Agent, Swing Line Lender and an Issuing Lender,

and

# THE OTHER LENDERS PARTY HERETO

# BARCLAYS CAPITAL,

as Syndication Agent

and

# HSBC BANK USA, N.A.,

# NORDEA BANK FINLAND PLC, NEW YORK BRANCH,

and

# U.S. BANK NATIONAL ASSOCIATION,

as Co-Documentation Agents

and

# MERRILL LYNCH, PIERCE, FENNER & SMITH INCORPORATED,

and

# BARCLAYS CAPITAL

as Joint Lead Arrangers and Joint Book Runners

# TABLE OF CONTENTS

			Page
1.	DEFI	NITIONS	1
	1.1.	Defined Terms	1
	1.2.	Other Definitional Provisions	25
	1.3.	Letter of Credit Amounts	25
	1.4.	Rounding	26
	1.5.	Times of Day	26
	1.6.	Accounting Terms; Utilization of GAAP for Purposes of Calculations Under Agreement	26
2.	AMO	UNT AND TERMS OF COMMITMENTS	26
	2.1.	Revolving Credit Commitments	26
	2.2.	Procedure for Revolving Credit Borrowing	27
	2.3.	Swing Line Commitment	27
	2.4.	Procedure for Swing Line Borrowing; Refunding of Swing Line Loans	28
	2.5.	Repayment of Loans; Evidence of Debt	30
	2.6.	Facility Fee, etc	31
	2.7.	Termination or Reduction of Revolving Credit Commitments	32

2.8.

Prepayments

	2.9.	Conversion and Continuation Options	3
	2.10.	Maximum Number of Eurodollar Loans	34
	2.11.	Interest Rates and Payment Dates	34
	2.12.	Computation of Interest and Fees; Retroactive Adjustments of Applicable Margin and Facility Fee Rate	34
	2.13.	Inability to Determine Interest Rate	35
	2.14.	Pro Rata Treatment and Payments	30
	2.15.	Requirements of Law	37
	2.16.	Taxes	39
	2.17.	Compensation for Losses	42
	2.18.	Illegality	42
	2.19.	Change of Office	43
	2.20.	Replacement of Lenders under Certain Circumstances	43
	2.21.	Guaranty of Payment and Performance	43
	2.22.	Increase in Commitments	44
	2.23.	Cash Collateral	44
	2.24.	Defaulting Lenders	46
3.	LETTE	RS OF CREDIT	48
	3.1.	L/C Commitment	48
	3.2.	Procedure for Issuance and Amendment of Letter of Credit; Auto-Extension Letters of Credit	49
		i	

# TABLE OF CONTENTS

(continued)

			Page
	3.3.	Drawings and Reimbursements; Funding of Participations	50
	3.4.	Repayment of Participations	52
	3.5.	Obligations Absolute	53
	3.6.	Role of Issuing Lender	53
	3.7.	Applicability of ISP98	5-
	3.8.	Fees and Other Charges	54
	3.9.	Letters of Credit Issued for Subsidiaries	55
	3.10.	Conflict with Issuer Documents	55
4.	COND	ITIONS PRECEDENT	55
	4.1.	Conditions to Closing	55
	4.2.	Conditions to Closing and Each Extension of Credit	57
5.	REPRI	ESENTATIONS AND WARRANTIES	57
	5.1.	Financial Statements	57
	5.2.	Corporate Existence; Compliance with Law	58
	5.3.	Corporate Power; Authorization; Enforceable Obligations	58
	5.4.	No Legal Bar	59
	5.5.	No Material Litigation	59
	5.6.	Ownership of Property; Liens	59
	5.7.	Intellectual Property	59
	5.8.	Taxes	59
	5.9.	Federal Regulations	59
	5.10.	ERISA	60
	5.11.	Investment Company Act; Other Regulations	60
	5.12.	Use of Proceeds	63
	5.13.	Accuracy of Information, etc	61
	5.14.	Insurance Regulatory Matters	61
	5.15.	Indebtedness and Liens	61
	5.16.	Taxpayer Identification Number	63
6.	AFFIR	MATIVE COVENANTS	62
	6.1.	Financial Statements	62
	6.2.	Certificates; Other Information	64
	6.3.	Payment of Obligations	65
	6.4.	Conduct of Business and Maintenance of Existence, etc	65
	6.5.	Maintenance of Property; Insurance	65
	6.6.	Inspection of Property; Books and Records; Discussions	65
	6.7.	Notices	66
	6.8.	Taxes	66
	6.9.	Use of Proceeds	67

# TABLE OF CONTENTS

(continued)

			Page
	6.10.	Further Assurances	67
7.	NEGA	TIVE COVENANTS	67
	7.1.	Financial Condition Covenants	67
	7.2.	Limitation on Indebtedness	69
	7.3.	Limitation on Liens	70
	7.4.	Limitation on Changes in Fiscal Periods	71
	7.5.	Limitation on Lines of Business	71
	7.6.	Guarantors	71
8.	EVEN	TS OF DEFAULT	71
	8.1.	Events of Default	71
	8.2.	Remedies Upon Event of Default	73
9.	THE A	DMINISTRATIVE AGENT	74
	9.1.	Appointment	74
	9.2.	Delegation of Duties	75
	9.3.	Exculpatory Provisions	75
	9.4.	Reliance by Administrative Agent	76
	9.5.	Non-Reliance on Administrative Agent and Other Lenders	76
	9.6.	Administrative Agent in its Individual Capacity	76
	9.7.	Successor Administrative Agent	77
	9.8.	Administrative Agent May File Proofs of Claim	78
	9.9.	Guarantee and Collateral Matters	78
	9.10.	Other Agents; Arrangers and Managers	79
10.	MISCE	ELLANEOUS	79
	10.1.	Amendments, Etc	79
	10.2.	Notices; Effectiveness; Electronic Communication	81
	10.3.	No Waiver; Cumulative Remedies	83
	10.4.	Survival of Representations and Warranties	84
	10.5.	Attorney Costs and Expenses	84
	10.6.	Indemnification	85
	10.7.	Successors and Assigns	86
	10.8.	Adjustments; Setoff	92
	10.9.	Counterparts	93
	10.10.	Severability	93
	10.11.	Integration	94
	10.12.	GOVERNING LAW	94
	10.13.	SUBMISSION TO JURISDICTION; WAIVERS	94
	10.14.	WAIVERS OF JURY TRIAL	95

# TABLE OF CONTENTS (continued)

		Page
10.15.	No Advisory or Fiduciary Responsibility	95
10.16.	Confidentiality	95
10.17.	Release of Guarantee Obligations	96
10.18.	USA PATRIOT Act Notice	97
10.19.	Interest Rate Limitation	97
10.20.	Entire Agreement	98

# SCHEDULES:

1	Commitment Schedule
1A	Existing Letters of Credit

Form of Compliance Certificate Form of Borrowing Request **B-1** B-2 Form of Swing Line Loan Notice Form of Revolving Credit Note C-1 Form of Swing Line Note C-2D Form of Exemption Certificate Е Form of Closing Certificate F Form of Legal Opinion of Robert Seelig, Esq. G Form of Legal Opinion of Conyers Dill & Pearman Form of Assignment and Assumption Η Form of Instrument of Accession Form of Guaranty

Notice Addresses

5.3 10.2

**EXHIBITS:** 

Consents, Authorizations, Filings and Notices

iv

#### **CREDIT AGREEMENT**

This **CREDIT AGREEMENT**, dated as of August 12, 2011, is among (i) **WHITE MOUNTAINS INSURANCE GROUP, LTD.**, a company existing under the laws of Bermuda (the "Borrower"), (ii) each lender from time to time party hereto (collectively, the "Lenders"), (iii) **BANK OF AMERICA, N.A.** ("Bank of America"), as Administrative Agent, Swing Line Lender and an Issuing Lender and (iv) **BARCLAYS CAPITAL**, as Syndication Agent.

#### PRELIMINARY STATEMENTS

**WHEREAS**, the Borrower has requested that the Lenders provide a revolving credit facility under which the Borrower may obtain revolving loans and other extensions of credit in an aggregate principal amount outstanding at any time not in excess of \$375,000,000 (subject to an option to increase such amount by an additional \$75,000,000), and the Lenders are willing to do so on the terms and conditions set forth herein.

**NOW THEREFORE**, in consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

#### 1. **DEFINITIONS**

1.1. <u>Defined Terms</u>. As used in this Agreement, the terms listed in this <u>Section 1.1</u> shall have the respective meanings set forth in this <u>Section 1.1</u>.

"Act of 1934" means the Securities Exchange Act of 1934 and the regulations issued thereunder.

"Administrative Agent" means Bank of America, in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent appointed in accordance with <u>Section 9.7</u>.

"<u>Administrative Agent's Office</u>" means the Administrative Agent's address and, as appropriate, account as set forth on <u>Schedule 10.2</u>, or such other address or account as the Administrative Agent may from time to time notify the Borrower and the Lenders.

"Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.

"Affiliate" means, as to any Person, any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" of a Person means the power, directly or indirectly, either to (a) vote 10% or more of the securities having ordinary voting power for the election of directors (or persons performing similar functions) of such Person or (b) direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.

"Agent Fee Letter" means that certain letter agreement dated as of June 20, 2011 by and between the Borrower, the Administrative Agent and Merrill Lynch, Pierce, Fenner & Smith Incorporated.

"Agent-Related Persons" means the Administrative Agent, together with its Affiliates (including, Bank of America, in its capacity as the Administrative Agent and Merrill Lynch, Pierce, Fenner & Smith Incorporated in its capacity as one of the Arrangers), and the officers, directors, employees, agents and attorneys-in-fact of such Persons and Affiliates.

"Agreement" means this Credit Agreement, as amended, restated, extended, supplemented or otherwise modified from time to time.

"Annual Statement" means the annual statutory financial statement of any Insurance Subsidiary required to be filed with the Department of its jurisdiction of incorporation or organization, which statement shall be in the form required by such Insurance Subsidiary's jurisdiction of incorporation or organization or, if no specific form is so required, in the form of financial statements permitted by such Department to be used for filing annual statutory financial statements and shall contain the type of information permitted or required by such Department to be disclosed therein, together with all exhibits or schedules filed therewith.

"<u>Applicable Margin</u>" means the applicable percentage per annum set forth below corresponding to the Total Consolidated Debt to Total Consolidated Capitalization Ratio as of the most recent fiscal quarter of the Borrower for which a Compliance Certificate has been or is required to have been delivered pursuant to <u>Section 6.2(b)</u>:

Pricing Level	Total Consolidated Debt to Total Consolidated Capitalization Ratio	Applicable Margin for Eurodollar Loans	Applicable Margin for Base Rate Loans
I	£ 10.0%	1.375%	0.375%
II	> 10.0% £ 15.0%	1.575%	0.575%
III	> 15.0% £ 22.5%	1.775%	0.775%
IV	> 22.5% < 30.0%	2.100%	1.100%
V	<sup>3</sup> 30.0%	2.500%	1.500%

The Applicable Margin in effect from the Closing Date through the first Business Day immediately following the date the first Compliance Certificate is delivered to the Administrative Agent pursuant to Section 6.2(b) shall be the Applicable Margin set forth in pricing level III. Any increase or decrease in the Applicable Margin resulting from a change in the Total Consolidated Debt to Total Consolidated Capitalization Ratio, as set forth on a Compliance Certificate delivered pursuant to Section 6.2(b), shall become effective as of the first Business Day immediately following delivery of such Compliance Certificate; provided, however, that if a Compliance Certificate is not delivered when due

2

in accordance with <u>Section 6.2(b)</u>, then the pricing level V Applicable Margin shall apply as of the first Business Day after the date on which such Compliance Certificate was required to have been delivered until the first Business Day after the date on which such Compliance Certificate is delivered.

Notwithstanding anything to the contrary contained in this definition, the determination of the Applicable Margin for any period shall be subject to the provisions of Section 2.12(b).

"Application" means an application and agreement for the issuance or amendment of a Letter of Credit in the form from time to time used by the applicable Issuing Lender, which shall not be inconsistent with this Agreement or impose additional obligations on the Borrower.

"Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"Arrangers" means Merrill Lynch, Pierce, Fenner & Smith Incorporated and Barclays Capital, in their respective capacities as joint lead arrangers and joint book runners.

"Assignee Group" means two or more Eligible Assignees that are Affiliates of one another or two or more Approved Funds managed by the same investment advisor.

"<u>Assignment and Assumption</u>" means an assignment and assumption entered into by a Lender and an Eligible Assignee, and accepted by the Administrative Agent, substantially in the form of <u>Exhibit H</u> or any other form approved by the Administrative Agent.

"Attorney Costs" means and includes all reasonable fees, expenses and disbursements of any law firm or other external counsel.

"Auto-Extension Letter of Credit" has the meaning specified in Section 3.2(c).

"Available Revolving Credit Commitment" means, with respect to any Lender at any time, an amount equal to the excess, if any, of (a) such Lender's Revolving Credit Commitment then in effect over (b) such Lender's Revolving Extensions of Credit then outstanding.

"Barclays Capital" means, Barclays Capital, the investment banking division of Barclays Bank PLC.

"Barclays Fee Letter" means that certain letter agreement dated as of June 20, 2011 by and between the Borrower and Barclays Bank PLC.

3

"Base Rate" means for any day a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate," and (c) the Eurodollar Rate plus 1.00%. The "prime rate" is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in the Base Rate due to a change in any of the foregoing shall take effect at the opening of business on the day specified in the public announcement of such change.

"Base Rate Loans" means Loans for which the applicable rate of interest is based upon the Base Rate.

"Berkshire Hathaway" means, Berkshire Hathaway Inc., or an Affiliate thereof.

"Board" means the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower Materials" has the meaning specified in Section 6.2(e).

"Borrower" has the meaning specified in the preamble hereto.

"Borrowing Date" means any Business Day specified by the Borrower as a date on which the Borrower requests the relevant Lenders to make Loans hereunder.

"<u>Borrowing Request</u>" means a notice of (a) a borrowing, (b) a conversion of Loans from one Type to the other, or (c) a continuation of Eurodollar Loans pursuant to <u>Sections 2.2</u> or <u>2.9</u> which, if in writing, shall be substantially in the form of <u>Exhibit B-1</u>.

"Business Day" means (i) with respect to any borrowing, payment or rate selection of Eurodollar Loans, a day (other than a Saturday or Sunday) on which banks generally are open in New York City for the conduct of substantially all of their commercial lending activities, interbank wire transfers can be made on the Fedwire system and dealings in Dollars are carried on in the London interbank market and (ii) for all other purposes, a day (other than a Saturday or Sunday) on which banks generally are open in New York City for the conduct of substantially all of the commercial lending activities, and interbank wire transfers can be made on the Fedwire system.

"<u>Capital Lease Obligations</u>" means, with respect to any Person, the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP; and, for the purposes of this Agreement, the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with GAAP.

4

"<u>Capital and Surplus</u>" means, as of any date, (a) as to any Insurance Subsidiary domiciled in the United States, the total surplus as regards to policyholders (or any successor line item description that contains the same information) as shown in its Annual Statement or Interim Statement, or an amount determined in a consistent manner for any date other than one as of which an Annual Statement or Interim Statement is prepared, (b) as to Sirius, the total solvency capital (or any successor line item description that contains the same information) as shown in its Annual Statement or Interim Statement, or an amount determined in a consistent manner for any date other than one as of which an Annual Statement or Interim Statement is prepared and (c) as to any other Insurance Subsidiary, the equivalent amount (determined in good faith by the Borrower).

"<u>Capital Stock</u>" means any and all shares, interests, participations or other equivalents (however designated) of capital stock or share capital of a corporation, any and all equivalent ownership interests in a Person (other than a corporation) and any and all warrants, rights or options to purchase any of the foregoing.

"Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Administrative Agent, the Issuing Lenders or the Swing Line Lender (as applicable) and the Lenders, as collateral for L/C Obligations, obligations in respect of Swing Line Loans, or obligations of Lenders to fund participations in respect of either thereof (as the context may require), cash or deposit account balances or, if the applicable Issuing Lender or the Swing Line Lender benefiting from such collateral shall agree in its sole discretion, other credit support, in each case pursuant to documentation in form and substance satisfactory to (a) the Administrative Agent and (b) the applicable Issuing Lender or the Swing Line Lender (as applicable). "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"Change of Control" means an event or series of events by which:

- (a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Act of 1934, but excluding any employee benefit plan of such person or any of its subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) other than (i) Berkshire Hathaway, (ii) Franklin Mutual or (iii) John J. Byrne or any Related Person with respect to John J. Byrne (together with, in the case of clauses (i), (ii) and (iii), their Affiliates) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Act of 1934, except that a person or group shall be deemed to have "beneficial ownership" of all securities that such person or group has the right to acquire, whether such right is exercisable immediately or only after the passage of time (such right, an "option right")), directly or indirectly, of 30% or more of the equity securities of the Borrower entitled to vote for members of the board of directors or equivalent governing body of the Borrower on a fully-diluted basis (and taking into account all such securities that such "person" or "group" has the right to acquire pursuant to any option right); or
- (b) during any period of 24 consecutive months, a majority of the members of the board of directors or other equivalent governing body of the Borrower

5

cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first day of such period, (ii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (iii) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body (excluding, in the case of both clause (ii) and clause (iii), any individual whose initial nomination for, or assumption of office as, a member of that board or equivalent governing body occurs as a result of an actual or threatened solicitation of proxies or consents for the election or removal of one or more directors by any person or group other than a solicitation for the election of one or more directors by or on behalf of the board of directors).

For the avoidance of doubt, none of the Capital Stock held by the entities listed in <u>clauses (a)(i)</u>, <u>(a)(ii)</u> and <u>(a)(iii)</u>, nor the Capital Stock held by any of their Affiliates, shall be included as being owned by a "person" or "group" when determining whether such "person" or "group" has met the 30% threshold set forth in <u>clause (a)</u>.

"Closing Date" means the first date on which all the conditions precedent in Section 4.1 are satisfied or waived in accordance with Section 10.1.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

" $\underline{\text{Commitments}}$ " means, collectively the Revolving Credit Commitments, the Swing Line Commitment, the L/C Commitment or as the context may require, any such Commitment.

"Compensation Period" has the meaning specified in Section 2.14(e)(ii).

"Compliance Certificate" means a certificate duly executed by a Responsible Officer on behalf of the Borrower substantially in the form of Exhibit A.

"Conditional Common Equity" means convertible preferred equity issued by the Borrower or any of its Subsidiaries which will convert to common equity of the Borrower or any of its Subsidiaries upon shareholder approval (provided that such shareholder approval is obtained within the period required by the terms thereof).

"Consolidated Net Income" means, the consolidated net income (or loss) of the Borrower and its consolidated Subsidiaries for such period, determined in accordance with GAAP; provided, however, that in calculating Consolidated Net Income, there shall be excluded for purposes of the calculation of Consolidated Net Income (x) the net income (or loss) of any Person accrued prior to the date it becomes a Subsidiary of the Borrower or is merged into or consolidated with the Borrower or any of its

6

Subsidiaries, and (y) the net unrealized investment gain (or loss), less the net unrealized foreign currency gain (or loss) on investments, less applicable related income tax provisions (or plus applicable related income tax benefits), in each case, included in the consolidated net income (or loss) of the Borrower.

"Consolidated Net Worth" means the sum of all amounts that would be included on a consolidated balance sheet of the Borrower and its consolidated Subsidiaries under shareholders' equity at such date, plus non-controlling interests, in each case, determined in accordance with GAAP; provided, however, that in calculating Consolidated Net Worth, there shall be excluded for purposes of the calculation of Consolidated Net Worth (x) the accumulated net unrealized investment gain (or loss), less the net unrealized foreign currency gain (or loss) on investments, less applicable related income tax provisions (or plus applicable related income tax benefits), in each case, as of June 30, 2011, (y) the net unrealized investment gain (or loss), less the net unrealized foreign currency gain (or loss) on investments, less applicable related income tax provisions (or plus applicable related income tax benefits), in each case, included in the consolidated net income (or loss) of the Borrower subsequent to June 30, 2011 and (z) the equity in the net unrealized investment gain (or loss), less applicable related income tax provisions (or plus applicable related income tax benefits), in each case, from investments in unconsolidated affiliates.

"Cure Period" has the meaning specified in Section 7.1(b).

"Contractual Obligation" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its Property is bound.

"DAC Write-off" has the meaning specified in Section 7.1(a).

"Debt" means indebtedness for borrowed money.

"<u>Debtor Relief Laws</u>" the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief laws of the United States or other applicable jurisdictions, domestic or foreign, from time to time in effect and affecting the rights of creditors generally.

"<u>Default</u>" means any of the events specified in <u>Section 8.1</u>, whether or not any requirement for the giving of notice, the lapse of time, or both, has been satisfied.

"<u>Defaulting Lender</u>" means, subject to <u>Section 2.24(b)</u>, any Lender that, as reasonably determined by the Administrative Agent, (a) has failed to perform any of its funding obligations hereunder, including in respect of its Loans or participations in respect of Letters of Credit or Swing Line Loans, within two (2) Business Days of the date required to be funded by it hereunder, unless such obligation is the subject of a good faith dispute, (b) has notified the Borrower, the Administrative Agent or any Lender that it does not intend to comply with its funding obligations or has made a public statement to that effect with respect to its funding obligations hereunder or under other agreements

-

in which it commits to extend credit (except, in each case, to the extent such funding obligations are the subject of a good faith dispute), (c) has failed, within two (2) Business Days after request by the Administrative Agent or the Borrower, to confirm in a manner satisfactory to the Administrative Agent that it will comply with its funding obligations hereunder, or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, (ii) had a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or a custodian appointed for it, or (iii) taken any action in furtherance of, or indicated its consent to, approval of or acquiescence in any such proceeding or appointment; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority.

"<u>Department</u>" means, with respect to any Insurance Subsidiary, the insurance commissioner or other Governmental Authority of such Insurance Subsidiary's jurisdiction of incorporation or organization.

"Dollars" and "\$" means lawful currency of the United States of America.

"Eligible Assignee" means any Person that meets the requirements to be an assignee under Section 10.7(b)(iii) and (v) (subject to such consents, if any, as may be required under Section 10.7(b)(iii)).

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, injunctive or equitable relief, fines, penalties or indemnities), of the Borrower or any of its Subsidiaries resulting from or based upon (a) a violation of any environmental law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) human exposure to any Hazardous Materials, (d) the release or threatened release of any Hazardous Materials into the environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time.

"ERISA Affiliate" means any trade or business (whether or not incorporated) under common control with the Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

"ERISA Event" means (a) a Reportable Event with respect to a Pension Plan (other than a Multiemployer Plan); (b) the withdrawal of the Borrower or any ERISA Affiliate from a Pension Plan subject to Section 4063 of ERISA during a plan year in which such entity was a "substantial employer" as defined in Section 4001(a)(2) of ERISA or a cessation of operations that is treated as such a withdrawal under Section

8

4062(e) of ERISA; (c) a complete or partial withdrawal by the Borrower or any ERISA Affiliate from a Multiemployer Plan or notification that a Multiemployer Plan is in Reorganization; (d) the filing of a notice of intent to terminate, the treatment of a Pension Plan amendment as a termination under Section 4041 or 4041A of ERISA; (e) the institution by the PBGC of proceedings to terminate a Pension Plan; (f) any event or condition which constitutes grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan (other than a Multiemployer Plan); (g) the determination that any Pension Plan (other than a Multiemployer Plan) is considered an at-risk plan or a plan in endangered or critical status within the meaning of Sections 430, 431 and 432 of the Code or Sections 303, 304 and 305 of ERISA; (h) the receipt by the Borrower or any ERISA Affiliate of any notice that a Multiemployer Plan is in endangered status under Section 305 of ERISA, or (i) the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, upon the Borrower or any ERISA Affiliate.

"Esurance Sale" shall mean the sale by the Borrower of White Mountains, Inc., its Subsidiaries and Answer Financial Inc. to the Allstate Corporation.

"Esurance Segment" shall mean the Esurance segment as described in the Borrower's Annual Report on Form 10-K for the fiscal year ended December 31, 2010.

"<u>Eurodollar Loans</u>" means Loans for which the applicable rate of interest is based upon clause (a) of the definition of "Eurodollar Rate."

# "Eurodollar Rate" means:

- (a) for any Interest Period with respect to a Eurodollar Loan, the rate per annum equal to (i) the British Bankers Association LIBOR Rate ("BBA LIBOR"), as published by Reuters (or such other commercially available source providing quotations of BBA LIBOR as may be designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period or, (ii) if such rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Loan being made, continued or converted and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 11:00 a.m. (London time) two Business Days prior to the commencement of such Interest Period; and
- (b) for any interest calculation with respect to a Base Rate Loan on any date, the rate per annum equal to (i) BBA LIBOR, at approximately 11:00 a.m., London time determined two Business Days prior to such date for Dollar deposits being delivered in the London interbank market for a term of one month commencing that day or (ii) if such published rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the

9

date of determination in same day funds in the approximate amount of the Base Rate Loan being made or maintained and with a term equal to one month would be offered by Bank of America's London Branch to major banks in the London interbank Eurodollar market at their request at the date and time of determination.

"Existing Credit Agreement" means that certain Credit Agreement, dated as of June 19, 2007 (as amended), among the Borrower, the lenders from time to time party thereto, Bank of America as administrative agent and the other agents and arrangers from time to time party thereto.

"Existing Letters of Credit" means those letters of credit set forth on Schedule 1A.

"Event of Default" means any of the events specified in <u>Section 8.1</u>, provided that any requirement for the giving of notice, the lapse of time, or both, has been satisfied.

"<u>Facility Fee Rate</u>" means the applicable percentage per annum set forth below corresponding to the Total Consolidated Debt to Total Consolidated Capitalization Ratio as of the most recent fiscal quarter of the Borrower for which a Compliance Certificate has been or is required to have been delivered pursuant to <u>Section 6.2(b</u>):

Pricing Level	Total Consolidated Debt to Total Consolidated Capitalization Ratio	Facility Fee Rate
I	£ 10.0%	0.250%
II	> 10.0% £ 15.0%	0.300%
III	> 15.0% £ 22.5%	0.350%
IV	> 22.5% < 30.0%	0.400%
V	<sup>3</sup> 30.0%	0.500%

The Facility Fee Rate in effect from the Closing Date through the first Business Day immediately following the date the first Compliance Certificate is delivered to the Administrative Agent pursuant to Section 6.2(b) shall be the Facility Fee Rate set forth in pricing level III. Any increase or decrease in the Facility Fee Rate resulting from a change in the Total Consolidated Debt to Total Consolidated Capitalization Ratio, as set forth on a Compliance Certificate delivered pursuant to Section 6.2(b), shall become effective as of the first Business Day immediately following delivery of such Compliance Certificate; provided, however, that if a Compliance Certificate is not delivered when due in accordance with Section 6.2(b), then the level V Facility Fee Rate shall apply as of the first Business Day after the date on which such Compliance Certificate was required to have been delivered until the first Business Day after the date on which such Compliance Certificate is delivered.

10

Notwithstanding anything to the contrary contained in this definition, the determination of the Facility Fee Rate for any period shall be subject to the provisions of Section 2.12(b).

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as reasonably determined by the Administrative Agent.

"Fee Letters" means, collectively, the Agent Fee Letter and the Barclays Fee Letter.

"Fitch" means Fitch, Inc. (or any successor thereto).

"<u>Franklin Mutual</u>" means any investment fund managed by Franklin Mutual Advisers LLC (or any successor thereto) or any of its Affiliates.

"Fronting Exposure" means, at any time there is a Lender that is a Defaulting Lender, (a) with respect to any Issuing Lender, such Defaulting Lender's Revolving Credit Percentage of the outstanding L/C Obligations other than L/C Obligations as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof, and (b) with respect to the Swing Line Lender, such Defaulting Lender's Revolving Credit Percentage of Swing Line Loans other than Swing Line Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof.

"<u>Fund</u>" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"Fundamental Change" means any of (a) the Borrower consolidating or amalgamating with or merging into any other Person, (b) the Borrower failing to preserve, renew and keep, in full force and effect, its corporate existence, (c) the Borrower, directly or indirectly through any of its Subsidiaries, conveying or transferring the properties and assets of the Borrower and its Subsidiaries (taken as a whole for the Borrower and its Subsidiaries) substantially as an entirety (other than to the Borrower or any of its Subsidiaries), or (d) the Borrower liquidating, winding up or dissolving itself, other than, in the case of clauses (a) through (d), any such transaction or transactions the sole purpose of which is to change the domicile of the Borrower (in any such

11

redomiciliation (x) the surviving, amalgamated or transferee entity shall expressly assume, by an agreement reasonably satisfactory to the Administrative Agent, the obligations of the Borrower to be performed or observed hereunder and deliver to the Administrative Agent such corporate authority documents and legal opinions as the Administrative Agent shall reasonably request, (y) the surviving, amalgamated or transferee entity shall succeed to, and be substituted for, and may exercise every right and power of, the Borrower under this Agreement with the same effect as

if such surviving, amalgamated or transferee entity had been named as the Borrower herein and (z) the surviving, amalgamated or transferee entity shall be organized under the laws of the United States of America, any state thereof, the District of Columbia or Bermuda).

"GAAP" means generally accepted accounting principles in the United States of America as in effect from time to time and set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination.

"Governmental Authority" means any nation or government, any state or other political subdivision thereof whether state or local and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, and any corporation or other entity owned or controlled, through stock or capital ownership or otherwise, by any of the foregoing, including any board of insurance, insurance department or insurance commissioner.

"Granting Lender" has the meaning specified in Section 10.7(h).

"Guarantee Obligation" means as to any Person (the "guaranteeing person"), any obligation of (a) the guaranteeing person or (b) another Person (including, without limitation, any bank under any letter of credit) to induce the creation of which the guaranteeing person has issued a reimbursement, counterindemnity or similar obligation, in either case guaranteeing or in effect guaranteeing any Indebtedness, leases, dividends or other obligations (the "primary obligations") of any other third Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of the guaranteeing person, whether or not contingent, (i) to purchase any such primary obligation or any Property constituting direct or indirect security therefor, (ii) to advance or supply funds (1) for the purchase or payment of any such primary obligation or (2) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase Property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (iv) otherwise to assure or hold harmless the owner of any such primary obligation against loss in respect thereof; provided, however, that the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation of any guaranteeing person shall be deemed to be the lower of (a) an amount equal to the stated

12

or determinable amount of the primary obligation in respect of which such Guarantee Obligation is made and (b) the maximum amount for which such guaranteeing person may be liable pursuant to the terms of the instrument embodying such Guarantee Obligation, unless such primary obligation and the maximum amount for which such guaranteeing person may be liable are not stated or determinable, in which case the amount of such Guarantee Obligation shall be such guaranteeing person's maximum reasonably anticipated liability in respect thereof as determined by the Borrower in good faith.

"Guarantors" means each Initial Guarantor and any other Subsidiary of the Borrower that is or becomes a party to the Guaranty at the election of the Borrower (each, an "Additional Guarantor"), in each case unless and until such Person is released from such Guaranty pursuant to Section 10.17.

"Guaranty" means that certain Guaranty, substantially in the form of Exhibit J, dated as of the Closing Date by the Initial Guarantors in favor of the Administrative Agent and the Lenders, as amended, restated, extended, supplemented or otherwise modified from time to time.

"<u>Hazardous Materials</u>" means all explosive or radioactive substances or wastes, hazardous or toxic substances or wastes, pollutants, solid, liquid or gaseous wastes, including petroleum or petroleum distillates, asbestos or asbestos containing materials, polychlorinated biphenyls ("<u>PCBs</u>") or PCB-containing materials or equipment, radon gas, infectious or medical wastes and all other substances or wastes of any nature regulated pursuant to any environmental law.

"<u>Hedge Agreements</u>" means all interest rate swaps, caps or collar agreements or similar arrangements entered into by the Borrower or any of its Subsidiaries providing for protection against fluctuations in interest rates or currency exchange rates or otherwise providing for the exchange of nominal interest obligations, either generally or under specific contingencies.

"Increase Effective Date" has the meaning specified in Section 2.22(b).

"Indebtedness" means, as to any Person at any date, without duplication, all of the following, whether or not included as Indebtedness or liabilities in accordance with GAAP (a) all Debt of such Person, (b) all obligations of such Person for the deferred purchase price of Property or services (other than trade payables incurred in the ordinary course of such Person's business), (c) all obligations of such Person evidenced by notes, bonds, debentures or other similar instruments, (d) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to Property acquired by such Person (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such Property), (e) all Capital Lease Obligations of such Person, (f) all obligations of such Person, contingent or otherwise, as an account party or applicant under acceptance, letter of credit, bank guarantees, surety bonds or similar facilities, (g) all obligations of such Person, contingent or otherwise, to purchase, redeem, retire, defease or otherwise acquire

13

for value any Capital Stock of such Person, (h) all Guarantee Obligations of such Person in respect of any of the foregoing, (i) all obligations of the kind referred to in clauses (a) through (h) above secured by (or for which the holder of such obligation has an existing right, contingent or otherwise, to be secured by) any Lien on Property (including, without limitation, accounts and contract rights) owned by such Person, whether or not such Person has assumed or become liable for the payment of such obligation and (j) for the purposes of Section 8.1(h) only, all obligations of such Person in respect of Hedge Agreements.

- "Indemnified Liabilities" has the meaning specified in Section 10.6.
- "Indemnitees" has the meaning specified in Section 10.6.
- "Initial Guarantors" means Lone Tree Insurance Group Ltd., a company organized under the laws of Bermuda, and Lone Tree Holdings Ltd., a company organized under the laws of Bermuda.
  - "Instrument of Accession" has the meaning specified in Section 2.22.
  - "Insurance Regulations" means any Law, directive or order applicable to an insurance company.
  - "Insurance Regulator" means any Person charged with the administration, oversight or enforcement of any Insurance Regulation.
- "Insurance Subsidiary" means any Subsidiary which is required to be licensed by any Department as an insurer or reinsurer and each direct or indirect Subsidiary of such Subsidiary.
- "Intellectual Property" means the collective reference to all rights, priorities and privileges relating to intellectual property, arising under Laws, including, without limitation, copyrights, copyright licenses, patents, patent licenses, trademarks, trademark licenses, technology, know-how and processes, and all rights to sue at law or in equity for any infringement or other impairment thereof, including the right to receive all proceeds and damages therefrom.
- "Interest Payment Date" means (a) as to any Base Rate Loan, the last Business Day of each of March, June, September and December and the last day of the Revolving Credit Commitment Period, (b) as to any Eurodollar Loan, the last day of each Interest Period applicable to such Loan and the last day of the Revolving Credit Commitment Period; provided, however, that if any Interest Period for a Eurodollar Loan exceeds three months, the respective dates that fall every three months after the beginning of such Interest Period shall also be Interest Payment Dates, and (c) as to any Loan (other than a Base Rate Loan), the date of any repayment or prepayment made in respect thereof.
- "Interest Period" means, as to any Eurodollar Loan, (a) initially, the period commencing on the borrowing or conversion date, as the case may be, with respect to

such Eurodollar Loan and ending one, two, three or six months (or, unless unavailable to any Lender, nine or twelve months) thereafter, as selected by the Borrower in its notice of borrowing or notice of conversion, as the case may be, given with respect thereto; and (b) thereafter, each period commencing on the last day of the next preceding Interest Period applicable to such Eurodollar Loan and ending one, two, three or six months (or, unless unavailable to any Lender, nine or twelve months) thereafter, as selected by the Borrower by irrevocable notice to the Administrative Agent not less than three Business Days prior to the last day of the then current Interest Period with respect thereto; <u>provided</u> that, all of the foregoing provisions relating to Interest Periods are subject to the following:

- (i) if any Interest Period would otherwise end on a day that is not a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless the result of such extension would be to carry such Interest Period into another calendar month in which event such Interest Period shall end on the immediately preceding Business Day;
- (ii) any Interest Period in respect of the Loans that would otherwise extend beyond the Revolving Credit Termination Date shall end on the Revolving Credit Termination Date, and
- (iii) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period.

"Interim Statement" means any interim statutory financial statement or financial report (whether quarterly, semiannually or otherwise) of any Insurance Subsidiary required to be filed with the Department of its jurisdiction of incorporation or organization, which statement or report shall be in the form required by such Insurance Subsidiary's jurisdiction of incorporation or organization or, if no specific form is so required, in the form of financial statements or financial reports permitted by such Department to be used for filing interim statutory financial statements or financial reports and shall contain the type of information permitted or required by such Department to be disclosed therein, together with all exhibits or schedules filed therewith.

"ISP" means, with respect to any Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice (or such later version thereof as may be in effect at the time of issuance).

"Issuer Documents" means with respect to any Letter of Credit, the Application, and any other document, agreement and instrument entered into by the applicable Issuing Lender and the Borrower (or any Subsidiary) or by the Borrower (or any Subsidiary) in favor of the applicable Issuing Lender and relating to any such Letter of Credit.

15

"<u>Issuing Lender</u>" means Bank of America and any other Lender from time to time designated by the Borrower as an Issuing Lender, with the consent of such Lender and the Administrative Agent.

"<u>L/C Advance</u>" means, with respect to each Lender, such Lender's funding of its participation in any L/C Borrowing in accordance with its Revolving Credit Percentage.

"L/C Borrowing" means an extension of credit resulting from a drawing under any Letter of Credit which has not been reimbursed on the date when made or refinanced as a borrowing.

"L/C Commitment" means \$100,000,000, as the same may be reduced from time to time pursuant to Section 2.7.

"<u>L/C Credit Extension</u>" means, with respect to any Letter of Credit, the issuance thereof or extension of the expiry date thereof or the increase of the amount thereof.

"L/C Fee Payment Date" means the first Business Day of each of January, April, July and October and the last day of the Revolving Credit Commitment Period.

"L/C Obligations" means, at any time, an amount equal to the sum of (a) the aggregate amount available to be drawn under all outstanding Letters of Credit and (b) the aggregate amount of drawings under Letters of Credit that have not then been reimbursed pursuant to Section 3.3. For purposes of computing the amount available to be drawn under any Letter of Credit, the amount of such Letter of Credit shall be determined in accordance with Section 1.3. For all purposes of this Agreement, if on any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"L/C Participants" means, with respect to any Letter of Credit, the collective reference to all of the Lenders, other than the Issuing Lender that issued such Letter of Credit.

"<a href="Lenders" has the meaning specified in the preamble hereto.</a>  $\!\!$ 

"Letters of Credit" means any standby letters of credit issued hereunder and shall include the Existing Letters of Credit.

"<u>License</u>" means any license, certificate of authority, permit or other authorization which is required to be obtained from any Governmental Authority in

16

connection with the operation, ownership or transaction of insurance or reinsurance business.

"Lien" means any mortgage, pledge, security interest, encumbrance, charge or security interest of any kind.

"<u>Loan</u>" means any loan made by any Lender to the Borrower pursuant to this Agreement, including any Revolving Credit Loan and any Swing Line Loan made by the Swing Line Lender.

"Loan Documents" means this Agreement, the Applications, the Guaranty, the Notes, any agreement creating or perfecting rights in Cash Collateral pursuant to the provisions of Section 2.23 of this Agreement and any Instrument of Accession executed hereunder pursuant to Section 2.22.

"Majority Lenders" means the holders of more than 50% of the Total Revolving Extensions of Credit (or, if no such Revolving Extensions of Credit are outstanding, prior to any termination of the Revolving Credit Commitments, the holders of more than 50% of the Total Revolving Credit Commitments). The Revolving Credit Commitment in effect (or, when applicable, Revolving Extensions of Credit outstanding) of any Defaulting Lender shall be excluded for purposes of making a determination of Majority Lenders.

"Mandatory Convertible Securities" means equity securities or subordinated debt securities (which subordinated debt securities, if issued by the Borrower or a Guarantor, will include subordination to the obligations of the Borrower and the Guarantors hereunder), issued by the Borrower or any of its Subsidiaries which (i) are not (x) Mandatory Redeemable Securities (other than Qualified Securities) or (y) Conditional Common Equity and (ii) provide, pursuant to the terms thereof, that the issuer of such securities (or an affiliate of such issuer) may cause (without the payment of additional cash consideration by the issuer thereof) the conversion or exchange of, or has agreed to convert or exchange, such securities to or for equity securities of the Borrower or any of its Subsidiaries upon the occurrence of a certain date or of certain events. A Mandatory Convertible Security that is also a Qualified Security shall be treated as a Mandatory Convertible Security.

"Mandatory Redeemable Securities" means debt or equity securities (other than Conditional Common Equity, so long as such Conditional Common Equity may not be required, by the holder thereof, to be repurchased or redeemed during the period provided for shareholder approval of conversion pursuant to the terms of such Conditional Common Equity) issued by the Borrower or any of its Subsidiaries which either (i) are subordinated debt securities (which subordinated debt securities, if issued by the Borrower or a Guarantor, will include subordination to the obligations of the Borrower and the Guarantors hereunder), or (ii) provide, pursuant to the terms thereof, that such securities must be repurchased or redeemed, or the holder of such securities may require the issuer of such securities to repurchase or redeem such securities, upon the occurrence of a certain date or of certain events.

17

- "<u>Material Insurance Subsidiary</u>" means any Insurance Subsidiary (whether existing on or acquired or formed after the Closing Date) having Capital and Surplus equal to 10% or more of the Consolidated Net Worth of the Borrower as of the most recent Annual Statement or Interim Statement of such Insurance Subsidiary.
  - "Maximum Rate" has the meaning specified in Section 10.19(a).
  - "Moody's" means Moody's Investors Service, Inc. (or any successor thereto).
- "Multiemployer Plan" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA, to which the Borrower or any ERISA Affiliate makes or is obligated to make contributions, or during the preceding five plan years, has made or been obligated to make contributions.
- "NAIC" means the National Association of Insurance Commissioners or any successor thereto, or in the absence of the National Association of Insurance Commissioners or such successor, any other association, agency or other organization performing advisory, coordination or other like functions among insurance departments, insurance commissioners and similar Governmental Authorities of the various states of the United States towards the promotion of uniformity in the practices of such Governmental Authorities.
  - "Non-Excluded Taxes" has the meaning specified in Section 2.16(a).
  - "Non-Extension Notice Date" has the meaning specified in Section 3.2(c).
- "Non-Regulated Operating Subsidiary." means each Subsidiary of the Borrower engaged directly (as opposed to indirectly through the ownership of Capital Stock of a Person engaged in a Principal Business) in a Principal Business, whether now owned or hereafter acquired, which is not an Insurance Subsidiary.
  - "Non-U.S. Lender" has the meaning specified in Section 2.16(d).
- "Note" means any promissory note, including any revolving credit note or swing line note, made by the Borrower in favor of a Lender evidencing any Loan, substantially in the forms of Exhibit C-1 and C-2, as the case may be and as any such Note may be amended, restated, supplemented, modified or replaced from time to time.
- "OneBeacon Insurance Group" means the grouping of Insurance Subsidiaries of OneBeacon Limited identified by NAIC Group Code 1129 (or any successor grouping equivalent thereto).

- "OneBeacon Limited" means OneBeacon Insurance Group, Ltd., a company existing under the laws of Bermuda.
- "OneBeacon Notes" means those certain 5.875% Senior Notes issued pursuant to the Senior Indenture, dated as of May 19, 2003, between the Borrower, OneBeacon U.S. Holdings and The Bank of New York Mellon Trust Company, N.A., as trustee, in the initial aggregate principal amount of \$700,000,000 due May 15, 2013.
- " $\underline{OneBeacon~U.S.~Holdings}$ " means OneBeacon~U.S. Holdings, Inc., a Delaware corporation (f/k/a Fund American Companies, Inc.).
- "Other Taxes" means any and all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.
  - "Participant" has the meaning specified in Section 10.7(d).
- "PBGC" means the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA (or any successor).
- "<u>Pension Funding Rules</u>" means the rules of the Code and ERISA regarding minimum required contributions (including any installment payment thereof) to Pension Plans and set forth in Section 412, 430, 431, 432 and 436 of the Code and Sections 302, 303, 304 and 305 of ERISA.
- "Pension Plan" means any employee pension benefit plan (including a Multiemployer Plan) that is maintained or is contributed to by the Borrower and any ERISA Affiliate and is either covered by Title IV of ERISA or is subject to the minimum funding standards under Section 412 of the Code.
- "Permitted Liens" means (a) any Lien upon Property to secure any part of the cost of development, construction, alteration, repair or improvement of such Property, or Indebtedness incurred to finance such cost; (b) any extension, renewal or replacement, in whole or in part, of any Lien referred to in the foregoing <a href="clause(a)">clause(a)</a>; (c) any Lien relating to a sale and leaseback transaction; (d) any Lien in favor of the Borrower or any Subsidiary granted by the Borrower or any Subsidiary in order to secure any intercompany obligations; (e) mechanic's, materialmen's, carriers' or other like Liens arising in the ordinary course of business (including construction of facilities) in respect of obligations which are not due or which are being contested in good faith; (f) any Lien arising in connection with any legal proceeding which is being contested in good faith; (g) Liens for taxes not yet subject to penalties for non-payment or which are being contested in good faith by appropriate proceedings; (h) minor survey exceptions, minor encumbrances, easements or reservations of, or rights of others for, licenses, rights-of-way, sewers, electric lines, telegraph and telephone lines and other similar purposes, or zoning or other restrictions as to the use of real property or Liens incidental to the conduct of the business of such Person or to the ownership of its properties which were not incurred in connection with Indebtedness and which do not in the aggregate

materially adversely affect the value of said properties or materially impair their use in the operation of the business of such Person; (i) pledges or deposits under workers' compensation Laws, unemployment insurance Laws or similar social security legislation; (j) any pledge or deposit to secure performance of letters of credit, bank guarantees, bids, leases, statutory obligations, surety and appeal bonds, performance bonds or other obligations of a like nature arising in the ordinary course of business; (k) any interest or title of a lessor under any lease entered into in the ordinary course of business; (l) Liens on assets of any Insurance Subsidiary securing (i) short-term Indebtedness (i.e. with a maturity of less than one year when issued, provided that such Indebtedness may include an option to extend for up to an additional one year period) incurred or issued to provide short-term liquidity to facilitate claims payments in the event of catastrophe, (ii) Indebtedness incurred or issued in the ordinary course of its business or in securing insurance-related obligations (that do not constitute Indebtedness) or (iii) insurance Subsidiary in the ordinary course of its business or in securing insurance-related obligations (that do not constitute Indebtedness) or (iii) insurance-related obligations (that do not constitute Indebtedness) or (iii) insurance-related obligations (that do not constitute Indebtedness) or (iii) insurance-related obligations (that do not constitute Indebtedness) or (iii) insurance-related obligations (that do not constitute Indebtedness) or (iii) insurance-related obligations (that do not constitute Indebtedness) (m) judgment liens in respect of judgments that do not constitute an Event of Default under Section 8.1(j); (n) Liens on securities that are lent pursuant to securities lending activities; and (o) Liens securing the obligations hereunder.

"<u>Person</u>" means an individual, partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture, Governmental Authority or other entity of whatever nature.

"Plan" means any employee benefit plan within the meaning of Section 3(3) of ERISA (including a Pension Plan), maintained for employees of the Borrower or any ERISA Affiliate or any such Plan to which the Borrower or any ERISA Affiliate is required to contribute on behalf of any of its employees.

"Platform" has the meaning specified in Section 6.2(e).

"<u>Preference Shares</u>" means those certain fixed/floating perpetual non-cumulative preference shares, liquidation preference \$1,000 per share, of White Mountains Re.

"<u>Principal Business</u>" means (a) a business of the type engaged in by the Borrower and its Subsidiaries on the date of this Agreement, (b) any other insurance, insurance services, insurance related or risk management related business and (c) any business reasonably incident to any of the foregoing.

"Property" means any property of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible.

"Qualified Securities" means (a) Mandatory Redeemable Securities issued by the Borrower or any of its Subsidiaries that, pursuant to the terms thereof, must be redeemed or repurchased or repaid, or may be required to be redeemed or repurchased or

20

repaid at the option of the holder of such securities (excluding redemption, repurchase or repayment upon the occurrence of one or more events or conditions but including redemption, repurchase or repayment upon the occurrence of a certain date), (i) if such Mandatory Redeemable Securities are equity securities or subordinated debt securities, not sooner than the Revolving Credit Termination Date (except to the extent permitted by clause (ii) below) or (ii) only in exchange for equity securities or other Qualified Securities of the Borrower or any of its Subsidiaries (except to the extent permitted by clause (i) above) and (b) any other debt or equity securities issued by the Borrower or any of its Subsidiaries whose proceeds are or would be accorded, at or about the time of issuance, equity treatment by S&P.

"Refunded Swing Line Loans" has the meaning specified in Section 2.4(b).

"Refunding Date" has the meaning specified in Section 2.4(c).

"Register" has the meaning specified in Section 10.7(c).

"Regulation U" means Regulation U of the Board as in effect from time to time.

"Reimbursement Obligation" means the obligation of the Borrower to reimburse an Issuing Lender pursuant to Section 3.3(a) for amounts drawn under Letters of Credit issued by such Issuing Lender for the account of the Borrower.

"Related Person" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents and advisors of such Person and of such Person's Affiliates.

"Reorganization" means, with respect to any Multiemployer Plan, the condition that such plan is in reorganization within the meaning of Section 4241 of ERISA.

"Reportable Event" means any of the events set forth in Section 4043(c) of ERISA, other than events for which the 30 day notice period has been waived.

"Requested Reimbursement Date" has the meaning specified in Section 3.3(a).

"Requirement of Law" means, as to any Person, the Certificate of Incorporation and By-Laws or other organizational or governing documents of such Person (excluding, in the case of Section 2.15(a)(i), any of the foregoing relating to the Administrative Agent or any Lender), and any Law, in each case applicable to or binding upon such Person or any of its Property or to which such Person or any of its Property is subject.

officer, any vice president or any managing director of the Borrower or any Subsidiary, as the context requires. Any document delivered hereunder that is signed by a Responsible Officer on behalf of the Borrower or a Subsidiary shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of the Borrower or such Subsidiary and such Responsible Officer shall be conclusively presumed to have acted on behalf of the Borrower or such Subsidiary.

"Revolving Credit Commitment" means, as to any Lender, the obligation of such Lender, if any, to make Revolving Credit Loans and participate in Swing Line Loans and Letters of Credit, in an aggregate principal or face amount not to exceed the amount set forth under the heading "Revolving Credit Commitment" opposite such Lender's name on Schedule 1 to this Agreement, or, as the case may be, in the Assignment and Assumption pursuant to which such Lender became a party hereto, as the same may be adjusted from time to time pursuant to the terms hereof.

"Revolving Credit Commitment Period" means the period from and including the Closing Date to the earliest of (a) the Revolving Credit Termination Date, (b) the date of termination of the Revolving Credit Commitments pursuant to Section 2.7, and (c) the date of termination of the commitment of each Lender to make Loans and of the obligation of the Issuing Lenders to make L/C Credit Extensions pursuant to Section 8.2.

"Revolving Credit Loans" has the meaning specified in Section 2.1.

"Revolving Credit Percentage" means, as to any Lender at any time, subject to Section 2.24, the percentage (carried out to the ninth decimal place) which such Lender's Revolving Credit Commitment then constitutes of the Total Revolving Credit Commitments (or, at any time after the commitment of each Lender to make Loans and the obligation of each Issuing Lender to make L/C Credit Extensions shall have terminated pursuant to Section 8.2 or if all of the Revolving Credit Commitments shall have expired, then the Revolving Credit Percentage of each Lender shall be determined based on the Revolving Credit Percentage of such Lender most recently in effect, giving effect to any subsequent assignments).

"Revolving Credit Termination Date" means August 12, 2015; provided, however, that, if such date is not a Business Day, the Revolving Credit Termination Date shall be the next succeeding Business Day.

"Revolving Extensions of Credit" means, as to any Lender at any time, an amount equal to the sum of (a) the aggregate principal amount of all Revolving Credit Loans made by such Lender then outstanding, (b) the principal amount equal to such Lender's Revolving Credit Percentage of the L/C Obligations then outstanding and (c) the principal amount equal to such Lender's Revolving Credit Percentage of the aggregate principal amount of Swing Line Loans then outstanding.

"S&P" means Standard & Poor's Rating Services (or any successor thereto).

22

"SAP" means with respect to any Insurance Subsidiary, the statutory accounting practices prescribed or permitted by the Department in the jurisdiction of incorporation or organization of such Insurance Subsidiary for the preparation of annual statements and other financial reports by insurance companies of the same type as such Insurance Subsidiary, which are applicable to the circumstances as of the date of determination.

"SEC" means the Securities and Exchange Commission (or successors thereto or an analogous Governmental Authority).

"SFAS" means the Statements of Financial Accounting Standards adopted by the Financial Accounting Standards Board.

"Sirius" means Sirius International Insurance Corporation, a Swedish corporation.

"SPC" has the meaning specified in Section 10.7(h).

8.1(c).

"Specified Event of Default" means an Event of Default pursuant to Sections 8.1(a), 8.1(b) (with respect to Section 7.1 only) or

"Stated Rate" has the meaning specified in Section 10.20(a).

"Subsidiary" of a Person means (a) any corporation more than 50% of the outstanding securities having ordinary voting power of which shall at the time be owned or controlled, directly or indirectly, by such Person or by one or more of its Subsidiaries or by such Person and one or more of its Subsidiaries, or (b) any partnership, limited liability company, association, joint venture or similar business organization more than 50% of the ownership interests having ordinary voting power of which shall at the time be so owned or controlled. Unless otherwise expressly provided, all references herein to a "Subsidiary" shall mean a Subsidiary of the Borrower.

"Swing Line Commitment" means the obligation of the Swing Line Lender to make Swing Line Loans pursuant to Section 2.3 in an aggregate principal amount at any one time outstanding not to exceed \$50,000,000.

"Swing Line Lender" means Bank of America, in its capacity as provider of Swing Line Loans, or any successor swing line lender hereunder.

"Swing Line Loans" has the meaning specified in Section 2.3(a).

"Swing Line Participation Amount" has the meaning specified in Section 2.4(c).

"Syndication Agent" means Barclays Capital, and any other Lender as may be designated from time to time by the Borrower as a syndication agent, with the consent of such Lender and the Arrangers.

23

"<u>Total Consolidated Capitalization</u>" means, as at any date, the sum, without duplication, of (a) Consolidated Net Worth plus (b) Total Consolidated Debt plus, (c) the amounts in respect of Trust Preferred Securities, Mandatory Convertible Securities, Mandatory Redeemable Securities, Conditional Common Equity and any other preferred equity that would, in conformity with GAAP, be reflected on a consolidated balance sheet of the Borrower and its consolidated Subsidiaries prepared as of such date and which are not already included in <u>clause (a)</u> or <u>(b)</u> above.

"Total Consolidated Debt" means, at any date, the sum, without duplication, of (a) all amounts that would, in conformity with GAAP, be reflected and classified as debt on a consolidated balance sheet of the Borrower and its consolidated Subsidiaries prepared as of such date (other than amounts excluded by clauses (b) and (c) below), (b) Indebtedness represented by (i) Trust Preferred Securities or Qualified Securities (in each case, owned by Persons other than the Borrower or any of its consolidated Subsidiaries) but only to the extent that such securities (other than Mandatory Convertible Securities) exceed 15% of Total Consolidated Capitalization or (ii) Mandatory Redeemable Securities (owned by Persons other than the Borrower or any of its consolidated Subsidiaries) but only to the extent that such Mandatory Convertible Securities (owned by Persons other than the Borrower or any of its consolidated Subsidiaries) but only to the extent that such Mandatory Convertible Securities plus Trust Preferred Securities and Qualified Securities (in each case, owned by Persons other than the Borrower or any of its consolidated Subsidiaries) exceed 25% of Total Consolidated Capitalization; provided, that in the event that the notes related to the Mandatory Convertible Securities remain outstanding following the exercise of forward purchase contracts related to such Mandatory Convertible Securities, then such outstanding notes will be included in Total Consolidated Debt thereafter. Total Consolidated Debt shall, in any event, not include (1) Hedge Agreements entered into in the ordinary course of business for non-speculative purposes, (2) Indebtedness of the type described in Sections 7.2(b), (c), (d), (f), (g) and (k), (3) Conditional Common Equity, (4) any other amounts in respect of Trust Preferred Securities, Mandatory Redeemable Securities, Mandatory Convertible Securities or Qualified Securities, or (5) any amounts resulting from any entity consolidated as a Variable Interest Entity in "ASC Topic 810 — Consolidations

"<u>Total Consolidated Debt to Total Consolidated Capitalization Ratio</u>" means, as at the end of any fiscal quarter of the Borrower, the ratio of (a) Total Consolidated Debt to (b) Total Consolidated Capitalization.

"<u>Total Revolving Credit Commitments</u>" means, at any time, the aggregate amount of Revolving Credit Commitments then in effect. The aggregate amount of the Total Revolving Credit Commitments on the Closing Date is \$375,000,000.

"<u>Total Revolving Extensions of Credit</u>" means, at any time, the aggregate amount of the Revolving Extensions of Credit of the Lenders outstanding at such time.

"<u>Transferee</u>" means a Participant or an assignee of any Lender's rights and obligations under this Agreement pursuant to an Assignment and Assumption.

24

"Trust Preferred Securities" means preferred equity issued by a special purpose entity, the proceeds of which are used to purchase subordinated debt securities of the Borrower or any of its Subsidiaries having terms that substantially mirror those of such preferred equity issued by the special purpose entity such that the subordinated debt securities constitute credit support for obligations in respect of such preferred equity and such preferred equity is reflected on a consolidated balance sheet of the Borrower and its consolidated Subsidiaries in accordance with GAAP.

"Type" means, as to any Loan, its nature as a Base Rate Loan or a Eurodollar Loan.

"Unreimbursed Amount" has the meaning specified in Section 3.3(a).

"White Mountains Re" means White Mountains Re Group, Ltd., a company organized under the laws of Bermuda.

- 1.2. Other Definitional Provisions. Unless otherwise specified therein, all terms defined in this Agreement shall have the defined meanings when used in the other Loan Documents or any certificate or other document made or delivered pursuant hereto or thereto.
- (a) As used herein and in the other Loan Documents, and any certificate or other document made or delivered pursuant hereto or thereto, accounting terms relating to the Borrower or any of its Subsidiaries not defined in <u>Section 1.1</u> and accounting terms partly defined in <u>Section 1.1</u>, to the extent not defined, shall have the respective meanings given to them under GAAP or SAP, as the case may be.
- (b) References herein to particular pages, columns, lines or sections of any Person's Annual Statement shall be deemed, where appropriate, to be references to the corresponding page, column, line or section of such Person's Interim Statement, or if no such corresponding page, column, line or section exists or if any report form changes, then to the corresponding item referenced thereby.
- (c) The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section, Schedule and Exhibit references are to this Agreement unless otherwise specified.
  - (d) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.
- (e) The word "or" is not exclusive and the words "include", "includes" or "including" shall be deemed to be followed by the phrase

"without limitation".

- (f) References to "preferred equity" includes Capital Stock designated as preferred stock, preference shares, preferred shares or any similar term.
- 1.3. Letter of Credit Amounts. Unless otherwise specified herein, the amount of a Letter of Credit at any time shall be deemed to be the stated amount of such Letter of Credit in

effect at such time; <u>provided</u>, <u>however</u>, that with respect to any Letter of Credit that, by its terms or the terms of any Issuer Document related thereto, provides for one or more automatic increases in the stated amount thereof, the amount of such Letter of Credit shall be deemed to be the maximum stated amount of such Letter of Credit after giving effect to all such increases, whether or not such maximum stated amount is in effect at such time.

- 1.4. <u>Rounding</u>. Any financial ratios required to be maintained by the Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).
- 1.5. <u>Times of Day</u>. Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).
- 1.6. Accounting Terms; Utilization of GAAP for Purposes of Calculations Under Agreement. Except as otherwise expressly provided herein, all terms of an accounting or financial nature shall be construed in accordance with GAAP, as in effect from time to time; provided that, if either (x) the Borrower, by notice to the Administrative Agent or (y) the Administrative Agent or the Majority Lenders, by notice to the Borrower, requests an amendment to any provision in this Agreement or any other Loan Document to eliminate the effect of any change occurring after the Closing Date in GAAP or in the application thereof on the operation of such provision, regardless of whether any such notice is given before or after such change in GAAP or in the application thereof, then such provision shall be interpreted on the basis of GAAP as in effect and applied immediately before such change shall have become effective until such notice shall have been withdrawn or such provision amended in accordance the terms of Section 10.1.

# 2. AMOUNT AND TERMS OF COMMITMENTS

- 2.1. Revolving Credit Commitments. (a) Subject to the terms and conditions hereof each Lender severally agrees to make revolving credit loans ("Revolving Credit Loans") to the Borrower from time to time on any Business Day during the Revolving Credit Commitment Period in an aggregate principal amount at any one time outstanding for each Lender which, when added to such Lender's Revolving Credit Percentage of the sum of (A) the L/C Obligations then outstanding and (B) the aggregate principal amount of the Swing Line Loans then outstanding, does not exceed the amount of such Lender's Revolving Credit Commitment. During the Revolving Credit Commitment Period the Borrower may use the Revolving Credit Commitments by borrowing, prepaying the Revolving Credit Loans in whole or in part, and reborrowing, all in accordance with the terms and conditions hereof. The Revolving Credit Loans may from time to time be Eurodollar Loans or Base Rate Loans, as determined by the Borrower and notified to the Administrative Agent in accordance with Sections 2.2 and 2.9, provided that no Revolving Credit Loan shall be made as a Eurodollar Loan during the one month period immediately prior to the Revolving Credit Termination Date.
  - (b) The Borrower shall repay to the Lenders all outstanding Revolving Credit Loans on the Revolving Credit Termination Date.

26

- 2.2. Procedure for Revolving Credit Borrowing. The Borrower may borrow under the Revolving Credit Commitments on any Business Day during the Revolving Credit Commitment Period, provided that the Borrower shall give the Administrative Agent a borrowing request in the form of Exhibit B-1 hereto (hereinafter, a "Borrowing Request") (which Borrowing Request must be received by the Administrative Agent prior to 11:00 A.M., (a) three Business Days prior to the requested Borrowing Date, in the case of Eurodollar Loans, or (b) on the requested Borrowing Date, in the case of Base Rate Loans, provided that requests for Base Rate Loans not received prior to 11:00 A.M. on the requested Borrowing Date shall be deemed received on the following Business Day), and must specify (i) the amount and Type of Revolving Credit Loans to be borrowed, (ii) the requested Borrowing Date and (iii) in the case of Eurodollar Loans, the length of the initial Interest Period therefor; provided, however, that if the Borrower wishes to request Eurodollar Loans having an Interest Period of nine or twelve months in duration as provided in the definition of "Interest Period," the applicable notice must be received by the Administrative Agent not later than 11:00 A.M., four Business Days prior to the requested date of such borrowing, whereupon the Administrative Agent shall give prompt notice to the Lenders of such request and determine whether the requested Interest Period is unavailable to any of them. Not later than 10:00 A.M., three Business Days before the requested date of such borrowing, the Administrative Agent shall notify the Borrower (which notice may be by telephone) whether or not the requested Interest Period is unavailable to any Lender. If the Borrower requests a borrowing of Eurodollar Loans in any Borrowing Request, but fails to specify an Interest Period, it will be deemed to have specified an Interest Period of one month. Each borrowing of Revolving Credit Loans under the Revolving Credit Commitments shall be in an amount equal to (x) in the case of Base Rate Loans, \$1,000,000 or a whole multiple thereof (or, if the then aggregate Available Revolving Credit Commitments are less than \$1,000,000, such lesser amount) and (y) in the case of Eurodollar Loans, \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof; provided, that the Swing Line Lender may request, on behalf of the Borrower, borrowings of Base Rate Loans under the Revolving Credit Commitments in other amounts pursuant to Section 2.4. Upon receipt of any such notice from the Borrower, the Administrative Agent shall promptly notify each Lender thereof. Each Lender will make its Revolving Credit Percentage of the amount of each borrowing of Revolving Credit Loans available to the Administrative Agent for the account of the Borrower at the Administrative Agent's Office prior to 12:00 Noon, on the Borrowing Date requested by the Borrower in funds immediately available to the Administrative Agent. Such borrowing will then be made available to the Borrower by the Administrative Agent in like funds as received by the Administrative Agent.
- 2.3. Swing Line Commitment. (a) Subject to the terms and conditions hereof, the Swing Line Lender agrees, in reliance on the agreements of the other Lenders set forth in Section 2.4, that during the Revolving Credit Commitment Period, it will make available to the Borrower in the form of swing line loans ("Swing Line Loans") a portion of the credit otherwise available to the Borrower under the Revolving Credit Commitments; provided that (i) all Swing Line Loans shall be made at the sole and absolute discretion of the Swing Line Lender, (ii) the aggregate principal amount of Swing Line Loans outstanding at any time shall not exceed the Swing Line Commitment then in effect (notwithstanding that the Swing Line Loans outstanding at any time,

and the Swing Line Lender shall not make, any Swing Line Loan if, after giving effect to the making of such Swing Line Loan, the aggregate amount of the Available Revolving Credit Commitments would be less than zero. During the Revolving Credit Commitment Period, the Borrower may use the Swing Line Commitment by borrowing, repaying and reborrowing, all in accordance with the terms and conditions hereof. Swing Line Loans shall be Base Rate Loans only.

- (b) The Borrower shall repay all outstanding Swing Line Loans on the Revolving Credit Termination Date. Each payment in respect of Swing Line Loans shall be made to the Swing Line Lender.
- 2.4. Procedure for Swing Line Borrowing; Refunding of Swing Line Loans. (a) The Borrower may borrow under the Swing Line Commitment on any Business Day during the Revolving Credit Commitment Period, provided, the Borrower shall give the Swing Line Lender irrevocable telephonic notice confirmed promptly in writing in the form of Exhibit B-2 (which telephonic notice must be received by the Swing Line Lender not later than 1:00 P.M. on the proposed Borrowing Date), specifying (i) the amount to be borrowed and (ii) the requested Borrowing Date. Each borrowing under the Swing Line Commitment shall be in an amount equal to \$500,000 or a whole multiple of \$100,000 in excess thereof. If the Swing Line Lender shall elect to fund a requested Swing Line Loan, not later than 3:00 P.M. time, on the Borrowing Date specified in the borrowing notice in respect of such Swing Line Loan, the Swing Line Lender shall make available to the Administrative Agent at the Administrative Agent's Office an amount in immediately available funds equal to the amount of such Swing Line Loan. The Administrative Agent shall make the proceeds of such Swing Line Loan available to the Borrower on such Borrowing Date in like funds as received by the Administrative Agent. Notwithstanding the foregoing, if the Swing Line Lender shall elect not to fund a requested Swing Line Loan for any reason, the Swing Line Lender shall promptly, and in any event not later than 3:00 P.M., on the date that the borrowing notice in respect of such Swing Line Loan is received, notify the Borrower and the Administrative Agent of such election.
- (b) The Swing Line Lender, not less frequently than once each week shall, and at any other time, from time to time, as the Swing Line Lender elects in its sole and absolute discretion, may, on behalf of the Borrower (which hereby irrevocably directs the Swing Line Lender to act on its behalf), on one Business Day's notice given by the Swing Line Lender no later than 12:00 Noon request each Lender to make, and each Lender hereby agrees to make, a Revolving Credit Loan, in an amount equal to such Lender's Revolving Credit Percentage of the aggregate amount of the Swing Line Loans (the "Refunded Swing Line Loans") outstanding on the date of such notice, to repay the Swing Line Lender. Each Lender shall make an amount equal to its Revolving Credit Percentage of such Revolving Credit Loan available to the Administrative Agent at the Administrative Agent's Office in immediately available funds, not later than 10:00 A.M. one Business Day after the date of such notice. The proceeds of such Revolving Credit Loans (and any Cash Collateral available with respect to the applicable Swing Line Loan, if so applied by the Administrative Agent to refund such Swing Line Loan) shall be made immediately available by the Administrative Agent to the Swing Line Lender for application by the Swing Line Lender to the repayment of the Refunded Swing Line Loans. Upon the written request of any Lender, the Administrative Agent will, within three Business

28

Days of such request, inform such Lender of the aggregate amount of Swing Line Loans outstanding on the date of such request.

- (c) If prior to the time a Revolving Credit Loan would have otherwise been made pursuant to Section 2.4(b), one of the events described in Section 8.1(c) shall have occurred and be continuing with respect to the Borrower, or if for any other reason, as determined by the Swing Line Lender in its sole discretion, Revolving Credit Loans may not be made as contemplated by Section 2.4(b), each Lender shall, on the date such Revolving Credit Loan was to have been made pursuant to the notice referred to in Section 2.4(b) (the "Refunding Date"), purchase for cash an undivided participating interest in the then outstanding Swing Line Loans by paying to the Swing Line Lender an amount (the "Swing Line Participation Amount") equal to (i) such Lender's Revolving Credit Percentage times (ii) the sum of the aggregate principal amount of Swing Line Loans then outstanding which were to have been repaid with such Revolving Credit Loans.
- (d) If any Lender fails to make available to the Administrative Agent for the account of the Swing Line Lender any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.4(b) by the time specified in Section 2.4(b), the Swing Line Lender shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the Swing Line Lender at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the Swing Line Lender in accordance with banking industry rules on interbank compensation, plus any administrative, processing or similar fees customarily charged by the Swing Line Lender in connection with the foregoing. If such Lender pays such amount (with interest and fees as aforesaid), the amount so paid shall constitute such Lender's Loan included in the relevant borrowing or funded participation in the relevant Swing Line Loan, as the case may be. A certificate of the Swing Line Lender submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this paragraph (d) shall be conclusive absent manifest error.
- (e) Each Lender's obligation to make the Loans referred to in Section 2.4(b) and to purchase participating interests pursuant to Section 2.4(c) shall be absolute and unconditional and shall not be affected by any circumstance, including, without limitation, (i) any setoff, counterclaim, recoupment, defense or other right which such Lender may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever; (ii) the occurrence or continuance of a Default or an Event of Default or the failure to satisfy any of the other conditions specified in Section 4; (iii) any adverse change in the condition (financial or otherwise) of the Borrower; (iv) any breach of this Agreement or any other Loan Document by the Borrower or any Lender; or (v) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing. No such funding of risk participations shall relieve or otherwise impair the obligation of the Borrower to repay Swing Line Loans, together with interest as provided herein.
- (f) Whenever, at any time after the Swing Line Lender has received from any Lender such Lender's Swing Line Participation Amount, the Swing Line Lender receives any payment on account of the Swing Line Loans, the Swing Line Lender will distribute to such

Lender its Swing Line Participation Amount (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's participating interest was outstanding and funded and, in the case of principal and interest payments, to reflect such Lender's <u>pro rata</u> portion of such payment if such payment is not sufficient to pay the principal of and interest on all Swing Line Loans then due); <u>provided</u>, <u>however</u>, that in the event that such payment received by the Swing Line Lender is required to be returned, such Lender will return to the Swing Line Lender any portion thereof previously distributed to it by the Swing Line Lender. The obligation of the Lenders under this <u>paragraph (f)</u> shall survive the payment in full of the Loans and other obligations hereunder and the termination of this Agreement.

- (g) The Swing Line Lender shall be responsible for invoicing the Borrower for interest on the Swing Line Loans. Until each Lender funds its Refunded Swing Line Loan or risk participation pursuant to this <u>Section 2.4</u> to refinance such Lender's Revolving Credit Percentage of any Swing Line Loan, interest in respect of such Revolving Credit Percentage shall be solely for the account of the Swing Line Lender.
- (h) The Borrower shall make all payments of principal and interest in respect of the Swing Line Loans directly to the Administrative Agent for the account of the Swing Line Lender.
- 2.5. Repayment of Loans; Evidence of Debt. (a) The Borrower hereby unconditionally promises to pay to the Administrative Agent for the account of the appropriate Lender (i) the then unpaid principal amount on the Revolving Credit Termination Date (or on such earlier date on which the Loans become due and payable pursuant to Section 8.2) of each Revolving Credit Loan of such Lender, and (ii) the then unpaid principal amount on the Revolving Credit Termination Date (or on such earlier date on which the Loans become due and payable pursuant to Section 8.2) of each Swing Line Loan of the Swing Line Lender. The Borrower hereby further agrees to pay interest to the Administrative Agent for the account of the appropriate Lender on the unpaid principal amount of the Loans made to it from time to time outstanding from the Closing Date until payment in full thereof at the rates per annum, and on the dates, set forth in Section 2.11.
- (b) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing indebtedness of the Borrower to such Lender resulting from each Loan of such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time under this Agreement.
- (c) The Administrative Agent, on behalf of the Borrower, shall maintain the Register pursuant to Section 10.7(c), and a subaccount therein for each Lender, in which shall be recorded (i) the amount of each Loan to the Borrower made hereunder and any Note evidencing such Loan, the Type of such Loan and each Interest Period applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder and (iii) both the amount of any sum received by the Administrative Agent hereunder from or for the account of the Borrower and each Lender's share thereof. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of the Administrative Agent shall control in the absence of manifest error.

30

- (d) The entries made in the Register and the accounts of each Lender maintained pursuant to <u>Section 2.5(b)</u> shall, to the extent permitted by applicable Law, be <u>prima facie</u> evidence of the existence and amounts of the obligations of the Borrower therein recorded; <u>provided</u>, <u>however</u>, that the failure of any Lender or the Administrative Agent to maintain the Register or any such account, or any error therein, shall not in any manner affect the obligation of the Borrower to repay (with applicable interest) the Loans made to it by such Lender in accordance with the terms of this Agreement.
- (e) The Borrower agrees that, upon the request to the Administrative Agent by any Lender, it will execute and deliver to such Lender a promissory note of the Borrower evidencing any Revolving Credit Loans or Swing Line Loans, as the case may be, made by such Lender to the Borrower, substantially in the forms of Exhibit C-1 or C-2, respectively, with appropriate insertions as to date and principal amount. Each Lender may attach schedules to its Note and endorse thereon the date, Type (if applicable), amount and maturity of its Loans and payments with respect thereto.
- (f) In addition to the accounts and records referred to herein above, each Lender and the Administrative Agent shall maintain in accordance with its usual practice accounts or records evidencing the purchases and sales by such Lender of participations in Letters of Credit and Swing Line Loans. In the event of any conflict between the accounts and records maintained by the Administrative Agent and the accounts and records of any Lender in respect of such matters, the accounts and records of the Administrative Agent shall control in the absence of manifest error.
- 2.6. Facility Fee, etc. (a) The Borrower agrees to pay to the Administrative Agent for the account of each Lender in accordance with its applicable Revolving Credit Percentage a facility fee, computed at the Facility Fee Rate on the average daily amount of the Revolving Credit Commitment (or, if the Revolving Credit Commitments have been terminated, on the outstanding amount of all Revolving Credit Loans plus such Lender's applicable Revolving Credit Percentage of Swing Line Loans and L/C Obligations then outstanding), of such Lender during the period for which payment is made, regardless of usage, subject to adjustment as provided in Section 2.24. The facility fee shall accrue at all times during the Revolving Credit Commitment Period (and thereafter so long as any Revolving Credit Loans, Swing Line Loans or L/C Obligations remain outstanding), including at any time during which one or more of the conditions in Section 4.2 is not met, and shall be payable quarterly in arrears on the last Business Day of each of March, June, September and December and on the last day of the Revolving Credit Commitment Period (and, if applicable, thereafter on demand), commencing on the first of such dates to occur after the Closing Date. The facility fee shall be calculated quarterly in arrears, and if there is any change in the Facility Fee Rate during any quarter, the actual daily amount shall be computed and multiplied by the Facility Fee Rate separately for each period during such quarter that the Facility Fee Rate was in effect.
- (b) The Borrower agrees to pay to Barclays Bank PLC for the account of Barclays Capital the fees in the amounts and on the dates from time to time agreed to in the Barclays Fee Letter.

- (c) The Borrower agrees to pay to the Administrative Agent and Merrill Lynch, Pierce, Fenner & Smith Incorporated the fees in the amounts and on the dates from time to time agreed to in the Agent Fee Letter.
- 2.7. Termination or Reduction of Revolving Credit Commitments. The Borrower shall have the right, upon notice to the Administrative Agent, to terminate the Revolving Credit Commitments or, from time to time, to reduce the aggregate amount of the Revolving Credit Commitments; provided that (a) no such termination or reduction of Revolving Credit Commitments shall be permitted if, after giving effect thereto and to any prepayments of the Revolving Credit Loans and Swing Line Loans made on the effective date thereof, the Total Revolving Extensions of Credit would exceed the Total Revolving Credit Commitments, (b) any such reduction shall be in an amount equal to \$1,000,000, or a whole multiple thereof (or the remaining amount of the Revolving Credit Commitments), (c) any such notice shall be received by the Administrative Agent not later than 11:00 A.M. three Business Days prior to the date of termination or reduction and (d) if, after giving effect to any reduction of the Revolving Credit Commitments, the L/C Commitment or the Swing Line Commitment exceeds the amount of the Revolving Credit Commitment shall be automatically reduced by the amount of such excess; provided, further, that a notice of termination of the Revolving Credit Commitments delivered by the Borrower may state that such notice is conditioned upon the effectiveness of other credit facilities, transactions or borrowings in general, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified effective date) if such condition is not satisfied. The Administrative Agent will promptly notify the Lenders of any notice of termination or reduction of the Revolving Credit Commitments shall be applied to the Revolving Credit Commitment of each Lender according to its Revolving Credit Percentage. All fees accrued until the effective date of any termination of the Revolving Credit Commitment shall be paid on the effective date of such termination. Any reduction shall reduce permanently the Revo
- 2.8. Prepayments. (a) The Borrower may at any time and from time to time prepay the Loans made to the Borrower, in whole or in part, without premium or penalty, upon notice delivered to the Administrative Agent at least three Business Days prior thereto in the case of Eurodollar Loans and on the date of prepayment in the case of Base Rate Loans, which notice shall specify the date and amount of prepayment and whether the prepayment is of Eurodollar Loans or Base Rate Loans; provided, that (i) if a Eurodollar Loan is prepaid on any day other than the last day of the Interest Period applicable thereto, the Borrower shall also pay any amounts owing pursuant to Section 2.17 and (ii) no prior notice is required for the prepayment of Swing Line Loans; provided, further, that, if a notice of prepayment is given in connection with a conditional notice of termination of the Revolving Credit Commitments as contemplated by Section 2.7, then such notice of prepayment may be revoked if such notice of termination is revoked in accordance with Section 2.7. Upon receipt of any such notice the Administrative Agent shall promptly notify the Lenders thereof. If any such notice is given, the amount specified in such notice shall be due and payable on the date specified therein, together with (except in the case of Base Rate Loans) accrued interest to such date on the amount prepaid. Partial prepayments of Revolving Credit Loans shall be in an aggregate principal amount of \$1,000,000 or a whole multiple thereof. Partial prepayments of Swing Line Loans shall be in an aggregate principal amount of \$1,000,000 or a whole multiple thereof.

- (b) If for any reason the Total Revolving Extensions of Credit at any time exceed the Total Revolving Credit Commitments then in effect, the Borrower shall immediately prepay the Loans and/or Cash Collateralize the L/C Obligations in an aggregate amount equal to such excess; provided, however, that the Borrower shall not be required to Cash Collateralize the L/C Obligations pursuant to this Section 2.8(b) unless after the prepayment in full of the Loans the Total Revolving Extensions of Credit exceed the Total Revolving Credit Commitments then in effect.
- 2.9. Conversion and Continuation Options. (a) The Borrower may elect from time to time to convert Eurodollar Loans made to the Borrower to Base Rate Loans by giving the Administrative Agent at least two Business Days' prior irrevocable notice (which may be telephonic) of such election. The Borrower may elect from time to time to convert Base Rate Loans made to the Borrower to Eurodollar Loans by giving the Administrative Agent at least three Business Days' prior irrevocable notice (which may be telephonic) of such election (which notice shall specify the length of the initial Interest Period therefor); provided, however, that if the Borrower wishes to request Eurodollar Loans having an Interest Period of nine or twelve months in duration as provided in the definition of "Interest Period", the applicable notice must be received by the Administrative Agent not later than 11:00 A.M. four Business Days prior to the requested date of such conversion or continuation, whereupon the Administrative Agent shall give prompt notice to the Lenders of such request and determine whether the requested Interest Period is unavailable to any of them. Not later than 10:00 A.M. three Business Days before the requested date of such conversion or continuation, the Administrative Agent shall notify the Borrower (which notice may be by telephone) whether or not the requested Interest Period is unavailable to any of the Lenders, provided, further that no Base Rate Loan may be converted to a Eurodollar Loan (i) when any Event of Default has occurred and is continuing and the Administrative Agent or the Majority Lenders have determined in its or their sole discretion not to permit such conversions or (ii) during the one month period immediately prior to the Revolving Credit Termination Date. Each telephonic notice by the Borrower pursuant to this Section 2.9 must be confirmed promptly by delivery to the Administrative Agent of a written Borrowing Request appropriately completed and signed by a Responsible Officer of the Borrower. If the Borrower requests a conversion to a Eurodollar Loan in any Borrowing Request, but fails to specify an Interest Period, it will be deemed to have specified an Interest Period of one month. Upon receipt of any such notice the Administrative Agent shall promptly notify the Lenders thereof.
- (b) The Borrower may elect to continue any Eurodollar Loan made to the Borrower as such upon the expiration of the then current Interest Period with respect thereto by giving irrevocable notice (which may be telephonic) to the Administrative Agent, in accordance with the applicable provisions of the term "Interest Period" set forth in Section 1.1, of the length of the next Interest Period to be applicable to such Loans, provided that no Eurodollar Loan may be continued as such (i) when any Event of Default has occurred and is continuing and the Administrative Agent or the Majority Lenders have determined in its or their sole discretion not to permit such continuations or (ii) during the one month period immediately prior to the Revolving Credit Termination Date, and provided, further, that if the Borrower shall fail to give any required notice as described above in this paragraph or if such continuation is not permitted pursuant to the preceding proviso, such Loans shall be converted automatically to Base Rate Loans on the last day of such then expiring Interest Period. Each telephonic notice by the

33

Borrower pursuant to this <u>Section 2.9</u> must be confirmed promptly by delivery to the Administrative Agent of a written Borrowing Request appropriately completed and signed by a Responsible Officer of the Borrower. Upon receipt of any such notice the Administrative Agent shall promptly notify the Lenders thereof.

elections so that no more than ten Eurodollar Loans shall be outstanding at any one time.

- 2.11. <u>Interest Rates and Payment Dates</u>. (a) Subject to the provisions of <u>paragraph (c)</u> below, each Eurodollar Loan shall bear interest on the outstanding principal amount thereof for each day during each Interest Period with respect thereto at a rate per annum equal to the Eurodollar Rate determined for such day plus the Applicable Margin for Eurodollar Loans.
- (b) Each Base Rate Loan, including Swing Line Loans, shall bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Base Rate plus the Applicable Margin for Base Rate Loans.
- (c) (i) If all or a portion of the principal amount of any Loan or Reimbursement Obligation shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue amount shall bear interest at a rate per annum that is equal to (x) in the case of the Loans, the rate that would otherwise be applicable thereto pursuant to the foregoing provisions of this Section 2.11 plus 2% or (y) in the case of Reimbursement Obligations, the rate applicable to Base Rate Loans plus 2%, and (ii) if all or a portion of any interest payable on any Loan or Reimbursement Obligation or any facility fee or other amount payable hereunder shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue amount shall bear interest at a rate per annum equal to the rate then applicable to Base Rate Loans plus 2%, in each case, with respect to clauses (i) and (ii) above, from the date of such non-payment until such amount is paid in full (each of the foregoing collectively, the "Default Rate").
- (d) Interest shall be payable in arrears on each Interest Payment Date, <u>provided</u> that interest accruing pursuant to <u>paragraph (c)</u> of this <u>Section 2.11</u> shall be payable from time to time on demand (after as well as before judgment and before and after the commencement of any proceeding under any Debtor Relief Law).
- 2.12. Computation of Interest and Fees; Retroactive Adjustments of Applicable Margin and Facility Fee Rate. (a) Interest (including interest on Base Rate Loans determined by reference to the Eurodollar Rate), fees and commissions payable pursuant hereto shall be calculated on the basis of a 365-day (or 366-day, as the case may be) year for the actual days elapsed, except that, with respect to Eurodollar Loans, the interest thereon shall be calculated on the basis of a 360-day year for the actual days elapsed. The Administrative Agent shall as soon as practicable notify the Borrower and the Lenders of each determination of a Eurodollar Rate. Any change in the interest rate on a Loan resulting from a change in the Base Rate shall become

34

effective as of the opening of business on the day on which such change becomes effective. The Administrative Agent shall as soon as practicable notify the Borrower and the Lenders of the effective date and the amount of each such change in any interest rate. Interest shall accrue on each Loan for the day on which the Loan is made, and shall not accrue on a Loan, or any portion thereof, for the day on which the Loan or such portion is paid, <u>provided</u> that any Loan that is repaid on the same day on which it is made shall, subject to <u>Section 2.14(d)</u>, bear interest for one day.

- (b) If, as a result of any restatement of or other adjustment to the financial statements of the Borrower or for any other reason, the Borrower, the Administrative Agent or the Majority Lenders determine that (i) the Total Consolidated Debt to Total Consolidated Capitalization Ratio as calculated by the Borrower as of any applicable date was inaccurate and (ii) a proper calculation of the Total Consolidated Debt to Total Consolidated Capitalization Ratio would have resulted in higher pricing for such period, the Borrower shall immediately and retroactively be obligated to pay to the Administrative Agent for the account of the applicable Lenders or the applicable Issuing Lenders, as the case may be, promptly on demand by the Administrative Agent (or, after the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under any Debtor Relief Law, automatically and without further action by the Administrative Agent, any Lender or any Issuing Lender), an amount equal to the excess of the amount of interest and fees that should have been paid for such period over the amount of interest and fees actually paid for such period. This paragraph shall not limit the rights of the Administrative Agent, any Lender or any Issuing Lender under this Agreement. The Borrower's obligations under this paragraph shall survive the termination of the Revolving Credit Commitments and the repayment of all Loans and other obligations hereunder.
- (c) Each determination of an interest rate by the Administrative Agent pursuant to any provision of this Agreement shall be conclusive and binding on the Borrower and the Lenders in the absence of manifest error.
  - 2.13. <u>Inability to Determine Interest Rate</u>. If prior to the first day of any Interest Period:
  - (a) the Administrative Agent shall have determined (which determination shall be conclusive and binding upon the Borrower) that, by reason of circumstances affecting the relevant market, adequate and reasonable means do not exist for ascertaining the Eurodollar Rate for such Interest Period or in connection with an existing or proposed Base Rate Loan, or
  - (b) the Administrative Agent shall have received notice from the Majority Lenders that the Eurodollar Rate determined or to be determined for such Interest Period will not adequately and fairly reflect the cost to such Lenders (as conclusively certified by such Lenders) of making or maintaining their affected Loans during such Interest Period,

the Administrative Agent shall give telecopy or telephonic notice thereof to the Borrower and the relevant Lenders as soon as practicable thereafter. If such notice is given (x) any Eurodollar Loans requested to be made on the first day of such Interest Period shall be made as Base Rate

35

Loans, (y) any Loans that were to have been converted on the first day of such Interest Period to Eurodollar Loans shall be continued as Base Rate Loans and (z) any outstanding Eurodollar Loans shall be converted, on the last day of the then current Interest Period with respect thereto, to Base Rate Loans. Until such notice has been withdrawn by the Administrative Agent (x) no further Eurodollar Loans shall be made or continued as such, (y) the Borrower shall not have the right to convert Loans to Eurodollar Loans and (z) the utilization of the Eurodollar Rate component in determining the Base Rate shall be suspended.

2.14. <u>Pro Rata Treatment and Payments</u>. (a) Each borrowing, other than borrowings of Swing Line Loans, by the Borrower from the Lenders hereunder, each payment by the Borrower on account of any facility fee or Letter of Credit fee, and any reduction of the Revolving Credit Commitments of the Lenders, shall be made pro rata according to the respective Revolving Credit Percentages of the Lenders.

- (b) Each payment (including each prepayment) by the Borrower on account of principal of and interest on the Revolving Credit Loans of the Borrower shall be made <u>pro rata</u> according to the respective outstanding principal amounts of the Revolving Credit Loans of the Borrower then held by the Lenders. Each payment in respect of Reimbursement Obligations in respect of any Letter of Credit shall be made to the relevant Issuing Lender.
- (c) The application of any payment of Loans shall be made, <u>first</u>, to Base Rate Loans and, <u>second</u>, to Eurodollar Loans. Each payment of the Eurodollar Loans shall be accompanied by accrued interest to the date of such payment on the amount paid.
- (d) All payments (including prepayments) to be made by the Borrower hereunder, whether on account of principal, interest, fees or otherwise, shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff and shall be made prior to 12:00 Noon on the due date thereof to the Administrative Agent, for the account of the relevant Lenders, at the Administrative Agent's Office, in Dollars and in immediately available funds. Any payment made by the Borrower after 12:00 Noon on any Business Day shall be deemed to have been made on the next following Business Day. The Administrative Agent shall distribute such payments to the Lenders promptly upon receipt in like funds as received. If any payment hereunder (other than payments on the Eurodollar Loans) becomes due and payable on a day other than a Business Day, such payment shall be extended to the next succeeding Business Day unless the result of such extension would be to extend such payment into another calendar month, in which event such payment shall be made on the immediately preceding Business Day. In the case of any extension of any payment of principal pursuant to the preceding two sentences, interest thereon shall be payable at the then applicable rate during such extension.
- (e) Unless the Borrower or any Lender has notified the Administrative Agent, prior to the time any payment is required to be made by it to the Administrative Agent hereunder, that the Borrower or such Lender, as the case may be, will not make such payment, the Administrative Agent may assume that the Borrower or such Lender, as the case may be, has timely made such payment and may (but shall not be so required to), in reliance thereon, make

available a corresponding amount to the Person entitled thereto. If and to the extent that such payment was not in fact made to the Administrative Agent in immediately available funds, then:

- (i) if the Borrower failed to make such payment, each Lender shall forthwith on demand repay to the Administrative Agent the portion of such assumed payment that was made available to such Lender in immediately available funds, together with interest thereon in respect of each day from and including the date such amount was made available by the Administrative Agent to such Lender to the date such amount is repaid to the Administrative Agent in immediately available funds at the Federal Funds Rate from time to time in effect; and
- (ii) if any Lender failed to make such payment, such Lender shall forthwith on demand pay to the Administrative Agent the amount thereof in immediately available funds, together with interest thereon for the period from the date such amount was made available by the Administrative Agent to the Borrower to the date such amount is recovered by the Administrative Agent (the "Compensation Period") at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation, plus any administrative, processing or similar fees customarily charged by the Administrative Agent in connection with the foregoing. If such Lender pays such amount to the Administrative Agent, then such amount shall constitute such Lender's Revolving Credit Percentage of the Loan included in the applicable borrowing. If such Lender does not pay such amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent may make a demand therefor upon the Borrower, and the Borrower shall pay such amount to the Administrative Agent, together with interest thereon for the Compensation Period at a rate per annum equal to the rate of interest applicable to the applicable borrowing. Nothing herein shall be deemed to relieve any Lender from its obligation to fulfill its Revolving Credit Commitment or to prejudice any rights which the Administrative Agent or the Borrower may have against any Lender as a result of any default by such Lender hereunder.

A notice of the Administrative Agent to any Lender or the Borrower with respect to any amount owing under this <u>subsection (e)</u> shall be conclusive, absent manifest error.

- (f) The obligations of the Lenders hereunder to make Loans, to fund participations in Letters of Credit and Swing Line Loans and to make payments under Section 10.6 are several and not joint. The failure of any Lender to make any Loan, to fund any such participation or to make any payment under Section 10.6 on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its Loan, to purchase its participation or make its payment under Section 10.6.
- 2.15. <u>Requirements of Law</u>. (a) If the adoption of or any change in any Requirement of Law or in the interpretation or application thereof or compliance by any Lender

37

with any request or directive (whether or not having the force of law) from any central bank or other Governmental Authority made subsequent to the Closing Date:

- (i) shall subject any Lender to any tax of any kind whatsoever with respect to this Agreement, any Letter of Credit, any Application or any Eurodollar Loan made by it, or change the basis of taxation of payments to such Lender in respect thereof (except for Non-Excluded Taxes or Other Taxes covered by Section 2.16 and the imposition of, or any change in, the rate of any Excluded Tax payable by such Lender);
- (ii) shall impose, modify or hold applicable any reserve, special deposit, compulsory loan or similar requirement against assets held by, deposits or other liabilities in or for the account of, advances, loans or other extensions of credit by, or any other acquisition of funds by, any office of such Lender that is not otherwise included in the determination of the Eurodollar Rate hereunder; or
  - (iii) shall impose on such Lender any other condition;

and the result of any of the foregoing is to increase the cost to such Lender, by an amount which such Lender deems to be material, of making, converting into, continuing or maintaining any Eurodollar Loan to the Borrower or issuing or participating in Letters of Credit issued at the request of the Borrower, or to reduce any amount receivable hereunder in respect thereof, then, in any such case, the Borrower shall promptly pay such Lender, upon its demand, any additional amounts necessary to compensate such Lender for such increased cost or reduced amount receivable. If any Lender becomes entitled to claim any additional amounts pursuant to this Section 2.15, it shall promptly notify the Borrower (with a copy to the Administrative Agent) of the event by reason of which it has become so entitled.

- (b) If any Lender shall have determined that the adoption of or any change in any Requirement of Law regarding capital adequacy or in the interpretation or application thereof or compliance by such Lender or any corporation controlling such Lender with any request or directive regarding capital adequacy (whether or not having the force of law) from any Governmental Authority made subsequent to the Closing Date shall have the effect of reducing the rate of return on such Lender's or such corporation's capital as a consequence of its obligations hereunder or under or in respect of any Letter of Credit to a level below that which such Lender or such corporation could have achieved but for such adoption, change or compliance (taking into consideration such Lender's or such corporation's policies with respect to capital adequacy) by an amount deemed by such Lender to be material, then from time to time, after submission by such Lender to the Borrower (with a copy to the Administrative Agent) of a written request therefor, the Borrower shall pay to such Lender such additional amount or amounts as will compensate such Lender or such corporation for such reduction.
- (c) In addition to, and without duplication of, amounts which may become payable from time to time pursuant to <u>paragraphs (a)</u> and (b) of this <u>Section 2.15</u>, the Borrower agrees to pay to each Lender which requests compensation under this <u>paragraph (c)</u> by notice to the Borrower, on the last day of each Interest Period with respect to any Eurodollar Loan made by such Lender to the Borrower, at any time when such Lender shall be required to maintain

38

reserves against "Eurocurrency liabilities" under Regulation D of the Board of Governors of the Federal Reserve System (or, at any time when such Lender may be required by the Board of Governors of the Federal Reserve System or by any other Governmental Authority, whether within the United States or in another relevant jurisdiction, to maintain reserves against any other category of liabilities which includes deposits by reference to which the Eurodollar Rate is determined as provided in this Agreement or against any category of extensions of credit or other assets of such Lender which includes any such Eurodollar Loans), an additional amount (determined by such Lender's calculation or, if an accurate calculation is impracticable, reasonable estimate using such reasonable means of allocation as such Lender shall determine) equal to the actual costs, if any, incurred by such Lender during such Interest Period as a result of the applicability of the foregoing reserves to such Eurodollar Loans.

(d) A certificate as to any additional amounts payable pursuant to this Section 2.15 submitted by any Lender to the Borrower (with a copy to the Administrative Agent) shall be conclusive in the absence of manifest error. No Lender shall be entitled to compensation under this Section 2.15 from the Borrower for any costs incurred or reductions suffered more than 180 days prior to the date that such Lender notifies the Borrower of the circumstances giving rise to such increased costs or reductions and of such Lender's intention to claim compensation therefor; provided that if a change of law giving rise to such increased costs or reductions is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof. The obligations of the Borrower pursuant to this Section 2.15 shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

For purposes this Agreement, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, regulations, guidelines, interpretations or directives thereunder or issued in connection therewith (whether or not having the force of Law) and (y) all requests, rules, regulations, guidelines, interpretations or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities (whether or not having the force of Law), in each case pursuant to Basel III, shall, in the case of the foregoing clauses (x) and (y), be deemed to have been adopted after the Closing Date, regardless of the date enacted or adopted.

2.16. Taxes. (a) Except as required by Law, all payments made by the Borrower under this Agreement shall be made free and clear of, and without deduction or withholding for or on account of, any present or future income, stamp or other taxes, levies, imposts, duties, charges, fees, deductions or withholdings, now or hereafter imposed, levied, collected, withheld or assessed by any Governmental Authority, excluding net income taxes and franchise and doing business taxes (imposed in lieu of net income taxes) imposed on the Administrative Agent or any Lender as a result of a present or former connection between the Administrative Agent or such Lender and the jurisdiction of the Governmental Authority imposing such tax or any political subdivision or taxing authority thereof or therein (other than any such connection arising solely from the Administrative Agent's or such Lender's having executed, delivered or performed its obligations or received a payment under, or enforced, this Agreement or any other Loan Document) (such net income taxes and franchise or doing business taxes imposed in lieu of net income taxes being referred to hereinafter as "Excluded Taxes"). If any such taxes, levies,

39

imposts, duties, charges, fees, deductions or withholdings other than Excluded Taxes ("Non-Excluded Taxes") or any Other Taxes are required to be withheld from any amounts payable to the Administrative Agent or any Lender hereunder, the amounts so payable to the Administrative Agent or such Lender shall be increased to the extent necessary to yield to the Administrative Agent or such Lender (after payment of all Non-Excluded Taxes and Other Taxes) interest or any such other amounts payable hereunder at the rates or in the amounts specified in this Agreement; provided, however, that the Borrower shall not be required to increase any such amounts payable to any Lender with respect to any Non-Excluded Taxes (i) that are attributable to such Lender's failure to comply with the requirements of paragraph (d) or (e) of this Section 2.16 or (ii) that are withholding taxes imposed on amounts payable to such Lender at the time such Lender becomes a party to this Agreement or designates a new lending office, except to the extent that such Lender (or its assignor, if any) was entitled, at the time of designation of a new lending office or assignment, to receive additional amounts from the Borrower with respect to such Non-Excluded Taxes pursuant to this Section 2.16(a). For the avoidance of doubt, all references to taxes shall be deemed to include any interest, additions to tax or penalties applicable thereto.

- (c) Whenever any Non-Excluded Taxes or Other Taxes are paid or payable by the Borrower, as soon as practicable thereafter the Borrower shall send to the Administrative Agent for its own account or for the account of the relevant Lender, as the case may be, a certified copy of an official receipt received by the Borrower showing payment thereof (or other evidence of such payment reasonably satisfactory to the Administrative Agent). If the Borrower fails to pay any Non-Excluded Taxes or Other Taxes when due to the appropriate taxing authority, the Borrower shall indemnify the Administrative Agent and the Lenders for any incremental taxes, interest or penalties that may become payable by the Administrative Agent or any Lender as a result of any such failure. The agreements in this Section 2.16 shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.
- (d) Each Lender (or Transferee) that is not (i) a citizen or resident of the United States of America, (ii) a corporation, partnership or other entity created or organized in or under the laws of the United States of America (or any jurisdiction thereof), or (iii) an estate or trust that is subject to U.S. federal income taxation regardless of the source of its income (a "Non-U.S. Lender") that may lawfully do so shall deliver to the Borrower and the Administrative Agent (or, in the case of a Participant, to the Lender from which the related participation shall have been purchased) two copies of either U.S. Internal Revenue Service Form W-8BEN or Form W-8ECI (or other applicable form), or, in the case of a Non-U.S. Lender claiming exemption from U.S. federal withholding tax under Section 871(h) or 881(c) of the Code with respect to payments of "portfolio interest", a statement substantially in the form of Exhibit D and a Form W-8BEN (or other applicable form), or to the extent such Lender may lawfully do so, it shall deliver any subsequent versions thereof or successors thereto properly completed and duly executed by such Non-U.S. Lender claiming complete exemption from, or a reduced rate of, U.S. federal withholding tax on all payments by the Borrower under this Agreement and the other Loan Documents. Such forms shall be delivered by each Non-U.S. Lender on or before the date

it becomes a party to this Agreement (or, in the case of any Participant, on or before the date such Participant purchases the related participation). In addition, to the extent it may lawfully do so, each Non-U.S. Lender shall deliver such forms promptly upon the obsolescence or invalidity of any form previously delivered by such Non-U.S. Lender. Each Non-U.S. Lender shall, as soon as reasonably practicable, notify the Borrower at any time it determines that it is no longer in a position to provide any previously delivered certificate to the Borrower (or any other form of certification adopted by the U.S. taxing authorities for such purpose). Notwithstanding any other provision of this paragraph, a Non-U.S. Lender shall not be required to deliver any form pursuant to this paragraph that such Non-U.S. Lender is not legally able to deliver.

- (e) A Lender that is entitled to an exemption from or reduction of non-U.S. withholding tax under the Law of the jurisdiction in which the Borrower is located, or any treaty to which such jurisdiction is a party, with respect to payments under this Agreement shall deliver to the Borrower (with a copy to the Administrative Agent), at the time or times prescribed by applicable Law or reasonably requested by the Borrower, such properly completed and executed documentation prescribed by applicable Law or reasonably requested by the Borrower as will permit such payments to be made without withholding or at a reduced rate, <u>provided</u> that such Lender is legally entitled to complete, execute and deliver such documentation. If a payment made to a Lender or any assignee of any Lender under this Agreement would be subject to United States federal withholding tax imposed by Sections 1471 through 1474 of the Code and any regulations or interpretations thereof ("<u>FATCA</u>") if such Lender or assignee were to fail to comply with the applicable reporting requirements of FATCA, such Lender shall deliver to the Borrower and the Administrative Agent, at the time or times prescribed by law and at such time or times reasonably requested by the Borrower and the Administrative Agent, such documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C) (i) of the Code) and such additional documentation reasonably requested by the Borrower and the Administrative Agent to comply with its obligations under FATCA, to determine that such Lender or assignee has or has not complied with its obligations under FATCA or to determine the amount to deduct and withhold from such payment.
- (f) Unless required by applicable Laws, at no time shall the Administrative Agent have any obligation to file for or otherwise pursue on behalf of a Lender or an Issuing Lender, or have any obligation to pay to any Lender or any Issuing Lender, any refund of taxes withheld or deducted from funds paid for the account of such Lender or such Issuing Lender, as the case may be. If the Administrative Agent, any Lender or any Issuing Lender determines, in its sole discretion, that it has received a refund of any Non-Excluded Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to this Section, it shall pay to the Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section with respect to the Non-Excluded Taxes giving rise to such refund), net of all out-of-pocket expenses incurred by the Administrative Agent, such Lender or such Issuing Lender, as the case may be, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), provided that the Borrower, upon the request of the Administrative Agent, such Lender or such Issuing Lender, agrees to repay the amount paid over to the Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Administrative Agent, such Lender or such Issuing

41

Lender in the event the Administrative Agent, such Lender or such Issuing Lender is required to repay such refund to such Governmental Authority. This subsection shall not be construed to require the Administrative Agent, any Lender or any Issuing Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to the Borrower or any other Person.

2.17. Compensation for Losses. The Borrower agrees to, upon demand of any Lender (with a copy to the Administrative Agent) from time to time, to indemnify each Lender for, and to hold each Lender harmless from, any loss or expense that such Lender sustains or incurs as a consequence of (a) default by the Borrower in making a borrowing of, conversion to or continuation of Eurodollar Loans after the Borrower has given a notice requesting the same in accordance with the provisions of this Agreement, (b) default by the Borrower in making any prepayment after the Borrower has given a notice thereof in accordance with the provisions of this Agreement or (c) the making by the Borrower of a prepayment or conversion of Eurodollar Loans on a day that is not the last day of an Interest Period with respect thereto; provided that any request for indemnification made by a Lender pursuant to this Section 2.17 shall be made within six months of the incurrence of the loss or expense requested to be indemnified. Such indemnification may include an amount equal to the excess, if any, of (i) the amount of interest that would have accrued on the amount so prepaid, or not so borrowed, converted or continued, for the period from the date of such prepayment or of such failure to borrow, convert or continue to the last day of such Interest Period (or, in the case of a failure to borrow, convert or continue, the Interest Period that would have commenced on the date of such failure) in each case at the applicable rate of interest for such Loans provided for herein (excluding, however, the Applicable Margin included therein, if any) over (ii) the amount of interest (as reasonably determined by such Lender) that would have accrued to such Lender on such amount by placing such amount on deposit for a comparable period with leading banks in the interbank Eurodollar market. A certificate as to any amounts payable pursuant to this Section 2.17 submitted to the Borrower by any Lender shall be

conclusive in the absence of manifest error. This covenant shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

2.18. Illegality. Notwithstanding any other provision herein, if the adoption of or any change in any Requirement of Law or in the interpretation or application thereof shall make it unlawful for any Lender to make or maintain Loans whose interest is determined by reference to the Eurodollar Rate as contemplated by this Agreement, (a) the commitment of such Lender hereunder to make or maintain such Loans and convert Base Rate Loans to Eurodollar Loans shall forthwith be canceled and (b) such Lender's Loans then outstanding as Eurodollar Loans, if any, shall be converted automatically to Base Rate Loans (without reference to the Eurodollar Rate component thereof) on the respective last days of the then current Interest Periods with respect to such Loans or within such earlier period as required by Law and all Base Rate Loans of such Lender whose interest is determined by reference to the Eurodollar Rate shall forthwith be determined without reference to the Eurodollar Rate component of the definition of "Base Rate." If any such conversion of a Eurodollar Loan to a Base Rate Loan occurs on a day which is not the last day of the then current Interest Period with respect thereto, the Borrower shall pay to such Lender such amounts, if any, as may be required pursuant to Section 2.17.

42

- 2.19. Change of Office. Each Lender agrees that, upon the occurrence of any event that it knows to give rise to the operation of Sections 2.15, 2.16(a) or 2.18 with respect to such Lender, it will use all commercially reasonable efforts (subject to overall policy considerations of such Lender) to designate another lending office for any Loans affected by such event, or to assign its rights and obligations hereunder with respect to such Loans to another of its offices, branches or affiliates with the object of avoiding the consequences of such event; provided, that such designation is made on terms that, in the reasonable sole judgment of such Lender, cause such Lender and its lending office(s) to suffer no economic, legal or regulatory disadvantage, and provided, further, that nothing in this Section 2.19 shall affect or postpone any of the obligations of the Borrower or the rights of any Lender pursuant to Sections 2.15, 2.16(a) or 2.18. The Borrower hereby agrees to pay all reasonable out-of-pocket costs and expenses incurred by any Lender in connection with any such designation or assignment.
- Replacement of Lenders under Certain Circumstances. The Borrower may, at its sole expense and effort, upon notice to the applicable Lender and the Administrative Agent, require any Lender (a) that requests reimbursement for amounts owing pursuant to Section 2.15. (b) with respect to which the Borrower is required to pay any amounts under Sections 2.16 or 2.18, (c) that is a Defaulting Lender or (d) that fails to approve any amendment which, pursuant to Section 10.1, requires the approval of each Lender or each affected Lender, provided, that such amendment is approved by at least the Majority Lenders, to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 10.7), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), provided that (i) such assignment does not conflict with any Requirement of Law, (ii) with respect to a condition described in clause (a) or (b) above, such assignment will result in a reduction in such compensation or payments thereafter, (iii) the assignee shall purchase, at par, all Loans and other amounts owing to such replaced Lender on or prior to the date of such assignment, (iv) the Borrower shall be liable to such replaced Lender under Section 2.17 (as though Section 2.17 were applicable) if any Eurodollar Loan to the Borrower owing to such replaced Lender shall be purchased other than on the last day of the Interest Period relating thereto, (v) the Borrower shall have paid to the Administrative Agent the assignment fee specified in Section 10.7(b), (vii) the Borrower shall pay all additional amounts (if any) required pursuant to Section 2.15, 2.16 or 2.18, as the case may be, in respect of any period prior to the date on which such assignment shall be consummated, and (viii) any such assignment shall not be deemed to be a waiver of any rights that the Borrower, the Administrative Agent or any other Lender shall have against the replaced Lender. A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.
- 2.21. <u>Guaranty of Payment and Performance</u>. Subject to <u>Section 10.17</u>, the payment and performance of the obligations of the Borrower hereunder shall be guaranteed by each Guarantor pursuant to the terms of the Guaranty.

43

## 2.22. <u>Increase in Commitments</u>.

- (a) Request for Increase. Upon notice to the Administrative Agent (which shall promptly notify the Lenders), the Borrower may from time to time, increase the Total Revolving Credit Commitments by an amount not to exceed (x) \$75,000,000 minus (z) the aggregate amount of all prior increases of the Total Revolving Credit Commitment pursuant to this Section 2.22. Such increase in the Total Revolving Credit Commitments may be effectuated by obtaining additional Revolving Credit Commitments from existing Lenders or new Revolving Credit Commitments from Eligible Assignees designated by the Borrower to become Lenders (pursuant to an instrument of accession in the form of Exhibit I hereto, an "Instrument of Accession"); provided that (i) any such increase shall be in a minimum amount of \$5,000,000, (ii) the aggregate amount of the Total Revolving Credit Commitments after giving effect to any such increase shall not at any time exceed \$450,000,000 and (iii) that any such new Lender so designated by the Borrower shall be reasonably acceptable to the Administrative Agent. Nothing contained herein shall constitute, or otherwise be deemed to be, a commitment on the part of any Lender to increase its Revolving Credit Commitment hereunder.
- (b) <u>Effective Date and Allocations</u>. If the Total Revolving Credit Commitments are increased in accordance with this <u>Section 2.22</u>, the Administrative Agent and the Borrower shall determine the effective date (the "<u>Increase Effective Date</u>") and the Borrower, in consultation with the Administrative Agent, shall determine the final allocation of such increase. The Administrative Agent shall promptly notify the Lenders of the final allocation of such increase and the Increase Effective Date.
- (c) <u>Conditions to Effectiveness of Increase</u>. As a condition precedent to such increase, (i) no Default or Event of Default shall exist, (ii) the Borrower shall (x) deliver to the Administrative Agent (1) an Instrument of Accession executed by the Borrower and the applicable Lender(s), and (2) a certificate dated as of the Increase Effective Date (in sufficient copies for each Lender) signed by a Responsible Officer of the Borrower (A) certifying and attaching the resolutions adopted by the Borrower approving or consenting to such increase, and (B) certifying that, before and after giving effect to such increase no Default exists and (iii) pursuant to the terms of the Agent Fee Letter, pay the fees to the applicable Persons. The Borrower shall prepay any Revolving Credit Loans outstanding on the Increase Effective Date (and pay any additional amounts required pursuant to Section 2.17) to the extent necessary to keep the outstanding Revolving Credit Loans ratable with any revised Revolving Credit Percentages arising from any nonratable increase in the Total Revolving Credit Commitments under this Section 2.22.

- (d) <u>Conflicting Provisions</u>. This <u>Section 2.22</u> shall supersede any provisions in <u>Section 2.14</u> or <u>10.1</u> to the contrary.
- 2.23. Cash Collateral.
- (a) <u>Certain Credit Support Events</u>. Upon the request of the Administrative Agent or any Issuing Lender, if (i) as of the day that is one month prior to the Revolving Credit Termination Date, Letters of Credit with expiry dates occurring after the Revolving Credit Termination Date remain outstanding, the Borrower shall immediately Cash Collateralize the then outstanding amount of all L/C Obligations in respect of such Letters of Credit and (ii) as of the Revolving Credit Termination Date, any L/C Obligations (which are not Cash Collateralized in accordance with clause (i) above or otherwise in accordance with the terms of this Agreement) for any reason remain outstanding, the Borrower shall immediately Cash Collateralize the

then outstanding amount of all such L/C Obligations. At any time that there shall exist a Lender that is a Defaulting Lender, promptly (and in any event within three (3) Business Days thereafter) upon the request of the Administrative Agent, any Issuing Lender or the Swing Line Lender, the Borrower shall deliver to the Administrative Agent Cash Collateral in an amount sufficient to cover all Fronting Exposure (after giving effect to Section 2.24(a)(iv) and any Cash Collateral provided by the Defaulting Lender).

- (b) Grant of Security Interest. All Cash Collateral (other than credit support not constituting funds subject to deposit) shall be maintained in blocked, non-interest bearing deposit accounts at Bank of America. The Borrower, and to the extent provided by any Lender, such Lender, hereby grants to (and subjects to the control of) the Administrative Agent, for the benefit of the Administrative Agent, the Issuing Lenders and the Lenders (including the Swing Line Lender), and agrees to maintain, a first priority security interest in all such cash, deposit accounts and all balances therein, and all other property so provided as collateral pursuant hereto, and in all proceeds of the foregoing, all as security for the obligations to which such Cash Collateral may be applied pursuant to Section 2.23(c). If at any time the Administrative Agent reasonably determines that Cash Collateral is subject to any priority right or claim of any Person other than the Administrative Agent as herein provided, or that the total amount of Cash Collateral provided in respect of any applicable Fronting Exposure and/or relevant L/C Obligations, in each case, for which Cash Collateral is required to be provided pursuant to the terms of this Agreement, is less than the amount of such Fronting Exposure and/or such L/C Obligations, as the case may be, the Borrower or, in the case of any applicable Fronting Exposure, the relevant Defaulting Lender will, promptly upon demand by the Administrative Agent (and in any event, within three (3) Business Days thereafter), pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency.
- (c) <u>Application</u>. Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under any of this <u>Section 2.23</u> or <u>Sections 2.4</u>, <u>2.8</u>, <u>2.24</u>, <u>3.1</u> or <u>8.2</u> in respect of Letters of Credit or Swing Line Loans shall be held and applied to the satisfaction of the specific L/C Obligations, Swing Line Loans, obligations to fund participations therein (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) and, in the case of any Defaulting Lender, other funding obligations, for which the Cash Collateral was so provided (or in the case of any Defaulting Lender, deemed provided).
- (d) Release. Cash Collateral (or the appropriate portion thereof) provided to reduce Fronting Exposure, to collateralize L/C Obligations or, in the case of Cash Collateral provided or deemed provided by any Defaulting Lender, to collateralize funding obligations of such Defaulting Lender shall, in each case, be released promptly following (i) the elimination of the applicable Fronting Exposure, L/C Obligations or such funding obligations giving rise thereto (including by the termination of Defaulting Lender status of the applicable Revolving Credit Lender (or, as appropriate, its assignee following compliance with Section 10.7(b)(vi))) or

45

(ii) the Administrative Agent's good faith determination that there exists excess Cash Collateral; <u>provided</u>, <u>however</u>, (x) that Cash Collateral furnished by or on behalf of the Borrower shall not be released during the continuance of a Default or Event of Default, and (y) the Person providing Cash Collateral and the applicable Issuing Lender or Swing Line Lender, as applicable, may agree that Cash Collateral shall not be released but instead held to support future anticipated Fronting Exposure or, in the case of Cash Collateral provided or deemed provided by any Defaulting Lender, funding obligations of such Defaulting Lender.

## 2.24. <u>Defaulting Lenders</u>.

- (a) <u>Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as that Lender is no longer a Defaulting Lender, to the extent permitted by applicable Law:
  - (i) <u>Waivers and Amendments</u>. That Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in <u>Section 10.1</u>.
  - (ii) Reallocation of Payments. Any payment of principal, interest, fees or other amounts received by the Administrative Agent under any Loan Document for the account of that Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 8.2 or otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to Section 10.8), shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by that Defaulting Lender to the Issuing Lenders or Swing Line Lender hereunder; third, if so reasonably determined by the Administrative Agent or requested by any Issuing Lender or Swing Line Lender, to be held as Cash Collateral for future funding obligations of that Defaulting Lender of any participation in any Swing Line Loan or Letter of Credit; fourth, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement; fifth, if so determined by the Administrative Agent and the Borrower, to be held as Cash Collateral in a non-interest bearing deposit account and released in order to satisfy obligations of that Defaulting Lender to fund Loans or provide Cash Collateral for Fronting Exposure under this Agreement; sixth, to the payment of any amounts owing to the Lenders, the Issuing Lenders or Swing Line Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender, any Issuing Lender or Swing Line Lender against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a

court of competent jurisdiction; <u>provided</u> that if (x) such payment is a payment of the principal amount of any Loans or L/C Borrowings in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) such Loans or L/C Borrowings were made at a time when the conditions set forth in <u>Section 4.2</u> were satisfied or waived, such payment shall be applied solely to pay the Loans of, and L/C Borrowings owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or L/C Borrowings owed to, that Defaulting Lender. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this <u>Section 2.24(a)(ii)</u> shall be deemed paid to and redirected by that Defaulting Lender, and each Lender irrevocably consents hereto.

- (iii) Certain Fees. That Defaulting Lender (x) shall be entitled to receive any facility fee pursuant to Section 2.06 for any period during which that Lender is a Defaulting Lender only to the extent allocable to the sum of (1) the outstanding amount of the Loans funded by it and (2) its Revolving Credit Percentage of the stated amount of Letters of Credit and Swing Line Loans for which it has provided Cash Collateral pursuant to Section 2.4, Section 2.23, Section 2.24(a)(ii) or Section 3.1, as applicable (and the Borrower shall (A) be required to pay to each of the Issuing Lenders and the Swing Line Lender, as applicable, the amount of such fee allocable to its Fronting Exposure arising from that Defaulting Lender and (B) not be required to pay the remaining amount of such fee that otherwise would have been required to have been paid to that Defaulting Lender) and (y) shall be limited in its right to receive letter of credit fees as provided in Section 3.8.
- (iv) Reallocation of Applicable Percentages to Reduce Fronting Exposure. During any period in which there is a Lender that is a Defaulting Lender, for purposes of computing the amount of the obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit or Swing Line Loans pursuant to Sections 2.4 and 3.3, the "Revolving Credit Percentage" of each non-Defaulting Lender shall be computed without giving effect to the Revolving Credit Commitment of that Defaulting Lender; provided, that, (i) each such reallocation shall be given effect only if, at the date such reallocation is made, no Default or Event of Default exists; and (ii) the aggregate obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit and Swing Line Loans shall not exceed the positive difference, if any, of (1) the Revolving Credit Commitment of that non-Defaulting Lender minus (2) the aggregate outstanding amount of Loans of that Lender.
- (b) <u>Defaulting Lender Cure</u>. If the Borrower, the Administrative Agent, and in the case of the Lender that is a Defaulting Lender, the Swing Line Lender and the Issuing Lenders, agree in writing in their sole discretion that a Defaulting Lender should no longer be deemed to be a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth

47

therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Loans and funded and unfunded participations in Letters of Credit and Swing Line Loans to be held on a pro rata basis by the Lenders in accordance with their Revolving Credit Percentages (without giving effect to Section 2.24(a)(iv)), whereupon that Lender will cease to be a Defaulting Lender; provided that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.

# 3. LETTERS OF CREDIT

3.1. L/C Commitment. (a) Subject to the terms and conditions hereof, each Issuing Lender, in reliance on the agreements of the other Lenders set forth in Section 3.3, agrees to issue Letters of Credit for the account of the Borrower or any of its Subsidiaries and to amend or extend Letters of Credit previously issued by it, in accordance with Section 3.2(b), on any Business Day during the Revolving Credit Commitment Period in such form as may be approved from time to time by such Issuing Lender; provided, that no Issuing Lender shall issue any Letter of Credit if, after giving effect to such issuance, (i) the L/C Obligations would exceed the L/C Commitment or (ii) the aggregate amount of the Available Revolving Credit Commitments would be less than zero. Each Letter of Credit shall (i) be denominated in Dollars and (ii) expire no later than the earlier of (x) the first anniversary of its date of issuance and (y) the date which is one year after the Revolving Credit Termination Date, provided that any Letter of Credit with a one-year term may provide for the renewal thereof for additional one-year periods (it being understood that in the event the expiry date of any requested Letter of Credit would occur after the Revolving Credit Termination Date, all L/C Obligations in respect of such Letters of Credit shall be Cash Collateralized no later than one month prior to the Revolving Credit Termination Date in accordance with Section 2.23) as set forth in Section 3.2(c). All Existing Letters of Credit shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof.

- (b) No Issuing Lender shall at any time be obligated to issue any Letter of Credit hereunder if:
- (i) such issuance would conflict with, or cause such Issuing Lender or any L/C Participant to exceed any limits imposed by, any applicable Requirement of Law;
  - (ii) such issuance would violate one or more policies of such Issuing Lender applicable to letters of credit generally;
- (iii) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain such Issuing Lender from issuing such Letter of Credit, or any Law applicable to such Issuing Lender or any request or directive (whether or not having the force of Law) from any

Governmental Authority with jurisdiction over such Issuing Lender shall prohibit, or request that such Issuing Lender refrain from, the issuance of letters of credit generally, or such Letter of Credit in particular;

- (iv) any Lender is at that time a Defaulting Lender, unless such Issuing Lender has entered into arrangements, including the delivery of Cash Collateral, satisfactory to such Issuing Lender (in its sole discretion) with the Borrower or such Lender to eliminate such Issuing Lender's actual or potential Fronting Exposure (after giving effect to Section 2.24(a)(iv)) with respect to the Defaulting Lender arising from either the Letter of Credit then proposed to be issued or that Letter of Credit and all other L/C Obligations as to which such Issuing lender has actual or potential Fronting Exposure, as it may elect in its sole discretion; provided, however, that in the event that the Borrower delivers Cash Collateral in accordance with the terms hereof, such Issuing Bank shall not be entitled to rely on this clause (iv) as justification for not issuing such Letter of Credit.
- 3.2. Procedure for Issuance and Amendment of Letter of Credit; Auto-Extension Letters of Credit. (a) Each Letter of Credit shall be issued or amended, as the case may be, upon the request of the Borrower delivered to an Issuing Lender (with a copy to the Administrative Agent) in the form of an Application, completed and signed by a Responsible Officer of the Borrower. Such Application must be received by the applicable Issuing Lender and the Administrative Agent not later than 11:00 A.M. at least two Business Days (or such later date and time as the Administrative Agent and the applicable Issuing Lender may agree in a particular instance in their sole discretion) prior to the proposed issuance date or date of amendment, as the case may be. In the case of a request for an initial issuance of a Letter of Credit, such Application shall specify in form and detail reasonably satisfactory to the applicable Issuing Lender: (A) the proposed issuance date of the requested Letter of Credit (which shall be a Business Day); (B) the amount thereof; (C) the expiry date thereof; (D) the name and address of the beneficiary thereof; (E) the documents to be presented by such beneficiary in case of any drawing thereunder; and (G) such other matters as the applicable Issuing Lender may reasonably require. In the case of a request for an amendment of any outstanding Letter of Credit, such Application shall specify in form and detail reasonably satisfactory to the applicable Issuing Lender (A) the Letter of Credit to be amended; (B) the proposed date of amendment thereof (which shall be a Business Day); (C) the nature of the proposed amendment; and (D) such other matters as the applicable Issuing Lender may reasonably require. Additionally, the Borrower shall furnish to the applicable Issuing Lender and the Administrative Agent such other documents and information pertaining to such requested Letter of Credit issuance or amendment, including any Issuer Documents, as such Issuing Lender or the Administrative Agent may reasonably requ
- (b) Promptly after receipt of any Application, the applicable Issuing Lender will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has received a copy of such Application from the Borrower and, if not, the applicable Issuing Lender will provide the Administrative Agent with a copy thereof. Unless the applicable Issuing Lender has received written notice from any Lender or the Administrative Agent, at least one Business Day prior to the requested date of issuance or amendment of the applicable Letter

49

of Credit, that one or more applicable conditions contained in Section 4 shall not then be satisfied, then, subject to the terms and conditions hereof, the applicable Issuing Lender shall, on the requested date, issue a Letter of Credit for the account of the Borrower or the applicable Subsidiary of the Borrower or enter into the applicable amendment, as the case may be, in each case in accordance with such Issuing Lender's usual and customary business practices. Immediately upon the issuance of each Letter of Credit, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the applicable Issuing Lender a risk participation in such Letter of Credit in an amount equal to the product of such Lender's Revolving Credit Percentage times the amount of such Letter of Credit.

- (c) If the Borrower so requests in any applicable Application, the relevant Issuing Lender may, in its sole and absolute discretion, agree to issue a Letter of Credit that has automatic extension provisions (each, an "Auto-Extension Letter of Credit"); provided that any such Auto-Extension Letter of Credit must permit such Issuing Lender to prevent any such extension at least once in each twelve-month period (commencing with the date of issuance of such Letter of Credit) by giving prior notice to the beneficiary thereof not later than a day (the "Non-Extension Notice Date") in each such twelve-month period to be agreed upon at the time such Letter of Credit is issued. Unless otherwise directed by an Issuing Lender, the Borrower shall not be required to make a specific request to such Issuing Lender for any such extension. Once an Auto-Extension Letter of Credit has been issued, the Lenders shall be deemed to have authorized (but may not require) such Issuing Lender to permit the extension of such Letter of Credit; provided, however, that such Issuing Lender shall not permit any such extension if (A) such Issuing Lender has determined that it would not be permitted, or would have no obligation at such time to issue such Letter of Credit in its revised form (as extended) under the terms hereof (by reason of the provisions of Section 3.1 or otherwise), or (B) it has received notice (which may be by telephone or in writing) on or before the day that is seven Business Days before the Non-Extension Notice Date (1) from the Administrative Agent that the Majority Lenders have elected not to permit such extension or (2) from the Administrative Agent, any Lender or the Borrower that one or more of the applicable conditions specified in Section 4.2 is not then satisfied, and in each such case directing such Issuing Lender not to permit such extension.
- (d) Promptly after its delivery of any Letter of Credit or any amendment to a Letter of Credit to an advising bank with respect thereto or to the beneficiary thereof, the applicable Issuing Lender will also deliver to the Borrower and the Administrative Agent a true and complete copy of such Letter of Credit or amendment.
- 3.3. <u>Drawings and Reimbursements; Funding of Participations.</u> (a) Upon receipt from the beneficiary of any Letter of Credit of any notice of a drawing under such Letter of Credit, the applicable Issuing Lender shall notify the Borrower and the Administrative Agent thereof. The Borrower shall reimburse the applicable Issuing Lender, through the Administrative Agent, for the amount of any drawing under a Letter of Credit not later than 1:00 P.M. on the date that such drawing is made, if the Borrower has received notice from such Issuing Lender of such drawing prior to 10:00 A.M. on such date, or, if the Borrower has not received notice of such drawing prior to such time on such date, then not later than 1:00 P.M. on (i) the Business Day that the Borrower receives such notice, if such notice is received prior to 10:00 A.M. on the day of receipt, or (ii) the Business Day immediately following the day that the Borrower receives

such notice, if such notice is not received prior to 10:00 A.M. on the day of such receipt (the date on which such reimbursement by the Borrower is due pursuant to this sentence being referred to herein as the "Requested Reimbursement Date"). If the Borrower fails to so reimburse such Issuing Lender by such time, the Administrative Agent shall promptly notify each Lender of the Requested Reimbursement Date, the amount of the unreimbursed drawing (the "Unreimbursed Amount"), and the amount of such Lender's Revolving Credit Percentage thereof. In such event, the Borrower shall be deemed to have requested a borrowing of Base Rate Loans to be disbursed on the Requested Reimbursement Date in an amount equal to the Unreimbursed Amount, without regard to the minimum and multiples, and notice periods, specified in Section 2.2 for the principal amount of Base Rate Loans. Such Base Rate Loans may from time to time be converted to Eurodollar Loans, as determined by the Borrower and notified to the Administrative Agent in accordance with Section 2.9, provided that no Revolving Credit Loan shall be made as a Eurodollar Loan after the day that is one month prior to the Revolving Credit Termination Date. Any notice given by any Issuing Lender or the Administrative Agent pursuant to this Section 3.3(a) may be given by telephone if immediately confirmed in writing; provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice.

- (b) Each Lender (including any Lender acting as an Issuing Lender) shall upon any notice pursuant to Section 3.3(a) make funds available (and the Administrative Agent may apply Cash Collateral provided for this purpose) for the account of the applicable Issuing Lender at the Administrative Agent's Office in an amount equal to its Revolving Credit Percentage of the Unreimbursed Amount not later than 4:00 P.M. on the Business Day specified in such notice by the Administrative Agent, whereupon, subject to the provisions of Section 3.3(a), each Lender that so makes funds available shall be deemed to have made a Base Rate Loan to the Borrower in such amount. The Administrative Agent shall remit the funds so received and any Cash Collateral so applied to such Issuing Lender.
- (c) If any drawing is made under a Letter of Credit and is not reimbursed or refinanced on the date such drawing is made, for any reason, the Borrower shall be deemed to have incurred from the applicable Issuing Lender an L/C Borrowing in the amount of the Unreimbursed Amount that is not so reimbursed or refinanced, which L/C Borrowing (i) shall bear interest at the rate applicable to Base Rate Loans from and including the date that such drawing is paid by such Issuing Lender to but excluding the earlier of the date that such Unreimbursed Amount is so reimbursed or refinanced or the date that is the next Business Day following the Requested Reimbursement Date and, if not so reimbursed or refinanced on or prior to the date that is the next Business Day following the Requested Reimbursement Date to but excluding the date so reimbursed or refinanced, the rate applicable to Base Rate Loans <u>plus</u> 2% and (ii) shall, on and after the date that is the next Business Day following the Requested Reimbursement Date, be due and payable on demand. In such event, each Lender's payment to the Administrative Agent for the account of the applicable Issuing Lender pursuant to <u>Section 3.3(b)</u> shall be deemed payment in respect of its participation in such L/C Borrowing and shall constitute an L/C Advance from such Lender in satisfaction of its participation obligation under this <u>Section 3.3</u>.

- (d) Until each Lender funds its Loan or L/C Advance pursuant to this <u>Section 3.3</u> to reimburse the applicable Issuing Lender for any amount drawn under any Letter of Credit, interest in respect of such Lender's Revolving Credit Percentage of such amount shall be solely for the account of such Issuing Lender.
- (e) Each Lender's obligation to make Loans or L/C Advances to reimburse each Issuing Lender for amounts drawn under Letters of Credit, as contemplated by this Section 3.3, shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any set-off, counterclaim, recoupment, defense or other right which such Lender may have against any Issuing Lender, the Borrower or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Revolving Credit Loans pursuant to this Section 3.3 is subject to the conditions set forth in Section 4.2 (other than delivery by the Borrower of a Loan Notice). No such making of an L/C Advance shall relieve or otherwise impair the obligation of the Borrower to reimburse each Issuing Lender for the amount of any payment made by such Issuing Lender under any Letter of Credit, together with interest as provided herein.
- (f) If any Lender fails to make available to the Administrative Agent for the account of the applicable Issuing Lender any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 3.3 by the time specified in Section 3.3(b), then, without limiting the other provisions of this Agreement, such Issuing Lender shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to such Issuing Lender at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by such Issuing Lender in accordance with banking industry rules on interbank compensation, plus any administrative, processing or similar fees customarily charged by such Issuing Lender in connection with the foregoing. If such Lender pays such amount (with interest and fees as aforesaid), the amount so paid shall constitute such Lender's Loan included in the relevant borrowing or L/C Advance in respect of the relevant L/C Borrowing, as the case may be. A certificate of an Issuing Lender submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this paragraph (f) shall be conclusive absent manifest error.
- 3.4. Repayment of Participations. (a) At any time after an Issuing Lender has made a payment under any Letter of Credit and has received from any Lender such Lender's L/C Advance in respect of such payment in accordance with Section 3.3(b), if the Administrative Agent receives for the account of such Issuing Lender any payment in respect of the related Unreimbursed Amount or interest thereon (whether directly from the Borrower or otherwise, including proceeds of Cash Collateral applied thereto by the Administrative Agent), the Administrative Agent will distribute to such Lender its Revolving Credit Percentage thereof (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's L/C Advance was outstanding) in the same funds as those received by the Administrative Agent.
- (b) If any payment received by the Administrative Agent for the account of an Issuing Lender pursuant to Section 3.3(b) is required to be returned under any of the

- 3.5. <u>Obligations Absolute</u>. The obligation of the Borrower to reimburse each Issuing Lender for each drawing under each Letter of Credit and to repay each L/C Borrowing shall be absolute, unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including the following:
  - (i) any lack of validity or enforceability of such Letter of Credit, this Agreement, or any other Loan Document;
  - (ii) the existence of any claim, counterclaim, set-off, defense or other right that the Borrower or any of its Subsidiaries may have at any time against any beneficiary or any transferee of such Letter of Credit (or any Person for whom any such beneficiary or any such transferee may be acting), such Issuing Lender or any other Person, whether in connection with this Agreement, the transactions contemplated hereby or by such Letter of Credit or any agreement or instrument relating thereto, or any unrelated transaction;
  - (iii) any draft, demand, certificate or other document presented under such Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or any loss or delay in the transmission or otherwise of any document required in order to make a drawing under such Letter of Credit;
  - (iv) any payment by such Issuing Lender under such Letter of Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Letter of Credit; or any payment made by such Issuing Lender under such Letter of Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Letter of Credit, including any arising in connection with any proceeding under any Debtor Relief Law; or
  - (v) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Borrower or any of its Subsidiaries.
- 3.6. <u>Role of Issuing Lender</u>. Each Lender and the Borrower agree that, in paying any drawing under a Letter of Credit, no Issuing Lender shall have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by the Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or

53

the authority of the Person executing or delivering any such document. None of the Issuing Lenders, any Agent-Related Person nor any of the respective correspondents, participants or assignees of any Issuing Lender shall be liable to any Lender for (a) any action taken or omitted in connection herewith at the request or with the approval of the Lenders or the Majority Lenders, as applicable; (b) any action taken or omitted in the absence of gross negligence or willful misconduct; or (c) the due execution, effectiveness, validity or enforceability of any document or instrument related to any Letter of Credit or Application. The Borrower hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Letter of Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Borrower's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuing Lenders, any Agent-Related Person, nor any of the respective correspondents, participants or assignees of any Issuing Lender, shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 3.5; provided, however, that anything in such clauses (i) through (v) to the contrary notwithstanding, the Borrower may have a claim against an Issuing Lender, and such Issuing Lender may be liable to the Borrower, to the extent, but only to the extent, of any direct, as opposed to consequential or exemplary, damages suffered by the Borrower which the Borrower proves were caused by such Issuing Lender's willful misconduct or gross negligence or such Issuing Lender's willful failure to pay under any Letter of Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Letter of Credit. In furtherance and not in limitation of the foregoing, each Issuing Lender may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and no Issuing Lender shall be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

- 3.7. <u>Applicability of ISP98</u>. Unless otherwise expressly agreed by the applicable Issuing Lender and the Borrower when a Letter of Credit is issued including any such agreement as applicable to an Existing Letter of Credit, the rules of the ISP shall apply to each standby Letter of Credit.
- 3.8. Fees and Other Charges. (a) The Borrower will pay to the Administrative Agent, for the account of the Lenders, a fee on the daily amount available to be drawn under all outstanding Letters of Credit at a per annum rate equal to the Applicable Margin then in effect with respect to Eurodollar Loans, to be shared ratably among the Lenders in accordance with their respective Revolving Credit Percentages and payable quarterly in arrears on each L/C Fee Payment Date after the issuance date; provided, however, that (i) any letter of credit fees otherwise payable for the account of a Defaulting Lender with respect to any Letter of Credit as to which such Defaulting Lender has not provided Cash Collateral satisfactory to the applicable Issuing Lender pursuant to Section 3.1 shall be payable, to the maximum extent permitted by applicable Law, to the other Lenders in accordance with the upward adjustments in their respective Revolving Credit Percentages allocable to such Letter of Credit pursuant to Section 2.24(a)(iv), with the balance of such fee, if any, payable to the applicable Issuing Lender for its own account and (ii) the Borrower shall not be required to pay letter of credit fees pursuant to this Section 3.8 in respect of such portion of the outstanding Letters of Credit for which it has furnished Cash Collateral in accordance with the terms hereof to reduce the Fronting Exposure of

54

any Defaulting Lender. In addition, the Borrower shall pay to the relevant Issuing Lender for its own account a fronting fee on the daily amount available to be drawn under all outstanding Letters of Credit issued by such Issuing Lender at a rate and at the times to be agreed upon by the Borrower and such Issuing Lender. For purposes of computing the daily amount available to be drawn under the Letters of Credit, the amount of such Letters of Credit shall be determined in accordance with Section 1.3.

(b) In addition to the foregoing fees, the Borrower shall pay or reimburse each Issuing Lender for such normal and customary costs and expenses as are incurred or charged by such Issuing Lender in issuing, negotiating, effecting payment under, amending or otherwise administering any Letter of Credit.

3.9. Letters of Credit Issued for Subsidiaries. Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, a Subsidiary, the Borrower shall be obligated to reimburse the applicable Issuing Lender hereunder for any and all drawings under such Letter of Credit. The Borrower hereby acknowledges that the issuance of Letters of Credit for the account of Subsidiaries is being made at the request of the Borrower and in reliance on the Borrower's agreement to reimburse the applicable Issuing Lender hereunder for any and all drawings under any such Letter of Credit.

3.10. <u>Conflict with Issuer Documents</u>. In the event of any conflict between the terms hereof and the terms of any Issuer Document, the terms hereof shall control.

#### 4. CONDITIONS PRECEDENT

- 4.1. <u>Conditions to Closing</u>. The occurrence of the Closing Date is subject to the satisfaction on such date of the following conditions precedent:
- (a) The Administrative Agent's receipt of the following, each of which shall be originals or telecopies (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer of the Borrower, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance reasonably satisfactory to the Administrative Agent:
  - (i) executed counterparts of this Agreement, sufficient in number for distribution to the Administrative Agent, each Lender party hereto on the Closing Date and the Borrower;
  - (ii) a Note executed by the Borrower in favor of each Lender requesting a Note so long as such request is made at least three (3) Business Days prior to the Closing Date;
  - (iii) such certificates of resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of the Borrower as the Administrative Agent may reasonably require evidencing the identity, authority and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Documents to which the Borrower is a party;

55

- (iv) such documents and certifications as the Administrative Agent may reasonably require to evidence that the Borrower is duly organized or formed, and that the Borrower is validly existing, in good standing and qualified to engage in business in the jurisdiction where the Borrower is organized;
  - (v) a Closing Certificate of the Borrower with appropriate insertions and attachments, if any;
- (vi) a certificate signed by a Responsible Officer on behalf of the Borrower either (A) attaching copies of all consents, licenses and approvals required in connection with the execution, delivery and performance by the Borrower and the validity against the Borrower of the Loan Documents to which it is a party, and such consents, licenses and approvals shall be in full force and effect, or (B) stating that no such consents, licenses or approvals are so required;
- (vii) a certificate signed by a Responsible Officer on behalf of the Borrower certifying that there has been no event or circumstance since December 31, 2010 that has had or could be reasonably expected to have a Material Adverse Effect; and
  - $(viii) \qquad \hbox{the Guaranty executed by each Guarantor.}$
- (b) <u>Fees.</u> (i) The Administrative Agent and the Arrangers shall have received all fees required to be paid by the Borrower on or prior to the Closing Date.
  - (ii) The Borrower shall have paid all fees, charges and disbursements of Morgan, Lewis & Bockius LLP, as counsel to the Administrative Agent (directly to such counsel if requested by the Administrative Agent), to the extent required to be paid by the Borrower and invoiced prior to the Closing Date.
- (c) <u>Legal Opinions</u>. The Administrative Agent shall have received (i) the legal opinion of Robert Seelig, Esquire, counsel to the Borrower and each of the Initial Guarantors, substantially in the form of <u>Exhibit F</u> and (ii) the legal opinion of Conyers Dill & Pearman, Bermuda counsel to the Borrower and each of the Initial Guarantors, substantially in the form of Exhibit G.
- (d) <u>Termination of Existing Credit Facility</u>. The Administrative Agent shall have received evidence (including, without limitation, payoff letters), reasonably satisfactory to the Administrative Agent in its reasonable discretion, that the Existing Credit Agreement has been or concurrently with the Closing Date is being terminated, and all Loans (as defined in the Existing Credit Agreement) and all accrued but unpaid interest and fees payable under the Existing Credit Agreement have been, or concurrently with the effectiveness hereof will be, paid in full; <u>provided</u> that any such payments shall be permitted to be made with the proceeds of concurrent Borrowings made hereunder that may be made upon notice by 1:00 P.M. on the Closing Date.

- (e) <u>Material Adverse Effect</u>. Up to and including the Closing Date, since December 31, 2010 there has been no event or circumstance that has had or could reasonably be expected to have a Material Adverse Effect.
- 4.2. <u>Conditions to Closing and Each Extension of Credit</u>. The occurrence of the Closing Date and the agreement of each Lender to make any extension of credit requested to be made by it hereunder on any date (including, without limitation, its initial extension of credit or any issuance, or

increase in the amount of, any Letter of Credit but excluding conversions or continuations of Loans) is subject to the satisfaction of the following conditions precedent:

- (a) <u>Representations and Warranties</u>. Each of the representations and warranties made by the Borrower in <u>Section 5</u> (other than <u>Section 5.5</u>) or pursuant to any of the other Loan Documents shall be true and correct in all material respects on and as of such date as if made on and as of such date, except to the extent that they expressly relate to an earlier date, in which case they shall be true and correct in all material respects as of such earlier date.
- (b) <u>No Default</u>. No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the extensions of credit requested to be made on such date.
- (c) <u>Borrowing Request</u>. Except as provided in <u>Section 3.3</u>, the Administrative Agent shall have received a Borrowing Request or, as applicable, an Application.

Each borrowing by and issuance of a Letter of Credit on behalf of the Borrower hereunder shall constitute a representation and warranty by the Borrower as of the date of such extension of credit (including any issuance, or increase in the amount of, any Letter of Credit) that the conditions contained in this <u>Section 4.2 (a)</u> and (b) have been satisfied on and as of the date of the applicable extension of credit.

## 5. REPRESENTATIONS AND WARRANTIES

To induce the Administrative Agent and the Lenders to enter into this Agreement and to make the Loans and issue or participate in the Letters of Credit, the Borrower hereby represents and warrants to the Administrative Agent and each Lender that:

#### 5.1. Financial Statements.

(a) The audited consolidated balance sheet of the Borrower and its consolidated Subsidiaries, as at December 31, 2010 and the related consolidated statements of income and of cash flows for the fiscal year ended on such date, reported on and accompanied by unqualified reports from PricewaterhouseCoopers LLP or another independent certified public accounting firm of nationally recognized standing, present fairly in all material respects the consolidated financial condition of the Borrower and its consolidated Subsidiaries, as at such date, and the consolidated results of their operations and their consolidated cash flows for such fiscal year then ended in accordance with GAAP applied consistently throughout the periods involved (except as approved by the aforementioned firm of accountants and disclosed therein).

57

- (b) The unaudited consolidated balance sheet of the Borrower and its consolidated Subsidiaries, as of and for the fiscal quarter ended June 30, 2011, and the related unaudited consolidated statements of income and cash flows for such fiscal quarters ended on such dates, present fairly in all material respects the consolidated financial condition of the Borrower and its consolidated Subsidiaries as at such dates, and the consolidated results of their operations and their consolidated cash flows for the fiscal quarters then ended in accordance with GAAP applied consistently throughout the periods involved (except (x) as approved by the aforementioned firms of accountants and disclosed therein or (y) for normal year-end audit adjustments and the absence of footnotes).
- 5.2. Corporate Existence; Compliance with Law. The Borrower and each of its Subsidiaries (a) is duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization, except to the extent that the failure of the Subsidiaries to be so organized, validly existing and in good standing could not, in the aggregate, reasonably be expected to have a Material Adverse Effect, (b) has the corporate or other power and authority, and the legal right, to own and operate its Property, to lease the Property it operates as lessee and to conduct the business in which it is currently engaged, except to the extent that the failure to have such power, authority and legal right could not, in the aggregate, reasonably be expected to have a Material Adverse Effect, (c) is duly qualified as a foreign corporation and in good standing under the Laws of each jurisdiction where its ownership, lease or operation of Property or the conduct of its business requires such qualification, except to the extent failure to so qualify or be in good standing could not, in the aggregate, reasonably be expected to have a Material Adverse Effect, and (d) is in compliance with all Requirements of Law, including, without limitation, with respect to environmental laws, except to the extent that the failure to comply therewith could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 5.3. Corporate Power; Authorization; Enforceable Obligations. The Borrower has the corporate or other power and authority, and the legal right, to make, deliver and perform the Loan Documents to which it is a party and to borrow hereunder. The Borrower has taken all necessary corporate or other action to authorize the execution, delivery and performance of the Loan Documents to which it is a party and to authorize the borrowings on the terms and conditions of this Agreement. No consent or authorization of, filing with, notice to or other act by or in respect of, any Governmental Authority or any other Person is required in connection with the borrowings hereunder or the execution, delivery, performance, validity or enforceability of this Agreement or any of the other Loan Documents, except consents, authorizations, filings and notices have been obtained or made and are in full force and effect and except to the extent failure to obtain any consents, authorizations, filings, and notices could not, in the aggregate, reasonably be expected to have a Material Adverse Effect. Each Loan Document to which the Borrower is a party has been duly executed and delivered on behalf of the Borrower. This Agreement constitutes, and each other Loan Document to which the Borrower is a party upon execution will constitute, a legal, valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether enforcement is sought by proceedings in equity or at law).

- 5.5. No Material Litigation. No litigation, investigation or proceeding of or before any arbitrator or Governmental Authority is pending or, to the knowledge of the Borrower, threatened by or against the Borrower or any of its Subsidiaries or against any of their respective properties or assets that (a) purport to affect or pertain to this Agreement or any other Loan Document or any of the transactions contemplated hereby or thereby, or (b) could reasonably be expected to have a Material Adverse Effect.
- 5.6. Ownership of Property; Liens. The Borrower and each of its Subsidiaries has title in fee simple to, or a valid leasehold interest in, all its real property, and good title to, or a valid leasehold interest in, all its other Property, and none of such Property is subject to any Lien except as permitted by Section 7.3, except to the extent such defects in title could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 5.7. Intellectual Property. The Borrower and each of its Subsidiaries owns, or is licensed to use, all Intellectual Property material to the conduct of its business as currently conducted. No material claim has been asserted and is pending by any Person challenging or questioning the use of any Intellectual Property or the validity or effectiveness of any Intellectual Property, nor does the Borrower know of any valid basis for any such claim, other than claims that could not, in the aggregate, reasonably be expected to have a Material Adverse Effect. The use of Intellectual Property by the Borrower and each of its Subsidiaries does not infringe on the rights of any Person in any material respect, except for infringements that could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 5.8. Taxes. The Borrower and each of its Subsidiaries has filed or caused to be filed all material Federal, state and other tax returns that are required to be filed (taking into account any applicable extensions) and has paid all taxes shown to be due and payable on said returns or on any assessments made against it or any of its Property and all other material taxes, fees or other charges imposed on it or any of its Property by any Governmental Authority and, to the knowledge of the Borrower, no tax Lien has been filed, and no claim is being asserted, with respect to any such tax, fee or other charge, except (i) those in respect of which the amount or validity are currently being contested in good faith by appropriate proceedings and with respect to which reserves in conformity with SAP or GAAP, as applicable, have been provided on the books of the Borrower or any of its Subsidiaries, as the case may be, and (ii) any amount the failure of which to pay could not reasonably be expected to result in a Material Adverse Effect.
- 5.9. <u>Federal Regulations</u>. No part of the proceeds of any Loans will be used for "purchasing" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulation U as now and from time to time hereafter in effect or for any purpose that violates the provisions of the Regulations of the Board. If requested by any Lender

59

or the Administrative Agent, the Borrower will furnish to the Administrative Agent and each Lender a statement to the foregoing effect in conformity with the requirements of FR Form G-3 or FR Form U-1 referred to in Regulation U.

- 5.10. <u>ERISA</u>. (a) Except as could not reasonably be expected to result in a Material Adverse Effect, each Plan is in compliance in all material respects with the applicable provisions of ERISA, the Code and other Federal or state laws. Each Pension Plan that is intended to be a qualified plan under Section 401(a) of the Code has received a favorable determination letter from the Internal Revenue Service to the effect that the form of such Plan is qualified under Section 401(a) of the Code and the trust related thereto has been determined by the Internal Revenue Service to be exempt from federal income tax under Section 501(a) of the Code, or an application for such a letter is currently being processed by the Internal Revenue Service. To the best knowledge of the Borrower, nothing has occurred that would prevent or cause the loss of such tax-qualified status.
- (b) There are no pending or, to the best knowledge of the Borrower, threatened claims, actions or lawsuits, or action by any Governmental Authority, with respect to any Plan that could reasonably be expected to have a Material Adverse Effect. There has been no prohibited transaction or violation of the fiduciary responsibility rules with respect to any Plan that has resulted or could reasonably be expected to result in a Material Adverse Effect.
- (c) (i) No ERISA Event has occurred, and neither the Borrower nor any ERISA Affiliate is aware of any fact, event or circumstance that could reasonably be expected to constitute or result in an ERISA Event with respect to any Pension Plan; (ii) the Borrower and each ERISA Affiliate has met all applicable requirements under the Pension Funding Rules in respect of each Pension Plan (other than any Multiemployer Plan), and no waiver of the minimum funding standards under the Pension Funding Rules has been applied for or obtained; (iii) as of the most recent valuation date for any Pension Plan, the funding target attainment percentage (as defined in Section 430(d)(2) of the Code) is 60% or higher and neither the Borrower nor any ERISA Affiliate knows of any facts or circumstances that could reasonably be expected to cause the funding target attainment percentage for any such plan to drop below 60% as of the most recent valuation date; (iv) neither the Borrower nor any ERISA Affiliate has incurred any liability to the PBGC other than for the payment of premiums, and there are no premium payments which have become due that are unpaid; (v) neither the Borrower nor any ERISA Affiliate has engaged in a transaction that could be subject to Section 4069 or Section 4212(c) of ERISA; and (vi) no Pension Plan has been terminated by the plan administrator thereof nor by the PBGC, and to the best knowledge of the Borrower, no event or circumstance has occurred or exists that could reasonably be expected to cause the PBGC to institute proceedings under Title IV of ERISA to terminate any Pension Plan, in each case, except as could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- 5.11. <u>Investment Company Act; Other Regulations</u>. The Borrower is not an "investment company" within the meaning of the Investment Company Act of 1940, as amended. The Borrower is not subject to regulation under any Requirement of Law (other than Regulation X of the Board) which limits its ability to incur Indebtedness hereunder.

- 5.12. <u>Use of Proceeds</u>. The proceeds of the Loans and the Letters of Credit shall be used for working capital and general corporate purposes of the Borrower or any of its Subsidiaries, including, without limitation, (a) acquisitions, (b) the issuance of Letters of Credit, (c) refinancings of outstanding indebtedness, if any, of the Borrower or any of its Subsidiaries (including under the Existing Credit Agreement and the Existing Letters of Credit), (d) share repurchases, (e) special dividends and (f) for payment of fees and expenses incurred in connection with this Agreement.
- 5.13. <u>Accuracy of Information, etc.</u> No statement or information contained in any document, certificate or statement furnished to the Administrative Agent or the Lenders or any of them, by or on behalf of the Borrower for use in connection with the transactions contemplated by this Agreement or the other Loan Documents, taken as a whole contained, as of the date such statement, information, document or certificate was so furnished,

any untrue statement of a material fact or omitted to state a material fact necessary in order to make the statements contained therein not materially misleading in light of the circumstances under which such statement, information, document or certificate was made or furnished. The projections and pro forma financial information contained in the materials referenced above were prepared in good faith based on assumptions believed by management of the Borrower to be reasonable at the time made, it being recognized by the Lenders that such financial information as it relates to future events is not to be viewed as fact and that actual results during the period or periods covered by such financial information may differ from the projected results set forth therein by a material amount.

- 5.14. <u>Insurance Regulatory Matters</u>. No License of any Insurance Subsidiary, the loss of which could reasonably be expected to have a Material Adverse Effect, is the subject of a proceeding for suspension or revocation. To the knowledge of the Borrower, there is no sustainable basis for such suspension or revocation, and no such suspension or revocation has been threatened by any Governmental Authority.
- 5.15. Indebtedness and Liens. As of the Closing Date, (i) no Subsidiary of the Borrower had outstanding any Indebtedness that was created, incurred or assumed after June 30, 2011, except Indebtedness that would have been permitted by Section 7.2 (without giving effect to the Indebtedness permitted by Section 7.2(a), except for refinancings, refundings, renewals or extensions permitted by such Section) if created, incurred or assumed by such Subsidiary on the Closing Date and (ii) there does not exist (a) any Lien that was created, incurred or assumed after June 30, 2011, upon any stock or Indebtedness of any Subsidiary to secure any Debt of the Borrower or any of its Subsidiaries or any other Person (other than the obligations hereunder) or (b) any Lien that was created, incurred or assumed after June 30, 2011, upon any other Property, to secure any Debt of the Borrower or any of its Subsidiaries or any other Person (other than the obligations hereunder), except, in the case of (a) or (b), Liens that would have been permitted by Section 7.3 (i)(x)) if so created, incurred or assumed on the Closing Date.
- 5.16. <u>Taxpayer Identification Number</u>. As of the Closing Date, the Borrower's true and correct U.S. taxpayer identification number is set forth on <u>Schedule 10.02</u>.

61

#### 6. AFFIRMATIVE COVENANTS

The Borrower hereby agrees that, from and after the Closing Date and so long as, the Commitments remain in effect, any Letter of Credit remains outstanding, there exists any unpaid Reimbursement Obligations or any principal or interest on any Loan or any fee payable hereunder is owing to any Lender or the Administrative Agent hereunder, the Borrower shall and shall cause each of its Subsidiaries to:

- 6.1. <u>Financial Statements</u>. Furnish to the Administrative Agent (either electronically or with sufficient copies for distribution by the Administrative Agent to each Lender):
- (i) not later than the date required to be filed pursuant to the Act of 1934 (after giving effect to any extension permitted or granted by the SEC), but in any event not later than 95 days after the end of each fiscal year of the Borrower ending subsequent to the Closing Date, a copy of the audited consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at the end of such fiscal year, and the related audited consolidated statements of income and of cash flows for such fiscal year, setting forth in each case in comparative form the figures as of the end of and for the previous fiscal year, accompanied by an opinion by PricewaterhouseCoopers LLP, or other independent certified public accounting firm of nationally recognized standing, which report shall be prepared in accordance with generally accepted auditing standards and applicable securities laws and shall not be subject to a "going concern" or like qualification or exception, or qualification as to the scope of the audit (for purposes hereof, delivery of the Borrower's annual report on Form 10-K (which shall be deemed delivered on the date when such document is posted on the SEC's website at www.sec.gov or any replacement website) will be sufficient in lieu of delivery of such financial statements); and (ii) not later than the date required to be filed pursuant to the Act of 1934 (after giving effect to any extension permitted or granted by the SEC), but in any event not later than 60 days after the end of each of the first three fiscal quarters of each fiscal year of the Borrower ending subsequent to the Closing Date, a copy of the unaudited consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at the end of such fiscal quarter and the related unaudited consolidated statements of income and of cash flows for such fiscal quarter and the portion of the fiscal year through the end of such fiscal quarter, setting forth in each case in comparative form the figures as of the end of and for the corresponding period in the previous year, certified by a Responsible Officer on behalf of the Borrower as being fairly stated in all material respects in accordance with GAAP (subject to normal year-end audit adjustments and the absence of footnotes) (for purposes hereof, delivery of the Borrower's Quarterly Report on Form 10-Q (which shall be deemed delivered on the date when such document is posted on the SEC's website at www.sec.gov or any replacement website) will be sufficient in lieu of delivery of such financial statements and certifications); all such financial statements, together with notes to such financial statements, to fairly present in all material respects the financial condition and income and cash flows of the subject thereof as at the dates and for the periods covered thereby in accordance with GAAP applied consistently throughout the periods reflected therein and with prior periods (except (x) as approved by such accountants or officer, as the case may be, and disclosed therein or (y) in the case of unaudited financial statements, subject to normal yearend adjustments and the absence of footnotes);

- (b) not later than the date required by Law to be prepared (or such later date as may be allowed by the applicable Governmental Authority), but in any event not later than (i) 150 days after the end of each fiscal year of OneBeacon Insurance Group, copies of the unaudited combined Annual Statement of OneBeacon Insurance Group, certified by a Responsible Officer on behalf of OneBeacon Limited; provided, that any such combined Annual Statement shall only be required to be so furnished to the extent such statement is required by Law to be prepared and on any date if OneBeacon Limited is required to be consolidated on the balance sheet of the Borrower in accordance with GAAP on such date, and (ii) 95 days after the end of each fiscal year of a Material Insurance Subsidiary (as of the date of delivery pursuant hereto), copies of the unaudited Annual Statement of such Material Insurance Subsidiary, certified by a Responsible Officer on behalf of such Material Insurance Subsidiary; all such statements to be prepared in accordance with SAP consistently applied throughout the periods reflected therein and, if required by the applicable Governmental Authority, audited and certified by independent certified public accounting firm of recognized national standing (it being understood that delivery of audited statements shall be made within 10 days following the delivery of such statements to the applicable Governmental Authority); provided, however, that in the case of Sirius, such Annual Statement shall be due not later than 150 days after the end of each fiscal year;
- (c) not later than the date required by Law to be prepared (or such later date as may be allowed by the applicable Governmental Authority), but in any event not later than 70 days after the end of each interim financial period of each Material Insurance Subsidiary in respect of which an

Interim Statement is required to be prepared (as of the date delivery of such Interim Statement is required), copies of the Interim Statement of such Material Insurance Subsidiary for such interim financial period, all such statements to be prepared in accordance with SAP consistently applied throughout the period reflected herein; provided, that, in the case of Sirius, such Interim Statement shall be due not later than 90 days after the end of such interim period;

- (d) within 15 days after being delivered to any Material Insurance Subsidiary subsequent to the Closing Date, any final Report on Examination issued by the applicable Department or the NAIC that results in material adjustments to the financial statements referred to in <u>paragraphs (b)</u> or (c) above;
- (e) to the extent such a statement is required by Law to be prepared, within 10 days following the delivery to the applicable Department, a copy of each "Statement of Actuarial Opinion" and "Management Discussion and Analysis" for a Material Insurance Subsidiary which is provided to the applicable Department as to the adequacy of loss reserves of such Material Insurance Subsidiary, such opinion to be in the format prescribed by the insurance code of the state of domicile of such Material Insurance Subsidiary; and
- (f) promptly after the Borrower's receipt thereof, subject to any restrictions imposed by such independent accountants, copies of any management letters submitted to the board of directors (or the audit committee of the board of directors) of the Borrower by independent accountants in connection with the annual audit of the Borrower or any of its Subsidiaries.

63

- 6.2. <u>Certificates; Other Information</u>. Furnish to the Administrative Agent (either electronically or with sufficient copies for distribution by the Administrative Agent to each Lender) or, in the case of <u>clause (d)</u>, to the relevant Lender:
- (a) within 5 Business Days after the delivery of the audited financial statements referred to in Section 6.1(a)(i), a certificate of the independent certified public accounting firm reporting on such financial statements stating that in making the examination necessary therefor no knowledge was obtained of any Default or Event of Default (it being understood that (i) such certificate shall only be required if delivery by such independent certified public accounting firm of such a certificate is not prohibited by its policies and (ii) any such certificate may be limited in scope and qualified in accordance with customary practices of the accounting profession), except as specified in such certificate;
- (b) within 5 Business Days after the deadline for the delivery of any financial statements pursuant to Section 6.1(a), (i) a certificate of a Responsible Officer of the Borrower stating that such Responsible Officer has obtained no knowledge of any continuing Default or Event of Default except as specified in such certificate and (ii) a Compliance Certificate containing all information and calculations necessary for determining compliance by the Borrower with Section 7.1 as of the last day of the fiscal quarter or fiscal year of the Borrower (which delivery may, unless the Administrative Agent, or a Lender requests executed originals, be by electronic communication including fax or email and shall be deemed to be an original authentic counterpart thereof for all purposes);
- (c) within 10 days after the same are filed with the SEC (unless posted on the SEC's website at www.sec.gov or any replacement website), all reports and filings on Forms 10-K, 10-Q and 8-K that the Borrower may make to, or file with, the SEC, including any request of an extension of time for the filing of any such reports; and
- (d) promptly, such additional financial and other information as the Administrative Agent or any Lender may from time to time reasonably request.
- (e) The Borrower hereby acknowledges that (a) unless otherwise directed by the Borrower, the Administrative Agent and/or the Arrangers will make available to the Lenders and the Issuing Lender materials and/or information provided by or on behalf of the Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar electronic system (the "Platform"), subject to confidentiality undertakings reasonably acceptable to the Borrower and the Arrangers, and (b) certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Borrower or its securities) (each, a "Public Lender"). The Borrower hereby agrees that (w) all Borrower Materials that are to be made available to Public Lenders shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Borrower shall be deemed to have authorized the Administrative Agent, the Arrangers, the Issuing Lender and the Lenders to treat such Borrower Materials as either publicly available information or not material information (although it may be sensitive and proprietary) with respect to the Borrower or its securities for purposes of United States Federal and state securities laws; (y) all Borrower Materials marked "PUBLIC" are permitted to

64

be made available through a portion of the Platform designated "Public Investor;" and (z) the Administrative Agent and the Arrangers shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Investor." Notwithstanding any of the foregoing, if the Borrower also delivers any materials and/or information pursuant to this Section 6.2(e) in paper format to the Administrative Agent, such paper materials shall be deemed to be Borrower Materials for all purposes. Nothing in this Section 6.2(e) shall limit the obligations of the Administrative Agent and the Lenders under Section 10.16.

- 6.3. <u>Payment of Obligations</u>. Pay, discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all its material obligations of whatever nature (other than Indebtedness), except where the amount or validity thereof is currently being contested in good faith by appropriate proceedings and reserves in conformity with GAAP with respect thereto have been provided on the books of the Borrower or any of its Subsidiaries, as the case may be; <u>provided</u>, that the Borrower may, in the ordinary course of business, extend payments on those payables if beneficial to the operation of their businesses.
- 6.4. <u>Conduct of Business and Maintenance of Existence, etc.</u> (a) Except as otherwise would not be a Fundamental Change (i) with respect to each Subsidiary of the Borrower, preserve, renew and keep in full force and effect its corporate existence and (ii) with respect to the Borrower and each of its Subsidiaries, take all reasonable action to maintain all licenses, permits, rights, privileges and franchises necessary or desirable in the normal conduct of its business, except, in the case of <u>clause (i)</u> above and <u>clause (ii)</u> above, to the extent that failure to do so could not reasonably be expected to have a Material

Adverse Effect; and (b) comply with all Contractual Obligations (other than in respect of Indebtedness) and Requirements of Law, except to the extent that failure to comply therewith could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

- 6.5. Maintenance of Property; Insurance. (a) Keep all Property and systems useful and necessary in its business in good working order and condition, ordinary wear and tear excepted and (b) maintain with financially sound and reputable insurance companies (other than with the Borrower or any of its Subsidiaries) insurance on all its Property in at least such amounts and against at least such risks (but including in any event public liability, product liability and business interruption) as are usually insured against in the same general area by companies engaged in the same or a similar business (it being understood that, to the extent consistent with prudent business practices of Persons carrying on a similar business in a similar location, a program of self-insurance for first and other loss layers may be utilized).
- 6.6. <u>Inspection of Property; Books and Records; Discussions</u>. (a) Keep proper books of records and account in which full, true and correct entries in conformity with GAAP (or SAP as applicable) and all Requirements of Law shall be made of all material dealings and transactions in relation to its business and activities and (b) upon reasonable prior notice, permit representatives of the Administrative Agent (who may be accompanied by representatives of other Lenders) and, during the continuance of an Event of Default, any Lender to (x) visit and inspect any of its properties, (y) during the continuance of an Event of Default, conduct reasonable examinations of (and, with the consent of the Borrower, such consent not to be unreasonably withheld, make abstracts from) any of its books and records at any reasonable time

65

and as often as may reasonably be requested and (z) discuss the business, operations, properties and financial and other condition of the Borrower with officers and employees of the Borrower. It is understood that (i) any information obtained by the Administrative Agent or any Lender in any visit or inspection pursuant to this Section 6.6 shall be subject to the confidentiality requirements of Section 10.16, (ii) the Borrower may impose, with respect to any Lender or any Affiliate of any Lender reasonably deemed by the Borrower to be engaged significantly in a business which is directly competitive with any material business of the Borrower and its Subsidiaries, reasonable restrictions on access to proprietary information of the Borrower or any of its Subsidiaries and (iii) the Lenders will coordinate their visits through the Administrative Agent with a view to preventing the visits provided for by this Section 6.6 from becoming unreasonably burdensome to the Borrower or any of its Subsidiaries.

- 6.7. <u>Notices</u>. Give notice to the Administrative Agent (it being agreed that the Administrative Agent shall, upon receipt of such notice, notify each Lender thereof) of the following within the time periods specified:
- (a) Promptly after any Responsible Officer of the Borrower obtains knowledge thereof, the occurrence of any Default or Event of Default;
  - (b) Within five days after any Responsible Officer of the Borrower obtains knowledge thereof, the occurrence of:
  - (i) any default or event of default under any Contractual Obligation (other than in respect of Indebtedness) of the Borrower or any of its Subsidiaries or any litigation, investigation or proceeding which may exist at any time between the Borrower or any of its Subsidiaries and any Governmental Authority, that in either case, if not cured or if adversely determined, as the case may be, could reasonably be expected to have a Material Adverse Effect; and
  - (ii) (A) any litigation or proceeding affecting the Borrower or any of its Subsidiaries (other than claims-related litigation involving an Insurance Subsidiary) in which (x) the amount involved (and not covered by insurance) is \$50,000,000 or more or (y) in which injunctive or similar relief is sought that could reasonably be expected to have a Material Adverse Effect and (B) any claims-related litigation affecting any Insurance Subsidiary which could reasonably be expected to have a Material Adverse Effect; and
- (c) As soon as possible and, in any event, within 30 days after a Responsible Officer of the Borrower obtains knowledge thereof, the occurrence of any ERISA Event which could reasonably be expected to result in liability to the Borrower in excess of \$50,000,000.

Each notice pursuant to this <u>Section 6.7</u> shall be accompanied by a statement of a Responsible Officer on behalf of the Borrower setting forth details of the occurrence or such default referred to therein and stating what action the Borrower or the relevant Subsidiary proposes to take with respect thereto.

6.8. <u>Taxes</u>. Pay, discharge, or otherwise satisfy before the same shall become overdue, all taxes, assessments and other governmental charges imposed upon it and its real

66

estate, sales and activities, or any part thereof, or upon the income or profits therefrom, other than where failure to pay such taxes could not reasonably be expected to result in a Material Adverse Effect; <u>provided</u> that any such tax, assessment, charge, levy or claim need not be paid if the validity or amount thereof shall currently be contested in good faith by appropriate proceedings and reserves in conformity with SAP or GAAP, as applicable, have been provided on the books of the Borrower or any of its Subsidiaries, as the case may be.

- 6.9. Use of Proceeds. Use the proceeds of the Loans and the Letters of Credit solely for the purposes set forth in Section 5.12.
- 6.10. <u>Further Assurances</u>. The Borrower will, and will cause each of its Subsidiaries to, cooperate with the Lenders and the Administrative Agent and execute such further instruments and documents as the Lenders or the Administrative Agent shall reasonably request to give effect to the transactions contemplated by this Agreement and the other Loan Documents.

# 7. NEGATIVE COVENANTS

The Borrower hereby agrees that, from and after the Closing Date and so long as the Commitments remain in effect, any Letter of Credit remains outstanding, there exist any unpaid Reimbursement Obligations or any principal or interest on any Loan or any fee payable hereunder is owing to any Lender or the Administrative Agent hereunder:

#### 7.1. Financial Condition Covenants.

- (a) <u>Maintenance of Consolidated Net Worth.</u> The Borrower shall not permit its Consolidated Net Worth, as of the end of any fiscal quarter, commencing with the first fiscal quarter ending after the Closing Date, to be less than an amount equal to:
  - (i) sixty-five percent (65%) of Consolidated Net Worth of the Borrower as at the fiscal quarter ended June 30, 2011 (provided that if any of OneBeacon Limited, White Mountains Re or any other entity in which the Borrower holds a non-controlling interest as of June 30, 2011 is not required to be consolidated on the consolidated balance sheet of the Borrower in accordance with GAAP as of the end of any such fiscal quarter after the fiscal quarter ended June 30, 2011, the amount in this clause (i) shall be adjusted to remove sixty-five percent (65%) of the Borrower's non-controlling interests related to OneBeacon Limited, White Mountains Re or such other entity, as applicable, as at the fiscal quarter ended June 30, 2011), plus
    - (ii) fifty percent (50%) of positive Consolidated Net Income for each fiscal quarter ended following the Closing Date, minus
    - (iii) the sum of:
    - (A) sixty-five percent (65%) of the sum (not to exceed \$300,000,000) of the following (hereinafter, "<u>Specified Redemptions/Distributions</u>"): (1) the aggregate consideration paid by the

67

Borrower for repurchases of or exchanges for its Capital Stock, <u>plus</u> (2) special dividends paid by the Borrower in respect of its Capital Stock, <u>plus</u> (3) to the extent OneBeacon Limited is required to be consolidated on the consolidated balance sheet of the Borrower in accordance with GAAP, the aggregate consideration paid by OneBeacon Limited for repurchases of or exchanges for its Capital Stock (which Capital Stock is not held by the Borrower or any Subsidiary of the Borrower), <u>plus</u> (4) to the extent OneBeacon Limited is required to be consolidated on the consolidated balance sheet of the Borrower in accordance with GAAP, special dividends paid by OneBeacon Limited in respect of its Capital Stock (which Capital Stock is not held by the Borrower or any Subsidiary of the Borrower), <u>plus</u> (5) to the extent White Mountains Re is required to be consolidated on the consolidated balance sheet of the Borrower in accordance with GAAP, the aggregate consideration paid by White Mountains Re for repurchases of or exchanges for its Preference Shares (which Preference Shares are not held by the Borrower or any Subsidiary of the Borrower), in each case of this clause (A), since the fiscal quarter ended June 30, 2011, <u>plus</u>

- (B) with respect to any Specified Redemptions/Distributions in excess of \$300,000,000, 50% of the sum (not to exceed the amount of net after tax gain realized in connection with the Esurance Sale) of such Specified Redemptions/Distributions, <u>plus</u>
- (C) with respect to any Specified Redemptions/Distributions in excess of the sum of (i) \$300,000,000 <u>plus</u> (ii) the amount of net after tax gain realized in connection with the Esurance Sale, 65% of the sum (not to exceed the Borrower's total common shareholders' equity allocated to the Esurance Segment (consistent with the method of calculation at March 31, 2011) as of the most recent fiscal quarter ended prior to the closing of the Esurance Sale) of such Specified Redemptions/Distributions made by the Borrower.

Notwithstanding the foregoing, when the Borrower adopts Accounting Standards Update ("<u>ASU</u>") 2010-26, Accounting for Costs Associated with Acquiring or Renewing Insurance Contracts, a consensus of the FASB Emerging Issues Task Force (or any successor or similar accounting pronouncement, regardless of its title), which results in a write-off of deferred acquisition costs for the Borrower (the "<u>DAC Write-off</u>"), the minimum required Consolidated Net Worth pursuant to this <u>Section 7.1(a)</u>, as at the last day of the fiscal quarter in which the DAC Write-off is taken, shall permanently be reduced by the lesser of (a) 65% of such DAC Write-off, after tax, and (b) \$25,000,000.

(b) <u>Maintenance of Total Consolidated Debt to Total Consolidated Capitalization Ratio</u>. The Borrower shall not permit its Total Consolidated Debt to Total Consolidated Capitalization Ratio, as of the end of any fiscal quarter, commencing with the first fiscal quarter ending after the Closing Date, to exceed thirty-five percent (35%).

68

Notwithstanding the foregoing, in the event there shall be a breach of this financial covenant, at any time that the (i) OneBeacon Notes are guaranteed by the Borrower, (ii) the Borrower no longer consolidates OneBeacon Limited in its consolidated financial statements in accordance with GAAP, and (iii) such breach would not have occurred if the OneBeacon Notes were no longer guaranteed by the Borrower, then such breach shall not result in the occurrence of an Event of Default under Section 8.1(c) until a period of up to 180 days ("Cure Period") has passed since the occurrence of such breach; provided that, during such Cure Period, (x) the Total Consolidated Debt to Total Consolidated Capitalization Ratio shall not exceed thirty-five percent (35%) on a pro forma basis excluding the Indebtedness attributable to the guarantee of the OneBeacon Notes and (y) OneBeacon U.S. Holdings maintains investment grade senior unsecured credit ratings on the OneBeacon Notes from at least two of the following: Moody's, S&P and Fitch.

- 7.2. <u>Limitation on Indebtedness</u>. The Borrower will not permit any of its Subsidiaries to create, incur or assume or suffer to exist any Indebtedness, except:
- (a) Indebtedness outstanding as of the Closing Date and any refinancings, refundings, renewals or extensions thereof (without any increase in the principal amount thereof, other than by the amount of any necessary pre-payment premiums, unpaid accrued interest and other costs of refinancing, or any shortening of the final maturity of any principal amount thereof to a date prior to the Revolving Credit Termination Date);

- (b) Indebtedness of any Insurance Subsidiary incurred or issued in the ordinary course of its business or in securing insurance-related obligations (that do not constitute Indebtedness) of such Insurance Subsidiary and letters of credit, bank guarantees, surety bonds or similar instruments issued for the account of any Insurance Subsidiary in the ordinary course of its business or in securing insurance-related obligations (that do not constitute Indebtedness) of such Insurance Subsidiary;
- (c) Indebtedness in respect of letters of credit, bank guarantees, bids, leases, statutory obligations, surety and appeal bonds, or performance bonds or other obligations of a like nature arising in the ordinary course of business and not for capital raising purposes and issued for the account of any Non-Regulated Operating Subsidiary;
- (d) short-term Indebtedness (i.e. with a maturity of less than one year when issued, provided that such Indebtedness may include an option to extend for up to an additional one year period) of any Insurance Subsidiary incurred or issued to provide short-term liquidity to facilitate claims payment in the event of catastrophe;
- (e) Indebtedness of a Subsidiary acquired after the Closing Date or a corporation merged into or consolidated with a Subsidiary after the Closing Date and Indebtedness assumed in connection with the acquisition of assets, which Indebtedness, in each case, exists at the time of such acquisition, merger or consolidation and is not created in contemplation of such event, as well as any refinancings, refunds, renewals or extensions of such Indebtedness (without increase in the principal amount thereof other than by the amount of any necessary pre-payment premiums, unpaid accrued interest and other costs of refinancing);

69

- (f) Indebtedness owing or issued by a Subsidiary to any other Subsidiary or to the Borrower;
- (g) Guarantee Obligations made by the Guarantors in respect of the obligations of the Borrower or by a Subsidiary in respect of obligations of another Subsidiary;
  - (h) Indebtedness under the Loan Documents;
  - (i) Indebtedness of any Guarantor;
- (j) Indebtedness represented by Qualified Securities, Trust Preferred Securities or Mandatory Convertible Securities (except to the extent such Indebtedness is included in the calculation of Total Consolidated Debt);
- (k) to the extent constituting Indebtedness, liabilities representing collateral held with respect to securities lending activities and not exceeding 10% of the Borrower's consolidated investment assets as of the end of the most recent fiscal quarter for which consolidated financial statements have been furnished pursuant to Section 6.1(a); and
- (l) other Indebtedness of such Subsidiaries, <u>provided</u> that at the time such Indebtedness is incurred or issued, the aggregate principal amount of such Indebtedness when added to all other Indebtedness incurred or issued pursuant to this <u>clause (l)</u> and then outstanding, does not exceed 15% of the Consolidated Net Worth of the Borrower.
- 7.3. <u>Limitation on Liens</u>. The Borrower will not, and will not permit any of its Subsidiaries to, create, incur, assume or suffer to exist (a) any Lien upon any stock or Indebtedness of any Subsidiary, whether owned on the date of this Agreement or hereafter acquired, to secure any Debt of the Borrower or any of its Subsidiaries or any other Person (other than the obligations hereunder) or (b) any Lien upon any other Property of the Borrower or any of its Subsidiaries, whether owned or leased on the date of this Agreement, or thereafter acquired, to secure any Debt of the Borrower or any of its Subsidiaries or any other Person (other than the obligations hereunder), except:
  - (i) (x) any Lien existing as of the Closing Date or (y) any Lien upon any stock or Indebtedness or other Property of any Person existing at the time such Person becomes a Subsidiary or existing upon stock or Indebtedness of a Subsidiary or any other Property at the time of acquisition of such stock or Indebtedness or other Property (provided that such Lien was not created in connection with such Person becoming a Subsidiary or the acquisition of such stock or Indebtedness or other Property), and any extension, renewal or replacement (or successive extensions, renewals or replacements) in whole or in part of any such Lien in clauses (x) or (y) above; provided, however, that the principal amount of Debt secured by such Lien shall not exceed the principal amount of Debt so secured at the time of such extension, renewal or replacement; and provided, further, that such Lien shall be limited to all or such part of the stock or Indebtedness or other Property which secured the Lien so extended, renewed or replaced;

- (ii) any Permitted Liens; and
- (iii) any Lien upon any Property if the aggregate amount of all Debt then outstanding secured by such Lien and all other Liens permitted pursuant to this <u>clause (iii)</u> does not exceed 10% of the total consolidated shareholders' equity (including preferred equity) of the Borrower as shown on the audited consolidated balance sheet contained in the latest annual report to shareholders of the Borrower; <u>provided</u> that Debt secured by Liens permitted by <u>clauses (i)</u> and <u>(ii)</u> shall not be included in the amount of such secured Debt.
- 7.4. <u>Limitation on Changes in Fiscal Periods</u>. The Borrower shall not permit its fiscal year to end on a day other than December 31 or change its method of determining fiscal quarters.
- 7.5. <u>Limitation on Lines of Business</u>. The Borrower shall not engage to any extent that is material for the Borrower and its Subsidiaries, taken as a whole, in any business, either directly or through any Subsidiary, other than a Principal Business.

7.6. <u>Guarantors</u>. The Borrower shall not permit any Guarantor or any Subsidiary of any Guarantor to merge, consolidate, liquidate, wind-up or dissolve itself, or make a transfer of material assets, other than any such transaction or transactions pursuant to which (a) the surviving Person of any such merger or consolidation is, or any assets transferred in connection with such liquidation, winding-up, dissolution or transfer are transferred to, a Guarantor (including a Subsidiary of the Borrower that becomes a Guarantor in connection with such transaction) or a Subsidiary of a Guarantor (including a person that will become a Subsidiary of a Guarantor in connection with such transaction) or (b) after giving effect to any such merger, consolidation, liquidation, winding-up, dissolution or transfer, the aggregate Consolidated Net Worth of all Guarantors (including any Person that becomes a Guarantor or Subsidiary of a Guarantor in connection with such transaction but excluding for purposes of calculating the aggregate Consolidated Net Worth of all Guarantors any outstanding Indebtedness of the Borrower to a Guarantor (to the extent reflected as an asset of such Guarantor in the calculation of such Guarantor's Consolidated Net Worth)) is not less than the amount then required by <u>Section 7.1(a)</u>, on a pro forma basis as of the end of the most recently completed fiscal quarter for which financial statements have been delivered to the Administrative Agent pursuant to <u>Section 6.1(a)</u>.

#### 8. EVENTS OF DEFAULT

- 8.1. Events of Default. If any of the following events shall occur and be continuing:
- (a) The Borrower shall fail to pay any principal of any Loan made to the Borrower or Reimbursement Obligation owing by the Borrower when due in accordance with the terms hereof; or the Borrower shall fail to pay any interest on any Loan made to the Borrower or Reimbursement Obligation owing by the Borrower, or any other amount payable by the Borrower hereunder or under any other Loan Document, within three Business Days after any such interest or other amount becomes due in accordance with the terms hereof; or

71

- (b) The Borrower shall default in the observance or performance of any agreement contained in Section 6.7(a) or Section 7; or
- (c) (i) The Borrower, any Guarantor, White Mountains Re or any of the Borrower's Material Insurance Subsidiaries shall voluntarily commence any case, proceeding or other action (A) under any Debtor Relief Law, (B) seeking appointment of a receiver, trustee, custodian, conservator or other similar official for it or for all or any substantial part of its assets, or the Borrower, any Guarantor, White Mountains Re or any of the Borrower's Material Insurance Subsidiaries shall make a general assignment for the benefit of its creditors; or (ii) there shall be commenced against the Borrower, any Guarantor, White Mountains Re or any of the Borrower's Material Insurance Subsidiaries any case, proceeding or other action under any Debtor Relief Law that (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed, undischarged or unbonded for a period of 60 days; or (iii) the Borrower, any Guarantor, White Mountains Re or any of the Borrower's Material Insurance Subsidiaries shall take any corporate action to authorize or effect any of the acts set forth in clause (i), or (ii), above; or (iv) the Borrower, any Guarantor, White Mountains Re or any of the Borrower's Material Insurance Subsidiaries shall generally not, or shall be unable to, or shall admit in writing its inability to, pay its debts as they become due; or
  - (d) A Change of Control; or
  - (e) A Fundamental Change; or
- (f) Any representation or warranty made or deemed made by the Borrower herein or in any other Loan Document or that is contained in any certificate, document or financial or other statement furnished by it at any time under or in connection with this Agreement or any such other Loan Document shall prove to have been inaccurate in any material respect on or as of the date made or deemed made or furnished; or
- (g) The Borrower shall default in the observance or performance of any other agreement, covenant, term or condition contained in this Agreement or any other Loan Document (not specified in <u>Sections 8.1(a)</u>, <u>8.1(b)</u> or <u>8.1(f)</u>); or
- (h) The Borrower or any of its Subsidiaries shall (i) default in making any payment of any principal of any Indebtedness (including, without limitation, any Guarantee Obligation, but excluding the Loans and Reimbursement Obligations) on the scheduled or original due date with respect thereto (after giving effect to any applicable grace periods); or (ii) default in making any payment of any interest on any such Indebtedness beyond the period of grace, if any, provided in the instrument or agreement under which such Indebtedness was created; or (iii) default in the observance or performance of any other agreement or condition relating to any such Indebtedness or contained in any instrument or agreement evidencing, securing or relating thereto, the effect of which default is to cause, or to permit the holder or beneficiary of such Indebtedness (or a trustee or agent on behalf of such holder or beneficiary) to cause, with the giving of notice if required, such Indebtedness to become due prior to its stated maturity or to become subject to a mandatory offer to purchase by the obligor thereunder as a result of the occurrence of such default thereunder or (in the case of any such Indebtedness constituting a Guarantee Obligation) to become payable; provided, that a default described in

72

<u>clause (i), (ii)</u> or (<u>iii)</u> of this <u>paragraph (h)</u> shall not at any time constitute an Event of Default unless, at such time, one or more defaults of the type described in <u>clauses (i), (ii)</u> and (<u>iii)</u> of this <u>paragraph (h)</u> shall have occurred and be continuing with respect to Indebtedness the outstanding principal amount of which exceeds in the aggregate \$75,000,000; or

- (i) An ERISA Event occurs with respect to a Pension Plan or Multiemployer Plan which has resulted or could reasonably be expected to result in liability of the Borrower under Title IV of ERISA to the Pension Plan, Multiemployer Plan or the PBGC in an aggregate amount in excess of \$75,000,000, or (ii) the Borrower or any ERISA Affiliate fails to pay when due, after the expiration of any applicable grace period, any installment payment with respect to its withdrawal liability under Section 4201 of ERISA under a Multiemployer Plan in an aggregate amount in excess of \$75,000,000; or
- (j) One or more judgments or decrees shall be entered against the Borrower or any of its Subsidiaries involving for the Borrower and its Subsidiaries taken as a whole a liability (to the extent not paid or fully covered by insurance above applicable deductions) of \$75,000,000 or more, and shall not have been vacated, discharged, stayed or bonded pending appeal within 45 days from the entry thereof; or

- (k) The Guaranty shall cease, for any reason (other than as provided in <u>Section 10.17</u>) to be in full force and effect or any Guarantor or any Affiliate of a Guarantor shall so assert in writing; or
- (l) Any License of any Insurance Subsidiary (i) shall be revoked by the Governmental Authority which issued such License, or any action (administrative or judicial) to revoke such License shall have been commenced against such Insurance Subsidiary and shall not have been dismissed within thirty days after the commencement thereof, (ii) shall be suspended by such Governmental Authority for a period in excess of thirty days or (iii) shall not be reissued or renewed by such Governmental Authority upon the expiration thereof following application for such reissuance or renewal of such Insurance Subsidiary, which, in the case of each of clauses (i), (ii) and (iii) above, could reasonably be expected to have a Material Adverse Effect.

Notwithstanding the foregoing, in the case of each of paragraphs (f) through (l) of this <u>Section 8.1</u>, such event shall not constitute an Event of Default unless such event continues unremedied for a period of 30 days after the Borrower shall have received written notice of such event from the Administrative Agent or the Majority Lenders.

8.2. Remedies Upon Event of Default. If any Event of Default specified in Section 8.1 occurs and is continuing, then, and in any such event, (a) if such event is an Event of Default specified in clause (i) or (ii) of Section 8.1(c) above with respect to the Borrower, automatically the commitment of each Lender to make Loans and any obligation of the Issuing Lender to make L/C Credit Extensions shall immediately terminate and the Loans hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the other Loan Documents (including, without limitation, all amounts of L/C Obligations, whether or not the beneficiaries of the then outstanding Letters of Credit shall have presented the documents required thereunder) shall immediately become due and payable, and (b) if such event is any other Event of Default specified in Section 8.1, either or both of the following actions may be

73

taken: (i) with the consent of the Majority Lenders, the Administrative Agent may, or upon the request of the Majority Lenders, the Administrative Agent shall, by notice to the Borrower declare the Revolving Credit Commitments and the obligation of the Issuing Lender to issue Letters of Credit to be terminated forthwith, whereupon the Revolving Credit Commitments and the L/C Commitment shall immediately terminate; and (ii) with the consent of the Majority Lenders, the Administrative Agent shall, by notice to the Borrower, declare the Loans hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the other Loan Documents (including, without limitation, all amounts of L/C Obligations, whether or not the beneficiaries of the then outstanding Letters of Credit shall have presented the documents required thereunder) to be due and payable forthwith, whereupon the same shall immediately become due and payable. In the case of any Letter of Credit issued for the account of the Borrower with respect to which presentment for honor shall not have occurred at the time of an acceleration pursuant to this paragraph, the Borrower shall at such time Cash Collateralize such L/C Obligations in an amount equal to the aggregate then undrawn and unexpired amount of such Letters of Credit. Such Cash Collateral shall be applied by the Administrative Agent to the payment of drafts drawn under such Letters of Credit, and the unused portion thereof after all such Letters of Credit shall have expired or been fully drawn upon, if any, shall be applied to repay other obligations of the Borrower hereunder and under the other Loan Documents shall have been paid in full or (b) all Defaults and Events of Default hereunder and under the other Loan Document shall have been cured or waived, the balance, if any, in such Cash Collateral account shall be returned to the Borrower (or such other Person as may be lawfully entitled thereto).

#### 9. THE ADMINISTRATIVE AGENT

- 9.1. <u>Appointment</u>. (a) Each of the Lenders and Issuing Lenders hereby irrevocably appoints Bank of America to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. Except for the Borrower's rights expressly set forth in <u>Section 9.7</u>, the provisions of this Article are solely for the benefit of the Administrative Agent, the Lenders and the Issuing Lenders, and the Borrower shall not have rights as a third party beneficiary of any of such provisions.
- (b) The Administrative Agent shall also act as the "collateral agent" under the Loan Documents, and each of the Lenders and the Issuing Lenders hereby irrevocably appoints and authorizes the Administrative Agent to act as the agent of such Lender and such Issuing Lender for purposes of acquiring, holding and enforcing any and all Liens on collateral granted by any of the Borrower or any Guarantor to secure any Loan or other obligation under this Agreement, together with such powers and discretion as are reasonably incidental thereto. In this connection, the Administrative Agent, as "collateral agent" and any co-agents, sub-agents and attorneys-in-fact appointed by the Administrative Agent pursuant to <u>Section 9.2</u> for purposes of holding or enforcing any Lien on any collateral (or any portion thereof) granted under the Loan

74

Documents, or for exercising any rights and remedies thereunder at the direction of the Administrative Agent), shall be entitled to the benefits of all provisions of this  $\underline{\text{Article IX}}$  and  $\underline{\text{Article X}}$  (including  $\underline{\text{Section 10.6(c)}}$ , as though such co-agents, sub-agents and attorneys-in-fact were the "collateral agent" under the Loan Documents) as if set forth in full herein with respect thereto.

- 9.2. <u>Delegation of Duties</u>. The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Persons. The exculpatory provisions of this Article shall apply to any such sub-agent and to the Related Persons of the Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.
- 9.3. <u>Exculpatory Provisions</u>. The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:
  - (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;

- (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Majority Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), provided that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable law; and
- (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.
- (d) The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Majority Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in Sections 10.1 and 8.2) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default unless and until notice describing such Default is given to the Administrative Agent by the Borrower, a Lender or an Issuing Lender.
- (e) The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection

75

with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or the creation, perfection or priority of any Lien purported to be created by the Loan Documents, (v) the value or the sufficiency of any collateral, or (vi) the satisfaction of any condition set forth in Article IV or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

- 9.4. Reliance by Administrative Agent. (a) The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, or the issuance of a Letter of Credit, that by its terms must be fulfilled to the satisfaction of a Lender or an Issuing Lender, the Administrative Agent may presume that such condition is satisfactory to such Lender or such Issuing Lender unless the Administrative Agent shall have received notice to the contrary from such Lender or such Issuing Lender prior to the making of such Loan or the issuance of such Letter of Credit. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.
- (b) For purposes of determining compliance with the conditions specified in Section 4.1, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.
- 9.5. Non-Reliance on Administrative Agent and Other Lenders. Each Lender and the each Issuing Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Persons and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender and each Issuing Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Persons and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.
- 9.6. <u>Administrative Agent in its Individual Capacity</u>. The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a

76

Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.

9.7. Successor Administrative Agent. The Administrative Agent may at any time give notice of its resignation to the Lenders, the Issuing Lenders and the Borrower (which resignation shall be effective upon the earlier to occur of (x) the appointment of a successor Administrative Agent and (y) 30 days after the date such notice is received by the Borrower). Upon receipt of any such notice of resignation, the Majority Lenders shall have the right, with the consent of the Borrower at all times other than during the continuance of a Specified Event of Default (which consent of the Borrower shall not be unreasonably withheld or delayed), to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank with an office in the United States. If no such successor shall have been so appointed by the Majority Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders and the Issuing Lenders, appoint a successor Administrative Agent meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (a) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan

Documents (except that in the case of any collateral security held by the Administrative Agent on behalf of the Lenders or any Issuing Lender under any of the Loan Documents, the retiring Administrative Agent shall continue to hold such collateral security until such time as a successor Administrative Agent is appointed) and (b) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender and the Issuing Lenders directly, until such time as the Majority Lenders appoint a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Sections 10.5 and 10.6 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Persons in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

Any resignation by Bank of America as Administrative Agent pursuant to this Section shall also constitute its resignation as an Issuing Lender and Swing Line Lender. Upon

77

the acceptance of a successor's appointment as Administrative Agent hereunder, (i) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Issuing Lender and Swing Line Lender, (ii) the retiring Issuing Lender and Swing Line Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (iii) the successor Issuing Lender shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring Issuing lender to effectively assume the obligations of the retiring Issuing Lender with respect to such Letters of Credit.

- 9.8. <u>Administrative Agent May File Proofs of Claim</u>. In case of the pendency of any proceeding under any Debtor Relief Law or any other judicial proceeding relative to the Borrower or any Guarantor, the Administrative Agent (irrespective of whether the principal of any Loan or L/C Obligation shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise:
- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, L/C Obligations and all other obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders, the Issuing Lenders and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders, the Issuing Lenders and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders, the Issuing Lenders and the Administrative Agent under Sections 2.6, 3.8, 10.5, and 10.6) allowed in such judicial proceeding; and
  - (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender and each Issuing Lender to make such payments to the Administrative Agent and, if the Administrative Agent shall consent to the making of such payments directly to the Lenders and the Issuing Lenders, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.6(c), 10.5 and 10.6.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender or any Issuing Lender any plan of reorganization, arrangement, adjustment or composition affecting the obligations of the Borrower hereunder or the rights of any Lender or any Issuing Lender to authorize the Administrative Agent to vote in respect of the claim of any Lender or any Issuing Lender or in any such proceeding.

9.9. Guarantee and Collateral Matters. The Lenders irrevocably authorize the Administrative Agent, at its option and in its discretion,

78

- (a) to release any Lien on any property granted to or held by the Administrative Agent under any Loan Document (i) upon termination of the Total Revolving Credit Commitments and payment in full of all obligations of the Borrower hereunder or under any of the other Loan Documents (other than contingent indemnification obligations) and the expiration or termination of all Letters of Credit, (ii) that is sold or to be sold as part of or in connection with any sale permitted hereunder or under any other Loan Document, or (iii) subject to Section 10.1, if approved, authorized or ratified in writing by the Majority Lenders; and
- (b) to subordinate any Lien on any property granted to or held by the Administrative Agent under any Loan Document to the holder of any Lien on such property that is permitted by <u>Section 7.3</u>, and
  - (c) to effect any release of Guarantee Obligations contemplated by Section 10.17.

Upon request by the Administrative Agent at any time, the Majority Lenders will confirm in writing the Administrative Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Guarantor from its obligations under the Guaranty pursuant to this Section 9.9. In each case as specified in this Section 9.9, the Administrative Agent will, at the Borrower's expense, execute and deliver such documents as the Borrower may reasonably request to evidence the release of such item of collateral from the assignment and security interest granted under any Loan Document or to subordinate its interest in such item, or to release such Guarantor from its obligations under the Guaranty, in each case in accordance with the terms of the Loan Documents and this Section 9.9.

9.10. Other Agents; Arrangers and Managers. None of the Lenders or other Persons identified on the facing page or signature pages of this Agreement as a "syndication agent," "documentation agent," "co-agent," "book manager," "lead manager," "arranger," "lead arranger" or "co-arranger" shall

have any right, power, obligation, liability, responsibility or duty under this Agreement other than, in the case of such Lenders, those applicable to all Lenders as such. Without limiting the foregoing, none of the Lenders or other Persons so identified shall have or be deemed to have any fiduciary relationship with any Lender. Each Lender acknowledges that it has not relied, and will not rely, on any of the Lenders or other Persons so identified in deciding to enter into this Agreement or in taking or not taking action hereunder.

#### 10. MISCELLANEOUS

10.1. <u>Amendments, Etc.</u> No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by the Borrower therefrom, shall be effective unless in writing signed by the Majority Lenders and the Borrower and delivered to the Administrative Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; <u>provided, however</u>, that no such amendment, waiver or consent shall:

(a) extend the expiration date of or increase the Revolving Credit Commitment of any Lender (or reinstate any Revolving Credit Commitment terminated pursuant to Section 8.2) without the written consent of such Lender;

79

- (b) postpone any date fixed by this Agreement or any other Loan Document for any payment (excluding mandatory prepayments) of principal, interest or fees payable hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby;
- (c) reduce the principal of, or the rate of interest specified herein on, any Loan or L/C Borrowing, or, subject to <u>clause (v)</u> of the second proviso to this <u>Section 10.1</u>, any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby; <u>provided</u>, <u>however</u>, that only the consent of the Majority Lenders shall be necessary to amend the definition of "Default Rate" or to waive any obligation of the Borrower to pay interest or Letter of Credit Fees at the Default Rate;
- (d) change Section 2.14 in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender directly affected thereby; or
- (e) change any provision of this <u>Section 10.1</u> or the percentage in the definition of "Majority Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender;
  - (f) amend, modify or waive any provision of Section 2.3 or 2.4 without the written consent of the Swing Line Lender;
  - (g) amend, modify or waive any provision of Section 3 without the consent of the Issuing Lender;
- (h) amend, modify or waive the provisions of the definition of Interest Period regarding nine or twelve month Interest Periods for Eurodollar Loans without the consent of each relevant Lender;
- (i) consent to the assignment or transfer by the Borrower of any of its rights and obligations under this Agreement and the other Loan Documents without the written consent of each Lender; or
- (j) release any Guaranter from its Guarantee Obligations under the Guaranty except as provided in <u>Section 10.17</u>, without the consent of all Lenders;

and, <u>provided further</u>, that (i) no amendment, waiver or consent shall, unless in writing and signed by the Issuing Lender in addition to the Lenders required above, modify the rights or duties of the Issuing Lender under this Agreement or any Application relating to any Letter of Credit issued or to be issued by it; (ii) no amendment, waiver or consent shall, unless in writing and signed by the Swing Line Lender in addition to the Lenders required above, modify the rights or duties of the Swing Line Lender under this Agreement; (iii) no amendment, waiver or consent shall, unless in writing and signed by the Administrative Agent in addition to the Lenders required above, modify the rights or duties of the Administrative Agent under this Agreement or any other Loan Document; and (iv) Section 10.7(h) may not be amended, waived

80

or otherwise modified without the consent of each Granting Lender all or any part of whose Loans are being funded by an SPC at the time of such amendment, waiver or other modification; and (v) the Fee Letters may be amended, or rights or privileges thereunder waived, in a writing executed only by the parties thereto. Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder, except that the Revolving Credit Commitment of such Lender may not be increased or extended without the consent of such Lender.

Any such waiver and any such amendment, supplement or modification shall apply equally to each of the Lenders and shall be binding upon the Borrower, the Lenders, the Administrative Agent and all future holders of the Loans. In the case of any waiver, the Borrower, the Lenders and the Administrative Agent shall be restored to their former position and rights hereunder and under the other Loan Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing; but no such waiver shall extend to any subsequent or other Default or Event of Default, or impair any right consequent thereon. Any such waiver, amendment, supplement or modification shall be effected by a written instrument signed by the parties required to sign pursuant to the foregoing provisions of this Section 10.1; provided, that delivery of an executed signature page of any such instrument by facsimile transmission shall be effective as delivery of a manually executed counterpart thereof.

For the avoidance of doubt, this Agreement may be amended (or amended and restated) with the written consent of the Majority Lenders, the Administrative Agent and the Borrower party to each relevant Loan Document (x) to add one or more additional credit facilities to this Agreement and to permit the extensions of credit from time to time outstanding thereunder and the accrued interest and fees in respect thereof to share ratably in the benefits of

this Agreement and the other Loan Documents with the Loans, the L/C Obligations and the accrued interest and fees in respect thereof and (y) to include appropriately the Lenders holding such credit facilities in any determination of the Majority Lenders.

- 10.2. Notices; Effectiveness; Electronic Communication.
- (a) <u>Notices Generally.</u> Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in <u>subsection (b)</u> below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail, sent by telecopier or by electronic mail as follows, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:
  - (i) if to the Borrower, the Administrative Agent, the Issuing Lender or the Swing Line Lender, to the address, telecopier number, electronic mail address or telephone number specified for such Person on <u>Schedule 10.2</u>; and
  - (ii) if to any other Lender, to the address, telecopier number, electronic mail address or telephone number specified in its Administrative Questionnaire.

81

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by telecopier shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through electronic communications to the extent provided in <u>subsection (b)</u> below, shall be effective as provided in such <u>subsection (b)</u>.

(b) <u>Electronic Communications</u>. Notices and other communications to the Lenders and the Issuing Lender hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent, <u>provided</u> that the foregoing shall not apply to service of process or to notices to any Lender or the Issuing Lender pursuant to <u>Section 2</u>. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, <u>provided</u> that approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), <u>provided</u> that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing <u>clause (i)</u> of notification that such notice or communication is available and identifying the website address therefor.

(c) The Platform. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT-RELATED PERSONS DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT-RELATED PERSONS OR THE BORROWER IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Agent-Related Persons or the Borrower have any liability to any Agent-Related Person, the Borrower, any Lender, or the Issuing Lender for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent-Related Persons or the Borrower; provided, however, that in no event shall any Agent-Related Persons or the Borrower have any liability to any Agent-Related Person, the Borrower, any Lender, or the Issuing Lender

82

for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages). Each Lender agrees that the Borrower shall be responsible only for the Borrower Materials and shall not have any liability (unless otherwise agreed in writing by the Borrower) for any other materials made available to the Lenders and shall not have any liability for any errors or omissions other than errors or omissions in the materials delivered to the Administrative Agent by the Borrower. Nothing in this Section 10.2(c) shall limit the obligation of the Administrative Agent and the Lenders under Section 10.16.

- (d) <u>Change of Address, Etc.</u> The Borrower, the Administrative Agent, the Issuing Lender and the Swing Line Lender may change its address, telecopier or telephone number for notices and other communications hereunder by notice to the other parties hereto. Each other Lender may change its address, telecopier or telephone number for notices and other communications hereunder by notice to the Borrower, the Administrative Agent, the Issuing Lender and the Swing Line Lender. In addition, each Lender agrees to notify the Administrative Agent from time to time to ensure that the Administrative Agent has on record (i) an effective address, contact name, telephone number, telecopier number and electronic mail address to which notices and other communications may be sent and (ii) accurate wire instructions for such Lender.
- (e) Reliance by Administrative Agent, Issuing Lender and Lenders. The Administrative Agent, the Issuing Lender and the Lenders shall be entitled to rely and act upon any notices (including telephonic and written Borrowing Requests and notices of Swing Line Loans) purportedly given by or on behalf of the Borrower; provided that the foregoing shall not apply to losses, costs, expenses and liabilities caused by the gross negligence or willful misconduct of the relevant Lender or any Agent-Related Person even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Borrower shall indemnify the Administrative Agent, the Issuing Lender, each Lender and the Agent-Related Persons from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Borrower; provided that the foregoing shall not apply to losses, costs, expenses and liabilities caused by the gross negligence or willful misconduct of the relevant Lender or any Agent-

Related Person. All telephonic notices to and other telephonic communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

(f) <u>Effectiveness of Facsimile Documents and Signatures</u>. Loan Documents may be transmitted and/or signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable Law, have the same force and effect as manually-signed originals and shall be binding on the Borrower, the Administrative Agent and the Lenders. The Administrative Agent may also require that any such documents and signatures be confirmed by a manually-signed original thereof; <u>provided</u>, <u>however</u>, that the failure to request or deliver the same shall not limit the effectiveness of any facsimile document or signature.

10.3. No Waiver; Cumulative Remedies. No failure to exercise and no delay in exercising, on the part of the Administrative Agent or any Lender, any right, remedy, power or privilege hereunder or under the other Loan Documents shall operate as a waiver thereof; nor

83

shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by Law.

Notwithstanding anything to the contrary contained herein or in any other Loan Document, the authority to enforce rights and remedies hereunder and under the other Loan Documents against the Borrower, the Subsidiaries of the Borrower or any of them shall be vested exclusively in, and all actions and proceedings at law in connection with such enforcement shall be instituted and maintained exclusively by, the Administrative Agent in accordance with Section 8.2 for the benefit of all the Lenders and the Issuing Lenders; provided, however, that the foregoing shall not prohibit (a) the Administrative Agent from exercising on its own behalf the rights and remedies that inure to its benefit (solely in its capacity as Administrative Agent) hereunder and under the other Loan Documents, (b) an Issuing Lender or the Swing Line Lender from exercising the rights and remedies that inure to its benefit (solely in its capacity as an Issuing Lender or Swing Line Lender, as the case may be) hereunder and under the other Loan Documents, (c) any Lender from exercising setoff rights in accordance with Section 10.8(b) (subject to the terms of Section 10.8(a)), or (d) any Lender from filing proofs of claim or appearing and filing pleadings on its own behalf during the pendency of a proceeding relative to the Borrower or any Guarantor under any Debtor Relief Law; and provided, further, that if at any time there is no Person acting as Administrative Agent hereunder and under the other Loan Documents, then (i) the Majority Lenders shall have the rights otherwise ascribed to the Administrative Agent pursuant to Section 8.2 and (ii) in addition to the matters set forth in clauses (b), (c) and (d) of the preceding proviso and subject to Section 10.8(a), any Lender may, with the consent of the Majority Lenders, enforce any rights and remedies available to it and as authorized by the Majority Lenders.

10.4. <u>Survival of Representations and Warranties</u>. All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent and each Lender, regardless of any investigation made by the Administrative Agent or any Lender or on their behalf and notwithstanding that the Administrative Agent or any Lender may have had notice or knowledge of any Default at the time of any extension of credit, and shall continue in full force and effect as long as any Loan or any other obligation hereunder shall remain unpaid or unsatisfied or any Letter of Credit shall remain outstanding.

10.5. Attorney Costs and Expenses. The Borrower agrees (a) to pay or reimburse the Administrative Agent and the Arrangers for all reasonable out-of-pocket costs and expenses incurred in connection with the development, preparation, negotiation and execution of this Agreement and the other Loan Documents and any amendment, waiver, consent or other modification of the provisions hereof and thereof (whether or not the transactions contemplated hereby are consummated), and the consummation and administration of the transactions contemplated hereby and thereby, including all Attorney Costs, and (b) to pay or reimburse the Administrative Agent and each Lender for all reasonable out-of-pocket costs and expenses (which may include, to the extent reasonably incurred, all search, filing, recording, title

84

insurance and appraisal charges and fees and taxes related thereto, and other out-of-pocket expenses incurred by the Administrative Agent and the cost of independent public accountants and other outside experts) incurred in connection with the enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Loan Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the obligations of the Borrower hereunder or under any of the other Loan Documents and during any legal proceeding, including any proceeding under any Debtor Relief Law), including all Attorney Costs. All amounts due under this Section 10.5 shall be payable not later than 30 days following written demand. The agreements in this Section 10.5 shall survive the termination of the Total Revolving Credit Commitments and repayment of all other obligations.

10.6. Indemnification. (a) Whether or not the transactions contemplated hereby are consummated, the Borrower shall indemnify and hold harmless each Agent-Related Person, each Arranger, each Lender and their respective Affiliates, directors, officers, employees, counsel, agents, shareholders and attorneys-in-fact (collectively the "Indemnitees") from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, settlement payments and causes of action of any kind or nature whatsoever and reasonable related out-of-pocket costs and expenses which may at any time be imposed on, incurred, suffered, sustained, required to be paid by or asserted against any such Indemnitee in any way relating to or arising out of or in connection with (a) the execution, delivery, enforcement, performance or administration of any Loan Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby, (b) any Revolving Credit Commitment, Loan or Letter of Credit or the use or proposed use of the proceeds therefrom (including any refusal by the Issuing Lender to honor a demand for payment under a Letter of Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Letter of Credit), (c) any actual or alleged presence or release of Hazardous Materials on or from any property currently or formerly owned or operated by the Borrower or any Subsidiary, or any Environmental Liability related in any way to the Borrower or any of its Subsidiaries, or (d) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory (including any investigation of, preparation for, or defense of any pending or threatened claim, investigation, litigation or proceeding) and regardless of whether any Indemnitee is a party thereto (all the foregoing, collectively, the "Indemnified Liabilities"), IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY OR SOLE NEGLIGENCE OF THE INDEMNITEE; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such liabilities, obligations, losses, damages, penalties, claims,

- (b) To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under Section 10.5 and Section 10.6(a) to be paid by it to the Administrative Agent (or any sub-agent thereof), the Issuing Lender or any Agent-Related Person of any of the foregoing, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), the Issuing Lender or such Agent-Related Person, as the case may be, such Lender's Revolving Credit Percentage (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought) of such unpaid amount, provided that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) or the Issuing Lender in its capacity as such, or against any Agent-Related Person of any of the foregoing acting for the Administrative Agent (or any such sub-agent) or Issuing Lender in connection with such capacity. The obligations of the Lenders under this Section 10.6(b) are subject to the provisions of Section 2.14(f).
- (c) To the fullest extent permitted by applicable law, the Borrower shall not assert, and hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Loan or Letter of Credit or the use of the proceeds thereof. No Indemnitee referred to in subsection (a) above shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to such unintended recipients by such Indemnitee through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby other than for direct or actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final and nonappealable judgment of a court of competent jurisdiction.
- (d) The agreements in this Section shall survive the resignation of the Administrative Agent, the replacement of any Lender, the termination of the Total Revolving Credit Commitments and the repayment, satisfaction or discharge of all the other obligations.

#### 10.7. Successors and Assigns.

(a) <u>Successors and Assigns Generally.</u> The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of their rights or obligations hereunder without the prior written consent of the Administrative Agent and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of <u>Section 10.7(b)</u>, (ii) by way of participation in accordance with the provisions of <u>Section 10.7(f)</u>, or (iv) to an SPC in accordance with the provisions of <u>Section 10.7(h)</u> (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in

86

<u>Section 10.7(d)</u> and, to the extent expressly contemplated hereby, the Indemnitees) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Revolving Credit Commitment and the Loans (including for purposes of this <u>subsection (b)</u>, participations in L/C Obligations and in Swing Line Loans) at the time owing to it); <u>provided</u> that any such assignment shall be subject to the following conditions:

# (i) Minimum Amounts.

- (A) in the case of an assignment of the entire remaining amount of the assigning Lender's Revolving Credit Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (B) in any case not described in <u>subsection (b)(i)(A)</u> of this <u>Section 10.7</u>, the aggregate amount of the Revolving Credit Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Revolving Credit Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as no Specified Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); <u>provided, however</u>, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met.
- (ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Revolving Credit Commitment assigned, except that this <u>clause (ii)</u> shall not apply to the Swing Line Lender's rights and obligations in respect of Swing Line Loans;

- (iii) <u>Required Consents</u>. No consent shall be required for any assignment except to the extent required by <u>subsection (b)(i)</u> (<u>B</u>) of this <u>Section 10.7</u> and, provided that:
  - (A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (1) a Specified

87

Event of Default has occurred and is continuing at the time of such assignment or (2) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund;

- (B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required if such assignment is to a Person that is not a Lender or an Affiliate of the assigning Lender;
- (C) the consent of each Issuing Lender (such consent not to be unreasonably withheld or delayed) shall be required for any assignment that increases the obligation of the assignee to participate in exposure under one or more Letters of Credit (whether or not then outstanding); and
- (D) the consent of the Swing Line Lender (such consent not to be unreasonably withheld or delayed) shall be required for any assignment.
- (iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee in the amount of \$3,500; <u>provided</u>, <u>however</u>, that the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.
- (v) No Assignment to Certain Persons. No such assignment shall be made (A) to the Borrower or any of the Borrower's Affiliates or Subsidiaries, or (B) to any Defaulting Lender or any of its Subsidiaries or Affiliates, or any Person who, upon becoming a Lender hereunder, would constitute any of the foregoing Persons described in this clause (B), (C) to a natural person, (D) to an Approved Fund unless a Specified Event of Default has occurred and is continuing, or (E) to an Affiliate of a Lender or an Approved Fund unless it is a financial institution having a senior unsecured debt rating of not less than "A-", or its equivalent, by S&P.
- (vi) <u>Certain Additional Payments</u>. In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then

88

owed by such Defaulting Lender to the Administrative Agent or any Lender hereunder (and interest accrued thereon) and (y) acquire (and fund as appropriate) its full pro rata share of all Loans and participations in Letters of Credit and Swing Line Loans in accordance with its Revolving Credit Percentage. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable Law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to Section 10.7(c), from and after the effective date specified in each Assignment and Assumption, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 2.15, 2.16, 2.17, 2.18, 10.5 and 10.6 with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request, the Borrower (at its expense) shall execute and deliver a Note to the assignee Lender and a replacement Note, as applicable, to the assigning Lender.

- (c) Register. The Administrative Agent, acting solely for this purpose as an agent of the Borrower (and such agency being solely for tax purposes), shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Revolving Credit Commitments of, and principal amounts of the Loans and L/C Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, in the absence of manifest error, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. In addition, the Administrative Agent shall maintain on the Register information regarding the designation, and revocation of designation, of any Lender as a Defaulting Lender. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (d) <u>Participations</u>. Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural person, a Defaulting Lender or any Subsidiary or Affiliate of a Defaulting Lender, or the Borrower or any Affiliates or Subsidiaries of the Borrower) (each, a "<u>Participant</u>") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Revolving Credit Commitment and/or the Loans (including such Lender's participations in L/C Obligations and/or Swing Line Loans) owing to it); <u>provided</u> that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely

solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; <u>provided</u> that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to <u>Section 10.1</u> that directly affects such Participant. Subject to <u>Section 10.7(e)</u>, the Borrower agrees that each Participant shall be entitled to the benefits of <u>Sections 2.15</u>, <u>2.16</u> or <u>2.17</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to <u>Section 10.7(b)</u>. To the extent permitted by Law, each Participant also shall be entitled to the benefits of <u>Section 10.8</u> as though it were a Lender, <u>provided</u> such Participant agrees to be subject to <u>Section 2.14</u> as though it were a Lender.

- (e) <u>Limitations upon Participant Rights.</u> A Participant shall not be entitled to receive any greater payment under <u>Sections 2.15, 2.16</u> or <u>2.17</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent. A Participant that would be a Non-U.S. Lender if it were a Lender shall not be entitled to the benefits of <u>Section 2.16</u> unless the Borrower is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Borrower, to comply with <u>Sections 2.16(d)</u> and <u>(e)</u> as though it were a Lender.
- (f) <u>Certain Pledges</u>. Notwithstanding anything to the contrary contained herein, any Lender may, with notice to, but without prior consent of the Borrower and the Administrative Agent, at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank (provided that notice to the Borrower and the Administrative Agent shall not be required in the case of a pledge or assignment to secure obligations to a Federal Reserve Bank); <u>provided</u> further that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute, or permit the substitution of, any such pledgee or assignee for such Lender as a party hereto.
- (g) <u>Electronic Execution of Assignments</u>. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.
- (h) <u>Special Purpose Funding Vehicles</u>. Notwithstanding anything to the contrary contained herein, any Lender (a "<u>Granting Lender</u>") may, with notice to, but without prior consent of the Borrower and the Administrative Agent grant to a special purpose funding

90

vehicle identified as such in writing from time to time by the Granting Lender to the Administrative Agent and the Borrower (an "SPC") the option to provide all or any part of any Loan that such Granting Lender would otherwise be obligated to make pursuant to this Agreement; provided that (i) nothing herein shall constitute a commitment by any SPC to fund any Loan, and (ii) if an SPC elects not to exercise such option or otherwise fails to make all or any part of such Loan, the Granting Lender shall be obligated to make such Loan pursuant to the terms hereof and, if it fails to do so, to make such payment to the Administrative Agent as is required under Section 2.14(e)(ii). Each party hereto hereby agrees that (i) neither the grant to any SPC nor the exercise by any SPC of such option shall increase the costs or expenses or otherwise increase or change the obligations of the Borrower under this Agreement (including its obligations under Section 2.15), (ii) no SPC shall be liable for any indemnity or similar payment obligation under this Agreement for which a Lender would be liable, and (iii) the Granting Lender shall for all purposes, including the approval of any amendment, waiver or other modification of any provision of any Loan Document, remain the lender of record hereunder. The making of a Loan by an SPC hereunder shall utilize the Revolving Credit Commitment of the Granting Lender to the same extent, and as if, such Loan were made by such Granting Lender. In furtherance of the foregoing, each party hereto hereby agrees (which agreement shall survive the termination of this Agreement) that, prior to the date that is one year and one day after the payment in full of all outstanding commercial paper or other senior debt of any SPC, it will not institute against, or join any other Person in instituting against, such SPC any bankruptcy, reorganization, arrangement, insolvency, or liquidation proceeding under any Debtor Relief Laws or any other Laws. Notwithstanding anything to the contrary contained herein, any SPC may (i) with notice to, but without prior consent of the Borrower and the Administrative Agent and with the payment of a processing fee in the amount of \$3,500 (which processing fee may be waived by the Administrative Agent in its sole discretion), assign all or any portion of its right to receive payment with respect to any Loan to the Granting Lender and (ii) disclose on a confidential basis any non-public information relating to its funding of Loans to any rating agency, commercial paper dealer or provider of any surety or guarantee or credit or liquidity enhancement to such SPC.

(i) Resignation as Issuing Lender or Swing Line Lender after Assignment. Notwithstanding anything to the contrary contained herein, if at any time Bank of America assigns all of its Revolving Credit Commitment and Loans pursuant to Section 10.7(b), Bank of America may, (i) upon 30 days' notice to the Borrower and the Lenders, resign as Issuing Lender, so long as a successor Issuing Lender (consented to by the Borrower, such consent not to be unreasonably withheld or delayed) has been appointed and/or (ii) upon 30 days' notice to the Borrower, resign as Swing Line Lender, so long as a successor Swing Line Lender (consented to by the Borrower, such consent not to be unreasonably withheld or delayed) has been appointed. In the event of any such resignation as Issuing Lender or Swing Line Lender, the Borrower shall be entitled to appoint from among the Lenders a successor Issuing Lender or Swing Line Lender, however, that no failure by the Borrower to appoint any such successor shall affect the resignation of Bank of America as Issuing Lender or Swing Line Lender, as the case may be. If Bank of America resigns as Issuing Lender, it shall retain all the rights and obligations of the Issuing Lender hereunder with respect to all Letters of Credit outstanding as of the effective date of its resignation as Issuing Lender and all L/C Obligations with respect thereto (including the right to require the Lenders to make Base Rate Loans or fund risk participations in Unreimbursed Amounts pursuant to Section 3.3). If Bank of America resigns as Swing Line

Lender, it shall retain all the rights of the Swing Line Lender provided for hereunder with respect to Swing Line Loans made by it and outstanding as of the effective date of such resignation, including the right to require the Lenders to make Base Rate Loans or fund risk participations in outstanding Swing Line Loans pursuant to Section 2.4. Upon the appointment of a successor Issuing Lender and/or Swing Line Lender, (a) such successor shall succeed to and become vested with all of the rights and obligations of the retiring Issuing Lender or Swing Line Lender, as the case may be, and (b) the successor Issuing Lender shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to Bank of America to effectively assume the obligations of Bank of America with respect to such Letters of Credit.

10.8. Adjustments; Setoff. (a) If any Lender shall, by exercising any right of setoff or counterclaim or otherwise, obtain payment in respect of (a) any Loans and other amounts due and payable to such Lender hereunder and under the other Loan Documents at such time in excess of its ratable share (according to the proportion of (i) such Loans and other amounts due and payable to such Lender at such time to (ii) the aggregate amount of Loans and other amounts due and payable to all Lenders hereunder and under the other Loan Documents at such time) of payments on account of Loans and other amounts owing (but not due and payable) to such Lender hereunder and under the other Loan Documents at such time in excess of its ratable share (according to the proportion of (i) the amount of such Loans and other amounts owing (but not due and payable) to such Lender at such time to (ii) the aggregate amount of Loans and other amounts owing (but not due and payable) to all Lenders hereunder and under the other Loan Documents at such time to (ii) the aggregate amount of Loans and other amounts owing (but not due and payable) to all Lenders hereunder and under the other Loan Documents at such time obtained by all of the Lenders at such time then the Lender receiving such greater proportion shall (a) notify the Administrative Agent of such fact, and (b) purchase (for cash at face value) participations in the Loans and subparticipations in L/C Obligations and Swing Line Loans of the other Lenders, or make such other adjustments as shall be equitable, so that the benefit of all such payments shall be shared by the Lenders ratably in accordance with the aggregate amount of Loans and other amounts then due and payable to the Lenders or owing (but not due and payable) to the Lenders, as the case may be, provided that:

- (i) if any such participations or subparticipations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations or subparticipations shall be rescinded and the purchase price restored to the extent of such recovery, without interest; and
- (ii) the provisions of this Section shall not be construed to apply to (x) any payment made by or on behalf of the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender), (y) the application of Cash Collateral provided for in Section 2.23, or (z) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans or subparticipations in L/C Obligations or Swing Line Loans to any assignee or

92

participant, other than an assignment to the Borrower or any Subsidiary or Affiliate thereof (as to which the provisions of this Section shall apply).

The Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Borrower rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of the Borrower in the amount of such participation.

(b) In addition to any rights and remedies of the Lenders provided by Law, each Lender shall have the right, without prior notice to the Borrower, any such notice being expressly waived by the Borrower to the extent permitted by applicable Law, upon any amount becoming due and payable by the Borrower hereunder (whether at the stated maturity, by acceleration or otherwise), to set-off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Lender or any branch or agency thereof to or for the credit or the account of the Borrower; provided, that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so setoff shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of Section 2.24 and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the obligations owing to such Defaulting Lender as to which it exercised such right of setoff. Each Lender agrees promptly to notify the Borrower, as the case may be, and the Administrative Agent after any such set-off and application made by such Lender, provided that the failure to give such notice shall not affect the validity of such set-off and application.

10.9. <u>Counterparts</u>. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts, and all of said counterparts taken together shall be deemed to constitute one and the same instrument. Delivery of an executed signature page of this Agreement by facsimile transmission shall be effective as delivery of a manually executed counterpart hereof. A set of the copies of this Agreement signed by all the parties shall be lodged with the Borrower and the Administrative Agent.

10.10. Severability. Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. Without limiting the foregoing provisions of this Section 10.10, if and to the extent that the enforceability of any provisions in this Agreement relating to Defaulting Lenders shall be limited by Debtor Relief Laws, as determined in good faith by the Administrative Agent, any Issuing Lender or the Swing Line Lender, as applicable, then such provisions shall be deemed to be in effect only to the extent not so limited.

undertakings, representations or warranties by the Arrangers, the Administrative Agent, the Syndication Agent or any Lender relative to subject matter hereof not expressly set forth or referred to herein, in the other Loan Documents or in the Fee Letters. The Borrower agrees that its obligations under the Fee Letters shall survive the execution and delivery of this Agreement.

- 10.12. <u>GOVERNING LAW</u>. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK (EXCLUDING THE LAWS APPLICABLE TO CONFLICTS OR CHOICE OF LAW TO THE EXTENT THAT THE APPLICATION OF THE LAW OF ANOTHER JURISDICTION WOULD BE REQUIRED THEREBY).
  - 10.13. SUBMISSION TO JURISDICTION; WAIVERS. THE BORROWER HEREBY IRREVOCABLY AND UNCONDITIONALLY:
- (a) SUBMITS FOR ITSELF AND ITS PROPERTY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS TO WHICH IT IS A PARTY, OR FOR RECOGNITION AND ENFORCEMENT OF ANY JUDGMENT IN RESPECT THEREOF, TO THE NON-EXCLUSIVE GENERAL JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK, THE COURTS OF THE UNITED STATES OF AMERICA FOR THE SOUTHERN DISTRICT OF NEW YORK, AND APPELLATE COURTS FROM ANY THEREOF;
- (b) CONSENTS THAT ANY SUCH ACTION OR PROCEEDING MAY BE BROUGHT IN SUCH COURTS AND WAIVES ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE VENUE OF ANY SUCH ACTION OR PROCEEDING IN ANY SUCH COURT OR THAT SUCH ACTION OR PROCEEDING WAS BROUGHT IN AN INCONVENIENT COURT AND AGREES NOT TO PLEAD OR CLAIM THE SAME;
- (c) AGREES THAT SERVICE OF PROCESS IN ANY SUCH ACTION OR PROCEEDING MAY BE EFFECTED BY MAILING A COPY THEREOF BY REGISTERED OR CERTIFIED MAIL (OR ANY SUBSTANTIALLY SIMILAR FORM OF MAIL), POSTAGE PREPAID, TO THE BORROWER AT ITS ADDRESS SET FORTH IN <u>SECTION 10.2</u> OR AT SUCH OTHER ADDRESS OF WHICH THE ADMINISTRATIVE AGENT SHALL HAVE BEEN NOTIFIED PURSUANT THERETO;
- (d) AGREES THAT NOTHING HEREIN SHALL AFFECT THE RIGHT TO EFFECT SERVICE OF PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR SHALL LIMIT THE RIGHT TO SUE IN ANY OTHER JURISDICTION; AND
- (e) WAIVES, TO THE MAXIMUM EXTENT NOT PROHIBITED BY LAW, ANY RIGHT IT MAY HAVE TO CLAIM OR RECOVER IN ANY LEGAL ACTION

94

OR PROCEEDING REFERRED TO IN THIS SECTION 10.13 ANY SPECIAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES.

10.14. WAIVERS OF JURY TRIAL. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY LOAN DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO ANY LOAN DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION 10.14 WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

10.15. No Advisory or Fiduciary Responsibility. In connection with all aspects of each transaction contemplated hereby (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document), the Borrower acknowledges and agrees, and acknowledges its Subsidiaries' understanding, that: (i) (A) the arranging and other services regarding this Agreement provided by the Administrative Agent and the Arrangers are arm's-length commercial transactions between the Borrower and its Affiliates, on the one hand, and the Administrative Agent and the Arrangers, on the other hand, (B) the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate, and (C) the Borrower is capable of evaluating, and understands and accepts, the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents; (ii) (A) the Administrative Agent and the Arrangers each is and has been acting solely as a principal and, except as expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for the Borrower or any of its Affiliates, or any other Person and (B) neither the Administrative Agent nor any Arranger has any obligation to the Borrower or any of its Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; and (iii) the Administrative Agent and the Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower and its Affiliates, and neither the Administrative Agent nor any Arranger has any obligation to disclose any of such interests to the Borrower or its Affiliates. To the fullest extent permitted by law, the Borrower hereby waives and releases any claims that it may have against the Administrative Agent and any Arranger with respect to any breach or alleged brea

10.16. <u>Confidentiality</u>. Each of the Administrative Agent and the Lenders agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any

any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations, (g) with the consent of the Borrower or (h) to the extent such Information (x) becomes publicly available other than as a result of a breach of this Section 10.16 or (y) becomes available to the Administrative Agent or any Lender on a nonconfidential basis from a source other than the Borrower, provided that the Administrative Agent or any such Lender, as applicable, will notify the Borrower as soon as practical in advance of any proposed disclosure pursuant to clause (c) above, unless such notification shall be prohibited by applicable law or legal process, so that the Borrower may seek a protective order or other appropriate remedy and the Administrative Agent or any such Lender, as applicable, will disclose only that portion of the Information that it is advised by its counsel is legally required or otherwise necessary to disclose. For purposes of this Section 10.16, "Information" means all information received from the Borrower or any of its Subsidiaries relating to the Borrower or any Subsidiary or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower or any Subsidiary, provided that, in the case of information received from the Borrower or any Subsidiary after the Closing Date, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section 10.16 shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

Each of the Administrative Agent, the Lenders and the Issuing Lender acknowledges that (a) the Information may include material non-public information concerning the Borrower or a Subsidiary thereof, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Law, including Federal and state securities Laws.

10.17. Release of Guarantee Obligations. Notwithstanding anything to the contrary contained herein or any other Loan Document (a) when all obligations of the Borrower hereunder and under the other Loan Documents that are guaranteed under the terms of the Guaranty have been paid in full, all Revolving Credit Commitments have terminated, there exist no unpaid Reimbursement Obligations and no Letter of Credit issued for the account of the Borrower shall be outstanding or (b) so long as no Event of Default has occurred and is continuing, at such time as the aggregate outstanding principal amount of Indebtedness of Subsidiaries of OneBeacon Limited that is guaranteed by the Borrower is \$70,000,000 or less (excluding (i) any series of such Indebtedness if all of such series has been defeased (including legal or covenant defeasance) or discharged and (ii) such Indebtedness to the extent it is held by the Borrower or any of its Subsidiaries), upon request of the Borrower, the Administrative Agent

96

shall (without vote or consent of any Lender) take such actions as may be required to release all Guarantee Obligations of each Guarantor under any Loan Document, including, without limitation, its Guaranty. Any release of Guarantee Obligations pursuant to clause (a) above shall be deemed subject to the provision that such Guarantee Obligations shall be reinstated as to the Guarantors at the time of such release, if after such release any portion of any payment in respect of the obligations guaranteed thereby shall be rescinded or must otherwise be restored or returned upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Borrower or such Guarantor, or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or similar officer for, the Borrower or such Guarantor or any substantial part of its property, or otherwise, all as though such payment had not been made. In addition, upon the request of the Borrower, the Administrative Agent shall (without vote or consent of any Lender) so release all Guarantee Obligations of any Guarantor, if such Guarantor is merged or consolidated into another Person and such Guarantor is not the surviving Person of such merger or consolidation, or such Guarantor is liquidated, wound-up or dissolved, or such Guarantor otherwise ceases to be a Subsidiary of the Borrower; provided, however, that (x) the surviving Person of any such merger or consolidation is, or any assets transferred in connection with such liquidation, winding-up or dissolution are transferred to, a Guarantor (including a Subsidiary of the Borrower that becomes a Guarantor in connection with such transaction) or a Subsidiary of a Guarantor (including a Person that will become a Subsidiary of a Guarantor in connection with such transaction) or (y) after giving effect to any such merger, consolidation, liquidation, winding-up, dissolution, or cessation of such Guarantor's status as a Subsidiary of the Borrower, the aggregate Consolidated Net Worth of all other Guarantors (including any Person that becomes a Guarantor or Subsidiary of a Guarantor in connection with such transaction but excluding for purposes of calculating the aggregate Consolidated Net Worth of all Guarantors any outstanding Indebtedness of the Borrower to a Guarantor (to the extent reflected as an asset of such Guarantor in the calculation of such Guarantor's Consolidated Net Worth)) is not less than the amount then required by Section 7.1(a), on a pro forma basis as of the end of the most recently completed fiscal quarter for which financial statements have been delivered to the Administrative Agent pursuant to Section 6.1(a).

10.18. <u>USA PATRIOT Act Notice</u>. Each Lender that is subject to the Act (as hereinafter defined) and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "<u>Act</u>"), it is required to obtain, verify and record information that identifies the Borrower, which information includes the name and address of the Borrower and other information that will allow such Lender or the Administrative Agent, as applicable, to identify the Borrower in accordance with the Act.

## 10.19. Interest Rate Limitation.

(a) Notwithstanding anything to the contrary contained in any Loan Document, if at any time the rate of interest payable under any Loan Document (the "Stated Rate") would exceed the rate of interest permitted to be charged under any applicable Law (the "Maximum Rate"), then for so long as the Maximum Rate would be so exceeded, the rate of interest payable shall be equal to the Maximum Rate; provided that if at any time thereafter, the Stated Rate is less than the Maximum Rate, the Borrower shall, to the extent permitted by applicable Law, continue to pay interest at the Maximum Rate until such time as the total interest

97

received is equal to the total interest which would have been received had the Stated Rate been (but for the operation of this provision) the interest rate payable. Thereafter, the interest rate payable shall be the Stated Rate unless and until the Stated Rate again would exceed the Maximum Rate, in which event this provision shall again apply.

(b) In no event shall the total interest received by a Lender exceed the amount which it could lawfully have received had the interest been calculated for the full term hereof at the Maximum Rate.

10.20. Entire Agreement. THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

[Remainder of Page Left Intentionally B	lank]
---	-------

IN WITNESS WHEREOF, the parties hereto have ca authorized officers as of the day and year first above written.	used this Agreement to be duly executed and delivered by their p	proper and duly
, ,	The Borrower:	
	WHITE MOUNTAINS INSURANCE GROUP, LTD.	
	By:	
	Name: Title:	
	BANK OF AMERICA, N.A., as Administrative Agent	
	By: Name: Title:	
	BANK OF AMERICA, N.A., as a Lender, Issuing Lender and Swing Line Lender	
	By: Name: Title:	
	THC.	
	BARCLAYS BANK PLC, as a Lender	
	Ву:	
	Name: Title:	
TVICTING	C LETTERS OF CREDIT	SCHEDULE 1A
EXISTING	None.	
		SCHEDULE 5
CONSENTS, AUTHOR	IZATIONS, FILINGS AND NOTICES	
	None.	

## ADMINISTRATIVE AGENT'S OFFICE; CERTAIN ADDRESSES FOR NOTICES

## **BORROWER:**

White Mountains Insurance Group, Ltd.

80 South Main Street Hanover, NH 03755

Attention: Robert Seelig, Esq. Telephone: 603-640-2202 Telecopier: 603-643-4592

Electronic Mail: rseelig@whitemountains.com

U.S. Taxpayer Identification Number (White Mountains Insurance Group, Ltd.): 94-2708455

## with a copy to:

Cravath, Swaine and Moore LLP

WorldWide Plaza 825 Eighth Avenue

New York, NY 10019-7475 Attention: James Vardell Telecopier: 212-474-3700

## **ADMINISTRATIVE AGENT:**

#### Administrative Agent's Office

(for payments and Requests for Credit Extensions):

Bank of America, N.A. 2001 Clayton Road Mail Code: CA4-702-02-25 Concord, CA 94520

Attention: Faizan Hafeez Telephone: 925-675-8815 Telecopier: 866-540-7550

Electronic Mail: faizan.hafeez@baml.com

Account No.: 3750836479

Ref: White Mountains Insurance Group

ABA# 026009593

#### **EXHIBIT A-2**

Other Notices as Administrative Agent:

Bank of America, N.A. Agency Management 1455 Market Street, 5<sup>th</sup> Floor Mail Code: CA5-701-05-19 San Francisco, CA 94103 Attention: Aamir Saleem Telephone: 415-436-2769

Telecopier: 415-503-5089

Electronic Mail: aamir.saleem@baml.com

# ISSUING LENDER:

Bank of America, N.A. Trade Operations 1000 W. Temple Street Mail Code: CA9-705-07-05 Los Angeles, CA 90012-1514 Attention: Stella Rosales Telephone: 213-481-7828 Telecopier: 213-580-8441

Electronic Mail: stella.rosales@baml.com

#### **SWING LINE LENDER:**

Bank of America, N.A. 2001 Clayton Road

Mail Code: CA4-702-02-25 Concord, CA 94520 Attention: Faizan Hafeez Telephone: 925-675-8815 Telecopier: 866-540-7550

Electronic Mail: faizan.hafeez@baml.com

Account No.: 3750836479

Ref:	White Mountains	Insurance	Group
ABA	# 026009593		

#### EXHIBIT A-3

EXHIBIT A

#### FORM OF COMPLIANCE CERTIFICATE

This Compliance Certificate (this "<u>Certificate</u>") is delivered pursuant to <u>Section 6.2(b)</u> of the Credit Agreement, dated as of August 12, 2011 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "<u>Agreement</u>;" the terms defined therein being used herein as therein defined), among White Mountains Insurance Group, Ltd., a company organized under the laws of Bermuda (the "<u>Borrower</u>"), the Lenders from time to time party thereto and Bank of America, N.A., as Administrative Agent, Swing Line Lender and an Issuing Lender.

The undersigned hereby certifies to the Administrative Agent and the Lenders as follows:

- 1. I am the duly elected, qualified and acting [the Borrower to insert title of relevant Responsible Officer] of the Borrower.
- 2. I have reviewed and am familiar with the contents of this Certificate.
- 3. I have no knowledge of the existence, as of the date of this Certificate, of any continuing Default or Event of Default [, except as set forth below].
- 4. Attached hereto as <u>Attachment 1</u> are the computations showing compliance by the Borrower with the covenants set forth in <u>Section 7.1</u> of the Credit Agreement.

IN WITNESS WHEREOF, the undersigned has executed this Compliance Certificate as of the date set forth below.

ĺ	WHITE M	OUNTAINS INSURANCE GROU	JP, LTD.
	By: Name: Title:		
Date: , 20			
EXHIBIT A-4			
			Attachment 1 to Exhibit A
The information described herein is as	s of , 20 , and pertains to the period  [Set forth Covenant Calculate		, .

EXHIBIT B-1

## FORM OF BORROWING REQUEST

Date: ,

To: Bank of America, N.A., as Administrative Agent

Ladies and Gentlemen:

Reference is made to that certain Credit Agreement, dated as of August 12, 2011 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Agreement;" the terms defined therein being used herein as therein defined), among White Mountains Insurance Group, Ltd., a company organized under the laws of Bermuda (the "Borrower"), the Lenders from time to time party thereto and Bank of America, N.A., as Administrative Agent, Swing Line Lender and an Issuing Lender.

[Pursuant to Section 2.2 of the Credit Agreement, the Borrower her principal amount of \$\ \] [Eurodollar Loan in the principal amount of \$\ \], 20 and maturing on  20 ] be made on	reby requests that a Revolving Credit Loan consisting of a [Base Rate Loan in the nount of \$ with an Interest Period commencing on , 20 ].
	reby requests that a Revolving Credit Loan in a principal amount of ted to] [continued as] a [Base Rate] [Eurodollar] Loan [with an Interest Period , 20].
	ount of the aggregate outstanding principal amount of the Loans <u>plus</u> the L/C Revolving Credit Commitment available for borrowing pursuant to <u>Section 2.1</u>
[The Borrower hereby represents and warrants that the conditions of the date of the applicable extension of credit.](1)	contained in Section 4.2(a) and (b) of the Agreement shall be satisfied on and as
(1) To be included in the case of any borrowing made pursuant to Section 2.	2 of the Agreement.
	WHITE MOUNTAINS INSURANCE GROUP, LTD.
	By:
	Name:
	Title:
	EXHIBIT B-2
FORM OF SWING	LINE LOAN NOTICE
	Date: ,
To: Bank of America, N.A., as Swing Line Lender Bank of America, N.A., as Administrative Agent	
Ladies and Gentlemen:	
modified in writing from time to time, the "Agreement;" the terms defined the	gust [ ], 2011 (as amended, restated, extended, supplemented or otherwise lerein being used herein as therein defined), among White Mountains Insurance <u>r</u> "), the Lenders from time to time party thereto and Bank of America, N.A., as
The undersigned hereby requests a Swing Line Loan:	
1. On (a B	usiness Day).
2. In the amount of \$ .	
The borrowing of the Swing Line Loan requested herein complies was Agreement.	with the requirements of the provisos to the first sentence of <u>Section 2.4(a)</u> of the
The Borrower hereby represents and warrants that the conditions co the date of the applicable extension of credit.	ntained in Section 4.2(a) and (b) of the Agreement shall be satisfied on and as of
	WHITE MOUNTAINS INSURANCE GROUP, LTD.
	By:
	Name:
	Title:

# FORM OF REVOLVING CREDIT NOTE

THIS NOTE AND THE OBLIGATIONS REPRESENTED HEREBY MAY NOT BE TRANSFERRED EXCEPT IN COMPLIANCE WITH THE TERMS AND PROVISIONS OF THE CREDIT AGREEMENT REFERRED TO BELOW. TRANSFERS OF THIS NOTE AND THE OBLIGATIONS

Date	Amount of Base Rate Loans	Amount Converted to Base Rate Loans	Amount of Principal of Base Rate Loans Repaid	Amount of Base Rate Loans Converted to Eurodollar Loans	Unpaid Principal Balance of Base Rate Loans	Notation Made By
		LOANS, CONVERSIC	ONS AND REPAYMENTS	OF BASE RATE LOANS		Schedule A
			Name Title:	<u>:</u>		
			By:			
			WHITE M	OUNTAINS INSURANG	CE GROUP, LTD.	
		[Remair	nder of Page Left Intention	ally Blank]		
	OF NEW YORK (EX	LL BE GOVERNED BY, CLUDING THE LAWS A W OF ANOTHER JURIS	APPLICABLE TO CONF EDICTION WOULD BE	FLICTS OR CHOICE O REQUIRED THEREBY	F LAW TO THE EXT	
	NOT BE TRANSFER	DING ANYTHING TO T RED EXCEPT PURSUA OF THE CREDIT AGRE	NT TO AND IN ACCOR			•
Agreement.	Unless otherwise de	efined herein, terms defined	d in the Credit Agreement a	and used herein shall have	the meanings given to t	hem in the Credi
shall become,		re of any one or more of the be, immediately due and pa			rest then remaining unpa	aid on this Note
extent of the g	samanty, the terms and t	conditions upon which the	Somany was granted and t	ine rights of the holder of	and riote in respect their	
party thereto ( the provisions guaranteed as	plemented or otherwise (collectively, the " <u>Lende</u> s of the Credit Agreement provided in the Credit A	e of the promissory notes re modified in writing from the ers") and Bank of America, int and (c) is subject to option Agreement and the Guarant conditions upon which the	ime to time, the " <u>Credit Ag</u> N.A., as Administrative A onal and mandatory prepay ty. Reference is hereby ma	greement"), among the Bo agent, Swing Line Lender rments as provided in the G de to the Credit Agreemen	rrower, the lenders from and an Issuing Lender, ( Credit Agreement. This nt and the Guaranty for t	time to time b) is subject to Note is he nature and
date and amou and, in the cas the accuracy of	ned hereto and made a p unt of each payment or p se of Eurodollar Loans,	Note is authorized to endors art hereof the date, Type an prepayment of principal the the length of each Interest lesed. The failure to make a wing Credit Loan.	nd amount of each Revolvin Preof, each continuation the Period with respect thereto	ng Credit Loan made purs ereof, each conversion of a . Each such endorsement	uant to the Credit Agree all or a portion thereof to shall constitute <u>prima f</u> a	ment and the another Type <u>cie</u> evidence of
Agent's Office Revolving Cre Revolving Cre like money at	"Borrower"), hereby ure specified in the Credit edit Termination Date the dit Loans made by the	EIVED, the undersigned, Vaconditionally promises to part Agreement (as hereinafter the principal amount of (a) Lender to the Borrower purent's Office on the unpaid part.	pay to defined) in lawful money DOLLARS (\$ rsuant to <u>Section 2.1</u> of the	(the " <u>Lender</u> ") or its of the United States and in ) or, if less, (b) the age Credit Agreement. The I	s registered assigns at the n immediately available gregate unpaid principal Borrower further agrees	e Administrative funds, on the amount of all to pay interest in
\$						, 20
REPRESENT		THAGREEMENT REFER BE RECORDED IN THE R EMENT.				

#### LOANS, CONTINUATIONS, CONVERSIONS AND REPAYMENTS OF EURODOLLAR LOANS

			Interest		Amount of		
		Amount	Period and	Amount of	Eurodollar	Unpaid	
	Amount	Converted	Eurodollar	Principal of	Loans	Principal	
	of	to	Rate with	Eurodollar	Converted to	Balance of	
	Eurodollar	Eurodollar	Respect	Loans	Base Rate	Eurodollar	Notation
Date	Loans	Loans	Thereto	Repaid	Loans	Loans	Made By
	·						

**EXHIBIT C-2** 

## FORM OF SWING LINE NOTE

THIS NOTE AND THE OBLIGATIONS REPRESENTED HEREBY MAY NOT BE TRANSFERRED EXCEPT IN COMPLIANCE WITH THE TERMS AND PROVISIONS OF THE CREDIT AGREEMENT REFERRED TO BELOW. TRANSFERS OF THIS NOTE AND THE OBLIGATIONS REPRESENTED HEREBY MUST BE RECORDED IN THE REGISTER MAINTAINED BY THE ADMINISTRATIVE AGENT PURSUANT TO THE TERMS OF SUCH CREDIT AGREEMENT.

\$ , 20

FOR VALUE RECEIVED, the undersigned, WHITE MOUNTAINS INSURANCE GROUP, LTD., a company organized under the laws of Bermuda (the "Borrower"), hereby unconditionally promises to pay to Bank of America, N.A. (the "Swing Line Lender") or its registered assigns at the Administrative Agent's Office specified in the Credit Agreement (as hereinafter defined) in lawful money of the United States and in immediately available funds, on the Revolving Credit Termination Date the principal amount of (a) DOLLARS (\$ ) or, if less, (b) the aggregate unpaid principal amount of all Swing Line Loans made by the Lender to the Borrower pursuant to Section 2.3 of the Credit Agreement. The Borrower further agrees to pay interest in like money at the Administrative Agent's Office on the unpaid principal amount hereof from time to time outstanding at the rates and on the dates specified in Section 2.11 of the Credit Agreement.

The holder of this Note is authorized to endorse on the schedules annexed hereto and made a part hereof or on a continuation thereof which shall be attached hereto and made a part hereof the date and amount of each Swing Line Loan made pursuant to the Credit Agreement and the date and amount of each payment or prepayment of principal thereof. Each such endorsement shall constitute prima facie evidence of the accuracy of the information endorsed. The failure to make any such endorsement or any error in any such endorsement shall not affect the obligations of the Borrower in respect of any Swing Line Loan.

This Note (a) is one of the promissory notes referred to in the Credit Agreement, dated as of August 12, 2011 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "<u>Credit Agreement</u>"), among the Borrower, the lenders from time to time party thereto (collectively, the "<u>Lenders</u>") and Bank of America, N.A., as Administrative Agent, Swing Line Lender and an Issuing Lender, (b) is subject to the provisions of the Credit Agreement and (c) is subject to optional and mandatory prepayment as provided in the Credit Agreement. This Note is guaranteed as provided in the Credit Agreement and the Guaranty. Reference is hereby made to the Credit Agreement and the Guaranty for the nature and extent of the guaranty, the terms and conditions upon which the guaranty was granted and the rights of the holder of this Note in respect thereof.

Upon the occurrence of any one or more of the Events of Default, all principal and all accrued interest then remaining unpaid on this Note shall become, or may be declared to be, immediately due and payable, all as provided in the Credit Agreement.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

NOTWITHSTANDING ANYTHING TO THE CONTRARY CONTAINED HEREIN OR IN THE CREDIT AGREEMENT, THIS NOTE MAY NOT BE TRANSFERRED EXCEPT PURSUANT TO AND IN ACCORDANCE WITH THE REGISTRATION AND OTHER PROVISIONS OF SECTION 10.7 OF THE CREDIT AGREEMENT.

THIS NOTE SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK (EXCLUDING THE LAWS APPLICABLE TO CONFLICTS OR CHOICE OF LAW TO THE EXTENT THAT THE APPLICATION OF THE LAW OF ANOTHER JURISDICTION WOULD BE REQUIRED THEREBY).

[Remainder of Page Left Intentionally Blank]

		By: Name: Title:		-
				Schedule A to Swing Line Note
	I	OANS AND REPAYMENTS OF SWIN	G LINE LOANS	
Date	Amount of Swing Line Loans	Amount of Principal of Swing Line Loans Repaid	Unpaid Principal Balance of Swing Line Loans	Notation Made By
Date	Amount of Swing Line Loans	Amount of Principal of Swing Line Loans Repaid	Unpaid Principal Balance of Swing Line Loans	Notation Made By
		2		
				EXHIBIT D
Insurance Gro N.A., as Adm	dified in writing from time to time, the pup, Ltd., a company organized under the linistrative Agent, Swing Line Lender ander hereby represents and warrants the lines are the lines and warrants the lines are the	(the "Non-U.S. Lender") is providing the	n being used herein as therein define the Lenders from time to time party uis certificate pursuant to Section 2.3	ed), among White Mountains thereto and Bank of America,  16(d) of the Agreement. The
providing this	s certificate.	ank" for purposes of Section 881(c)(3)(A		· ·
" <u>Code</u> "). In t	his regard, the Non-U.S. Lender furth	-		
	(b) the Non-U.S. Lender ha	not subject to regulatory or other legal reconstructions or other legal reconstructions are subject to regulatory or other legal reconstruction made to a rating agency or other legal reconstructions.	of any tax, securities law or other fi	ling or submission made to
3.	The Non-U.S. Lender is not a 10	-percent shareholder of the Borrower with	hin the meaning of Section 881(c)(3	(B) of the Code; and
4. (3)(C) of the (		ntrolled foreign corporation receiving into	erest from a related person within th	ne meaning of Section 881(c)
	IN WITNESS WHEREOF, the u	ndersigned has executed this certificate a	s of the date set forth below.	
		[NAME C	DF NON-U.S. LENDER]	
		By: Nan Title		
Date:				

**EXHIBIT E** 

#### FORM OF CLOSING CERTIFICATE

August [ ], 2011

This Closing Certificate is delivered pursuant to Section 4.1(a)(v) and (vii) of the Credit Agreement, dated as of August 12, 2011 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Agreement;" the terms defined therein being used herein as therein defined), among White Mountains Insurance Group, Ltd., a company organized under the laws of Bermuda (the "Borrower"), the Lenders from time to time party thereto and Bank of America, N.A., as Administrative Agent, Swing Line Lender and an Issuing Lender.

The Borrower hereby certifies to the Administrative Agent and the Lenders as follows:

- 1. The representations and warranties of the Borrower set forth in, or made pursuant to, each of the Loan Documents to which it is a party are true and correct in all material respects on and as of the date hereof with the same effect as if made on the date hereof, except for representations and warranties expressly stated to relate to a specific earlier date, in which case such representations and warranties were true and correct in all material respects as of such earlier date.
- 2. No Default or Event of Default has occurred and is continuing as of the date hereof or after giving effect to the Loans to be made on the date hereof.
- 3. No event or circumstance has occurred since December 31, 2010, that has had or could reasonably be expected to have a Material Adverse Effect.
- 4. Assuming the required satisfaction of the Administrative Agent or the Lenders to the extent provided therein, the conditions precedent set forth in Section 4.1 of the Agreement will be satisfied as of the Closing Date, except as set forth on Schedule I attached hereto.

IN WITNESS WHEREOF, the undersigned has executed this Closing Certificate as of the date set forth above.

WHITE MOUNTAINS INSURANCE GROUP, LTD.  By: Name: Title:	
	Schedule I
[Form of Legal Opinion — R. Seelig]	EXHIBIT F
[Form of Legal Opinion — Conyers Dill & Pearman]	EXHIBIT G

EXHIBIT H

## ASSIGNMENT AND ASSUMPTION

This Assignment and Assumption (this "<u>Assignment and Assumption</u>") is dated as of the Effective Date set forth below and is entered into by and between the Assignor identified in item 1 below (the "<u>Assignor</u>") and the Assignee identified in item 2 below (the "<u>Assignee</u>"). Capitalized terms used but not defined herein shall have the meanings given to them in the Credit Agreement identified below (the "<u>Credit Agreement</u>"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Annex 1 attached hereto are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Standard Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below (i) all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto to the extent related to the amount and percentage interest identified below of all of such outstanding rights and obligations of the Assignor under the Credit Agreement (including, without limitation, participations in L/C Obligations and Swing Line Loans) and (ii) to the extent permitted to be assigned under applicable Law, all claims, suits, causes of action and any other right of the Assignor (in its capacity as a Lender) against any Person, whether known or unknown, arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby or in any way based on or related to any of the foregoing, including, but not limited to, contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity related to the rights and obligations sold and assigned pursuant to clause (i) above (the rights and obligations sold and assigned by the Assignor to the Assignee pursuant to clauses (i) and (ii) above being referred to herein collectively as the "Assigned Interest"). Each such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

	A . F 7
1	Assignor[s]:
1.	Transfirming.

2. <u>Assignee[s]</u>:

[for each Assignee, indicate [Affiliate][Approved Fund] of [identify Lender]]

- 3. <u>Borrower</u>: White Mountains Insurance Group, Ltd.
- 4. Administrative Agent: Bank of America, N.A., as the administrative agent under the Credit Agreement
- 5. <u>Credit Agreement</u>: Credit Agreement, dated as of August 12, 2011, among White Mountains Insurance Group, Ltd., the Lenders from time to time party thereto, and Bank of America, N.A., as Administrative Agent, Swing Line Lender, and Issuing Lender.
  - 6. Assigned Interest:

Assigno	or	Assignee	Aggregate Amount of Revolving Credit Commitment/ Loans for all Lenders(2)	Amour Revolv Crec Commit Loai Assig	ving lit ment/ ns	Percentage Assigned of Revolving Credit Commitment/ Loans(3)	CUSIP Number
			\$	\$		%	
			\$	\$		%	
			\$	\$		%	
[7.	Trade Date:		](4)				

Effective Date: , 20 [MUTUALLY AGREED TO BY THE ASSIGNOR AND ASSIGNEE AND WHICH SHALL BE THE EFFECTIVE DATE OF RECORDATION OF TRANSFER IN THE REGISTER THEREFOR.]

- (2) Amounts in this column and in the column immediately to the right to be adjusted by the counterparties to take into account any payments or prepayments made between the Trade Date and the Effective Date.
- (3) Set forth, to at least 9 decimals, as a percentage of the Commitments/Loans of all Lenders thereunder.
- (4) To be completed if the Assignor and the Assignee intend that the minimum assignment amount is to be determined as of the Trade Date.

The terms set forth in this Assignment and Assumption are hereby agreed to:

ASSIGNOR					
[NAME OF ASSIGNOR]					
Ву:					
Title:					
ASSIGNEE					
[NAME OF ASSIGNEE]					
By:					
Title:					

[Consented to and] Accepted:	
BANK OF AMERICA, N.A., as Administrative Agent	
By: Title:	
[Consented to:](5)	
WHITE MOUNTAINS INSURANCE GROUP, LTD.	
Ву:	
Name:	
Title:	
[Consented to:](6)	
[ ], as [Swing Line Lender][Issuing Lender]	
By: Title:	
(5) Not required if a Specified Event of Default has occurred and is continuing.	
(6) Include any additional consents required pursuant to Section 10.7(b)(iii) of the	Credit Agreement.

ANNEX 1 TO ASSIGNMENT AND ASSUMPTION

# STANDARD TERMS AND CONDITIONS FOR

# ASSIGNMENT AND ASSUMPTION

## 1. Representations and Warranties.

- 1.1. Assignor. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral thereunder, (iii) the financial condition of the Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.
- Assignee. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all the requirements to be an assignee under Sections 10.7(b)(iii), (y), and (vi) of the Credit Agreement (subject to such consents, if any, as may be required under Section 10.7(b)(iii) of the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it is sophisticated with respect to decisions to acquire assets of the type represented by the Assigned Interest and either it, or the Person exercising discretion in making its decision to acquire the Assigned Interest, is experienced in acquiring assets of such type, (v) it has received a copy of the Credit Agreement, and has received or has been accorded the opportunity to receive copies of the most recent financial statements delivered pursuant to Section 6.1 thereof, as applicable, and such other documents and information as it deems appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest, (vi) it has, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest, and (vii) if it is a Non-U.S. Lender, attached hereto is any documentation required to be delivered by it pursuant to the terms of the Credit Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance upon the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.

	strative Agent shall make all payments in respect of the Assigned Interest signor for amounts which have accrued to but excluding the Effective Date and to late.
respective successors and assigns. This Assignment and Assumption may be instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption may be instrument.	nall be binding upon, and inure to the benefit of, the parties hereto and their e executed in any number of counterparts, which together shall constitute one Assignment and Assumption by telecopy shall be effective as delivery of a ssignment and Assumption shall be governed by, and construed in accordance
	EXHIBIT I
FO	ORM OF
INSTRUMEN	T OF ACCESSION
the Credit Agreement, as defined below) provided to the Borrower in accord Agreement, hereby (a) agrees to become a Lender party to that certain Credi supplemented or otherwise modified in writing from time to time, the "Credi among the Borrower, the Lenders from time to time party thereto and Bank of Lender, a copy of which is attached hereto, and that its Revolving Credit Conwarrants that it meets all of the requirements of an Eligible Assignee under the Credit Agreement). The undersigned hereby agrees to perform all of Accession shall become a part of the Credit Agreement.  THIS INSTRUMENT OF ACCESSION AND THE RIGHTS AND OBLICOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCOVERNED BY, AND CONSTRUED TO CONFLICTS OR CHOLAW OF ANOTHER JURISDICTION WOULD BE REQUIRED THE	t Agreement, dated as of August 12, 2011 (as amended, restated, extended, it Agreement;" the terms defined therein being used herein as therein defined), of America, N.A., as Administrative Agent, Swing Line Lender and an Issuing mmitment amount is \$ and (b) represents and he Credit Agreement (subject to receipt of such consents as may be required duties and obligations of a Lender under the Credit Agreement. This Instrument LIGATIONS OF THE PARTIES UNDER HEREUNDER SHALL BE DRDANCE WITH, THE LAW OF THE STATE OF NEW YORK DICE OF LAW TO THE EXTENT THAT THE APPLICATION OF THE
Executed as of the date set forth below.	
	[LENDER]
	By: Name: Title:
Accepted as of this day of , 20 :	
WHITE MOUNTAINS INSURANCE GROUP, LTD.	
Ву:	
Name:	
Title:	

EXHIBIT J

# **GUARANTY**

FOR VALUE RECEIVED, the sufficiency of which is hereby acknowledged, and in consideration of credit and/or financial accommodation heretofore or hereafter from time to time made or granted to WHITE MOUNTAINS INSURANCE GROUP, LTD., a company organized under the laws of Bermuda (the "Borrower") by (a) BANK OF AMERICA, N.A. as administrative agent (the "Administrative Agent") for itself and the other lenders (the "Lenders") from time to time party to that certain Credit Agreement, dated as of August 12, 2011 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Credit Agreement"), by and among the Borrower, the Lenders, the Administrative Agent and Bank of America, N.A., as Swing Line Lender and an Issuing Lender and (b) the Lenders, each of the undersigned Guarantors (individually a "Guarantor" and if more than one

jointly and severally, collectively referred to herein as the "<u>Guarantors</u>"), hereby furnishes its guaranty of the Guaranteed Obligations (as hereinafter defined) to the Administrative Agent, for the benefit of itself and the Lenders, as set forth below.

**WHEREAS**, it is a condition precedent to the Lenders' making any loans or otherwise extending credit and each Issuing Lender issuing, extending or renewing letters of credit under the Credit Agreement that the Guarantors execute and deliver to the Administrative Agent, for the benefit of the Lenders, a continuing guaranty in substantially the form hereof;

**WHEREAS,** the Borrower and the Guarantors are members of a group of related entities, the success of any one of which is dependent in part on the success of the other members of such group;

**WHEREAS,** each Guarantor expects to receive, directly and indirectly, benefits from the extensions of credit to the Borrower by the Lenders pursuant to the Credit Agreement (which benefits are hereby acknowledged); and

**WHEREAS**, each Guarantor wishes to guarantee to the Administrative Agent, for the benefit of the Lenders the payment and performance in full of all of the Guaranteed Obligations (as hereinafter defined), as herein provided.

**NOW, THEREFORE**, in consideration of the promises contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. **Definitions.** Capitalized terms used herein without definition shall have the respective meaning provided therefor in the Credit Agreement.
- **2. Guaranty.** Subject to Section 2.21 of the Credit Agreement, each Guarantor hereby guarantees to the Lenders and the Administrative Agent the full and punctual payment when due (whether at stated maturity, by required pre-payment, by acceleration or otherwise) of all of the obligations of the Borrower under the Credit Agreement and the other Loan Documents (including, but not limited to, the principal of the Loans advanced to the Borrower, all Reimbursement Obligations of the Borrower in respect of Letters of Credit, and all interest, fees, expenses, indemnities and other amounts payable by the Borrower under the Credit Agreement), including all such which would become due but for the operation of the automatic stay pursuant

to §362(a) of the Federal Bankruptcy Code (Title 11, United States Code) and the operation of §502(b) of the Federal Bankruptcy Code (collectively, the "Guaranteed Obligations"). This Guaranty is an absolute, unconditional and continuing guaranty of the full and punctual payment and performance of all such Guaranteed Obligations, and not of their collectibility only and is in no way conditioned upon any requirement that the Administrative Agent or any Lender first attempt to collect any of the Guaranteed Obligations from the Borrower or resort to any other means of obtaining payment. Should an Event of Default occur with respect to the payment or performance of any such Guaranteed Obligations of the Borrower, the obligations of the Guarantors under this Guaranty with respect to such Guaranteed Obligations in default shall, upon demand by the Administrative Agent, become immediately due and payable to the Administrative Agent, for the benefit of the Lenders and the Administrative Agent, without demand or notice of any nature, all of which are expressly waived by each Guarantor. Payments by any Guarantor in respect of this Guaranty may be required by the Administrative Agent on any number of occasions. All payments by any Guarantor in respect of this Guaranty shall be made to the Administrative Agent, in the manner and at the place of payment specified in the Credit Agreement, for the account of the Lenders and the Administrative Agent. Anything contained herein to the contrary notwithstanding, the obligations of each Guarantor hereunder at any time shall be limited to an aggregate amount equal to the largest amount that would not render its obligations hereunder subject to avoidance as a fraudulent transfer or conveyance under §548 of the Federal Bankruptcy Code or any comparable provisions of any similar federal or state Law.

No Setoff or Deductions; Taxes; Payments. Except as required by Law, each Guarantor shall make all payments hereunder without set-off or counterclaim and free and clear of, and without deduction or withholding for, or on account of, any present or future income, stamp or other taxes, levies, imposts, duties, charges, fees, deductions or withholdings now or hereafter imposed, levied, collected, withheld or assessed by any Governmental Authority. If any such obligation (other than one arising with respect to taxes based on or measured by the income or profits of the Lenders) is imposed upon any Guarantor with respect to any amount payable by it hereunder, such Guarantor will pay to the Administrative Agent for the benefit of the Lenders, on the date on which such amount is due and payable hereunder, such additional amount in Dollars as shall be necessary to enable the Lenders to receive the same net amount which the Lender would have received on such due date had no such obligation been imposed upon such Guarantor. Such Guarantor will deliver promptly to the Administrative Agent a certified copy of any official receipt received by such Guarantor (or other evidence satisfactory to the Administrative Agent) showing all taxes or other charges deducted from or paid with respect to payments made by such Guarantor hereunder. The obligations of the Guarantors under this paragraph shall survive the payment in full of the Guaranteed Obligations and termination of this Guaranty. All payments under this Guaranty shall be made in the United States. The obligations hereunder shall not be affected by any acts of any legislative body or Governmental Authority affecting the Borrower, including but not limited to, any restrictions on the conversion of currency or repatriation or control of funds or any total or partial expropriation of the Borrower's property, or by economic, political, regulatory or other events in the countries where the Guarantor is located. A Lender that is entitled to an exemption from or reduction of non-U.S. withholding tax under the Law of the jurisdiction in which the Guarantor is located, or any treaty to which such jurisdiction is a party, with respect to payments under this Guaranty shall deliver to the Guarantor (with a copy to the Administrative Agent), at the time or times

3

prescribed by applicable Law or reasonably requested by the Guarantor, such properly completed and executed documentation prescribed by applicable Law or reasonably requested by the Guarantor as will permit such payments to be made without withholding or at a reduced rate, provided further that such Lender is legally entitled to complete, execute and deliver such documentation.

**4. Agreement to Pay Enforcement Costs, etc.** Each Guarantor further agrees, as the principal obligor and not as a guarantor only, to pay to the Administrative Agent, on demand, all reasonable out-of-pocket costs and expenses (including court costs and legal expenses) incurred or expended by the Administrative Agent or any Lender in connection with this Guaranty and the enforcement thereof, together with interest on amounts recoverable under this Section 3 from the time when such amounts become due until payment, whether before or after judgment, at the rate of interest for overdue principal set forth in the Credit Agreement, provided that if such interest would exceed the Maximum Rate, then such interest shall be reduced to such Maximum Rate.

Waivers by Guarantors; Lenders' Freedom to Act. To the fullest extent permitted by applicable Law, each Guarantor agrees that the Guaranteed Obligations will be paid and performed strictly in accordance with their respective terms, regardless of any Law now or hereafter in effect in any jurisdiction affecting any of such terms or the rights of the Administrative Agent or any Lender with respect thereto. To the fullest extent permitted by applicable Law, each Guarantor waives promptness, diligence, presentment, demand, protest, notice of acceptance, notice of any obligations incurred and all other notices of any kind, all defenses which may be available by virtue of any valuation, stay, moratorium law or other similar Law now or hereafter in effect, any right to require the marshalling of assets of the Borrower or any other entity or other person primarily or secondarily liable with respect to any of the Guaranteed Obligations, and all suretyship defenses generally. Without limiting the generality of the foregoing, each Guarantor agrees to the provisions of any instrument evidencing or otherwise executed in connection with any Guaranteed Obligation and agrees, to the fullest extent permitted by applicable Law, that its obligations in respect of its Guaranty shall not be released or discharged, in whole or in part, or otherwise affected by (i) the failure of the Administrative Agent or any Lender to assert any claim or demand or to enforce any right or remedy against the Borrower or any other entity or other person primarily or secondarily liable with respect to any of the Guaranteed Obligations; (ii) any extensions, compromise, refinancing, consolidation or renewals of any Guaranteed Obligation; (iii) any change in the time, place or manner of payment of any of the Guaranteed Obligations or any rescissions, waivers, compromise, refinancing, consolidation or other amendments or modifications of any of the terms or provisions of the Credit Agreement or the other Loan Documents or any other agreement evidencing or otherwise executed in connection with any of the Guaranteed Obligations, (iv) the addition, substitution or release of any entity or other person primarily or secondarily liable for any Guaranteed Obligation; (v) the adequacy of any rights which the Administrative Agent or any Lender may have against any means of obtaining repayment of any of the Guaranteed Obligations; or (vi) any other act or omission which might in any manner or to any extent vary the risk of any of the Guarantors or otherwise operate as a release or discharge of any of the Guarantors, all of which may be done without notice to any of the Guarantors. To the fullest extent permitted by applicable Law, each Guarantor hereby expressly waives any and all rights or defenses arising by reason of (A) any "one action" or "anti-deficiency" Law which

4

would otherwise prevent the Administrative Agent or any Lender from bringing any action, including any claim for a deficiency, or exercising any other right or remedy (including any right of set-off), against any of the Guarantors before or after the Administrative Agent's or such Lender's commencement or completion of any foreclosure action, whether judicially, by exercise of power of sale or otherwise, or (B) any other Law which in any other way would otherwise require any election of remedies by the Administrative Agent or any Lender.

- 6. Unenforceability of Obligations Against the Borrower. If for any reason the Borrower has no legal existence or is under no legal obligation to discharge any of the Guaranteed Obligations, or if any of such obligations have become irrecoverable from the Borrower by reason of the Borrower's bankruptcy or reorganization or by other operation of Law or for any other reason, this Guaranty shall, to the fullest extent permitted by applicable Law, nevertheless be binding on each Guarantor to the same extent as if the affected Guarantor at all times had been the principal obligor on all such Guaranteed Obligations. In the event that acceleration of the time for payment of any of the Guaranteed Obligations is stayed upon the insolvency, bankruptcy or reorganization of the Borrower, or for any other reason, all such Guaranteed Obligations otherwise subject to acceleration under the terms of the Credit Agreement and the other Loan Documents or any other agreement evidencing or otherwise executed in connection with any such obligation shall be immediately due and payable by the applicable Guarantor
- **7. Subrogation**. Until the final payment and performance in full of all of Guaranteed Obligations and any amounts payable under this Guaranty, no Guarantor shall exercise any rights against the Borrower arising as a result of payment by such Guarantor in respect of this Guaranty, by way of subrogation, reimbursement, restitution, contribution or otherwise, and will not prove any claim in competition with the Administrative Agent or any Lender in respect of any payment hereunder in any bankruptcy, insolvency or reorganization case or proceedings of any nature and such Guarantor will not claim any setoff, recoupment or counterclaim against the Borrower in respect of any liability of such Guarantor to the Borrower.
- **8. Obligations Independent; Provisions Supplemental.** The obligations of the Guarantors hereunder are those of primary obligor, and not merely as surety, and are independent of the Guaranteed Obligations and the obligations of any other guarantor, and a separate action may be brought against any Guarantor to enforce this Guaranty whether or not the Borrower or any other person or entity is joined as a party. The provisions of this Guaranty shall be supplemental to and not in derogation of any other rights and remedies of the Lenders and the Administrative Agent under the Credit Agreement, the other Loan Documents and any separate subordination agreement which the Administrative Agent may at any time and from time to time enter into with any of the Guarantors for the benefit of the Lenders and the Administrative Agent.
- **9. Further Assurances**. Each Guarantor agrees that it will from time to time, at the request of the Administrative Agent, do all such things and execute all such documents as the Administrative Agent may reasonably request to give full effect to this Guaranty and to preserve the rights and powers of the Lenders and the Administrative Agent in respect of this Guaranty. Each Guarantor acknowledges and confirms that it has established its own adequate means of obtaining from the Borrower on a continuing basis all information desired by such Guarantor

5

concerning the financial condition of the Borrower and that such Guarantor will look to the Borrower and not to the Administrative Agent or any Lender in order for such Guarantor to keep adequately informed of changes in the Borrower's financial condition.

- 10. Successors and Assigns. This Guaranty shall be binding upon each Guarantor, its successors and assigns, and shall inure to the benefit of the Administrative Agent and the Lenders and their respective permitted transferees and permitted assigns; provided that no Guarantor shall assign its rights and obligations under this Guaranty without the prior written consent of the Administrative Agent (and any attempted assignment without such consent shall be void). Without limiting the generality of the foregoing sentence, each Lender may, to the extent permitted by Section 10.7 of the Credit Agreement, assign or otherwise transfer the Credit Agreement, the other Loan Documents or any other agreement or Note held by it evidencing or otherwise executed in connection with the Guaranteed Obligations, or sell participations in any interest therein, to any other entity or other person, and such other entity or other person shall, to the extent provided by Section 10.7 of the Credit Agreement, thereupon become vested, to the extent set forth in the agreement evidencing such assignment, transfer or participation, with all the rights in respect thereof granted to such Lender herein, all in accordance with Section 10.7 of the Credit Agreement.
- 11. **Termination; Reinstatement.** Subject to Section 2.21 and 10.17 of the Credit Agreement, this Guaranty is a continuing and irrevocable guaranty of all Guaranteed Obligations now or hereafter existing and shall remain in full force and effect until (a) all Guaranteed Obligations and any other amounts payable under this Guaranty are paid in full and all Revolving Credit Commitments have terminated, there exist no unpaid Reimbursement

Obligations and no Letter of Credit issued for the account of the Borrower shall be outstanding or (b) so long as no Event of Default has occurred and is continuing, then at such time as the aggregate outstanding principal amount of Indebtedness of Subsidiaries of OneBeacon Limited that is guaranteed by the Borrower is \$70,000,000 or less (excluding (i) any series of such Indebtedness if all of such series has been defeased (including legal or covenant defeasance) or discharged and (ii) such Indebtedness to the extent it is held by the Borrower or any of its Subsidiaries). Notwithstanding the foregoing, any release of Guarantee Obligations pursuant to clause (a) above shall be deemed subject to the provision that such Guarantee Obligations shall be reinstated as to the Guarantors at the time of such release if after such release any portion of any payment in respect of the Guaranteed Obligations is rescinded or must otherwise be restored or returned upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Borrower or such Guarantor, or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or other similar officer for, the Borrower or such Guarantor or any substantial part of its property, or otherwise, all as though such payment had not been made and whether or not the Administrative Agent is in possession of or has released this Guaranty. In addition, upon request of the Borrower, the Administrative Agent shall (without vote or consent of any Lender) take such actions as may be required to release all Guarantee Obligations under this Guaranty of any Guarantor if such Guarantor is merged or consolidated into another Person and such Guarantor is not the surviving Person of such merger or consolidation, or such Guarantor otherwise ceases to be a Subsidiary of the Borrower; provided, however, that (x) the surviving Person of any such merger or consolidation is, or any assets transferred in connection with such liquidation, winding-up or dissolution

6

a Guarantor in connection with such transaction) or a Subsidiary of a Guarantor (including a Person that will become a Subsidiary of a Guarantor in connection with such transaction) or (y) after giving effect to any such merger, consolidation, liquidation, winding-up, dissolution, or cessation of such Guarantor's status as a Subsidiary of the Borrower, the aggregate Consolidated Net Worth of all other Guarantors (including any Person that becomes a Guarantor or Subsidiary of a Guarantor in connection with such transaction but excluding for purposes of calculating the aggregate Consolidated Net Worth of all Guarantors any outstanding Indebtedness of the Borrower to a Guarantor (to the extent reflected as an asset of such Guarantor in the calculation of such Consolidated Net Worth) is not less than the amount then required by Section 7.1(a) of the Credit Agreement, on a pro forma basis as of the end of the most recently completed fiscal quarter for which financial statements have been delivered to the Administrative Agent pursuant to Section 6.1(a) of the Credit Agreement. The obligations of the Guarantors under this paragraph shall survive termination of this Guaranty.

- **12. Miscellaneous.** No provision of this Guaranty may be waived, amended, supplemented or modified, except by a written instrument executed by the Administrative Agent and the Guarantors. No failure by the Administrative Agent or any Lender to exercise, and no delay in exercising, any right, remedy or power hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy or power hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies herein provided are cumulative and not exclusive of any remedies provided by law or in equity. The unenforceability or invalidity of any provision of this Guaranty shall not affect the enforceability or validity of any other provision herein.
- **13. Setoff.** If and to the extent any payment is not made when due hereunder, each Lender may setoff and charge from time to time any amount so due against any or all of any Guarantor's accounts or deposits with such Lender.
- 14. Representations and Warranties. Each Guarantor represents and warrants that (a) it is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (b) it has corporate or other power and authority and the legal right, to make, deliver and perform this Guaranty, (c) this Guaranty constitutes its legal, valid and binding obligation enforceable against such Guarantor in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether enforcement is sought by proceedings in equity or in law); (d) the making and performance of this Guaranty does not and will not violate any Requirement of Law or any Contractual Obligation of such Guarantor, except to the extent such violation could not, in the aggregate, reasonably be expected to have a Material Adverse Effect; and (e) all consents, approvals, licenses and authorizations of, and filings and registrations with, any Governmental Authority required under applicable Law for the making and performance of this Guaranty have been obtained or made and are in full force and effect except to the extent failure to obtain any such consent, approval, license, authorization, or to make such filing or registration, could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **15. Indemnification and Survival.** Without limitation on any other obligations of the Guarantors or remedies of the Administrative Agent and the Lenders under this Guaranty,

7

each Guarantor shall indemnify and hold harmless the Administrative Agent and the Lenders from and against any and all obligations, penalties, claims, demands, actions, judgments, suits, settlement payments, damages, losses, liabilities and causes of action of every kind or nature whatsoever and reasonable related out-of-pocket costs and expenses that may at any time be imposed on, sustained, required to be paid by, suffered, incurred or asserted against, the Administrative Agent or any Lender in connection with or as a result of any failure of any Guaranteed Obligations to be the legal, valid and binding obligations of the Borrower enforceable against the Borrower in accordance with their terms; provided that such indemnity shall not, as to the Administrative Agent or any Lender, be available to the extent that such obligations, penalties, claims, demands, actions, judgments, suits, settlement payments, damages, losses, liabilities, causes of action or costs or expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of the Administrative Agent or such Lender. In all such litigation, or the preparation therefor, the Administrative Agent and the Lenders shall be entitled to select counsel to the Administrative Agent and the Lenders. To the extent reasonably practicable and not disadvantageous to the Administrative Agent or any of the Lenders (as reasonably determined by the relevant party), it is anticipated that a single counsel selected by the Administrative Agent and the affected Lenders will be used. The obligations of the Guarantors under this paragraph shall survive the payment in full of the Guaranteed Obligations and termination of this Guaranty.

- 16. Governing Law. THIS GUARANTY AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS GUARANTY SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK (EXCLUDING THE LAWS APPLICABLE TO CONFLICTS OR CHOICE OF LAW TO THE EXTENT THAT THE APPLICATION OF THE LAW OF ANOTHER JURISDICTION WOULD BE REQUIRED THEREBY).
  - 17. Submission to Jurisdiction; Waivers. EACH GUARANTOR HEREBY IRREVOCABLY AND UNCONDITIONALLY:

- (a) SUBMITS FOR ITSELF AND ITS PROPERTY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS GUARANTY, OR FOR RECOGNITION AND ENFORCEMENT OF ANY JUDGMENT IN RESPECT THEREOF, TO THE NON EXCLUSIVE GENERAL JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK, THE COURTS OF THE UNITED STATES OF AMERICA FOR THE SOUTHERN DISTRICT OF NEW YORK, AND APPELLATE COURTS FROM ANY THEREOF;
- (b) CONSENTS THAT ANY SUCH ACTION OR PROCEEDING MAY BE BROUGHT IN SUCH COURTS AND WAIVES ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE VENUE OF ANY SUCH ACTION OR PROCEEDING IN ANY SUCH COURT OR THAT SUCH ACTION OR PROCEEDING WAS BROUGHT IN AN INCONVENIENT COURT AND AGREES NOT TO PLEAD OR CLAIM THE SAME;
- (c) AGREES THAT SERVICE OF PROCESS IN ANY SUCH ACTION OR PROCEEDING MAY BE EFFECTED BY MAILING A COPY THEREOF BY

8

REGISTERED OR CERTIFIED MAIL (OR ANY SUBSTANTIALLY SIMILAR FORM OF MAIL), POSTAGE PREPAID, TO THE GUARANTORS AT ITS ADDRESS SET FORTH ON THE SIGNATURE PAGE HEREOF OR AT SUCH OTHER ADDRESS OF WHICH THE ADMINISTRATIVE AGENT SHALL HAVE BEEN NOTIFIED PURSUANT HERETO;

- (d) AGREES THAT NOTHING HEREIN SHALL AFFECT THE RIGHT TO EFFECT SERVICE OF PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR SHALL LIMIT THE RIGHT TO SUE IN ANY OTHER JURISDICTION; AND
- (e) WAIVES, TO THE MAXIMUM EXTENT NOT PROHIBITED BY LAW, ANY RIGHT IT MAY HAVE TO CLAIM OR RECOVER IN ANY LEGAL ACTION OR PROCEEDING REFERRED TO IN THIS SECTION 17 ANY SPECIAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES.
- 18. Waivers of Jury Trial. EACH PARTY TO THIS GUARANTY HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING HEREUNDER OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO THIS GUARANTY, OR THE TRANSACTIONS RELATED HERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS GUARANTY MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION 18 WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.
- **19. Confidentiality; Notices.** Subject to the terms of Section 10.16 of the Credit Agreement, each Guarantor agrees that the Administrative Agent and the Lenders may disclose to any assignee of or participant in, or any prospective assignee of or participant in, any of its rights or obligations of all or part of the Guaranteed Obligations any and all information in the Administrative Agent's or any Lender's possession concerning the Guarantors, this Guaranty and any security for this Guaranty. All notices and other communications to the Guarantors under this Guaranty shall be delivered in accordance with the terms of Section 10.2 of the Credit Agreement and to the applicable Guarantor at its address set forth on the signature pages hereof or such other address of which the Administrative Agent shall have been notified pursuant hereto.
- **20. FINAL AGREEMENT.** THIS GUARANTY REPRESENTS THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS BETWEEN THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

9

[REMAINDER OF PAGE LEFT INTENTIONALLY BLANK]

10

[EXECUTION COPY]

Executed this day of August, 2011.

LONE TREE INSURANCE GROUP LTD.

By:			
Name:			

......

Title:

LONE TREE HOLDINGS LTD.

v:			

Name:
Title:
Address for each Guarantor:
14 Wesley Street 5th floor

Hamilton, HM 11, Bermuda

#### **GUARANTY**

FOR VALUE RECEIVED, the sufficiency of which is hereby acknowledged, and in consideration of credit and/or financial accommodation heretofore or hereafter from time to time made or granted to WHITE MOUNTAINS INSURANCE GROUP, LTD., a company organized under the laws of Bermuda (the "Borrower") by (a) BANK OF AMERICA, N.A. as administrative agent (the "Administrative Agent") for itself and the other lenders (the "Lenders") from time to time party to that certain Credit Agreement, dated as of August 12, 2011 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Credit Agreement"), by and among the Borrower, the Lenders, the Administrative Agent and Bank of America, N.A., as Swing Line Lender and an Issuing Lender and (b) the Lenders, each of the undersigned Guarantors (individually a "Guarantor" and if more than one jointly and severally, collectively referred to herein as the "Guarantors"), hereby furnishes its guaranty of the Guaranteed Obligations (as hereinafter defined) to the Administrative Agent, for the benefit of itself and the Lenders, as set forth below.

**WHEREAS**, it is a condition precedent to the Lenders' making any loans or otherwise extending credit and each Issuing Lender issuing, extending or renewing letters of credit under the Credit Agreement that the Guarantors execute and deliver to the Administrative Agent, for the benefit of the Lenders, a continuing guaranty in substantially the form hereof;

**WHEREAS,** the Borrower and the Guarantors are members of a group of related entities, the success of any one of which is dependent in part on the success of the other members of such group;

**WHEREAS**, each Guarantor expects to receive, directly and indirectly, benefits from the extensions of credit to the Borrower by the Lenders pursuant to the Credit Agreement (which benefits are hereby acknowledged); and

**WHEREAS**, each Guarantor wishes to guarantee to the Administrative Agent, for the benefit of the Lenders the payment and performance in full of all of the Guaranteed Obligations (as hereinafter defined), as herein provided.

**NOW, THEREFORE**, in consideration of the promises contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- **21. Definitions.** Capitalized terms used herein without definition shall have the respective meaning provided therefor in the Credit Agreement.
- **22. Guaranty.** Subject to Section 2.21 of the Credit Agreement, each Guarantor hereby guarantees to the Lenders and the Administrative Agent the full and punctual payment when due (whether at stated maturity, by required pre-payment, by acceleration or otherwise) of all of the obligations of the Borrower under the Credit Agreement and the other Loan Documents (including, but not limited to, the principal of the Loans advanced to the Borrower, all Reimbursement Obligations of the Borrower in respect of Letters of Credit, and all interest, fees, expenses, indemnities and other amounts payable by the Borrower under the Credit Agreement), including all such which would become due but for the operation of the automatic stay pursuant

12

to §362(a) of the Federal Bankruptcy Code (Title 11, United States Code) and the operation of §502(b) of the Federal Bankruptcy Code (collectively, the "Guaranteed Obligations"). This Guaranty is an absolute, unconditional and continuing guaranty of the full and punctual payment and performance of all such Guaranteed Obligations, and not of their collectibility only and is in no way conditioned upon any requirement that the Administrative Agent or any Lender first attempt to collect any of the Guaranteed Obligations from the Borrower or resort to any other means of obtaining payment. Should an Event of Default occur with respect to the payment or performance of any such Guaranteed Obligations of the Borrower, the obligations of the Guarantors under this Guaranty with respect to such Guaranteed Obligations in default shall, upon demand by the Administrative Agent, become immediately due and payable to the Administrative Agent, for the benefit of the Lenders and the Administrative Agent, without demand or notice of any nature, all of which are expressly waived by each Guarantor. Payments by any Guarantor in respect of this Guaranty may be required by the Administrative Agent on any number of occasions. All payments by any Guarantor in respect of this Guaranty shall be made to the Administrative Agent, in the manner and at the place of payment specified in the Credit Agreement, for the account of the Lenders and the Administrative Agent. Anything contained herein to the contrary notwithstanding, the obligations of each Guarantor hereunder at any time shall be limited to an aggregate amount equal to the largest amount that would not render its obligations hereunder subject to avoidance as a fraudulent transfer or conveyance under §548 of the Federal Bankruptcy Code or any comparable provisions of any similar federal or state Law.

23. No Setoff or Deductions; Taxes; Payments. Except as required by Law, each Guarantor shall make all payments hereunder without setoff or counterclaim and free and clear of, and without deduction or withholding for, or on account of, any present or future income, stamp or other taxes,
levies, imposts, duties, charges, fees, deductions or withholdings now or hereafter imposed, levied, collected, withheld or assessed by any Governmental
Authority. If any such obligation (other than one arising with respect to taxes based on or measured by the income or profits of the Lenders) is imposed upon
any Guarantor with respect to any amount payable by it hereunder, such Guarantor will pay to the Administrative Agent for the benefit of the Lenders, on the
date on which such amount is due and payable hereunder, such additional amount in Dollars as shall be necessary to enable the Lenders to receive the same
net amount which the Lender would have received on such due date had no such obligation been imposed upon such Guarantor. Such Guarantor will deliver
promptly to the Administrative Agent a certified copy of any official receipt received by such Guarantor (or other evidence satisfactory to the Administrative

Agent) showing all taxes or other charges deducted from or paid with respect to payments made by such Guarantor hereunder. The obligations of the Guarantors under this paragraph shall survive the payment in full of the Guaranteed Obligations and termination of this Guaranty. All payments under this Guaranty shall be made in the United States. The obligations hereunder shall not be affected by any acts of any legislative body or Governmental Authority affecting the Borrower, including but not limited to, any restrictions on the conversion of currency or repatriation or control of funds or any total or partial expropriation of the Borrower's property, or by economic, political, regulatory or other events in the countries where the Guarantor is located. A Lender that is entitled to an exemption from or reduction of non-U.S. withholding tax under the Law of the jurisdiction in which the Guarantor is located, or any treaty to which such jurisdiction is a party, with respect to payments under this Guaranty shall deliver to the Guarantor (with a copy to the Administrative Agent), at the time or times

13

prescribed by applicable Law or reasonably requested by the Guarantor, such properly completed and executed documentation prescribed by applicable Law or reasonably requested by the Guarantor as will permit such payments to be made without withholding or at a reduced rate, provided further that such Lender is legally entitled to complete, execute and deliver such documentation.

- **24. Agreement to Pay Enforcement Costs, etc.** Each Guarantor further agrees, as the principal obligor and not as a guarantor only, to pay to the Administrative Agent, on demand, all reasonable out-of-pocket costs and expenses (including court costs and legal expenses) incurred or expended by the Administrative Agent or any Lender in connection with this Guaranty and the enforcement thereof, together with interest on amounts recoverable under this Section 3 from the time when such amounts become due until payment, whether before or after judgment, at the rate of interest for overdue principal set forth in the Credit Agreement, <u>provided</u> that if such interest would exceed the Maximum Rate, then such interest shall be reduced to such Maximum Rate.
- Waivers by Guarantors; Lenders' Freedom to Act. To the fullest extent permitted by applicable Law, each Guarantor agrees that the Guaranteed Obligations will be paid and performed strictly in accordance with their respective terms, regardless of any Law now or hereafter in effect in any jurisdiction affecting any of such terms or the rights of the Administrative Agent or any Lender with respect thereto. To the fullest extent permitted by applicable Law, each Guarantor waives promptness, diligence, presentment, demand, protest, notice of acceptance, notice of any obligations incurred and all other notices of any kind, all defenses which may be available by virtue of any valuation, stay, moratorium law or other similar Law now or hereafter in effect, any right to require the marshalling of assets of the Borrower or any other entity or other person primarily or secondarily liable with respect to any of the Guaranteed Obligations, and all suretyship defenses generally. Without limiting the generality of the foregoing, each Guarantor agrees to the provisions of any instrument evidencing or otherwise executed in connection with any Guaranteed Obligation and agrees, to the fullest extent permitted by applicable Law, that its obligations in respect of its Guaranty shall not be released or discharged, in whole or in part, or otherwise affected by (i) the failure of the Administrative Agent or any Lender to assert any claim or demand or to enforce any right or remedy against the Borrower or any other entity or other person primarily or secondarily liable with respect to any of the Guaranteed Obligations; (ii) any extensions, compromise, refinancing, consolidation or renewals of any Guaranteed Obligation; (iii) any change in the time, place or manner of payment of any of the Guaranteed Obligations or any rescissions, waivers, compromise, refinancing, consolidation or other amendments or modifications of any of the terms or provisions of the Credit Agreement or the other Loan Documents or any other agreement evidencing or otherwise executed in connection with any of the Guaranteed Obligations, (iv) the addition, substitution or release of any entity or other person primarily or secondarily liable for any Guaranteed Obligation; (v) the adequacy of any rights which the Administrative Agent or any Lender may have against any means of obtaining repayment of any of the Guaranteed Obligations; or (vi) any other act or omission which might in any manner or to any extent vary the risk of any of the Guarantors or otherwise operate as a release or discharge of any of the Guarantors, all of which may be done without notice to any of the Guarantors. To the fullest extent permitted by applicable Law, each Guarantor hereby expressly waives any and all rights or defenses arising by reason of (A) any "one action" or "anti-deficiency" Law which

14

would otherwise prevent the Administrative Agent or any Lender from bringing any action, including any claim for a deficiency, or exercising any other right or remedy (including any right of set-off), against any of the Guarantors before or after the Administrative Agent's or such Lender's commencement or completion of any foreclosure action, whether judicially, by exercise of power of sale or otherwise, or (B) any other Law which in any other way would otherwise require any election of remedies by the Administrative Agent or any Lender.

- **26. Unenforceability of Obligations Against the Borrower**. If for any reason the Borrower has no legal existence or is under no legal obligation to discharge any of the Guaranteed Obligations, or if any of such obligations have become irrecoverable from the Borrower by reason of the Borrower's bankruptcy or reorganization or by other operation of Law or for any other reason, this Guaranty shall, to the fullest extent permitted by applicable Law, nevertheless be binding on each Guarantor to the same extent as if the affected Guarantor at all times had been the principal obligor on all such Guaranteed Obligations. In the event that acceleration of the time for payment of any of the Guaranteed Obligations is stayed upon the insolvency, bankruptcy or reorganization of the Borrower, or for any other reason, all such Guaranteed Obligations otherwise subject to acceleration under the terms of the Credit Agreement and the other Loan Documents or any other agreement evidencing or otherwise executed in connection with any such obligation shall be immediately due and payable by the applicable Guarantor
- **27. Subrogation**. Until the final payment and performance in full of all of Guaranteed Obligations and any amounts payable under this Guaranty, no Guarantor shall exercise any rights against the Borrower arising as a result of payment by such Guarantor in respect of this Guaranty, by way of subrogation, reimbursement, restitution, contribution or otherwise, and will not prove any claim in competition with the Administrative Agent or any Lender in respect of any payment hereunder in any bankruptcy, insolvency or reorganization case or proceedings of any nature and such Guarantor will not claim any setoff, recoupment or counterclaim against the Borrower in respect of any liability of such Guarantor to the Borrower.
- **28. Obligations Independent; Provisions Supplemental.** The obligations of the Guarantors hereunder are those of primary obligor, and not merely as surety, and are independent of the Guaranteed Obligations and the obligations of any other guarantor, and a separate action may be brought against any Guarantor to enforce this Guaranty whether or not the Borrower or any other person or entity is joined as a party. The provisions of this Guaranty shall be supplemental to and not in derogation of any other rights and remedies of the Lenders and the Administrative Agent under the Credit Agreement, the other Loan Documents and any separate subordination agreement which the Administrative Agent may at any time and from time to time enter into with any of the Guarantors for the benefit of the Lenders and the Administrative Agent.
- **29. Further Assurances**. Each Guarantor agrees that it will from time to time, at the request of the Administrative Agent, do all such things and execute all such documents as the Administrative Agent may reasonably request to give full effect to this Guaranty and to preserve the rights and powers

15

concerning the financial condition of the Borrower and that such Guarantor will look to the Borrower and not to the Administrative Agent or any Lender in order for such Guarantor to keep adequately informed of changes in the Borrower's financial condition.

- **30. Successors and Assigns.** This Guaranty shall be binding upon each Guarantor, its successors and assigns, and shall inure to the benefit of the Administrative Agent and the Lenders and their respective permitted transferees and permitted assigns; <u>provided</u> that no Guarantor shall assign its rights and obligations under this Guaranty without the prior written consent of the Administrative Agent (and any attempted assignment without such consent shall be void). Without limiting the generality of the foregoing sentence, each Lender may, to the extent permitted by Section 10.7 of the Credit Agreement, assign or otherwise transfer the Credit Agreement, the other Loan Documents or any other agreement or Note held by it evidencing or otherwise executed in connection with the Guaranteed Obligations, or sell participations in any interest therein, to any other entity or other person, and such other entity or other person shall, to the extent provided by Section 10.7 of the Credit Agreement, thereupon become vested, to the extent set forth in the agreement evidencing such assignment, transfer or participation, with all the rights in respect thereof granted to such Lender herein, all in accordance with Section 10.7 of the Credit Agreement.
- 31. **Termination; Reinstatement.** Subject to Section 2.21 and 10.17 of the Credit Agreement, this Guaranty is a continuing and irrevocable guaranty of all Guaranteed Obligations now or hereafter existing and shall remain in full force and effect until (a) all Guaranteed Obligations and any other amounts payable under this Guaranty are paid in full and all Revolving Credit Commitments have terminated, there exist no unpaid Reimbursement Obligations and no Letter of Credit issued for the account of the Borrower shall be outstanding or (b) so long as no Event of Default has occurred and is continuing, then at such time as the aggregate outstanding principal amount of Indebtedness of Subsidiaries of OneBeacon Limited that is guaranteed by the Borrower is \$70,000,000 or less (excluding (i) any series of such Indebtedness if all of such series has been defeased (including legal or covenant defeasance) or discharged and (ii) such Indebtedness to the extent it is held by the Borrower or any of its Subsidiaries). Notwithstanding the foregoing, any release of Guarantee Obligations pursuant to clause (a) above shall be deemed subject to the provision that such Guarantee Obligations shall be reinstated as to the Guarantors at the time of such release if after such release any portion of any payment in respect of the Guaranteed Obligations is rescinded or must otherwise be restored or returned upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Borrower or such Guarantor, or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or other similar officer for, the Borrower or such Guarantor or any substantial part of its property, or otherwise, all as though such payment had not been made and whether or not the Administrative Agent is in possession of or has released this Guaranty. In addition, upon request of the Borrower, the Administrative Agent shall (without vote or consent of any Lender) take such actions as may be required to release all Guarantee Obligations under this Guaranty of any Guarantor if such Guarantor is merged or consolidated into another Person and such Guarantor is not the surviving Person of such merger or consolidation, or such Guarantor is liquidated, wound-up or dissolved, or such Guarantor otherwise ceases to be a Subsidiary of the Borrower; provided, however, that (x) the surviving Person of any such merger or consolidation is, or any assets transferred in connection with such liquidation, winding-up or dissolution are transferred to, a Guarantor (including a Subsidiary of the Borrower that becomes

16

a Guarantor in connection with such transaction) or a Subsidiary of a Guarantor (including a Person that will become a Subsidiary of a Guarantor in connection with such transaction) or (y) after giving effect to any such merger, consolidation, liquidation, winding-up, dissolution, or cessation of such Guarantor's status as a Subsidiary of the Borrower, the aggregate Consolidated Net Worth of all other Guarantors (including any Person that becomes a Guarantor or Subsidiary of a Guarantor in connection with such transaction but excluding for purposes of calculating the aggregate Consolidated Net Worth of all Guarantors any outstanding Indebtedness of the Borrower to a Guarantor (to the extent reflected as an asset of such Guarantor in the calculation of such Consolidated Net Worth) is not less than the amount then required by Section 7.1(a) of the Credit Agreement, on a pro forma basis as of the end of the most recently completed fiscal quarter for which financial statements have been delivered to the Administrative Agent pursuant to Section 6.1(a) of the Credit Agreement. The obligations of the Guarantors under this paragraph shall survive termination of this Guaranty.

- **32. Miscellaneous.** No provision of this Guaranty may be waived, amended, supplemented or modified, except by a written instrument executed by the Administrative Agent and the Guarantors. No failure by the Administrative Agent or any Lender to exercise, and no delay in exercising, any right, remedy or power hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy or power hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies herein provided are cumulative and not exclusive of any remedies provided by law or in equity. The unenforceability or invalidity of any provision of this Guaranty shall not affect the enforceability or validity of any other provision herein.
- **33. Setoff.** If and to the extent any payment is not made when due hereunder, each Lender may setoff and charge from time to time any amount so due against any or all of any Guarantor's accounts or deposits with such Lender.
- **34. Representations and Warranties.** Each Guarantor represents and warrants that (a) it is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (b) it has corporate or other power and authority and the legal right, to make, deliver and perform this Guaranty, (c) this Guaranty constitutes its legal, valid and binding obligation enforceable against such Guarantor in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether enforcement is sought by proceedings in equity or in law); (d) the making and performance of this Guaranty does not and will not violate any Requirement of Law or any Contractual Obligation of such Guarantor, except to the extent such violation could not, in the aggregate, reasonably be expected to have a Material Adverse Effect; and (e) all consents, approvals, licenses and authorizations of, and filings and registrations with, any Governmental Authority required under applicable Law for the making and performance of this Guaranty have been obtained or made and are in full force and effect except to the extent failure to obtain any such consent, approval, license, authorization, or to make such filing or registration, could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **35. Indemnification and Survival.** Without limitation on any other obligations of the Guarantors or remedies of the Administrative Agent and the Lenders under this Guaranty,

each Guarantor shall indemnify and hold harmless the Administrative Agent and the Lenders from and against any and all obligations, penalties, claims, demands, actions, judgments, suits, settlement payments, damages, losses, liabilities and causes of action of every kind or nature whatsoever and reasonable related out-of-pocket costs and expenses that may at any time be imposed on, sustained, required to be paid by, suffered, incurred or asserted against, the Administrative Agent or any Lender in connection with or as a result of any failure of any Guaranteed Obligations to be the legal, valid and binding obligations of the Borrower enforceable against the Borrower in accordance with their terms; provided that such indemnity shall not, as to the Administrative Agent or any Lender, be available to the extent that such obligations, penalties, claims, demands, actions, judgments, suits, settlement payments, damages, losses, liabilities, causes of action or costs or expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of the Administrative Agent or such Lender. In all such litigation, or the preparation therefor, the Administrative Agent and the Lenders shall be entitled to select counsel to the Administrative Agent and the Lenders. To the extent reasonably practicable and not disadvantageous to the Administrative Agent or any of the Lenders (as reasonably determined by the relevant party), it is anticipated that a single counsel selected by the Administrative Agent and the affected Lenders will be used. The obligations of the Guarantors under this paragraph shall survive the payment in full of the Guaranteed Obligations and termination of this Guaranty.

- **36. Governing Law.** THIS GUARANTY AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS GUARANTY SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK (EXCLUDING THE LAWS APPLICABLE TO CONFLICTS OR CHOICE OF LAW TO THE EXTENT THAT THE APPLICATION OF THE LAW OF ANOTHER JURISDICTION WOULD BE REQUIRED THEREBY).
  - 37. Submission to Jurisdiction; Waivers. EACH GUARANTOR HEREBY IRREVOCABLY AND UNCONDITIONALLY:
  - (a) SUBMITS FOR ITSELF AND ITS PROPERTY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS GUARANTY, OR FOR RECOGNITION AND ENFORCEMENT OF ANY JUDGMENT IN RESPECT THEREOF, TO THE NON EXCLUSIVE GENERAL JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK, THE COURTS OF THE UNITED STATES OF AMERICA FOR THE SOUTHERN DISTRICT OF NEW YORK, AND APPELLATE COURTS FROM ANY THEREOF;
  - (b) CONSENTS THAT ANY SUCH ACTION OR PROCEEDING MAY BE BROUGHT IN SUCH COURTS AND WAIVES ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE VENUE OF ANY SUCH ACTION OR PROCEEDING IN ANY SUCH COURT OR THAT SUCH ACTION OR PROCEEDING WAS BROUGHT IN AN INCONVENIENT COURT AND AGREES NOT TO PLEAD OR CLAIM THE SAME;
  - (c) AGREES THAT SERVICE OF PROCESS IN ANY SUCH ACTION OR PROCEEDING MAY BE EFFECTED BY MAILING A COPY THEREOF BY

18

REGISTERED OR CERTIFIED MAIL (OR ANY SUBSTANTIALLY SIMILAR FORM OF MAIL), POSTAGE PREPAID, TO THE GUARANTORS AT ITS ADDRESS SET FORTH ON THE SIGNATURE PAGE HEREOF OR AT SUCH OTHER ADDRESS OF WHICH THE ADMINISTRATIVE AGENT SHALL HAVE BEEN NOTIFIED PURSUANT HERETO;

- (d) AGREES THAT NOTHING HEREIN SHALL AFFECT THE RIGHT TO EFFECT SERVICE OF PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR SHALL LIMIT THE RIGHT TO SUE IN ANY OTHER JURISDICTION; AND
- (e) WAIVES, TO THE MAXIMUM EXTENT NOT PROHIBITED BY LAW, ANY RIGHT IT MAY HAVE TO CLAIM OR RECOVER IN ANY LEGAL ACTION OR PROCEEDING REFERRED TO IN THIS SECTION 17 ANY SPECIAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES.
- 38. Waivers of Jury Trial. EACH PARTY TO THIS GUARANTY HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING HEREUNDER OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO THIS GUARANTY, OR THE TRANSACTIONS RELATED HERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS GUARANTY MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION 18 WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.
- **39. Confidentiality; Notices.** Subject to the terms of Section 10.16 of the Credit Agreement, each Guarantor agrees that the Administrative Agent and the Lenders may disclose to any assignee of or participant in, or any prospective assignee of or participant in, any of its rights or obligations of all or part of the Guaranteed Obligations any and all information in the Administrative Agent's or any Lender's possession concerning the Guarantors, this Guaranty and any security for this Guaranty. All notices and other communications to the Guarantors under this Guaranty shall be delivered in accordance with the terms of Section 10.2 of the Credit Agreement and to the applicable Guarantor at its address set forth on the signature pages hereof or such other address of which the Administrative Agent shall have been notified pursuant hereto.
- **40. FINAL AGREEMENT.** THIS GUARANTY REPRESENTS THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS BETWEEN THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

#### [EXECUTION COPY]

Executed this	day of August, 2011.

xecuted this	day of August, 2011.	
		LONE TREE INSURANCE GROUP LTD.
		By:
		Name:
		Title:
		LONE TREE HOLDINGS LTD.
		By:
		Name:
		Title:
		Address for each Guarantor:
		14 Wesley Street 5th floor Hamilton, HM 11, Bermuda

### PRINCIPAL EXECUTIVE OFFICER CERTIFICATION PURSUANT TO RULE 13a - 14(a) OF THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED

- I, Raymond Barrette, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of White Mountains Insurance Group, Ltd.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

October 28, 2011		
By:		
/s/ Raymond Barrette		
Chairman and Chief Executive Officer		

(Principal Executive Officer)

### PRINCIPAL FINANCIAL OFFICER CERTIFICATION PURSUANT TO RULE 13a - 14(a) OF THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED

- I, David T. Foy, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of White Mountains Insurance Group, Ltd.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

October 28, 2011 By:
/s/ David T. Foy
Executive Vice President and Chief Financial Officer
(Principal Financial Officer)

# PRINCIPAL EXECUTIVE OFFICER CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of White Mountains Insurance Group, Ltd. (the "Company") for the period ending September 30, 2011 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Raymond Barrette, Chairman and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that to my knowledge:

- · The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and,
- The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company as of and for the periods presented in the Report.

/s/ Raymond Barrette	
Chairman and Chief Executive Officer	
(Principal Executive Officer)	

October 28, 2011

# PRINCIPAL FINANCIAL OFFICER CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of White Mountains Insurance Group, Ltd. (the "Company") for the period ending September 30, 2011 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, David T. Foy, Executive Vice President and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that to my knowledge:

- · The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and,
- The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company as of and for the periods presented in the Report.

/s/ David T. Foy

Executive Vice President and Chief Financial Officer
(Principal Financial Officer)

October 28, 2011